Precursor Systems Analyses of Automated Highway Systems

#### RESOURCE MATERIALS

### Lateral and Longitudinal Control



U.S. Department of Transportation Federal Highway Administration Publication No. FHWA-RD-95-095 October 1994

#### FOREWORD

This report was a product of the Federal Highway Administration's Automated Highway System (AHS) Precursor Systems Analyses (PSA) studies. The AHS Program is part of the larger Department of Transportation (DOT) Intelligent Transportation Systems (ITS) Program and is a multi-year, multi-phase effort to develop the next major upgrade of our nation's vehicle-highway system.

The PSA studies were part of an initial Analysis Phase of the AHS Program and were initiated to identify the high level issues and risks associated with automated highway systems. Fifteen interdisciplinary contractor teams were selected to conduct these studies. The studies were structured around the following 16 activity areas:

(A) Urban and Rural AHS Comparison, (B) Automated Check-In, (C) Automated Check-Out, (D) Lateral and Longitudinal Control Analysis, (E) Malfunction Management and Analysis, (F) Commercial and Transit AHS Analysis, (G) Comparable Systems Analysis, (H) AHS Roadway Deployment Analysis, (I) Impact of AHS on Surrounding Non-AHS Roadways, (J) AHS Entry/Exit Implementation, (K) AHS Roadway Operational Analysis, (L) Vehicle Operational Analysis, (M) Alternative Propulsion Systems Impact, (N) AHS Safety Issues, (O) Institutional and Societal Aspects, and (P) Preliminary Cost/Benefit Factors Analysis.

To provide diverse perspectives, each of these 16 activity areas was studied by at least three of the contractor teams. Also, two of the contractor teams studied all 16 activity areas to provide a synergistic approach to their analyses. The combination of the individual activity studies and additional study topics resulted in a total of 69 studies. Individual reports, such as this one, have been prepared for each of these studies. In addition, each of the eight contractor teams that studied more than one activity area produced a report that summarized all their findings.

Lyle Saxton Director, Office of Safety and Traffic Operations Research and Development

#### NOTICE

This document is disseminated under the sponsorship of the Department of Transportation in the interest of information exchange. The United States Government assumes no liability for its contents or use thereof. This report does not constitute a standard, specification, or regulation.

The United States Government does not endorse products or manufacturers. Trade and manufacturers' names appear in this report only because they are considered essential to the object of the document.

1. Report No.	2. Government Accession No	).	3. Recipients Catalog No.
4. Title and Subtitle Precursor Systems Anal Volume II-I ateral and I	ysis for Automated Highway Syst	ems:	5. Report Date October 1994
Final Report	Longitudinal Control		6. Performing Organization Code
7. Author(s) Dickerson, J.A.	Damos, D.*		8. Performing Organization Report No
Lai, M.C.	Smith, D.* Shulman, M <sup>†</sup>		10. Work Unit No. (TRAIS)
Chien, C.C. Kanaris, A.	Eckert, S. <sup>†</sup>		
<b>9. Performing Organizat</b> University of Southern Calif	ion Name and Address ornia		11. Contract or Grant No.
Center for Advanced Tr	ansportation Technology	-	DTFH61-93-C-00196
EE–Systems Departmer Los Angeles, California	nt , 90089-2562		<b>13. Type of Report and Period Covered</b> Final Report
<ul> <li>* Human Factors Department, University of Southern Californ</li> <li>† Ford Motor Company, Electronics Division, 19540 Allen Ro</li> <li>Michigan 48122</li> </ul>		ia ad, Melvindale,	October 1993-September 1994
12. Sponsoring Agency Na	me and Address		14. Sponsoring Agency Code
Federal Highway Administration 400 Seventh Street, SW			
15. Supplementary Notes			
<b>16. Abstract</b> This report analyzes the requ AHS. This report presents a path is characterized by five longitudinal, lateral and final and deployment scenarios. The control and transitions betwee requirements for the control travel time reduction and con- the reliability requirements in the automatic systems used in required redundancy and stru- the next and difficulty with the set	tirements, issues, and risks associal possible evolutionary path for the evolutionary representative system lly combined lateral and longitudin The performance requirement anal- een automatic and manual control systems. Roadway traffic controlling estion avoidance. The reliability in various levels of vehicle automatic n lateral and longitudinal control. inctural complexity in implementin	tted with lateral and longit automation of lateral and n configurations (ERSCs) nal systems in terms of the ysis covers driver comfort in addition to investigating lers may improve traffic fly requirements analysis us tion. This report derives the The reliability functional g these automatic systems	tudinal control of vehicles operating on the longitudinal control. This evolutionary . This analysis looks at the development of e performance and reliability requirements and acceptance issues during automatic g the sensor, actuator and controller low through traffic networks in terms of es NHTSA's accident rates data to quantify the reliability functional requirements for requirements allows us to assess the a. This information can be used to estimate
<b>17. Key Words</b> vehicle lateral and longitudinal control, evolutionary representative system configurations, reliability requirements, redundancy, performance requirements, human factors, capacity banefits, autonomous vahicles		<b>18. Distribution State</b> No Restrictions.	ment
19. Security Classif. (of this report) Unclassified	20. Security Classif. (of this page) Unclassified	<b>21.</b> No. of Pages 299	22. Price

Form DoT F17007 (8-72) Reproduction of Completed Page Authorized

Raytheon

### **Table of Contents**

Section 1	: Introduction	1
Section 2	Evolutionary Representative System Configuration	s5 7
EF	SC 2	8
EF	SC 3	10
EF	SC 4	11
EF	SC 5	12
Section 3	: Headway Selection And Capacity	13
St	opping Scenario for Vehicle Following	13
Mi	nimum Headway For Collision-Free Vehicle Following	
		17
Ac	cident Severity And Headway Selection	20
St	eady State Highway Capacity	25
Section 4	: Reliability	27
Re	liability Indices	27
Lo	ngitudinal Reliability of Human Drivers	28
La	teral Reliability of Human Drivers	28
Section 5	: ERSC 1 Analysis	31
De	scription of ERSC 1	31
Pe	rformance Requirements	33
Re	liability Requirement Analysis	56
EF	SC 1 Driver Tasks and Workload	74
Ke	y Results	81
Section 6	: ERSC 2 Analysis	95
De	scription of ERSC 2	95
Pe	rformance Requirements	97
Re	liability Requirement Analysis	123
EF	SC 2 Driver Tasks and Workload	138
Ke	y Results	143

### **Table of Contents**

#### (continued)

Section 7:	ERSC 3 Analysis	151
Description	n of ERSC 3	151
Performan	ce Requirements	
Reliability	Requirement Analysis	
Driver Tasl	ks and Workload	
Key Result	S	
Section 8:	ERSC 4 Analysis	
Description	n of ERSC 4	191
Performan	ce Requirements	193
Reliability	Requirement Analysis	
Driver Tasl	ks and Workload	
Key Result	S	
Section 9.	ERSC 5 Analysis	225
Description	n of ERSC 5	
Performan	ce Requirements	226
Reliability	Requirement Analysis	235
Driver Tasl	ks and Workload	239
Key Result	S	241
Section 10	Conclusions	243
Appendix A:	Communication Systems	253
Appendix B:	Roadway Traffic Controller	
References		

1.	Main automatic vehicle functions for each ERSC. The arrow indicates the introduction of a new fully automated vehicle function
2.	Acceleration profile for the leading vehicle in the worst case14
3.	Longitudinal forces on the leading vehicle
4.	Acceleration profile of the trailing vehicle
5.	The effect of trailing vehicle reaction time
6.	The effect of deceleration difference between the leading and trailing vehicles19
7.	The effect of road-tire friction coefficient under constant $E A_m$
8.	The effect of initial velocity difference
9.	( <i>h</i> , $\mathcal{A} V^2$ ) curves for different values of $t_{hr}$
10.	( <i>h</i> , $\not \in V^2$ ) curves for $A_{lm} = 0.8$ g, $A_{fm}$ varies from 0.4 to 0.7g. As $A_{fm}$ decreases the collision damage increases
11.	Block diagram of the speed and headway maintenance function. The sensors measure the relative speed and spacing of the vehicles, then the controller sends commands to the brake and throttle actuators so that the vehicle maintains the selected speed and/or headway
12.	Sensor requirements for range and closing rate detection for the speed and headway maintenance system
13.	The maximum sensor range with respect to the maximum deceleration of the vehicles for two different velocities. These values come from the "brick wall" scenario when the lead vehicle or debris is stopped on the highway
14.	Minimum safe time headway with respect to differences in the maximum deceleration between vehicles $\mathcal{E}A_m = A_{lm} - A_{fm}$ . A negative acceleration value means that the follower car can brake faster than the lead car. T1 is the time that the following vehicle detects the lead vehicle deceleration and initiates soft braking
15.	Minimum safe distance between vehicles for different human reaction times. The human reaction time is measured from the time that the vehicle ahead begins braking. The solid line shows the headway for a communications system. The dashed line shows the headway for a sensor-based system. 39
16.	Block diagram for the rear-end collision warning system. The system senses the distance and closing rate of the target vehicle. Then the warning algorithm computes the headway between the vehicles and warns the driver if necessary

17.	Changes in the detection time for lane change/merge collisions have a small effect on the accident severity. T1=0.1s shows the case for the ideal sensor. T1=0.4s shows the case for the single beam sensor
18.	Changes in the human reaction time for lane change/merge collisions have a large effect on the accident severity. $T_{hr}$ =0.7s shows the best case for human reaction time to an unexpected hazard. $T_{hr}$ =2.0s shows the worst case for human reaction time to an unexpected hazard43
19.	The probability of correctly deciding that the vehicle ahead is decelerating depends on the space between the two signals $d$ . As $d$ increases the detection-false alarm trade-off improves45
20.	The deceleration difference between the vehicles $\mathcal{E}A_m = A_{fm} - A_{lm}$ has a large effect on accident severity. As $\mathcal{E}A_m$ increases the potential accident severity increases
21.	Block diagram for the blind spot warning system
22.	The shaded area shows the region of coverage for a blind spot sensor on the left side of the vehicle
23.	As the road-tire friction coefficient decreases (the road gets more slippery), the minimum safe time headway increases. $\mathcal{E}A_m = A_{lm} - A_{fm}$ is the deceleration difference between the vehicles. A positive value means that the lead car has a larger deceleration than the follower vehicle54
24.	The functional block diagram of the SHM system
25.	The reliability functional block diagram of the SHM-driver system
26.	The required SHM system failure rates very as the driver's reliability degrades60
27.	The functional block diagram of the RECW system
28.	The RECW system framework for reliability design
29.	The functional block diagram of the BSW system
30.	The BSW system framework for reliability design69
31.	A block diagram of the roadway to vehicle speed/highway command function71
32.	A freeway system subdivided into sections
33.	Traffic flow entering section 1 during case with a low input traffic flow

34.	(a) Evolution of traffic density when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8. (b) Evolution of traffic density when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8
35.	<ul> <li>(a) Evolution of traffic velocity when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8 causing changes of speed back to section 1.</li> <li>(b) Evolution of traffic velocity when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8 and the vehicles resume a constant speed.</li> </ul>
36.	(a) The traffic flow evolution with respect to time with no roadway control. It takes more than 30 minutes (2000 s) for the traffic flow disturbances to subside. (b) The traffic flow evolution with respect to time for the roadway with roadway control. The traffic flow quickly smooths out to the steady state velocity
37.	The traffic flow evolution with respect to time for the roadway with roadway control. The traffic flow quickly smooths out to the steady state velocity
38.	(a) Evolution of traffic density when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8. (b) Evolution of traffic density when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8
39.	<ul> <li>(a) Evolution of traffic velocity when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8 causing changes of speed back to section 1.</li> <li>(b) Evolution of traffic velocity when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8 and the vehicles resume a constant speed.</li> </ul>
40.	Block diagram of the speed and headway maintenance function. The sensors measure the relative speed and spacing of the vehicles, then the controller sends commands to the brake and throttle actuators so that the vehicle maintains the target speed or minimum headway
41	Sensor requirements for range and closing rate detection for the speed and headway maintenance system
42.	The maximum distance sensor range versus the maximum vehicle deceleration for the "brick wall" scenario at two different velocities. The maximum vehicle deceleration is the vehicle capability on the roadway at a given time. It reflects the road-tire friction coefficient and vehicle's capabilities
43.	Minimum safe time headway with respect to differences in the maximum deceleration between vehicles $\mathcal{E}A_m = A_{lm} - A_{fm}$ . A negative acceleration value means that the follower car can brake faster than the lead car. ERSC 1 is the headway when the driver is responsible for any emergency braking situations. ERSC 2 shows the minimum headway when the vehicle has a rear-end collision avoidance function

- 45. As the road-tire friction coefficient decreases (the road gets more slippery), the minimum safe time headway increases.  $A_m = A_{lm} A_{fm}$  is the deceleration difference between the vehicles in *g*. A positive value means that the lead car has a larger deceleration than the follower vehicle.104

46.	Minimum safe distance between vehicles for different velocities . ERSC 1 is the headway when the driver is responsible for any emergency braking situations. ERSC 2 shows the minimum headway when the vehicle has a rear-end collision avoidance function105
47.	Block diagram of the rear-end collision avoidance system. The sensor measures the distance to obstacles. The controller assesses collision potential. This controller overrides the SHM function if an emergency stop is required
48.	Different collision avoidance scenarios for the rear-end collision avoidance function. (a) The vehicle ahead begins an emergency stopping maneuver. (b) Another vehicle is stopped in the lane ahead. (c) The preceding vehicle performs an evasive maneuver by steering out of the lane to avoid a vehicle or debris ahead. (d) Another vehicle cuts-in ahead
49.	The maximum sensor range with respect to the maximum deceleration of the vehicle at 60 mph for ERSCs 1 and 2 (SHM without and with rear-end collision avoidance) These values come from the "brick wall" scenario when the lead vehicle is stationary or there is debris on the highway
50.	Changes in the detection time for lane change/merge collisions have a large effect on the potential accident severity
51.	The probability of correctly deciding that the vehicle ahead is decelerating depends on the space between the two signals $d$ . As $d$ increases the detection-false alarm trade-off improves
52.	The deceleration difference between the vehicles $\mathcal{E}A_m = A_{lm} - A_{fm}$ has a large effect on accident severity. As $\mathcal{E}A_m$ increases the potential accident severity increases
53.	Block diagram of the lane departure warning system. The lane position sensors give the vehicle's position in the lane. The warning system calculates the time-to-lane-crossing and warns the driver if necessary
54.	Variables used in the calculation of the time to lane crossing

#### (continued)

55.	Block diagram of the steering assist system. The steering assist system acts as an inner loop to correct vehicle steering in the presence of disturbances such as wind gusts
56.	The sensors for the steering assist system measure the heading deviation <i>y</i> and the side slip angle <i>b</i>
57.	The functional block diagram of the SHM and RECA system
58.	The SHM and RECA system framework for reliability design126
59	The block diagram of the steering assist system interfacing with the driver
60.	The LDW system interfacing with the driver and roadway
61.	The LDW system framework for reliability design
62.	Maximum steady state flow rates for different velocities and time headways143
63.	Traffic flow entering section 1 during the simulation145
64.	Evolution of traffic density when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8
65.	Evolution of traffic density when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8
66.	(a) Evolution of traffic velocity when there is no roadway controller. (b) Evolution of traffic velocity when the roadway controller is present. The disturbance quickly dies out and the vehicles stay at a uniform speed
67.	The traffic flow evolution with respect to time for the roadway with roadway control. The traffic flow quickly smoothes out to the steady state velocity
68.	Block diagram for the automatic lane keeping function
69.	Roadway geometry for the lane width calculations. (a) shows these calculations for lane width on a straight section of roadway. (b) shows the turning path for a vehicle on a curve156
70.	Roadway width for different roadway curvatures. (a) The track width for different types of vehicles. P denotes passenger vehicles. SU is a single unit truck. WB-50 is a semi-trailer truck. (b) Human drivers require extra width on turns. The extra width depends on the design speed of the highway
71.	Lane change crash subtypes and variations. (21) SV is the subject vehicle. POV is the Principal Other Vehicle. (a) Types of proximity crashes. Both vehicles have a small velocity differential. (b) Fast approach crashes. There is a large velocity difference between the vehicles that rapidly closes the longitudinal gap between the vehicles

72.	Block diagram of the lateral collision warning system. Sensors detect possible obstacles. Then the warning algorithm calculates the time to collision and warns the driver of potential collisions.160
73.	The shaded area shows the region of coverage for a blind spot sensor on the left side of the vehicle
74.	Geometry and operation for entry/exit maneuver coordination. The roadway adjusts the speed and headway of the vehicles to allow for easy entrance in segment $S_{i+1}$ . (a) Geometry for designated entry/exit points. (b) Geometry for dedicated ramps
75.	The functional block diagram of the SHM and RECA system in ERSC 3170
76.	The reliability block diagram of the automatic lane keeping system
77.	The block diagram of the proposed automated lane keeping design172
78.	The LCW system interfacing with the driver177
79.	The LCW system framework for reliability design
80.	The functional block diagram of the roadway to vehicle speed/headway command function181
81.	The functional block diagram of the roadway maneuver coordination181
82.	The functional block diagram of the roadway lane reference aids
83.	Block diagram for the automatic lane changing system. Sensors and communication systems give information about the surrounding vehicles and their intent. The vehicle navigation system tells the vehicle when it should change lanes to following the vehicle's course. The roadway gives the vehicle information on the number of lanes available for traffic and their status194
84.	Sensor range for the automatic lane changing system for a lane change to the left side. The vehicle must detect all vehicles that could affect its lane change
85.	Message timeline for lane change coordination messages. The lane changing vehicle (LCV) sends out a request to change lanes. The surrounding vehicles send back acknowledgments. The vehicle then begins the lane change
86.	Block diagram for the collision avoidance controller. This controller uses lateral and longitudinal control to avoid collisions
87.	The shaded area shows the sensor range for the primary vehicle
88.	Block diagram for the maneuver coordination system

#### (continued)

89.	The functional block diagram of the automatic lateral and longitudinal control system208
90.	The ALLC system framework for reliability design
91.	The functional block diagram of the lane reference aids in ERSC 4
92.	The functional block diagram of the lane reference aids in ERSC 4
93.	Block diagram for the maneuver coordination system
94.	The functional block diagram of the roadway navigation system
95.	The functional block diagram of the vehicle navigation system in ERSC 5238
96.	Structure of message for communications between vehicles and between vehicles and the roadway
97.	Possible architectures for vehicle to vehicle communications systems. (a) shows a direct point- to-point link between vehicles. (b) shows a system where the roadway acts as a relay between the two systems
98.	Different types of signal coding to avoid interference. (a) Frequency division multiplexing (FDMA,b) Time division multiplexing (TDMA,c) Code division multiplexing (CDMA)257
99.	Cellular communication system for VRC. Each cell has a base station (black dot) that communicates with the vehicles within its range
100.	The dashed line shows the actual deceleration curve of the lead vehicle relative to the following vehicle. The solid line shows the actual sensor measurements taken by the following vehicle. The measurement noise is due to sensor and controller errors. $t_{fa}$ shows a false alarm situation. $t_{miss}$ shows a case where the detector misses the emergency braking condition. $t_{det}$ shows when the detector can safely decide that the vehicle is braking with a low probability of false alarm. 259
101.	The curve centered at 0 is the null hypothesis that assumes that the vehicle ahead is not executing an emergency stop. (a) In this case, the vehicle has just begun to brake, the hypotheses are not far enough away (i.e. $d$ is small) to make a good decision. (b) The vehicle has been braking for some time(i.e. $d$ is large) and the detector can make a correct decision with a low probability of false alarm
102.	The probability of correctly deciding that the vehicle ahead is decelerating depends on the space between the two signals <i>d</i> . As <i>d</i> increases the detection-false alarm trade-off improves

### List of Tables

1.	Summary of the functions of the roadway, vehicle and driver in each ERSC.	6
2.	Parameter definitions from figure 2.	14
3.	Parameters for trailing vehicle	16
4.	Parameter values for simulation study	18
5.	Parameter values for simulation study	21
6.	ERSC 1 functions and driver/vehicle/roadway performance requirements.	32
7.	Summary of requirements for speed and headway maintenance system.	37
8.	Variable values used in the simulations in section 5-1 for ERSC 1	38
9.	Summary of requirements for rear-end collision warning system	41
10.	Summary of requirements for blind spot warning system.	49
11.	Driver task description for normal ERSC 1 driving	75
12	Driver task description for speed and headway adjustments while driving	76
13	Driver task description for hard or emergency braking.	78
14	Driver task description for blind spot warning	79
15	Driver task description for fall-back mode to manual driving.	80
16.	ERSC 2 functions and driver/vehicle/roadway performance requirements.	96
17	Variable values used in the simulations in SHM section for ERSC 2	102
18.	Data used in the message for vehicle to vehicle communications	113
19.	Potential status variables to be sent to the roadway by the vehicles for check-in	122
20.	Driver task description for ERSC 2 driving	139
21	Driver resumes control of rear-end collision avoidance function	140
22	Driver task description for hard or emergency braking.	141
23.	Driver task description for lane departure warning	142
24.	ERSC 3 functions and driver/vehicle/roadway performance requirements.	152

### List of Tables

#### (continued)

25.	Design vehicle parameters for highway design.(1)157
26.	Potential status variables to be sent to the roadway by the vehicles for check-in.168
27.	Driver task description for ERSC 3 driving
28.	Driver resumes control of lane-keeping function
29.	Driver task description for lateral collision warning
30.	Driver task description for maneuver coordination
31.	ERSC 4 functions and driver/vehicle/roadway performance requirements192
32.	Types of communications needed for maneuver coordination203
33.	Potential status variables to be sent to the roadway by the vehicles for check-in.206
34.	Driver task description for ERSC 4 driving
35.	Driver resumes control of lane-changing function
36.	Driver resumes control of maneuver coordination
37.	ERSC 5 functions and driver/vehicle/roadway performance requirements226
38.	Types of communications needed for maneuver coordination
39.	Potential status variables to be sent to the roadway by the vehicles for check-in.234
40.	Driver task description for ERSC 5 driving
41.	Possible decisions of a binary detector

### List of Acronyms

AASHTO	American Association of State Highway and Transportation Officials		
ABS	Anti-lock Braking System		
AHS	Automated Highway System		
AICC	Autonomous Intelligent Cruise Control		
ALK	Automatic Lane Keeping		
ALLC	Automatic Lateral and Longitudinal Control		
ATIS	Advanced Traffic Information Systems		
ATMS	Advanced Traffic Management Systems		
AVI	Advanced Vehicle Identification		
BSW	Blind Spot Warning		
CDMA	Code Division Multiple Access		
ERSC	Evolutionary Representative System Configuration		
FDMA	Frequency Division Multiple Access		
FHWA	Federal Highway Administration		
FMEA	Failure Mode and Effects Analysis		
GPS	Global Positioning System		
ICC	Intelligent Cruise Control		
ID	Identification		
LCD	Liquid Crystal Display		
LCW	Lateral Collision Warning		
LED	Light Emitting Diode		
LDW	Lane Departure Warning		
MTTF	Mean Time to Failure		
NHTSA	National Highway Traffic Safety Administration		
PATH	Program on Advanced Technology for the Highway		
PSA	Precursor Systems Analysis		
RECA	Rear-End Collision Avoidance		
RECW	Rear-End Collision Warning		
SA	Steering Assist		
SHM	Speed and Headway Maintenance		
SHMS	Speed and Headway Maintenance System		
tbd	to be determined		

### List of Acronyms

(continued)

- TCAS Traffic Alert and Collision Avoidance System
- TDMA Time Division Multiple Access
- TLC Time to Line Crossing
- TTC Time to Collision
- USC University of Southern California
- VRC Vehicle-Roadway Communications

#### EXECUTIVE SUMMARY

The report analyzes the requirements, issues, and risks associated with lateral and longitudinal control of vehicles operating on AHS. This report is part of the Precursor Systems Analysis (PSA) study sponsored by the Federal Highway Administration (FHWA).

The contractor team for this project consisted of the University of Southern California (USC), Ford and Daimler-Benz. The USC group led the effort with P. Ioannou as the principal investigator. Other investigators include faculty from the Department of Human Factors at USC. Ford Motor Company assisted the USC group by providing experimental data, technical guidance and information related to vehicle dynamics, hardware, lateral and longitudinal control functions.

This report presents a possible evolutionary path for the automation of vehicle lateral and longitudinal control systems. This path is characterized by five evolutionary representative system configurations (ERSCs). Each ERSC builds upon the earlier ones in terms of performance and vehicle complexity. If a failure occurs then the system falls back to the previous ERSC. ERSC 1 has a dedicated lane for vehicles with speed and headway maintenance systems. In ERSC 2 the vehicles have steering assist, more vehicle to vehicle communications, lane departure warnings, and rear-end collision avoidance. ERSC 3 has "hands-off" steering in a single lane, lateral collision warnings, and maneuver coordination near entry and exit points. In ERSC 4 vehicles can change lanes automatically. These vehicles also have collision avoidance capability in both the lateral and longitudinal directions. ERSC 5 corresponds to a fully automated system where the infrastructure can guide and route the vehicles. For each ERSC, we analyze the performance and reliability requirements for automatic lateral and longitudinal control.

The performance requirement analysis covers safety, driver comfort and acceptance issues during automatic control and transitions between automatic and manual control. The analysis also investigates the sensor, actuator and controller requirements for the automatic control systems. The reliability requirement analysis uses NHTSA's accident rates data to estimate the required reliability at various levels of automation. This report also derives reliability functional requirements for automatic lateral and longitudinal control systems. The reliability functional requirements provide a way to assess the required redundancy and complexity in designing these automatic systems. The driver task analysis highlights the causes and effects of human driver interactions with the automated systems. Finally the report discusses the benefits in terms of capacity, traffic congestion, and safety of each level of automation.

The summary of key findings and issues of this research are:

1. Communication systems and sensors such as radar will be used on vehicles in all stages of AHS deployment. When multiple radar operate in a small area at similar frequencies, the radar on different vehicles may interfere with one another. This interference could lead to shorter sensor ranges and poor performance. Such interference is a safety issue since the sensors or communications systems may not work as expected. This issue may also affect how the highways are built and what separation is required between lanes.

2. Vehicle-to-vehicle communication can reduce the number of false alarms in rear-end collision warning systems without reducing the detection rates. Each vehicle sends its braking data to the vehicle behind it. These communications can serve as a sort of electronic brake light for the vehicle behind. This lets the following vehicle anticipate the braking of the vehicle ahead. This warning can decrease the number of false alarms in the rear-end collision warning system. However, the government will need to set communication frequencies and protocols so that a common communication system can be used on all highways.

3. A roadway traffic controller can decrease traffic congestion even in cases of low levels of automation. This controller eases roadway congestion by regulating vehicle speed and traffic density along sections of the roadway. This can smooth traffic flow and decrease backups due to incidents on the roadway. Studies at USC show considerable improvements in congestion reduction and stability of traffic flow by using appropriate roadway traffic controllers.

4. Multiple types of driver warnings for lateral/longitudinal collisions will be installed on vehicles in the early stages of AHS deployment. Some examples are rear-end collision and lateral collision warnings. The driver may experience warning overload and warning confusion problems while interacting with other vehicle and roadway functions. More research needs to be done on the interaction and prioritization of these warnings for the driver.

5. Collision avoidance systems use braking and steering to avoid crashing into other vehicles or stationary obstacles in the lane. Under the current system, the vehicle manufacturer may be liable for collisions caused by vehicle failures. This may have an impact on the automobile manufacturer's willingness to produce automated vehicles. It may also affect the design and operation of the automated systems. The legal system may need to limit manufacturer's liability.

6. Collision avoidance requires combined lateral and longitudinal control. Our work proposes an evolutionary system for the development of AHS. In one partial automation scenario the driver can override the automatic lane keeping system but not the longitudinal control for the throttle. The driver may not be able to perform lateral collision avoidance maneuvers using steering control only. This issue needs more human factors studies.

7. Our original evolutionary design split the development of full lateral control into two parts. The first part was automated lane keeping. The second part was lane changing and lateral collision avoidance. These control modes should not be separated since the driver becomes a passenger once the vehicle takes over lane keeping. Human factors studies show that humans do not perform monitoring tasks well.

8. A roadway controller that sends speed commands to vehicles can smooth vehicle speeds and improve traffic flow when incidents occur. The vehicle can respond to the roadway controller commands automatically as proposed in the Prometheus project or the driver can change the vehicle's cruising speed. If the vehicle automatically responds to the speed commands from the roadway, the driver's reliability in detecting rear-end collision danger may be degraded. The driver may be put into a situation that he/she can not handle and may blame the roadway for any rear-end collision. The roadway may be liable for accident caused by incorrect speed commands. However, if the vehicle does not respond directly to roadway speed commands, the effectiveness of the roadway traffic flow control will be reduced since the driver may not follow the roadway commands.

9. At each stage of the AHS evolution, drivers are required to have a basic understanding about the system operation so they can interface with AHS safely and efficiently. Drivers may need to be trained to handle the necessary tasks and workload.

10. Vehicles require a distance and closing rate sensor for speed and headway maintenance or intelligent cruise control. This system can also warn the driver of potential rear-end collisions. The rear-end collision warning system computes the Time-to-Collision and warns the driver of a potential collision in time to avoid the collision (about 1.5 s before the collision). The sensors for this type of system only need to be accurate to within 3-5 km/hr.

11. As the AHS evolves to higher ERSCs, the driver's functions are gradually replaced by the automatic control systems of vehicle and roadway. These systems needs to be more reliable than human drivers so that the AHS can be accepted by the public. They also need to be fail-safe to guarantee the safety of drivers. These reliability requirements may substantially increase the component redundancy and design complexity.

12. The reliability requirements for automatic systems in each ERSC can be used to assess the required component redundancy and design complexity. This gives us a way to estimate the gaps between ERSCs in terms of technical difficulties, cost and risks.

13. Safety and efficiency benefits in higher ERSCs can only be realized by heavily instrumented vehicles, using roadway lane reference aids and a roadway navigation system. It will take joint efforts of the public, the automobile manufacturers and the government to settle the potential legal and liability problems arising from AHS operation.

14. The communication requirements for AHS increase rapidly with the level of automation. When the vehicles have only longitudinal control, each vehicle needs to communicate only with the vehicle behind and in front. In a fully automated system the vehicles must communicate with the roadway and the surrounding vehicles to coordinate maneuvers. This change requires increases in bandwidth and reliability and more sophisticated communication protocols.

15. The human factors issues are very complex in the ERSCs that mix automated and manual control systems. Humans do not perform well as supervisors of automated systems. Systems that rely on a human driver who is not actively involved in lateral or longitudinal control to respond to infrequent warnings for collision avoidance or other emergencies may not work well. Once the vehicle can perform hands-off, feet-off driving in the dedicated lane, the human factors issues may be less critical to system design.

Along the proposed AHS evolutionary path, the key findings and issues and risks of each of the five ERSCs are:

ERSC 1 Key Findings

1. There are significant safety and performance benefits even with low levels of automation. Simple systems like the roadway velocity controller can give large increases in capacity and traffic flow volumes. Furthermore, the driver can serve as a back up to the automated systems at this level. The speed and headway maintenance function does not have to be very accurate in this ERSC (5-10%) since it is limited by human perception and reaction times. This suggests that speed and headway maintenance systems at this level may be derived (74.26)

from the intelligent cruise control systems planned by several companies.<sup>(74,26)</sup>

2. There are significant human factors issues associated with warnings and driver interface that must be researched for even low levels of automation. The warning systems will require multiple levels of warnings. Vehicle to vehicle communication systems can reduce the number of false alarms in collision warning systems without reducing the detection rates.

ERSC 1 Issues and Risks

#### 1. Driver Interface

Research needs to be done on the driver interface with the speed and headway maintenance, rear end collision warning, and blind spot warning functions. In particular, we need to investigate if the driver's reliability as a back-up to the SHM system will be degraded when he/she is interfacing with the on board SHM system and the roadway speed commands. It may be too much of a burden on the driver to expect him/her to change lanes and enable the automated systems immediately. The driver may need to preset the target speed, minimum headway, and warning thresholds when he/she starts the vehicle.

#### 2. Dedicated Lane

The biggest social issue for AHS is how to get a separate lane for automated vehicles. This infrastructure change may have to wait until enough vehicles have speed and headway maintenance systems and traffic flow benefits can be easily seen.

#### 3. Legal/Liability

In this ERSC the roadway sends recommended target speeds and minimum headways directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the minimum recommended headway is too short.<sup>(102)</sup>

#### 4. Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

#### ERSC 2 Key Findings

1. The driver can not act as a back-up for an automated rear-end collision avoidance system that uses short time headways. In this case there is not enough time for the driver to react to dangerous situations. The reliability requirement for the rear-end collision avoidance system is: "Under no circumstances should a single component/point failure let the vehicle crash into any moving or stationary object in the lane, and there should be no common failure modes." One

possible design leads to multiple control paths and sensors to ensure system reliability. This increases the cost and complexity of automated rear-end collision avoidance systems.

2. Driver warnings from the lane departure warning system and the blind spot warning system must be prioritized to avoid driver confusion if multiple warnings occur. Great care must be taken in ERSCs that combine automated systems with manual driving to make sure that the driver interface is simple to understand and operate.

ERSC 2 Issues and Risks

#### 1. Legal/Liability

Rear end collision avoidance gives rise to new liability issues. If a vehicle collides with the vehicle in front, then who is at fault? The driver may set the time headway too small or the collision avoidance system may have not worked properly.

#### 2. Driver Override of Braking

The driver may need to override the automated braking system if he/she sees an object that the collision avoidance sensors do not. However, this could lead to dangerous situations and a loss of vehicular control. This issue needs further study.

#### 3. Fully automated longitudinal control with manual lateral control

When only the longitudinal control is automated, the speed/headway maintenance and rear-end collision avoidance system may have difficulties in avoiding crashing into stationary obstacles that suddenly appear in the lane. This can happen when, for example, the preceding vehicle abruptly changes lanes to avoid hitting a stationary object in the lane. In this case, the vehicle may not have enough distance to stop the vehicle without a collision. The probability of such collisions can be reduced if we increase the headway used in vehicle following control. However, the capacity of the AHS will be affected by increasing the headway.

#### 4. Warning Overload and Confusion

The driver may experience warning overload and warning confusion problems while interfacing with BSW, LDW and other vehicle and roadway functions.

#### ERSC 3 Key Findings

1. Automatic lane keeping (ALK) will reduce the collisions caused by vehicle lane departures. The most destructive lane departure accidents take place when the driver is fatigued during a long drive. The ALK system can help reduce this type of accidents. It will require redundant active control channels for fail-safe design since the driver can not back up the automatic lane keeping system if it fails. The reliability functional requirement of the automatic lane keeping system is: "Under no circumstances, should a single point/component failure let the vehicle depart from the lane, and there should be no common failure modes".

2. It may be impractical to have the driver perform lateral collision avoidance while giving him/her only steering override authority.

3. The operation of lane keeping control requires preview information about the roadway. The on-board automatic lane keeping (ALK) system obtains this data with the help of roadway lane reference aids. The sensing technologies used by the automatic lane keeping system determine the types of lane reference aids. Possible lane reference aids include vision based systems, magnetic nails, radar, and GPS with dead-reckoning. Some types of lane reference aids may not reliably provide preview information under certain environmental conditions such as rain or snow. Reliability studies show that two independent control paths may be necessary for reliable fail-safe lane keeping. Each control path needs to have an independent sensor type.

#### ERSC 3 Issues and Risks

#### 1. Cost

A potential automatic lane keeping system design to fulfill the reliability functional requirement will need two redundant, independent steering control channels. This substantially increases the cost of automated vehicles in ERSC 3 due to the design complexity and required redundant components.

2. Reliability of Lane Reference Aids

The operation of lane keeping control requires preview information. The on-board automatic lane keeping (ALK) system obtains it with the help of roadway lane reference aids. The sensing technologies used by the automatic lane keeping system determine the types of lane reference aids. It might not be easy for some types of lane reference aids, such as vision based, to reliably provide preview information under certain environmental conditions.

#### 3. Driver's Lateral Collision Avoidance

Lateral collision avoidance may require combined lateral and longitudinal control. In ERSC 3, the driver can override the automatic lane keeping system but not the longitudinal control. The effectiveness of the driver's lateral collision avoidance, using steering control only, has not yet been verified and requires further research.

#### 4. Legal/Liability

Since the roadway is involved in the dedicated lane entry and exit coordination, the roadway may be liable for accidents if it fails to send out proper speed/headway commands to coordinate the entry/exit maneuvers. The driver may still blame the roadway for an entry/exit accident even it is caused by the his/her improper merging/demerging maneuvers. In addition, the roadway may be liable for accidents caused by failures of the lane reference aids. The vehicle manufacturers may also be responsible for the failure of the ALK system.

#### ERSC 4 Key Findings

1. The driver will not be able to back up the automatic lateral and longitudinal control (ALLC) system if it fails. The reliability functional requirement of the ALLC system is: "Under no circumstances should a single component/point failure result in a lateral or longitudinal collision, and there should be no common failure modes.

2. A potential ALLC system design to achieve the reliability functional requirement will need two active redundant control channels with substantial doubling of components.

3. The fully automated vehicles perform maneuver coordination to further reduce accident rates and improve traffic flow rates. This will require complicated vehicle-to-vehicle communication and maneuver protocols.

ERSC 4 Issues and Risks

#### 1. Cost

The required reliability of the ALLC system enormously increases the system cost and design complexity. The required instrumentation on the vehicle will be very costly. In addition, there are multiple dedicated lanes in ERSC 4. Since each lane needs to be equipped with redundant independent lane reference aids, the cost of the roadway infrastructure will be increased considerably.

#### 2. Legal/Liability

The roadway may be liable for traffic accidents due to its involvement in the vehicle lateral and longitudinal control, such as sending speed/headway commands, assisting maneuver coordination, and providing lane reference aids, etc. The vehicle manufacturers may also be responsible for the ALLC system failures during operation.

3. Reliability of Lane Reference Aids

If a dedicated lane has lane reference aids failure in some roadway sections, the AHS may have to prohibit vehicles from entering those sections. This will reduce the highway flow rate and cause traffic control problems.

#### ERSC 5 Key Findings

1. Network controllers may improve traffic flow through traffic networks in terms of travel time reduction and congestion avoidance or limitation. The roadway to vehicle communication system must be highly reliable since incorrect speed or lane change messages could cause congestion or unsafe situations.

2. The roadway controller proposed in earlier ERSCs for vehicle velocity control was local in nature. That controller aimed to improve local traffic conditions on a single stretch of road without coordination with other related systems such as ramp metering. The roadway controller is also isolated from other traffic control systems for route selection and navigation. Freeway network control uses real-time and predicted information to smooth traffic flow throughout the traffic network. ERSC 5 proposes an integrated controller that makes route and course recommendations and ramp metering decisions to achieve a common goal. The goal of freeway network control can be either user optimality or network optimality.<sup>(71)</sup> User optimality can be in terms of trip time or fuel expended. Network optimality can be in terms of overall travel time at the cost of some vehicles taking longer paths to meet their goals.

3. Simulation studies by Messmer and Papageorgiou indicate that there is good potential for improving traffic flow in freeway networks in terms of travel time reduction and congestion avoidance or limitation.<sup>(71,7)</sup> More detailed controllers will select lanes and speeds for each vehicle on the highway.

#### ERSC 5 Issues and Risks

#### 1. Network Traffic Control Effectiveness

Network freeway control needs extensive study and simulation before appropriate controllers can be selected. The system must also offer demonstrable improvements from the driver's existing ERSC 4 route selection system. Otherwise drivers will not use the network controller and will rely directly on their own vehicle to choose their route.

#### 2. Driver's Navigation

Driver overrides would allow individual drivers and vehicles to disregard commands from the roadway and use their own navigation systems. While this may benefit one driver, it may hurt overall traffic flow and cause safety problems if one vehicle makes many lane changes in an attempt to go at a certain speed. The effect of having network guided vehicles and independent vehicles on the same traffic network needs to be studied.

#### 3. Communication

Some researchers <sup>(111)</sup> have proposed network controllers that send control signals directly to each individual vehicle. This puts stringent performance and reliability constraints on the communications system since the messages are frequent and time-sensitive. System reliability considerations dictate that the communications system must be fail-safe with no single point failures for autonomous vehicles. This could lead to multiple independent communications systems.

This report also identifies some key areas for further research that are described below.

- The evolution of the AHS will be governed by the safe and efficient interaction between the human and the automated systems in the vehicle and the roadway. The (potential) automation technologies that can lead to the fully automated highway system are avaliable. However, it is an engineering challenge to incorporate these useful technologies to build automated systems that can be accepted by the users. Thus, human interaction with automatic control systems and driver assist devices need to be studied further. It is unknown how much attention the driver will devote to the driving task. Research also needs to be done on when and how the driver can safely override automated systems for collision avoidance, lateral or longitudinal control. The experience in the airplane cockpit design for controller–human interface analysis will be helpful.

- Sensor and actuator capabilities need to be improved. Sensors are needed that can see everything that a human driver can see. This allows the automatic control system to take the place of the human driver and prevents unsafe overrides of the automatic control system. The state of the art in vision-based sensors falls woefully short of this goal. The detection systems must have low false alarm rates and high probabilities of detection to work well.

- Extensive communications between vehicles and between vehicles and the roadway will be needed to coordinate maneuvers. Further research needs to be done to characterize the data volume, speed, and reliability requirements of this system.

### LATERAL AND LONGITUDINAL CONTROL ANALYSIS

#### Section 1: Introduction

The first phase of the automated highway system (AHS) development program is an analysis that lays the groundwork for prototype system development. The goal of this analysis is to provide a better understanding of the issues and risks associated with the design, deployment, and operation of AHS facilities. Ford Motor Company and Raytheon Company have formed a consortium that addresses 15 AHS activity areas in order to provide a comprehensive, integrated study. The consortium consists of Ford Motor Company, Raytheon Company, the University of Southern California (USC), the Georgia Institute of Technology, Tufts University, Daimler-Benz, and the University of Tennessee.

The design of AHS where fully automated vehicles will be driven by computer control systems and guided to their destinations without any or much intervention by the driver is considered by many experts to be the ultimate and most efficient form of transportation. It is going to revolutionize highway driving and introduce considerable benefits in terms of safety, capacity, driving comfort, and automobile pollution reduction. The development of such a complex system will depend on driver acceptance, vehicle manufacturer's willingness to produce automated vehicles and the ability of the local, state and federal government to develop and maintain the roadway infrastructure and resolve the associated legal, liability and institutional issues. The success of AHS will be based on the its ability to improve traffic flow and highway safety.

The complexity of these issues suggests that the development of AHS is going to follow an evolutionary path, along which issues associated with technology, human factors, policy, etc. will be resolved gradually over time. Initial market introductions will be based on small, incremental increases in the level of automatic control added onto existing vehicle dynamic systems. If these products are successful then this will lead to a complete implementation of an automated highway system. Therefore, we examined each of the essential elements of automation separately. Once the issues and limits of each component was understood, a deployment road map was developed that combined the elements of automatic control to provide a complete AHS.

Our analysis uses two types of representative system configurations from a generic automated roadway configuration. These two configurations are a dedicated automated roadway and a mixed roadway, where "mixed" means dedicated automated and manual lanes on the same roadway. Manual lanes contain instrumented (for automation, but not automatically operating) vehicles and non-instrumented vehicles. Designated lanes contain only instrumented vehicles.

In this report we present five representative system configurations that follow an evolutionary path. We refer to them as evolutionary representative system configurations (ERSCs). These ERSCs allow us to study the issues and risks associated with AHS in an incremental fashion



### Figure 1. Main automatic vehicle functions for each ERSC. The arrow indicates the introduction of a new fully automated vehicle function.

starting from partial automation, close to today's driving, and building towards a fully automated vehicle/roadway system. The ERSCs could also represent stages of implementation of AHS. As a result, each ERSC is chosen based on the complexity of issues involved, feasibility of technology, and benefits in terms of capacity and safety. Our emphasis in the area of lateral and longitudinal control and vehicle operational analysis is the evolution of vehicle automation for AHS. As we go from one ERSC to the next, we automate additional driving functions until we arrive at a fully automated vehicle whose motion is largely dictated by the roadway. Figure 1 shows the main automatic functions of the vehicle for each ERSC.

In the following sections we present a detailed description of each ERSC and the roadway, vehicle and driver functions that we analyzed under the areas of lateral/longitudinal control and vehicle operational analysis. This research addresses the desired performance requirements of longitudinal, lateral, and fully automatic control. The desired performance requirements include inputs from the USC and Ford human factors groups. These requirements include driver comfort and acceptance. A major part of this study focuses on understanding the human factors issues involved in automatic control and transitions between manual and automatic control. Then we investigate the reliability and fault tolerance requirements for fail-safe system design at each ERSC. This task establishes a basis for quantifying the reliability of automated systems.

This basis will be used for assessing the reliability requirements for vehicles under different types of automated and manual vehicle control. Successful implementations of automatic controllers will involve multiple independent control channels to avoid single-point failures in longitudinal and lateral controllers. This requirement will pose a heavy burden on automobile manufacturers. The results from the reliability and performance requirements studies allow us to look at the gaps between today's vehicles and future automated vehicles.

The first two ERSC's concentrate on longitudinal control. In an AHS longitudinal control may be used for functions such as speed and headway maintenance, rear-end collision avoidance, and lane changing. Headway refers to a measure of inter-vehicle spacing either in distance (constant space headway) or in time (constant time headway). Space headway is the product of time headway and vehicle speed. Potential gains in traffic throughput have prompted several researchers in the field to suggest constant spacing headways of around 1 meter. This gap leads to small relative collision velocities that may lead to minor or no damages to colliding vehicles. However this analysis does not take into account lateral collision forces that may lead to extensive vehicle damage. One meter gaps between vehicles would increase the capacity of the freeway. It has also been argued that by organizing the vehicles into platoons of 10 to 30 vehicles with large gaps between them, minor intervehicle collisions can be contained within the platoon. This gap raises many important issues related to safely, human factors, and the capabilities of sensor and computer technologies. Therefore this analysis takes a collision-free approach in which each vehicle can come to a complete stop without hitting the vehicle in front of it.

The second two ERSC's concentrate on lateral control. Lateral control may be used for functions such as keeping the vehicle in the lane, lane changing, entry/exit maneuvers, and lateral collision avoidance. Lateral collision avoidance maneuvers and lane changing require a combination of lateral and longitudinal control.

Section 2 of this report describes the five ERSCs that we analyzed in this report. Section 3 covers the mathematical models used to determine safe stopping distance and freeway traffic densities. Section 4 gives background and definition of terms used in reliability analysis. Sections 5 through 9 cover each of the five ERSCs for performance requirements, reliability requirements, benefits, and risks. Section 10 summarizes these issues and risks for the ERSCs.

## Section 2. Evolutionary Representative System Configurations

The evolution of todays' vehicles to fully automated ones will be gradual. It will be dictated by market forces and liability obstacles. This evolution has already started. Automatic features such as cruise control, anti-lock braking systems (ABS) and, more recently, headway maintenance systems and other warning devices will be standard items on most vehicles in the very near future. There is also considerable research on steering assist devices, blind-spot detector devices, collision warning systems, etc. A continuing trend in the development of vehicles is the use of more electronics and computers. Products such as steer-by-wire, drive-by-wire and brake-by-wire have been experimentally tested and will gradually appear in vehicles. From the roadway and infrastructure point of view, developments such as automatic toll collection, automated traffic light control and emerging technologies for Advanced Traffic Management Systems (ATMS) and Advanced Traffic Information Systems (ATIS) are already implemented, or they are going to be implemented in the very near future.<sup>(55)</sup>

Based on the evolution of these technologies the following deployment scenarios or evolutionary representative system configurations (ERSC) for AHS may become feasible. Each ERSC builds upon the one before it in terms of safety and reliability. If a failure occurs then the system drops down to the previous ERSC. ERSC 1 has a dedicated lane for vehicles with speed and headway maintenance systems. In ERSC 2 the vehicles have steering assist, more vehicle-vehicle communications, lane departure warnings, and rear-end collision avoidance. ERSC 3 has "hands-off" steering in a single lane, lateral collision warnings, and communications between vehicles in separate lanes. In ERSC 4 vehicles can change lanes automatically. These vehicles also have a collision avoidance ability in both the lateral and longitudinal direction. ERSC 5 is the fully automated highway that can guide and route the vehicles. Table 1 summarizes these ERSCs. We define each ERSC with the roadway, the driver, and the vehicle. Each ERSC raises new issues and risks as the amount of automation increases.

ERSC	ROADWAY	VEHICLE	DRIVER
1	<ul> <li>Provides dedicated lane</li> <li>Sends target speeds to vehicles</li> <li>Sends minimum headway to vehicles</li> <li>Sends traffic status information to vehicles</li> <li>Receives vehicle speed, headway and operational status information and identification from vehicles (option)</li> </ul>	<ul> <li>Speed &amp; headway maintenance</li> <li>Rear-end collision warning</li> <li>Blind spot warning</li> <li>Receives speed, headway commands and traffic status information</li> <li>Sends speed, headway, operational status and i.d. to the roadway (option)</li> <li>Driver/roadway interface</li> <li>Fall-back mode</li> </ul>	<ul> <li>Lane keeping &amp; lane changing</li> <li>Responds to blind spot and rear-end collision warnings</li> <li>Collision avoidance (lateral/longitudinal)</li> <li>Route Planning</li> <li>Vehicle interface</li> </ul>
2	<ul> <li>Provides dedicated lane</li> <li>Sends target speeds to vehicles</li> <li>Sends minimum headway to vehicles</li> <li>Sends traffic status information to vehicles</li> <li>Receives speed, headway, operational status information, and i.d. from vehicles</li> <li>Provides lane reference aids for lane departure warning</li> </ul>	<ul> <li>Speed &amp; headway maintenance</li> <li>Rear-end collision avoidance</li> <li>Blind spot warning</li> <li>Receives target speed, headway commands and traffic information from roadway</li> <li>Sends speed, headway, operational status and i.d. to the roadway</li> <li>Lane departure warning</li> <li>Steering assist</li> <li>Driver/roadway interface</li> <li>Fall-back mode</li> </ul>	<ul> <li>Lane keeping &amp; lane changing</li> <li>Responds to blind spot and lane departure warnings</li> <li>Lateral collision avoidance</li> <li>Route Planning</li> <li>Vehicle interface</li> </ul>
3	<ul> <li>Provides dedicated lane</li> <li>Sends target speeds to vehicles</li> <li>Sends minimum headway to vehicles</li> <li>Sends traffic information to vehicles</li> <li>Provides lane reference aids for lane keeping</li> <li>Receives vehicle status speed, headway data</li> <li>May assist in vehicle maneuver coordination</li> </ul>	<ul> <li>Speed &amp; headway maintenance</li> <li>Rear-end collision avoidance</li> <li>Lateral collision warning</li> <li>Receives speed, headway and traffic information from roadway</li> <li>Transmits vehicle status and speed to roadway</li> <li>Lane keeping</li> <li>Maneuver coordination with communication (lane change/merge in AHS entry/exits)</li> <li>Driver interface</li> <li>Fall back mode</li> </ul>	<ul> <li>Lane changing</li> <li>Responds to lateral collision warning</li> <li>Route Planning</li> <li>Lateral collision avoidance</li> <li>Vehicle interface</li> </ul>

#### Table 1. Summary of the functions of the roadway, vehicle and driver in each ERSC.

4	• Provides multiple dedicated	• Speed & headway maintenance	Vehicle interface
	lanes	Rear-end collision avoidance	
	• Sends target speeds to vehicles	<ul> <li>Lateral collision avoidance</li> </ul>	
	<ul> <li>Sends minimum headway to</li> </ul>	<ul> <li>Receives speed, headway and</li> </ul>	
	vehicles	traffic information from roadway	
	• Sends traffic information to	<ul> <li>Transmit vehicle status and</li> </ul>	
	vehicles	speed to roadway	
	• Provides lane reference aids for	Lane keeping	
	lane keeping and lane changing	Lane changing	
	<ul> <li>Receives vehicle status data</li> </ul>	<ul> <li>Maneuver coordination</li> </ul>	
	<ul> <li>Assists in vehicle maneuver</li> </ul>	Route Planning	
	coordination	<ul> <li>Driver/roadway interface</li> </ul>	
		• Fall-back mode	
5	<ul> <li>Provides multiple dedicated</li> </ul>	• Speed & headway maintenance	Vehicle interface
	lanes	<ul> <li>Rear-end collision avoidance</li> </ul>	
	<ul> <li>Sends lateral/longitudinal</li> </ul>	Lane keeping	
	control commands	Lane changing	
	• Receives vehicle status, i.d.,	<ul> <li>Lateral collision avoidance</li> </ul>	
	position, speed, acceleration,	• Receives control commands from	
	headway, and destination	roadway	
	<ul> <li>Provides lane reference aids</li> </ul>	<ul> <li>Sends its operational status,</li> </ul>	
	• Coordinates vehicle maneuvers	identification. position, speed,	
	<ul> <li>Vehicle route planning</li> </ul>	acceleration, headway, and	
		destination to roadway	
		<ul> <li>Maneuver coordination</li> </ul>	
		<ul> <li>Driver/roadway interface</li> </ul>	
		• Fall-back mode	

#### ERSC 1

In ERSC 1 the automated vehicles use a dedicated lane provided by the roadway. The lane could be isolated and accessed by a dedicated ramp or it could be next to a manual lane and accessed at designated points or at any point along the manual/dedicated lane boundary. The roadway is responsible for maintaining the dedicated lane and providing emergency vehicle response. It could also assist in keeping non-fit vehicles off the dedicated lane. The roadway senses traffic flow (average speed, density) and environmental conditions on the dedicated lane and provides target speed commands, minimum time headway recommendations and traffic status information to the vehicles. The roadway also performs ramp metering and incident management if needed.

The automated vehicle can maintain a selected headway and speed relative to the preceding vehicle by using a computer control system to control the throttle and the brake. Automatic braking is limited to soft braking applied when the engine torque is not sufficient to maintain the selected headway. It does not include hard braking for emergency stops or other situations. The vehicle provides blind spot and rear-end collision warnings to the driver. It receives target speed commands, minimum time headway recommendations and traffic status information from the roadway. It responds to the speed commands automatically in a smooth way as long as the resulting headway is not smaller than the one selected by the driver. If the target speed is larger than the one the driver feels comfortable with, the driver may be required to exit the lane. The

vehicle responds to the recommended minimum time headway provided it is larger than the selected one. The vehicle provides the driver with the traffic status information received from the roadway. It responds to driver commands for changing the headway, and setting the cruising speed. It also enables or disables the speed headway maintenance system (SHMS) upon driver command. The vehicle may send its speed, headway and operational status and identification to the roadway to be used for traffic flow control purposes. If the SHMS fails the vehicle fall-back mode allows the driver to take over this function and continue the trip until the next exit.

The driver is responsible for driving the vehicle into the dedicated lane with the aid of the blind spot warning system. Once in the lane the driver switches the SHMS on and sets the desired headway for vehicle following or the cruising speed. The driver supervises the SHMS and overrides it when hard braking is required for emergency stops and other situations. The driver is aided by the rear-end collision warning in avoiding rear-end collisions. The driver is responsible for lane keeping, lane changing and lateral collision avoidance. The driver is also responsible for choosing the route of the vehicle and conforming to traffic regulations. In case of malfunction of the SHMS or at the end of the trip the driver drives the vehicle out of the dedicated lane by using the next available exit with the aid of the blind spot warning system.

The motivation behind ERSC 1 is:

- to improve safety through the use of warnings and roadway headway recommendations
- to smooth traffic flow through the use of roadway target speed commands
- to increase capacity through the use of roadway commands for smooth traffic flow control and possibly smaller headways selected by the drivers
- to reduce the number of possible liability issues by making the driver responsible for vehicle emergencies
- the feasibility of deployment based on current and near-term availability of technologies

#### ERSC 2

In ERSC 2, the roadway provides a dedicated lane with the same vehicle accessibility as in ERSC 1.

The roadway maintains the dedicated lane and provides emergency vehicle response. It assists in keeping non-fit vehicles off the dedicated lane. The roadway receives vehicle status data such as speed, headway, operational status and vehicle identification. It uses the vehicle speed and headway information for calculating the average speed and density of the traffic flow in the lane and computes the desired target speed at each section of the lane for smoother traffic flow. It sends information to the vehicles regarding minimum time headway based on environmental conditions. The operational status and identification of the vehicle are used for check-in purposes and for roadway emergency response in case of disabled vehicles. The roadway also assists drivers in route planning by providing them with traffic status information. The roadway may also assist vehicles merging into the automated lane by recommending larger headways at the entry points or sections or the dedicated lane. The roadway provides lane reference aids that support the vehicles' lane departure warning and steering assist function.

The vehicle has all the capabilities as in ERSC 1, with the rear-end collision warning being upgraded to rear-end collision avoidance. The SHMS and rear-end collision avoidance allow the driver complete "feet-off" driving, while in the dedicated lane. The vehicle provides steering assist to smooth the driver's steering by compensating for roadway disturbances, wind gusts and small driver steering errors. The vehicle also has lane departure warning to warn the driver during large deviations from the center of the lane. The vehicle sends its speed and headway, operational status and identification to the roadway. It sends its braking capabilities and acceleration intentions to the vehicle behind. It receives the braking capabilities and acceleration and deceleration intentions of the vehicle in front. The vehicle also receives and responds to target speed commands, headway recommendations, and traffic status information sent by the roadway in the same way as in ERSC 1.

The braking capabilities and acceleration/deceleration intentions of the vehicle in front as well as the headway recommendations sent by the roadway are used by the vehicle to calculate the minimum time headway for collision free vehicle following. The SHMS chooses a headway that is larger than the calculated one based on some tolerance for additional safety. Since the driver is not considered to be a back up in a rear-end collision avoidance situation the calculated headway does not take into account the driver reaction time and is therefore smaller than the one used for collision warning in ERSC 1. The vehicle responds to driver commands for changing the headway and setting the cruising speed. It also enables and disables the SHMS and rear-end collision avoidance functions upon driver commands. The driver disengagement of the rear-end collision avoidance function initiates a smooth transition sequence that does not put the driver in a situation of a short time headway relative to his reaction time. The vehicle provides a blind spot warning for aiding the driver during lane changing and a lane departure warning for aiding the driver during lane following. A steering assist vehicle function used in series with the driver improves the accuracy of steering in the presence of disturbances such as wind gust and road surface varying conditions. As in ERSC 1 the vehicle has on-board diagnostics that are used to detect malfunctions. The vehicle provides a fall-back mode to ERSC 1 in case the rear-end collision function or steering assist and lane departure warnings fail. In such a case the vehicle increases the headway and warns the driver to assume control of the failed function.

The driver drives the vehicle to the entrance of the dedicated lane, provided the vehicle is fit to operate in the lane, and looks for a safe gap in the lane for positioning his/her vehicle. The entrance could be a long lane boundary next to a transition lane that allows speed synchronization. Since the vehicle is equipped with vehicle-to-vehicle and vehicle-to-roadway communications its intentions to enter the lane could be communicated to the roadway and surrounding traffic and used to negotiate request or safe gaps.

Once in the lane the driver switches on the SHMS, the rear-end collision avoidance, the lane keeping warning and steering assist functions. The vehicle chooses a safe headway that does not take into account the driver's reaction time. As a result, the driver function in the longitudinal direction is feet-off. The driver may be given the option to increase the headway if he/she does not feel comfortable with the one chosen by the system. The driver however will not be able to reduce the headway chosen by the system. The driver is fully responsible for lane keeping, route

planning, lane changing, and for lateral collision avoidance. For these functions he is assisted by the lane departure, blind spot warnings and steering assist. In case of malfunction or at the end of the trip the driver is responsible for driving the vehicle out of the lane and for obeying all traffic regulations. When the vehicle falls back to ERSC 1 due to malfunctions of the collision avoidance or steering assist and lane departure warning functions the driver is responsible for taking over the failed functions.

The motivation behind ERSC 2 is:

- to increase capacity by using smaller headways made possible by the rear-end collision avoidance function, vehicle-to-vehicle and roadway-to-vehicle communications
- to improve safety and driver comfort by introducing lane keeping warning and steering assist
- to smooth traffic flow through the use of roadway traffic flow control achieved by conforming to roadway specified vehicle target speeds, and headways.
- the feasibility of deployment based on current and near-term availability of technologies

ERSC 2 allows us to focus on the analysis of the fully automated vehicle longitudinal control function.

#### ERSC 3

In ERSC 3 the roadway provides and maintains a dedicated lane with a similar accessibility to vehicles as in ERSC 1, 2. The roadway also provides lane reference aids that support the lane keeping function of the vehicle as well as preview lane information for smooth and accurate lane following.

The vehicle has all the capabilities as in ERSC 2, with the exception of the lane departure warning and steering assist that evolve into automatic lane keeping. The automatic lane keeping function keeps the vehicle in the center of the lane at highway speeds and curvatures. The blind spot warning evolves into a lateral collision warning. The vehicle is fully responsible for vehicle-following in the longitudinal direction, for rear-end collision avoidance and for keeping the vehicle in the center of the dedicated lane without any support from the driver. It sends its speed, headway and operational status to the roadway. It responds to target speed and headway commands and traffic information received from the roadway in a similar manner as in ERSC 1, 2. The vehicle notifies the driver in case of malfunctions and responds to commands for switching from the manual to automated mode and vice versa. The vehicle also responds to and coordinates its maneuvers with other vehicles during merging and exiting the dedicated lane.

The vehicle alerts the driver when is the time to assume manual control and transfers control to the driver in a smooth way that is compatible with driver's skills and reaction times. The vehicle uses its on-board diagnostics to notify the driver of the fitness of the vehicle to operate on the dedicated lane. In case of malfunctions or failures of some of the automated vehicle functions the vehicle has a fall-back mode that allows the vehicle to operate as in ERSC 2 or ERSC 1 or

the manual mode for a designated time. In such case it warns and aids the driver in assuming control of the failed functions.

The driver is responsible for merging the vehicle into the dedicated lane with the aid of lateral collision warning and vehicle-to-vehicle communication for maneuver coordination and for switching on and off the automated mode. The driver is also responsible for operating the vehicle in a fall-back mode and for driving the vehicle out of the dedicated lane in case of malfunctions of the automated functions and at the end of the trip. Since driving is hands-off and feet-off, the driver has no responsibility during vehicle operation in the dedicated lane apart from deciding the route of the vehicle and the end of the trip. The point at which the driver releases manual control could be in the dedicated ramp or in the dedicated lane depending on the entry configuration. Similarly the point at which the driver assumes manual control from the automated mode depends on the exit configuration.

The motivation behind ERSC 3 is:

- to smooth traffic flow and increase capacity by using the roadway commands and smaller headways
- to improve safety by introducing a lateral collision warning
- the feasibility of deployment by evolving from ERSC 2 to ERSC 3.

ERSC 3 allows us to focus on the analysis of both the fully automated longitudinal control and lane keeping functions, i.e., feet-off and hands-off operation without the complexity of automatic lane changing.

#### ERSC 4

In ERSC 4 the vehicle is fully automated. The vehicles of ERSC 3 add automatic lane changing and lateral collision avoidance capabilities. The roadway provides and maintains the dedicated lanes and provides emergency vehicle response. It also provides lane reference aids for lane keeping and lane reference aids for lane changing. The dedicated lanes could be segregated and accessed from dedicated ramps or could be next to manual lanes and accessed through transition lanes depending on the particular entry/exit configuration. As in ERSC 2 and 3 the roadway receives vehicle status information, assists in the check-in and merging process and sends target speed, headway commands and traffic information to the vehicles.

The additional capabilities of automatic lane-changing and lateral collision avoidance allow the vehicle to take over the entire driving task when the vehicle is in the dedicated lanes. All vehicle maneuvers, such as lane changes and merges, are coordinated with the surrounding automated vehicles. The vehicles use cooperative communications and their own sensor measurements for this coordination. Each vehicle broadcasts its position and heading to the vehicles around it. These communications may be routed through the roadway so that the roadway can monitor the highway status and coordinate communication traffic and protocols. The vehicle plans its route according to the trip destination entered or changed by the driver and the traffic information
received from the roadway. The vehicle may display the planned route and the estimated travel time to the driver. The vehicle alerts the driver and assists him in resuming manual control. It has a contingency plan if the driver cannot resume manual control at the end of the trip or during certain emergencies. The vehicle plans and performs all collision avoidance maneuvers. The vehicle is designed to be fail safe and completely reliable during automated driving. The vehicle uses its on-board diagnostics to check the fitness of the vehicle for operating in the automated mode and notify the driver accordingly.

The vehicle has a fall-back mode that allows the vehicle to operate as in lower ERSCs in case of malfunctions.

The driver enters the destination and manually drives the vehicle to the entry point. The vehicle passes a check-in test. The driver switches on the automatic mode and the vehicle assumes control. Once the automated mode is on the driver becomes a passenger until he/she initiates a request to switch back to a lower ERSC or to manual mode. In the case of switching back to the manual mode the vehicle initiates a check-out procedure and the driver gradually regains full manual control in a transition lane or at the exit ramp. When malfunctions occur and the fall-back mode of the vehicle becomes active the driver is responsible for operating the vehicle as in lower ERSCs. At the end of the trip, the driver gradually resumes control from the vehicle. Driver control resumption includes tests for alertness, coordination, etc.

The motivation behind ERSC 4 is:

- to analyze the benefits in terms of driver comfort, safety and capacity that could be achieved by full vehicle automation
- to study the evolution of technology from ERSC 3 to ERSC 4

ERSC 4 allows us to focus on the analysis of the vehicle fully automated functions as they interact with each other to perform driving and emergency tasks without relying too much on the roadway infrastructure.

## ERSC 5

In ERSC 5, the roadway provides multiple dedicated lanes as described in ERSC 4. The roadway is responsible for maintaining these dedicated lanes and the communications infrastructure.

The roadway receives position, speed, destination, and capabilities data from the vehicles. The roadway senses traffic flow and environmental conditions on the dedicated lanes. The roadway uses this data to provide detailed course commands to the vehicles. These commands tell the vehicle what lane to travel in, when to change lanes and what speed and headway to use. The roadway coordinates these maneuvers to avoid collisions and smooth traffic flow. Based on the indicated trip destination the roadway plans the vehicle's route.

The vehicle has all the capabilities for full lateral and longitudinal control described in ERSC 4. Additionally the vehicle receives detailed course commands from the roadway. The vehicle

responds to these commands automatically as long as it judges the maneuver to be safe. The vehicle sends its position, speed, maneuver status, and capabilities to the roadway. The vehicle coordinates each roadway-commanded maneuver with the vehicles around it. The vehicle provides the driver with traffic information and trip status data. It responds to driver commands for changes in destination. In case of malfunctions the vehicle may fall-back to a lower ERSC and notify the roadway and driver appropriately.

The driver manually drives the vehicle to the dedicated lane check-in point. The driver initiates the check-in procedure and releases control to the vehicle and roadway. The driver inputs a trip destination that is used by the vehicle and roadway for route planning. The driver, as in ERSC 4, can request a check-out procedure and resume manual control at the next chosen exit ramp or transition lane depending on the exit configuration used.

The motivation behind ERSC 5 is:

- to maximize highway system throughout by permitting full roadway control of all vehicle activities.
- to improve driver comfort and reduce trip time.
- to efficiently detour traffic around traffic incidents and roadwork.

ERSC 5 allows us to focus on the analysis of a highly coordinated AHS where the roadway plays an active role in the lateral and longitudinal control of the automated vehicles by planning their path and coordinating their maneuvers.

# Section 3: Headway Selection And Capacity

The inter-vehicle spacing or headway in vehicle following affects both safety and highway capacity. For collision-free vehicle following, the headway should be large enough so that under a worst case stopping scenario no collision can take place. For a high capacity highway system, the headway setting should be as small as possible. Since safety cannot be easily traded-off, the choice of the minimum safety headway for a collision-free environment is important both from safety and capacity points of view.

Traffic accidents involve various types of vehicle crashes, such as rear-end collisions, backing collisions, single vehicle road departure accidents, etc.<sup>(60)</sup> During vehicle following operations, rear-end collision is the most common type of accident. In 1990, 23 percent of all police-reported crashes were rear-end collisions that caused 4.7 percent of all fatalities.<sup>(60)</sup> Current statistics portray these rear-end crashes as resulting largely from driver delayed recognition and relatively long reaction time when driving under high speed and close inter-vehicle separation.<sup>(60)</sup>

In principle, the possibility of a rear-end collision can be reduced by reducing vehicle speed and increasing inter-vehicle spacing. Since roadway capacity is proportional to vehicle speed and inversely proportional to inter-vehicle spacing, a large reduction in speed or a large increase in spacing leads to a low capacity highway system. This is the so called safety/capacity trade-off that is well known in the area of transportation.

The choice of the operating vehicle speed (V) and inter-vehicle spacing (S) for maximum capacity under the constraint of collision-free vehicle following environment is a big challenge in the design of roadway/vehicle systems. The design of vehicles imposes an upper bound on the maximum velocity a vehicle can attain. State and federal regulations also impose upper bounds on the maximum allowable velocity. The spacing S can be reduced under the imposed safety constraints. One way to characterize these safety constraints is to consider a worst case stopping scenario in a vehicle following operation. Such a scenario may be used to calculate the minimum value of S for collision-free vehicle following.

## Stopping Scenario for Vehicle Following

We consider the following worst case stopping scenario in a single lane vehicle following situation.

**Leading vehicle** : At time instant t = 0, the leading vehicle brakes with maximum jerk ( $J_{lmax}$ ) until it reaches its maximum deceleration ( $-a_{lm}$ ), and then keeps this deceleration until a full stop is achieved. The acceleration profile of the leading vehicle is shown in figure 2. Table 2 describes the parameters used in figure 2.



Figure 2. Acceleration profile for the leading vehicle in the worst case.

Table 2.	Parameter	definitions	from	figure 2.
I dole I	I al allievel	actinitions		ingui e II

Parameter Name	Description
$a_{lm}$	the maximum deceleration of the leading vehicle
J <sub>lmax</sub>	the maximum jerk of the leading vehicle
$t_{la}$	the time required for the leading vehicle to reach $-a_{lm}$
$t_{lb}$	the time for the leading vehicle to come to a full stop

The jerk value  $J_{lmax}$  is constrained by the mechanical limits of the vehicle braking system. The maximum deceleration  $a_{lm}$  is determined by the condition and properties of the vehicle braking system as well as the condition of the road. The impact of the road geometry and condition on the braking ability of the leading vehicle is illustrated in figure 3.



Figure 3. Longitudinal forces on the leading vehicle.

In figure 3, *M* is the mass of the leading vehicle;  $\theta_l$  denotes the road slope angle;  $F_1$  denotes the maximum braking force;  $F_2$  is the force due to the gravity along the road slope. These two forces can be further described as:



Figure 4. Acceleration profile of the trailing vehicle.

$$F_{1} = \lim_{l \to \infty} MA_{lm} \cos l$$

$$F_{2} = Mg \sin l$$
(1)

where  $\mu_{lmax}$  denotes the maximum road-tire friction coefficient.  $A_{lm}$  denotes the maximum deceleration under normal dry road with zero slope angle.

Using Newton's Second Law, the maximum deceleration of the leading vehicle can be written as

$$a_{lm} = g \sin \theta_l + \mu_{lmax} A_{lm} \cos \theta_l \tag{2}$$

The time values  $t_{la}$  and  $t_{lb}$  in table 2 can be calculated from other parameters shown in figure 2.

$$t_{la} = \frac{a_{lm}}{J_{lmax}}$$
(3)  
$$V_l(0) + \int_0^{t_{lb}} a_l(t)dt = 0 \Longrightarrow t_{lb} = \frac{-V_l(0) + \frac{1}{2}J_{lmax}t_{la}^2}{a_{lm}} + t_{la}$$
(4)

Trailing vehicle :

At time t = 0, the trailing vehicle is accelerating with a constant acceleration  $(a_{fac})$ . After a certain time delay  $(T_1)$ , the driver or the trailing vehicle detects the braking maneuver of the leading vehicle. Then after some reaction delay  $(\tau_B)$ , the trailing vehicle starts to brake with certain jerk  $(J_{fc})$  and deceleration rate  $(a_{fauto})$  at  $t = t_{fa}$ . Since the trailing vehicle may not know that the leading vehicle is executing an emergency stop, its initial braking is done to control the speed and its spacing relative to that of the leading vehicle. After the trailing vehicle or the driver detects that the leading vehicle is in the emergency stopping mode, it brakes with its maximum jerk  $(J_{fmax})$  to achieve the maximum deceleration  $(-a_{fm})$  at  $t = t_{fc}$  until it reaches a full stop. The acceleration profile of the trailing vehicle is shown in figure 4. The parameters in figure 4 are explained in table 3.

Page 42

In table 3, the parameter  $J_{fc}$  and  $a_{fauto}$  indicate a soft braking stage in the stopping maneuver of the trailing vehicle. This braking stage may due to the response of a driver to the brake lights of the leading vehicle, or due to an automatic "soft" braking mode of an Intelligent Cruise Control (ICC) system .<sup>(50)</sup> The maximum deceleration  $a_{fm}$  can be obtained from a similar equation as (2).

The time parameters  $t_{fa}$ ,  $t_{fb}$ ,  $t_{fd}$  and  $t_{fe}$  in figure 4 can be expressed as follows:

$$t_{fa} = T_1 + \tau B \tag{5}$$

$$t_{fb} = \frac{a_{fac} - a_{fauto}}{J_c} + t_{fa}$$
(6)

Parameter Name	Description
$a_{fac}$	the acceleration value at time $t = 0$
$a_{fauto}$	the acceleration value for soft braking
$a_{fm}$	the maximum deceleration of the trailing vehicle
$J_{fc}$	the jerk value for soft braking
$t_{fmax}$	the maximum jerk of the trailing vehicle
$t_{fa}$	the time that the trailing vehicle initiates a braking maneuver
$t_{fb}$	the time that the trailing vehicle reaches $a_{fauto}$
$t_{fc}$	the time that the trailing vehicle starts to brake as hard as possible
$t_{fd}$	the time that the trailing vehicle reaches $-a_{fm}$
$t_{fe}$	the time that the trailing vehicle comes to a full stop without collision

Table 3.	Parameters	for	trailing	vehicle.
----------	------------	-----	----------	----------

$$t_{fd} = \frac{a_{fauto} + a_{fm}}{J_{f \max}} + t_{fc}$$

$$t_{fe} = t_{fd} + \frac{1}{a_{fm}} [V_f(0) + a_{fac} t_{fa} - \frac{1}{2} J_{fc} (t_{fa} - t_{fb})^2 + a_{fauto} (t_{fc} - t_{fb}) - \frac{1}{2} J_{f \max} (t_{fd} - t_{fe})^2]$$
(8)

where  $T_1$  denotes the detection delay of the driver (or the automated vehicle control system). The actuation delay of soft braking is denoted by  $\tau_B$ .  $t_{fc}$  denotes the time when the driver (or the automated vehicle control system) initiates a maximum deceleration maneuver. If the trailing vehicle is driven manually, the value of  $t_{fc}$  indicates how fast the driver perceives and reacts to

an emergency stopping maneuver of the leading vehicle. For automated vehicle driving, this  $t_{fc}$  includes the detection, data processing and actuation delays of the automated vehicle control system.

The above stopping scenario includes many cases that have been considered in the past as well as new cases that are relevant to the operation of automated vehicles. For example, by taking  $J_{lmax} = \infty$ ,  $a_{lm} = \infty$ , and  $t_{lb} = 0$ , we have the so called "brick-wall" scenario. In automated vehicle following operations, advanced hardware and software systems may be implemented to reduce the reaction time of the trailing vehicle. If the automated system of the trailing vehicle can perceive and react to the emergency situation before the soft braking stage ( $a_{fauto}$ ) is achieved, then we can have a scenario that  $t_{fc} < t_{fb}$ . If vehicle to vehicle communication is available then the leading vehicle can communicate its actions to the trailing vehicle. This gives a deceleration function of the trailing vehicle in which  $a_{fauto} = a_{fac}$ ,  $t_{fa} = t_{fb} = t_{fc}$  and  $t_{fa}$  is a small time value.

## Minimum Headway For Collision-Free Vehicle Following

In this subsection we calculate the minimum headway between the leading and trailing vehicles that will guarantee no rear-end collision under the worst case stopping scenario described above. If we let  $S_0$  to be the constant space headway at  $t = 0^-$  (just before the stopping scenario is initiated), then the inter-vehicle distance  $S_r(t)$  is given by

$$S_r(t) = S_0 + [V_l(0) - V_f(0)]t + \int_0^t \int_0 [a_l(s) - a_f(s)]dsd$$
(9)

If  $S_0$  is large enough, we could have  $S_r(t) > 0$ ,  $\forall t \in [0,T]$ , where [0,T] is the time interval of vehicle following under consideration. This situation means that no collision takes place. We are looking for the minimum value of  $S_0$  that will guarantee that a collision does not occur. This minimum value of  $S_0$  is *the minimum space headway for collision-free vehicle following* and is denoted by  $S_{min}$ . It can be calculated by solving for the marginal collision case as follows:

Consider the following minimization problem

$$\min_{t \in [0,T]} \{S_r(t)\} = \min_{t \in [0,T]} \{S_0 + [V_l(0) - V_f(0)]t + \int_0^t \int_0^t [a_l(s) - a_f(s)]dsd\}$$
(10)

and let  $S_{rmin} \triangleq \min_t \in [0,T] \{S_r(t)\}$ . We have marginal collision situation when  $S_{rmin} = 0$ , which indicates that the minimum space headway  $(S_{min})$  for avoiding collision is

$$S_{\min} = \max_{t \in [0,T]} \{ [V_f(0) - V_l(0)]t + \int_0^t \int_0 [a_l(s) - a_f(s)] dsd , 0 \}$$
(11)

Equation (11) indicates that  $S_{min}$  is a function of the initial vehicle speeds ( $V_l(0)$ ,  $V_f(0)$ ) and accelerations ( $a_l(t)$ ,  $a_f(t)$ ). The vehicle accelerations are functions of many variables, including time indices, jerks and acceleration limits as shown in figure 2 and 4. The values of these variables can be easily calculated from Equation (2) through (8).

From the value of  $S_{min}$  we can calculate the minimum time headway  $h_{min}$  in seconds, i.e.,

$$h_{\min} = \frac{S_{\min}}{V_f(0)} \tag{12}$$

Similarly, the minimum k-factor headway  $k_{min}$  is given by

$$k_{\min} = \frac{S_{\min}}{V_f^2(0)}$$
(13)

Figures 5 through 8 show how the minimum headway  $(h_{min})$  changes. The simulation analyses in this subsection are done using the parameter values shown in table 4. Figure 5 shows that  $h_{min}$  is linear with respect to the reaction delay of the trailing vehicle. If advanced technologies (e.g., vehicle-to-vehicle communication) are implemented to reduce  $t_{fc}$ ,  $h_{min}$  can be reduced accordingly.

$V_{l}(0)$	60mph	$A_{fm}$	0.80g
$V_{f}(0)$	60 <i>mph</i>	$T_1$	0.1second
$J_{lmax}$	$72 meter/sec^3$	$t_{fc}$	0.35second
$A_{lm}$	0.85g	τ	0.1second
$a_{fac}$	0.05g	$\mu_{lmax}$	1
$J_{fc}$	$20 meter/sec^3$	$\Theta_l$	0
$a_{fauto}$	-0.20g	$\mu_{fmax}$	1
$J_{fmax}$	$72 meter/sec^3$	$\theta_{f}$	0

#### Table 4. Parameter values for simulation study.



Figure 5. The effect of trailing vehicle reaction time.



Figure 6. The effect of deceleration difference between the leading and trailing vehicles.

Figure 6 shows a non-linear relationship between  $h_{min}$  and the deceleration difference between vehicles  $\Delta A_m$ . If the leading vehicle can brake faster than the trailing vehicle ( $\Delta A_m > 0$ ), then the headway setting should be increased compared to the case that both vehicles have the same maximum deceleration value ( $\Delta A_m = 0$ ) or the trailing vehicle can decelerate faster than the leading vehicle ( $\Delta A_m < 0$ ).

If we ignore the impact of road slope angle and other factors (e.g., loading effect, wind), Equation (2) shows that the maximum deceleration value is proportional to the maximum roadtire friction coefficient. In case that  $\mu_{max} \triangleq \mu_{lmax} = \mu_{fmax}$ ,  $A_{fm} = 0.8g$ , the effect of road-tire friction coefficients can be illustrated by a plot of  $h_{min}$  versus  $\mu_{max}$  (figure 7).  $h_{min}$  has non-linear relationship with the maximum road-tire coefficient. If the leading vehicle can brake faster than the trailing vehicle, and  $\Delta A_m$  is a constant, then  $h_{min}$  increases as the road surface becomes slippery.



Figure 7. The effect of road-tire friction coefficient under constant  $\Delta A_m$ 



#### Figure 8. The effect of initial velocity difference

Figure 8 shows that when a vehicle changes from a high speed lane to a low speed lane, a large headway is needed to avoid collision with its leading vehicle. Also, when a vehicle changes to a lane that has lower speed, a large separation between this vehicle and its trailing vehicle is needed to avoid possible collisions.

## Accident Severity And Headway Selection

The values of vehicle speed (V) and spacing (S) in the case of rear-end collision have a significant effect on the severity of the collision or the level of vehicle damage. Their effect manifests itself in the amount of kinetic energy dissipated at impact. Several studies have been performed to quantify and understand the severity of rear-end collisions. Rahimi *et al.* <sup>(88)</sup> defined a *safety index* as a function of vehicle operating characteristics that are related to the kinetic energy dissipated during the rear-end collision. Calson <sup>(13)</sup>, Lenard <sup>(63)</sup> and Glimm and Fenton <sup>(35)</sup> defined another measure of the dissipated kinetic energy called *accident severity index*. They formulated this accident severity index in different ways. Various functions of collision speeds are used to calculate the accident severity index.

In Equation (10),  $S_{rmin} < 0$  indicates a rear-end collision occurs during the vehicle following operation. If the leading and trailing vehicles have constant jerk and deceleration values during the entire stopping maneuver, then the negative value of  $S_{rmin}$  could be taken as an approximate measure of the kinetic energy of the trailing vehicle at impact. Rahimi *et al.* defined the quantity  $S_{rmin}$  as a safety index .<sup>(88)</sup>

Calson <sup>(13)</sup> defined the accident severity index by a function of both the relative speeds of the colliding vehicles and their absolute speeds. Lenard <sup>(63)</sup> described the severity of an accident as a function of the square of the collision velocity. Glimm and Fenton simplified Lenard's definition <sup>(35)</sup> They expressed the accident severity index ( $S^2$ ) for a platoon of (n+1) colliding vehicles as

$$S^{2} = \sum_{i=1}^{n} \Delta V_{i+1,i}^{2}(T_{coli})$$
(14)

where  $\Delta V_{i+1,i}(T_{coli})$  denotes the relative speed at impact between vehicle *i* and *i*+1. Here, we only consider two-vehicle collisions. By using Glimm and Fenton's definition, the accident severity index can be simplified as <sup>(35)</sup>

$$S^{2} = \Delta V^{2}(t_{c})$$
  
=  $[V_{f}(t_{c}) - V_{l}(t_{c})]^{2}$  (15)

where  $t_c$  is the time when a rear-end collision is initiated. If the headway is less than the minimum safety headway, then a rear-end collision would take place in the worst case stopping scenario (figure 2 and 4).

The accident severity index depends on the headway setting. The accident severity depends on parameters such as maximum deceleration values ( $A_{lm}$ ,  $A_{fm}$ ), trailing vehicle reaction time ( $T_{l}$ ,  $t_{fc}$ ). If not explicitly stated, the simulation analyses in this subsection are done with parameter values shown in table 5.

$V_l(0)$	60 <i>mph</i>	$A_{fm}$	0.70g
$V_f(0)$	60mph	$T_1$	0.1second
$J_{lmax}$	$72 meter/sec^3$	$t_{fc}$	0.85second
$A_{lm}$	0.80g	$ au_B$	0.1second
$a_{fac}$	0	$\mu_{lmax}$	1
$J_{fc}$	0	$\Theta_l$	0
$a_{fauto}$	0g	$\mu_{fmax}$	1
$J_{fmax}$	72 meter/sec <sup>3</sup>	$\theta_{f}$	0

#### Table 5. Parameter values for simulation study



Figure 9. (*h*,  $\Delta V^2$ ) curves for different values of  $t_{hr.}$ 

Figure 9 shows the effect of the trailing vehicle reaction time  $(t_{fc})$ . Figure 9 shows that when the headway setting is very small or large enough, the value of the accident severity index is small. There is a critical headway value  $(h_c)$  between 0 and the minimum safety headway that yields the maximum collision damage. As the reaction delay of the trailing vehicle increases, the potential collision damage increases.

Figure 10 shows the effect of deceleration difference between the leading and trailing vehicles  $(\Delta A_m \stackrel{\Delta}{=} A_{lm} - A_{fm}).$ 



Figure 10.  $(h, \Delta V^2)$  curves for  $A_{lm} = 0.8g$ ,  $A_{fm}$  varies from 0.4 to 0.7g. As  $A_{fm}$  decreases the collision damage increases.

As the value of  $\Delta A_m$  increases, the collision damage increases. When the leading and trailing vehicle have almost the same deceleration capability, the value of accident severity index is bounded by a small value.

The simulation results show that the maximum collision damage increases when the deceleration capability is reduced. If vehicles have and use the same high value of maximum deceleration value, then the probability and severity of rear-end collisions may be reduced.

Using figures 9 and 10 we can come up with the following possible approaches for choosing the minimum safe headway:

## (i) Conservative Approach.

We take into account all reasonable extreme conditions, such as maximum possible acceleration difference minimum possible friction coefficient, reaction times, velocities etc. and use a worse case stopping scenario to come up with a value for the minimum headway. This value will be relatively large when compared with the one an average driver uses in manual driving and will reduce highway capacity. For example, the California Department of Motor Vehicles Driver's Handbook suggests three seconds time headway at lower speeds and greater than three seconds

at higher speeds. The majority of today's drivers drive with headways much less than three seconds.

#### (ii) Vehicle Based Approach.

A vehicle can approximate its braking capabilities with the proper on-board sensors or from the driver's experience. It also knows or can assume a worse possible response for a leading vehicle, i.e., if vehicles can brake with deceleration ranging from 0.3g to 1g the vehicle can assume 1g deceleration for any leading vehicle. Using this knowledge the driver or the computer control system on board of the vehicle could select the appropriate headway for a collision free vehicle following environment. Many drivers follow this approach during manual driving. The more confident a driver feels about the braking capabilities and performance of his vehicle the smaller the headway he may try to maintain during vehicle following.

#### (iii) An Approach Based on Vehicle to Vehicle Communication.

In partial or fully automated vehicles the braking capabilities of each vehicle can be evaluated with proper sensors and communicated to other vehicles. By knowing its own braking capabilities and those of the leading vehicle the computer control system on board the vehicle could choose an appropriate headway. This approach allows vehicles with different capabilities, such as passenger vehicles, trucks and buses to drive on the same roadway. The value of the headway in this case will be smaller than that of approach (i'ii) when applied to similar situations. Vehicle to vehicle communication could also reduce delays and eliminate human reaction times allowing the minimum headway to be reduced further.

#### (iv) Uniform Performance.

Vehicles in an automated mode could be made to have similar performance characteristics by proper design of hardware and software tools. This similarity in performance especially with respect to braking capabilities will make the behavior of vehicles more predictable and lead to lower headways.

## (v) Aggressive Approach: Short Headway Platooning.

Several researchers argue<sup>(111,95)</sup> that since the minimum headway depends on many variables that may not be accurately evaluated, one should either choose approach (i) or choose a small intervehicle spacing of about 1 meter or less so that if a rear-end collision takes place the relative velocity at the time of impact is so small (as shown in figure 9) that the collision will be accommodated without any damage to the vehicle or occupants. For additional safety the vehicles could be organized in platoons of 8 to 25 vehicles with interplatoon spacings chosen according to the conservative approach (i). The short headway platooning approach is pursued by most researchers under PATH and is used to study capacity<sup>(56;89)</sup>, safety and different architectures for AHS. <sup>(45;46;111)</sup>

If a fully automated highway system is to evolve from today's "manual" system then it is unlikely that short headway platooning will be part of any initial deployment. Most likely it will be used towards the final stages of AHS to improve capacity further. The reason is that 1m spacing platooning is a high performance system that requires highly accurate and very reliable technologies that could be very costly to develop initially before the whole idea of vehicle automation is mature and in use. Furthermore the jump from today's driving of relatively large headways to something that appears to the driver or passenger as almost zero headway may pose serious human factors problems and is therefore risky.

We believe that the selection of headway for automatic vehicle following will evolve gradually from that described by approach (i) and (ii) used in today's highways to that described by approach (iii) and (iv), where human time delays will be replaced by the much smaller delays of sensors/actuators and communication devices. Headways based on approach (iii) and (iv) may not provide the capacity improvement that the short headway platooning proponents are advocating but have the potential of improving current capacities by 20% to 200% depending on the architecture of AHS.<sup>(50;89)</sup>

## Steady State Highway Capacity

The velocities of the leading and trailing vehicles are denoted by  $V_l$  (t) and  $V_f$  (t) and the accelerations by  $a_l$  (t),  $a_f$  (t) respectively. The leading vehicle is assumed to have a total length of L meters. The absolute position of the vehicles measured from a common reference point is denoted by  $S_l$  (t),  $S_f$  (t) respectively. The relative distance  $S_r$  between the two vehicles measured from the front of the trailing vehicle to the rear of the leading vehicle is given by

$$S_r(t) = S_l(t) - L - S_f(t)$$
 (15)

The headway between the vehicle can be defined for space or time. The inter-vehicle distance  $S_r$  is the space headway. Time headway (*h*) is the time required for the trailing vehicle to travel through the inter-vehicle distance ( $S_r$ ),  $S_r = hV_{f.}$ 

The *k*-factor headway (*k*) is proportional to the square of the trailing vehicle velocity. The constant of proportionality *k* is referred to as *k*-factor headway,  $S_r = kV_f^2$ . This is also called *safety factor* headway.

Reduced time headways increase the capacity of the highway. The equation for steady-state traffic flow in vehicles/hour/lane is

Flow = 
$$3600 \frac{V_{avg}}{S_d + L}$$

(16)

 $V_{avg}$  is the average vehicle speed in meters/second. *L* is the average length of a car in meters. S<sub>d</sub> is the average gap between vehicles in meters. The equation gives an estimate of steady-state traffic flow over long stretches of highway. It does not take entries and exits into account. Most urban highway trips are short so this model gives an upper-bound on capacity. More realistic traffic flow models show model entries and exits into the traffic stream. Raytheon

(17)

# Section 4: Reliability

As the AHS evolves, the driver's functions will be gradually automated by the vehicle and roadway. The automated functions provided by the vehicle and roadway need to be reliable so that safe and efficient AHS operation can be achieved otherwise automated functions will not be accepted by the general public. A major goal of this report is to characterize the reliability requirements of vehicle and roadway automated functions in the AHS evolution.

## **Reliability Indices**

A system's reliability is defined as the probability that the system will perform its intended function for a specified period of time under a given set of conditions. Indices used to characterize a system's reliability property include reliability function, failure rate and mean time to failure (MTTF), etc. The reliability function of a system is

R(t) := The probability that a system operates without failure for a length of time t.

The failure rate or hazard function  $\lambda(t)$  is defined such that, when the time interval  $\mathcal{E}t$  is small,  $\lambda(t) \mathcal{E}t$  is the probability that the system will fail at some time earlier than  $t + \mathcal{E}t$  under the condition that it has not yet failed at time t

$$I(t)Dt = P\left\{t < t + Dt \mid t > t\right\}$$

The failure rate  $\lambda(t)$  can be related to reliability function R(t) by the following equation

$$(t) = \frac{f(t)}{R(t)}$$
(18)

where f(t) is the failure probability density function of time *t*. Mean time to failure is defined as the expected value of the failure time

$$MTTF = \int_0^\infty t f(t) dt$$
(19)

In addition, MTTF can be related to the reliability function R(t) by

$$MTTF = \int_0^\infty R(t) dt$$
 (20)

A system with a constant failure rate  $\lambda$  gives

$$f(t) = e^{-t}$$

$$R(t) = e^{-t}$$

$$MTTF = \frac{1}{-t}$$

(21)

A system failure stops the system from performing its intended function. In human driving, the potential outcomes of a longitudinal or lateral control failure may be complicated. To use the reliability of today's manual driving to quantify reliability standards for AHS automation in each ERSC, we consider that each failure in manual driving causes a collision.

## Longitudinal Reliability of Human Drivers

United States drivers average around one reportable rear-end collision every 50 years.<sup>(60,28)</sup> Data from Ford Motor Company show that drivers brake about 50,000 times per year on average. The probability of a reportable rear-end collision under the condition that a brake action is needed in manual driving is

P <sub>b</sub>	_	_	$\frac{1}{50 \times 50.000}$	
	=	=	$50 \times 50,000$	
			1	
	=	=	2,500,000	
1		ъ	C	

This gives a reliability for an average driver  $R_b$  of  $R_b = 1 - R_b$ 

 $= 1 - P_b$ = 0.9999996

Human drivers are very reliable in longitudinal control.

The probability that a car will have a rear-end collision in its lifetime is  $P_{re} = 0.2262$ .<sup>(60)</sup> The average lifetime of a car is assumed to be 13 years.<sup>(60)</sup> If we assume a constant failure rate, the longitudinal control must have  $\lambda < 0.02$  year<sup>-1</sup>. If we assume a duty cycle *c*, the failure rate per hour of operation must satisfy

$$< \frac{0.2}{c} \cdot \frac{1}{365 \times 24} \text{ hour}^{-1}$$

If the driver operates the vehicle one hour per day in average, then c = 1/24. In this case, the driver's longitudinal control failure rate is

$$< \frac{0.2}{c} \cdot \frac{1}{365 \times 24} \text{ hour}^{-1}$$

## Lateral Reliability of Human Drivers

The probability that a car will have a lane departure collision in its lifetime is  $P_{re} = 0.09$  <sup>(114)</sup> The average lifetime of a car is assumed to be 13 years <sup>(59)</sup> If we assume a constant failure rate, the reliability of human lane keeping control has  $\lambda = 0.00667$  year<sup>-1</sup> or MTTF = 150 years. If we assume a duty cycle *c*, the human lane keeping failure rate is

$$< \frac{0.09}{c} \cdot \frac{1}{365 \times 24} \text{ hour}^{-1}$$

For c = 1/24, the reliability of manual lane keeping control can be approximated by  $< 1.8265 \times 10^{-5} \text{ hour}^{-1}$ 

The probability that a car will have a lane change/merge collision in its lifetime is  $P_{re} = 0.015.^{(113)}$  If we assume a constant failure rate, the reliability of human lane changing control has  $\lambda = 0.00125$  year<sup>-1</sup> or MTTF = 800 years. If we assume a duty cycle *c*, the human lane changing failure rate is

$$< \frac{0.015}{c} \cdot \frac{1}{365 \times 24} \text{ hour}^{-1}$$

For c = 1/24, the reliability of manual lane keeping control can be approximated by  $< 3.425 \times 10^{-6} \text{ hour}^{-1}$ 

## **Reliability Requirements of Automated Functions**

Automated functions provide no benefits unless they are reliable than today's manual driving. By quantifying the reliability of human drivers, we will be able to estimate the required reliability for automated functions using reliability indices, such as failure rate and mean time to failure. The results of the previous sections provide measures to determine the required reliability when longitudinal and lateral control is automated.

Automation is mainly achieved by automatic control systems from the vehicle and roadway. An automatic control system needs to be design in such a way that it fails in a safe manner. However, from the system design point of view, reliability indices provide little help in obtaining fail-safe system designs. To design a fail-safe system, we will first derive the reliability functional requirement. The reliability functional requirement of an automatic system

is concluded by analyzing the interactions between the driver and the system. The requirement guarantees the safety of the driver in case the system fails.

The reliability functional requirement of a system can be used to evaluate the necessity of redundant components and to come up with system design frameworks. This gives us the knowledge of the complexity and difficulty in designing an automatic system. By analyzing reliability requirements for vehicle and roadway functions in each ERSC, we can estimate the technical and cost gaps between ERSCs.

# Section 5: ERSC 1 Analysis

## Description of ERSC 1

In ERSC 1 the roadway provides a dedicated lane for the use of the automated vehicles. The lane may be isolated and accessed by a dedicated ramp or it could be next to a manual lane and accessed at designated points or at any point along the manual/dedicated lane boundary. The roadway senses traffic flow (average speed, density) and environmental conditions on the dedicated lane and provides target speed commands, minimum time headway recommendations and traffic status information to the vehicles.

The automated vehicle can maintain a selected headway and speed relative to the preceding vehicle by using a computer control system to control the throttle and the brake. Automatic braking is limited to soft braking that is applied when the engine torque is not sufficient to maintain the selected headway. It does not include hard braking for emergency stops or other situations. It receives target speed commands, minimum time headway recommendations and traffic status information from the roadway. It responds to the speed commands automatically in a smooth way as long as the resulting headway is not smaller than the one selected by the driver. If the target speed is larger than the one the driver feels comfortable with, the driver may be required to exit the lane. The vehicle responds to the recommended minimum time headway provided it is larger than the selected one. It responds to driver commands for changing the headway, and setting the cruising speed. It also enables and disables the speed and headway and operational status and identification to the roadway to be used for traffic flow control purposes. If the SHMS fails then the vehicle fall-back mode allows the driver to take over this function and continue the trip until the next exit.

The driver is responsible for driving the vehicle into the dedicated lane with the aid of the blind spot warning system. Once in the lane the driver switches the SHMS on and sets the desired headway for vehicle following or the cruising speed. The driver supervises the SHMS and overrides it when hard braking is required for emergency stops and other situations. The driver is aided by the rear-end collision warning in avoiding rear-end collisions. The driver is responsible for lane keeping, lane changing and lateral collision avoidance. The driver is also responsible for choosing the route of the vehicle and conforming to traffic regulations. In case of malfunction of the SHMS or at the end of the trip the driver drives the vehicle out of the dedicated lane by using the next available exit with the aid of the blind spot warning system.

A list of the roadway, vehicle, and driver functions that are to be studied for ERSC 1 is given in table 6. We will discuss each of these functions in detail in this section.

Function/Section	Requirement	Driver/Vehicle/Roadway Tasks
Speed and Headway Maintenance	<ul> <li>Maintain the selected cruising speed when no vehicle is ahead.</li> <li>Maintain a selected headway for vehicle following.</li> <li>Smoothly transition between vehicle following and cruising modes</li> </ul>	<ul> <li>Respond to driver commands for setting the cruising speed and minimum headway and maintain that cruising speed and chosen headway in a smooth manner</li> <li>Respond to driver commands for enabling and disabling the SHMS safely</li> <li>Respond to roadway commands for following target speeds and increasing headway in a smooth manner</li> </ul>
Rear-end Collision Warning	Warn the driver of a potential rear- end collision due to moving or stationary obstacles in the lane on time without false alarms	<ul> <li>Detect potential rear-end collisions</li> <li>Have a high detection rate and a low rate of false or nuisance alarms</li> <li>Respond to commands from the driver to adjust the sensor threshold</li> <li>Send driver warnings of rear-end collisions</li> <li>Communicate braking data from the preceding vehicle to reduce false alarms</li> </ul>
Blind spot warning	Warn the driver of moving or stationary obstacles in the vehicle's blind spot	<ul> <li>Detect objects in the blind spot on both sides of vehicle</li> <li>High detection rate and a low rate of false or nuisance alarms</li> <li>Warn driver in time to avoid collision</li> <li>Driver enables blind spot warning</li> <li>Respond to driver command to adjust warning thresholds</li> </ul>
Driver Interface	<ul> <li>Driver should be able to:</li> <li>» override the speed &amp; headway maintenance function</li> <li>» adjust headway and set cruising speed</li> <li>» adjust warning threshold</li> </ul>	<ul> <li>Adjust thresholds on warnings</li> <li>Enable/Disable automatic functions</li> <li>Adjust headway and cruising speed</li> <li>Interface should be simple to understand and use</li> </ul>
Roadway/Vehicle Speed & Headway Commands	Senses vehicle average speed and density and environmental conditions then sends commands to smooth traffic flow	<ul> <li>Roadway measures the average speed and traffic density, and assesses environmental conditions, and calculates vehicle target speeds and minimum safe headways</li> <li>Vehicles may display relevant traffic information to driver</li> </ul>
Fall-back Mode to Manual Driving	Driver should be warned of system failures or degradation in time to recover from dangerous situations	<ul> <li>sense status of different vehicle control channels</li> <li>Safely transfer control to the driver if SHM function fails</li> </ul>

nents.
n

## Performance Requirements

## Speed and Headway Maintenance

The speed and headway maintenance (SHM) function should maintain the selected cruising speed when no vehicle is ahead. It should maintain the selected headway during vehicle following. The SHM function operates by measuring the relative speed and spacing, then sending commands to the throttle and brake actuators. Figure 11 shows the structure of the SHM function. This section will discuss the sensor and actuation requirements, control algorithms, selection of a safe headway, and driver/roadway interface for the SHM function.



Figure 11. Block diagram of the speed and headway maintenance function. The sensors measure the relative speed and spacing of the vehicles, then the controller sends commands to the brake and throttle actuators so that the vehicle maintains the selected speed and/or headway.

## Sensor Requirements

To maintain a target velocity the vehicle must be able to measure its own speed. The sensor measures speeds between 5 and 100 mph. The sensor needs to be accurate within 2-5 mph since humans cannot perceive small changes in velocity.

To maintain a constant time headway the vehicle must be able to measure the distance to the target vehicle ahead. The vehicle must also estimate or measure the closing rate between vehicles. Figure 12 shows the sensor region for headway maintenance. The single beam sensor region is sufficient for vehicle following in the lane and for detecting stationary obstacles. The ideal sensor region shows the sensor requirement for detecting obstacles other than vehicles in the lane such as motorcycles, animals, etc. and vehicles during cut-in situations.



Figure 12 Sensor requirements for range and closing rate detection for the speed and headway maintenance system.

There are several competing requirements that determine the maximum sensor range. The sensor range needs to be the maximum value for these requirements. <sup>(29)</sup> Since drivers may rely on the SHM function to initiate braking when an obstacle appears in the range of the sensor, the sensor should detect obstacles at a range of at least 3 seconds. The *California Driver's Handbook* states that 3 seconds and higher is a safe time headway for human drivers. If we assume a speed of 60 mph (26 meters/s) this gives a distance of 78 meters. Therefore the vehicle range sensor should work over a distance of at least 80 meters. The second requirement is that the vehicle be able to stop when there is debris or another vehicle stopped in the roadway ahead. This scenario is known as the "brick wall" scenario. Rear-end collisions with a stationary vehicle outnumber rear-end collisions where the lead vehicle is moving by 2:1.<sup>(60)</sup> Figure 13 shows some range curves for the brick wall scenario for different velocities and a human reaction time  $T_{hr}$  of 2 seconds.<sup>(1;64)</sup> For example, at 60 mph we get a range of 100 m for a vehicle with a maximum braking capability of -0.55 g. Table 8 gives the parameters used in this calculation.



Figure 13. The maximum sensor range with respect to the maximum deceleration of the vehicles for two different velocities. These values come from the "brick wall" scenario when the lead vehicle or debris is stopped on the highway.

The sensor must cover the lane so it does not miss the target vehicle. But it must also minimize interference from adjacent lanes which means that the sensor beam cannot spread out too much or it must use multiple beams. The sensor must be able to track the vehicle around highway curves without losing the target.

The vehicle must also detect or estimate the closing rate between itself and the target vehicle. This may be done directly with a pulse-doppler radar or it may be estimated by taking the derivative of the distance measurement over time. The sensor accuracy needs to be with 5-10% again because drivers cannot tell the difference between small changes in closing rate.

The most common type of distance sensor measures the time-of-flight from the time that a pulse is sent out to the time that the pulse returns. The distance is the (time-of-flight)\*(pulse velocity)/2. These types of sensors include radar, ultrasound and laser rangers. These time-of-flight sensors can measure the range, relative speed, and angular position of the target vehicle. Other sensors use triangulation techniques such as stereo to measure the distance to the vehicle ahead. These sensors include vision-based systems and radar.<sup>(68,86)</sup> The time-of-flight sensors

are commercially available for automotive use today, (12,33,76) so they appear to be the best solution for ERSC 1.

Radar systems are time-of-flight systems. Range gates in the receiver look for the return pulse in different time intervals. The receiver can modulate the pulse in frequency to estimate the closing rate of the target. Radars are also long range and accurate. Their performance is not degraded by rain or snow. Since radars have a long range, they also have problems with interference. The beam of a radar spreads out as the pulse propagates as shown in figure 12 and it can hit vehicles or signs in other lanes or along the edge of the roadway. A narrow beam can reduce interference. Reliable performance may require a scanning or multiple beam radar that searches the lane and rejects interference.

Laser radars use high frequency pulses in the *near-infrared* region. They give extremely accurate range measurements. However their narrow beam width can make it difficult to track targets around curves and at large ranges. The performance of laser radars is degraded due to dirt on the transmitter or receiver, heavy rain, thick fog, and car exhaust emissions.<sup>(76)</sup> Experiments have shown that the maximum detection range of a laser radar can be reduced by 30% in wet weather compared to dry weather.<sup>(76)</sup>

The vehicle may also use combinations of different sensors to improve performance. This can reduce interference by taking measurements in multiple frequency bands and increase reliability by providing independent measurements of range.

## Control Algorithms

The vehicle must respond quickly and smoothly to commands from the driver and roadway. The controller must also limit the jerk and accelerations in normal operation. Humans find longitudinal accelerations above 0.17 g and jerks above 0.2 g/s uncomfortable at low speeds.<sup>(15)</sup> At high speeds the deceleration should be kept below 0.1g with jerk as low as possible. The maximum deceleration (or soft braking) should be limited to around 0.2 g with low jerk.<sup>(15)</sup>

The controller must switch smoothly between the vehicle following and cruise modes and avoid sudden accelerations when the preceding vehicle exits the dedicated lanes. Tests on an intelligent cruise control system by Fancher et al showed that abrupt accelerations by the automated vehicle when the sensor lost its target or the vehicle ahead change lanes made drivers feel uncomfortable and unsafe.<sup>(26)</sup> Thus the controller should not accelerate quickly. The controller must also switch smoothly between the throttle and brake without high frequency oscillations.

Studies on automatic vehicle following go as far back as the mid-70's where simple point mass vehicle models were used to design throttle and brake controllers,  $^{(20,87)}$  Hedrick et al  $^{(42)}$  and Sheikholesam and Desoer  $^{(93)}$  designed controllers that maintain constant intervehicle gaps. When three or more vehicles begin to follow each other at a constant intervehicle spacing the vehicle dynamics become coupled. Controller and actuator delays can cause instability unless there are vehicle-to-vehicle communications.<sup>(87)</sup> If the vehicles follow a constant time headway

policy instead, the stability of the vehicle dynamics can be achieved without relying on vehicle-to-vehicle communication.<sup>(18)</sup>

Researchers at the University of Southern California have designed several controllers for the SHM function. Ioannou and Xu designed several throttle and brake controllers for SHM.<sup>(51)</sup> These controllers showed an accuracy of 5 meters for vehicle following in simulations and real-world tests. The controller limited the acceleration and the braking to keep the ride smooth. The actuators had delays of 0.25-0.5 seconds. A logic switch ensured smooth transitions between the brake and throttle by inserting a neutral zone where neither the throttle nor the brake was active. An alternative controller is a fuzzy rule-based controller.<sup>(25)</sup> The fuzzy controller also limited acceleration and the change in throttle angle. Simulation results for a similar sensor gave an accuracy of 3 m. Tests on I-15 in California gave an accuracy of 3-5 meters on a hilly stretch of freeway. On a steep downhill the vehicle did not have enough torque to maintain a constant gap and the driver had to override the SHM function.

## Actuator Requirements

The brake and throttle actuators must respond quickly to commands from the controller. Faster actuators may give better control. If the delay is too long then vehicle following may be unsafe if there are many vehicles in a row following one another.<sup>(50)</sup> Tests show that delays less than or equal to 0.3 seconds are sufficient for ERSC 1.<sup>(51,25)</sup> Table 7 summarizes the requirements for the SHM function.

Requirement	Value	Background
Distance Sensor	>100 m	Value comes from a "brick wall" scenario, the warning system gives the driver 2 seconds to react to a warning to brake, see figure 13 for details
Closing Rate Sensor	Measure values between [-160,160] km/hour, with accuracy of 3-5 km/hr	Must be able to measure all velocity differentials between vehicles. System accuracy bounds are limited by human perception
Velocity Sensor	0-160 km/hour	Measure all possible speeds in dedicated lane
Controller distance accuracy	3-5 meters	Humans do not measure distance between vehicles accurately
Acceleration Limit	+/- 0.15 m/s <sup>2</sup>	Human factors studies show that drivers find accelerations and decelerations out of this range uncomfortable. <sup>(15)</sup>
Jerk Limit	+/- 0.2 g/s	Human factors studies show that drivers find jerks out of this range uncomfortable. <sup>(15)</sup>
Throttle Actuator Delay	<0.3 seconds	Controller studies show that the required SHM system accuracy can be achieved with delays in this range $.^{(51,25)}$

Table 7. Summary of requirements for speed and headway maintenance system.

Safe Headway Selection

The safe headway between vehicles depends on driver reaction time, the braking capabilities of the lead and following vehicles, and environmental conditions. Table 8 gives the variables used in the minimum time headway calculations for ERSC 1. Unless noted these values are used for all the figures. Figure 14 shows the minimum time headway for different maximum deceleration values of the follower car. When the follower vehicle decelerates faster than the lead vehicle the minimum time headway decreases. When the follower vehicle decelerates slower than the lead vehicle the minimum time headway increases. This suggests that proper minimum time headways should be set as if the lead vehicle has a high maximum possible acceleration.

Variable	Description	Value
Name	-	
$V_l, V_f$	Initial velocity of the lead and following vehicles	100 km/hr
$J_{lmax}, J_{fmax}$	Maximum jerk of the lead and following vehicles	72 m/s <sup>3</sup>
lmax, fmax	Maximum road-tire friction coefficient of the lead and following vehicles	1,1
$A_{lm}, A_{fm}$	Maximum deceleration of the lead and following vehicles respectively	0.85 g,0.8 g
$a_{fac}$	Acceleration under vehicle following situation	0.05 g
<i>a<sub>fauto</sub></i>	Acceleration value for soft braking	-0.20 g
$J_{fc}$	Jerk value for soft braking	20 m/s <sup>3</sup>
b	Braking actuation delay	0.1 s
<i>T</i> ( <i>1</i> )	Detection delay of the automated vehicle headway system	0.3s
T <sub>hr</sub>	Human reaction time from when vehicle warns driver of a potential rear-end collision	1.5 s

 Table 8 Variable values used in the simulations in section 5-1 for ERSC 1.



Figure 14. Minimum safe time headway with respect to differences in the maximum deceleration between vehicles  $\not EA_m = A_{lm} - A_{fm}$ . A negative acceleration value means that the follower car can brake faster than the lead car. T1 is the time that the following vehicle detects the lead vehicle deceleration and initiates soft braking.

Figure 15 shows the minimum time headway for different human reaction times. Longer reaction times lead to larger minimum time headways. Human factors studies and guidelines suggest that a human reaction time of 1.5 - 2 seconds is sufficient for most people to react to an unexpected braking situation <sup>(64,1)</sup> The recommended minimum headway for collision-free vehicle following needs to account for human reaction time.





## Driver/Roadway Interface

The vehicle responds to driver commands for switching the SHM function on and off, selecting the minimum headway and the cruising speed, and overriding the SHM function. It also responds to roadway speed commands, headway recommendations, and displays traffic status information received from the roadway to the driver.

The driver turns the SHM function on and off. The SHM function may be turned on before the driver enters the dedicated lane or once the driver has successfully merged into the lane. Tests on a vehicle with one experimental SHM system show that most drivers do not feel safe using this automated system while merging.<sup>(26)</sup> This switch needs to be standardized and clearly marked so that the vehicles can begin operations quickly and easily. The vehicle must also allow

the driver to override the SHM function by braking since the driver is responsible for emergency stops.

The roadway may send target speed commands directly to the vehicle or it may communicate with the driver who then sets the vehicle's speed. The roadway can send the headway and speed to the vehicle with variable message signs or directly to the driver with in-vehicle signing. This system implementation depends on driver cooperation and attention to work well. If the traffic and weather are variable, this implementation could impose a heavy workload on the driver.

#### Issues and Risks

1 The SHM function does not have to be very accurate in this ERSC (5-10%) since it is limited by human perceptions and reaction times. This suggests that SHM systems at this level may come from the

intelligent cruise control systems planned by several companies .<sup>(74,26)</sup> The next ERSC requires more accurate SHM. The components for a more accurate SHM function need to be designed with AHS requirements in mind instead of taking commercial technology off the shelf.

#### 2 Controller Algorithms

The controller algorithms need to smoothly adjust the brake and throttle to avoid sudden accelerations when the vehicle ahead leaves the lane. Driver tests indicate that drivers feel unsafe when the controller

suddenly accelerates.<sup>(26)</sup> The controller needs to use alternative algorithms that limit accelerations.<sup>(51,25)</sup>

#### 3 Electromagnetic Interference

Full deployment of ERSC 1 means that most vehicles will have active sensors such as radar. When multiple radar operate in a small area at similar frequencies the radar may interfere with one another. This interference could lead to shorter sensor ranges and poor performance.

#### 4 Driver Interface

Research needs to be done on the driver interface to the SHM function. This device may have separate controls for enable/disable and minimum time headway. It may be too much of a burden on the driver to expect him/her to change lanes and start the SHM immediately. This function may require the driver to preset an initial target speed and minimum headways when he/she starts the vehicle.

## Rear-End Collision Warning Function

A rear-end collision warning system needs to be able to accurately predict the likelihood of a collision. If a collision is likely, then the system warns the driver to respond to the threat. Analysis of minimum time headways between vehicles suggests that small reductions in driver reaction times can dramatically decrease the minimum safe headway. Figure 15 shows the effect of driver reaction time on the minimum time headway. Reductions in the reaction time may be achieved by warning the driver. Many accidents are due to failures in human information processing.<sup>(59)</sup> In many cases the other vehicle was clearly visible. This again suggests that a rear-end collision warning system may increase driver safety.

The rear-end collision warning (RECW) function should warn the driver of a potential rear-end collision due to moving or stationary obstacles in the lane. This function assesses the potential for rear-end collision using vehicle characteristics and driver reaction time, then warns the driver if necessary. Figure 16 below shows the principle blocks of the RECW function and their

relationship to one another. This section will discuss the sensor requirements, warning algorithms, vehicle-to-vehicle communications, and driver interface for the RECW function. Table 9 summarizes the requirements for the rear-end collision warning system.



Figure 16. Block diagram for the rear-end collision warning system. The system senses the distance and closing rate of the target vehicle. Then the warning algorithm computes the headway between the vehicles and warns the driver if necessary.

Requirement	Value	Background
Sensor Range	>100 m	Value comes from a "brick wall" scenario, the warning system gives the driver 2 seconds to react to a warning to brake, see figure 13 for details
Obstacles detected	All types of motor vehicles and debris	Sensor must detect obstacles in the lane and warn the driver to brake.
Cautionary Warning	0-15° from line of sight; visual warning	Warning must be non-intrusive; warnings should be near driver's primary line of sight for the task <sup>(64)</sup>
Imminent Warning	Dual warning modes (Auditory and visual); >1.4 s before collision	Warning needs to give driver enough time to react to possible collision. <sup>(64)</sup> Studies show that most (95%) drivers react within 1.4 s for braking

#### Sensor

The rear-end collision warning system can use a velocity sensor, a ranging sensor and a closing rate sensor to recognize potential collisions. These inputs allow the vehicle to estimate the time to collision that is the gap distance over the closing rate, the time headway, and the safe stopping distance. Sensors that estimate the roadway coefficient of friction may also improve the performance of a RECW system.<sup>(39)</sup>

The vehicle must be able to stop when there is debris or another vehicle stopped in the roadway ahead. This scenario is known as the "brick wall" scenario. Rear-end collisions with a stationary vehicle outnumber rear-end collisions where the lead vehicle is moving by 2:1.<sup>(60)</sup> Figure 13 shows some range curves for this requirement for different velocities and a human

reaction time  $T_{hr}$  of 1.5 s. For example, at 60 mph we get a range of 100 m for a vehicle with a maximum braking capability of -0.55 g. Table 7 gives the parameters used in this calculation.

The sensor must also detect "cut-ins" that may change lanes in between vehicles. A single beam sensor does not see the cut-in until it enters the beam shown in figure 11. The speed of detection depends on the beam width  $_b$  of the radar, the lateral velocity of the vehicle that is cutting-in,

and the distance of the cut-in from the vehicle. A typical lane change takes about 2-6 sec.<sup>(66)</sup> This gives a bound of less than 0.5 seconds for the detection time between the ideal forward looking sensor and the single beam sensor. In the case of manual braking, the detection time difference for the ideal sensor and a single beam sensor will not reduce the accident severity very much as shown in figure 17. The problem of "cut-ins" may preclude the continuous entry option for check-in.<sup>(108)</sup>



Figure 17. Changes in the detection time for lane change/merge collisions have a small effect on the accident severity. T1=0.1s shows the case for the ideal sensor. T1=0.4s shows the case for the single beam sensor.

#### Warning

The warning algorithm for rear-end collision warnings is based on the time-to-collision (TTC) or minimum headway between the lead and following vehicles. The response time requirements for the TTC are estimated based on the American Association of State Highway and

Transportation Officials (AASHTO) model for stopping sight distance.<sup>(1)</sup> The model assumes that the distance required to brake to a stop in reaction to an obstacle ahead is the summation of the distance traveled during the perception-reaction time of the driver and the distance required to decelerate to a complete stop. This gives a 3-5 second TTC criteria based on a 2.0 second driver reaction time.<sup>(64)</sup>

A warning criterion can also be based on the time headway separation. The minimum headway depends on the driver reaction time, the relative deceleration difference between vehicles and the relative velocity difference. Figure 18 shows the effect of driver reaction time on the collision severity. The reaction time of the driver varies with traffic conditions, past system performance and driver attention. Lerner et al recommend a time headway threshold of 1.0-1.5 seconds.<sup>(64)</sup> This warning is based on a study of braking responses in reaction to a lead vehicle's brake lights that found that very few drivers reacted by braking if the headway exceeded 1.4 seconds.<sup>(106)</sup>



Figure 18. Changes in the human reaction time for lane change/merge collisions affect accident severity.  $T_{hr}$ =0.7s shows the best case for human reaction time to an unexpected hazard.  $T_{hr}$ =2.0s shows the worst case for human reaction time to an unexpected hazard.

The collision warning system must warn the driver when a collision may occur without false alarms. High rates of false alarms may cause the driver to ignore the warnings or turn off the system. The follower vehicle must detect that the lead vehicle is starting to brake hard. This can be done either with the detection of a change in the relative velocity or acceleration between the

vehicles or with the reception of an emergency transmission from the leading vehicle (a sort of electronic brake light) A deployed system may require both measurements for an emergency warning.

Human factors studies show that there should be multiple levels of warning.<sup>(98)</sup> The severity and intensity of the warning depends on the immediacy of the threat. Lerner et al <sup>(64)</sup> suggest two levels of warning: cautionary and imminent. The warnings end when the driver takes action to end the threat.

A cautionary crash avoidance situation exists when another vehicle is sensed in front of the vehicle and the time-to-collision is estimated at greater than 3-5 seconds. The cautionary warning gives the driver time to recognize a hazard and correct the problem. The warning needs to give the driver informational feedback on his /her headway maintenance performance.<sup>(64)</sup>

An imminent crash avoidance situation exists when a vehicle or object is sensed the sensor detection zone with a short time-to-collision or a short time headway. The goal of the warning is to make the driver react immediately. The AASHTO models use a driver reaction time of 2 seconds for this warning. Other studies find that most drivers respond to warnings within 1.4 seconds. We use 1.5 seconds in our analyses.

The collision warning system needs to recognize potential collisions accurately. No warning system will ever be perfectly accurate so it is important to investigate how false alarms affect the driver. The number of false alarms depends on the accuracy of the collision detection system. Communication of braking data between vehicles can help lower the rate of false alarms while maintaining a high detection rate.

## Vehicle to Vehicle Communications

The vehicle needs to estimate the TTC or minimum headway as described above. When the car ahead begins to brake with a deceleration,  $a_{bmax}$ , the TTC decreases. However the measurements for the distance and closing rate between the vehicle itself and the target vehicle can be quite noisy and inaccurate for quick estimates of the TTC. The RECW function must distinguish between sensor measurement noise and a true braking maneuver. Appendix A gives the derivation of this relationship between detection time, false alarms, and detections.

Figure 19 shows the relationship for  $P_D$  and  $P_{FA}$  with respect to d. The curves change with

 $=\infty$ . This corresponds to the case where the receiver always decides no acceleration. =0 corresponds to the case where the receiver always decides that acceleration is present. In this case the cost for a false alarm is about the same as the cost of a correct detection . This gives a

around 1. This shows that as the time available to detect a deceleration ahead increases for constant noise and accelerations, the performance of the detector improves. A good rear-end collision detection system needs a low false alarm rate and a high probability of detection <sup>(64)</sup> This may require a *d* on the order of 4 or higher. For example if the maximum deceleration of the lead vehicle is 8 m/s<sup>2</sup>and the noise deviation is 3-5 m/s for the sensor and controller noise then t > 0.5 s gives d=4. This means that we require at least a half second after the vehicle ahead has started braking to make a good decision. Note that the brake actuator delay (0.2-0.3 s)

increases the total time delay to around 0.75 seconds before the RECW warns the driver of a potential collision.



Figure 19. The probability of correctly deciding that the vehicle ahead is decelerating depends on the space between the two signals d. As d increases the detection-false alarm trade-off improves.

A vehicle-to-vehicle communication system sends braking data directly between vehicles. This lets the vehicle warn the driver quickly and reliably that an emergency is needed just as brake lights reduce human response times. The probability of false alarm is low. As the lead vehicle brakes it sends its brake data directly to the vehicle behind. This braking data only needs to say that the vehicle in front is braking hard. This communications system can have a very short delay on the order of 0.05 to 0.1 seconds.

Each vehicle sends its deceleration to the car behind it. This lets the car behind anticipate any braking before changes in the distance or velocity can be measured. This data may be in the form of brake line pressure or an accelerometer measurement. Studies by Ford and USC show that before skidding or for low line pressures brake line pressure is linearly proportional to brake force.<sup>(117)</sup>

$$F_b \approx P_b$$

The brake line pressure anticipates the actual deceleration. These measurements may be filtered to get a more accurate estimate of the deceleration. An accelerometer measures the actual deceleration of the vehicle. Vehicle-to-vehicle communications may also send data on the
# Task D

maximum braking capabilities of the target vehicle. This data could be computed from on-line tests or by the factory where the vehicle was built. Figure 20 shows how accident severity is affected by the deceleration capabilities of the vehicles.



Time Headway (s)

Figure 20. The deceleration difference between the vehicles  $\pounds A_m = A_{fm} - A_{lm}$  affects accident severity. As  $\pounds A_m$  increases the potential accident severity increases.

Other forces that decelerate a vehicle are the aerodynamic drag coefficient, the friction between the car and the roadway, and the slope of the roadway. The aerodynamic drag stays constant over the life of the vehicle. It may be compensated for by calibrating each car model. The friction between the car and the road depends on the roadway, the tire condition, and the mass of the vehicle and passengers.<sup>(6)</sup> The road-tire friction coefficient may be estimated by measurements from an anti-lock braking system sensors .<sup>(39)</sup> This measurement coupled with an on-line estimate of the vehicle's maximum deceleration can give the vehicle's estimated braking capability.

The maximum deceleration for most passenger cars is around 0.9 g. This gives ten levels of quantization of braking data. This translates into 4 bits  $(16=2^4)$  of information. The maximum braking capability of the vehicle may also be encoded in a 4 bit field. This data may require online calibration to ensure that false braking signals are not sent out. The message between vehicles needs a length of around 50 bits in the worst case. This gives 16 bits for

synchronization.<sup>(100)</sup> 8 bits each for the source and destination IDs if local addressing for a cell is used. The data field it 4 bits. 16 bits of error correction can help detect transmission errors.<sup>(100)</sup> If directional systems such as infrared (IR) or radio are used then the source and destination fields may not be necessary.<sup>(6)</sup>

The communications system must transmit over a range of at least 40 meters in a single lane. This allows the driver a 1.4 second reaction time to a rear-end collision warning at 60 mph. The signal delay is 0.05 seconds. The transmission rate is 20\*M bits per second with a message length of M bits. The worst case transmission rate is 1000 bits per second. The transmission delays should be less than 20 ms since the brake data is time-critical. If the rear-end collision warning system decision time is 0.1 seconds, the rear-end collision warning system receives at least two messages before an emergency stop. The probability of message error ( $P_m$ ) should be very low. Research at California PATH has produced IR systems that meet this data rate and accuracy requirement.<sup>(31)</sup> The tested system can transmit at 19.2 Kbits/second over a range of 30-40 meters.<sup>(31)</sup> As the distance between vehicles decreases the allowable data rate increases. Driver Interface

The rear-end collision warning function provides levels of warning to the driver and provides a detector sensitivity adjustment for the driver. The different warning levels allow the vehicle to warn the driver without intrusive nuisance alarms. The detector sensitivity adjustment lets the driver adjust the range of the sensor to reduce sensitivity and false alarm rates.<sup>(64)</sup> Human factors studies show that there should be multiple levels of warning.<sup>(98)</sup> The severity and intensity of the warning depend on the immediacy of the threat. Lerner et al <sup>(64)</sup> suggest two levels of warning: cautionary and imminent. The warnings end when the driver takes action to end the threat.

The cautionary warning should be visual and located within 15-30<sup>th</sup> above the driver's line-ofsight for ordinary driving. <sup>(64)</sup> Visual signals are more detectable if they are near the center of the line-of-sight. The warning needs to give the driver informational feedback for headway maintenance.<sup>(37)</sup> A headway display could provide this type of feedback with a heads-up display.<sup>(116,83)</sup> This type of display could also assist the driver in selecting a safe headway.

The imminent rear-end collision warning should use two types of warnings. Lerner et al suggests a combination of auditory and visual warnings.<sup>(64)</sup> The auditory warning can attract the driver's attention if the visual warning is ignored. Other studies suggest that haptic warnings such as vibrations on the gas pedal or in the driver's seat may be effective for imminent collision warning.<sup>(105)</sup> However since ERSC 1 is designed to be "feet-off" during normal operations, haptic warnings may not be effective.

The rear-end collision warning system may have an adjustable threshold. This lets the driver reduce the sensor range to avoid excessive false alarms. For example, heavy traffic may cause the rear-end collision warning system to trigger frequently. Also the human reaction time of 1.5 s used for calculating the warning threshold is a worst case number. If a driver perceives himself as faster than average, then he/she may adjust the warning threshold at his own risk.

#### **Issues and Risks**

1 Sensor Capabilities

The RECW system may increase safety and reduce certain types of rear-end collisions by decreasing driver reaction time. But there are sensor and human factor issues that still must be resolved. The sensors must detect small vehicles or debris, such as motorcycles with few nuisance alarms. It will be very difficult for radar and ultrasonic sensors to decide if a motorcycle is present. A video system may be able to see small vehicles better during daylight. The video system sees what the driver sees so it may be more natural for human drivers to use. The performance of vision based systems may degrade at night and in rainy weather plus it is more expensive and complicated than radars of ultrasonic systems.

#### 2 Warning system design

First the designers need to decide when the RECW function should be turned on: always on, only when the SHM function is on, or on only at higher speeds. The warning thresholds for emergencies and driver awareness must also be set. The system must avoid nuisance alarms, yet detect vehicles accurately. The types of alarms and warnings must also be researched. The emergency alarm needs to get consistent and fast driver responses.

#### 3 Multiple Warnings

ERSC 1 introduces multiple types of driver warnings for blind spot and rear-end collision avoidance. Research needs to be done on the interaction and prioritization of these warnings for the human driver.

# Blind Spot Warning

Most (95%) of lane change/merge crashes are angle or sideswipe collisions.<sup>(113)</sup> The driver of the merging car either did not see the other vehicle or misjudged the distance between the

vehicles. These crashes usually happen in dry, clear, daylight conditions.<sup>(113)</sup> The driver "did not see" the other vehicles until too late. The blind spot or "lateral encroachment" warning system on the vehicle detects vehicles in adjacent lanes that are traveling in the same

direction.<sup>(59)</sup> The blind spot detection warning system has a sensor and a system that warns the driver. Figure 21 shows a block diagram of the blind spot warning system.



#### Figure 21. Block diagram for the blind spot warning system.

The blind spot warning system helps avoid "proximity" crashes in which the relative longitudinal locations and velocity differences between the vehicle and object are small.<sup>(21)</sup> This system

Requirement	Value	Background
Sensor Range	3-4 meters laterally 6-10 meters on side of vehicle	Sensors must cover both sides of the vehicle since accidents occurs equally on both sides of vehicle. Sensor looks for vehicle in blind spot. $(64,21)$
Obstacles detected	All types of motor vehicles	Sensor must detect small vehicles such as motorcycles and large vehicles such as trucks and buses. <sup>(64)</sup>
Cautionary Warning	0-15° from side mirror; visual warning	Warning must be non-intrusive; warnings should be near driver's primary line of sight <sup>(64)</sup>
Imminent Warning	Dual warning modes (Auditory and visual); >1.2 s before collision	Warning needs to give driver enough time to react to possible collision $^{(64)}$ Studies show that most(95%) drivers react within 1.2 s $^{(21)}$ for steering back into their lane

Table 10	Summary	of requirem	ents for blind	snot warning	system
Table 10.	Summary	of requirem	ents for philu	spot warning	system.

does not help with "fast approach" crashes. In this case there is a longitudinal gap between vehicles when the lane change begins and the relative velocity difference is large.<sup>(21)</sup> Sensor

Since change/merge accidents occur with equal frequency on both sides of the vehicle there should be blind spot detection sensors on both sides of the vehicle.<sup>(64)</sup> Figure 22 shows the left-side blind spot for a vehicle. These sensors should increase safety as vehicles enter and exit the automated lane. The sensor needs to detect vehicles accurately with few unnecessary warnings or false alarms. Too many warnings will cause the driver to turn off the system or ignore its warnings.



Figure 22. The shaded area shows the region of coverage for a blind spot sensor on the left side of the vehicle.

The blind spot detector should cover the adjacent lane to the side and behind the vehicle. This means a range of no more than 10-12 feet to the side and 20-30 feet rearward and  $along^{(64,21)}$  the side of the vehicle. The size of the detection zone depends on the vehicle type and the size of its blind spot. The sensor should reliably detect small vehicles such as motorcycles in all likely lane positions.<sup>(64)</sup> The lateral detection zone comes from the standard highway lane width of 12 feet. If the lateral range is too large, the sensor could detect vehicles two lanes over leading to false alarms during lane changes.

Currently available blind spot sensor systems use ultrasound, radar and video systems.<sup>(76)</sup> Ultrasonic systems transmit ultrasound pulses to detect objects and measure their distance. Radar systems use frequency modulation and range-gating to find the range and relative velocity of objects. Video systems can show the driver a shot of the blind spot or they may add in computer vision technology to search for objects.

Ultrasonic sensors detect objects and their range with pulses of ultrasound. The sensor estimates the pulse time-of-flight. Sensor accuracy depends on local air properties, such as humidity and temperature. These sensors are short range  $(0.3-10 \text{ m})^{(76)}$  with accurate range measurements.

Radar sensors measure the range and relative velocity of objects. The radar estimates the pulse time-of-flight with range gates that look for return pulses at different time ranges. Pulse doppler radars measure the relative velocity between vehicles. This lets the radar tell which vehicles are approaching and which are dropping back. Radar sensors have a longer range than do ultrasound sensors (15-45 m), but they are not as accurate. Extra range may not be good for the blind spot detection problem since the sensor may detect vehicles two lanes away and cause a nuisance alarm.

Video systems can show the driver an image of his/her blind spot. This system has no image processing that searches for vehicles. Newer systems use imaging devices mounted by the side view mirrors to track vehicles in the adjacent lanes. These systems use the continuity between image frames to track vehicles and estimate their distance.

## Warning

The blind spot warning system may be active whenever the vehicle ignition is on and the vehicle is going forward.<sup>(64)</sup> This system may have many nuisance alarms caused by slow traffic or parked cars. Nuisance alarms occur when the sensor detects an object that is not there or an object more than one lane away. The system could also be active only if the driver indicates an intent to change course, such as with a turn signal. This system relies on the driver to signal lane changes. The blind spot warnings may also be triggered by a change in the vehicle's steering angle that signals a lane change. This warning would probably occur too late to avoid an accident.

The blind spot warning system should have at least two levels of warning: an imminent crash avoidance warning and a cautionary warning.<sup>(98,64)</sup> A cautionary warning tells the driver that there is a vehicle in the blind spot. This warning may operate when the vehicle is moving forward or only when the automated systems have been turned on by the driver. An imminent crash warning alerts the driver to an imminent crash avoidance situation that exists when an object is sensed in the blind spot and there is an indication that the vehicle's path may cause a

collision with that object.<sup>(64)</sup> A turn signal or changes in the vehicle's steering angle can indicate a course change for the vehicle.

The imminent blind spot warnings should turn off when the hazard disappears or the driver responds to the warning. The cautionary warning stays on until there is no vehicle in the sensor field. The warning should give information on the location and speed of the threat without causing the driver to look away from the roadway. Directional audio systems can tell the driver the location of the threat.

The blind spot detector may turn on only when the driver turns on the turn signal to change lanes or it could be on the entire time the vehicle is in the automated lane. But if the blind spot sensor was on continuously, it would sound warnings every time the vehicle passed another vehicle. Another option is have the warnings sound only when the turn signal is on.

# Driver Interface

The blind spot warning system provides levels of warning to the driver and provides a detector sensitivity adjustment for the driver. The different warning levels allow the vehicle to warn the driver without intrusive nuisance alarms. The detector sensitivity adjustment lets the driver adjust the range of the sensor to reduce sensitivity and false alarm rates.<sup>(64)</sup>

The cautionary warning should be visual and located within 15<sup>th</sup> above the line-of-sight of the side view mirror.<sup>(64)</sup> Visual signals are more detectable if they are near the center of the line-of-sight. Several of the commercial blind spot sensors mentioned in Najm use visual displays located by the side view mirror.<sup>(76)</sup> The imminent blind spot crash warning should use two types of warnings. Lerner et al suggests a combination of auditory and visual warnings.<sup>(64)</sup> The auditory warning can help the driver determine the direction of the imminent collision with a directional signal.

The blind spot warning system may have an adjustable threshold. This lets the driver reduce the sensor range to avoid excessive false alarms in which the driver perceives frequent hazards. For example, on multiple-lane highways the sensors may be triggered by objects across the adjacent lane or by vehicles two lanes away.

## Issues and Risks

#### 1 Sensor Capabilities

The blind spot sensor system may increase safety and reduce certain types of lane change/merge crashes. But there are sensor and human factor issues that still must be resolved. The sensors must detect small vehicles, such as motorcycles with few nuisance alarms. It will be very difficult for radar and ultrasonic sensors to decide if a motorcycle is present or if there is a large vehicle two lanes over. A video system may be able to track small vehicles better during daylight. The video system sees what the driver sees so it may be more natural for human drivers to use. The performance of vision based systems may degrade at night and in rainy weather plus it is more expensive and complicated than radars of ultrasonic systems.

2 Warning system design

First the designers need to decide when the system should be turned on: always on, only when turn signal on, or turned on when steering column shows lane change. The warning thresholds for emergencies and driver awareness must also be set. The system must avoid nuisance alarms, yet detect vehicles accurately. The types of alarms and warnings must also be researched. The emergency alarm needs to be directional to get fast driver response.

#### 3 Multiple Warnings

ERSC 1 introduces multiple types of driver warnings for blind spot and rear-end collision avoidance. Research needs to be done on the interaction and prioritization of these warnings for the human driver.

# Driver Interface

The driver should be able to override the speed & headway maintenance function and adjust headway and set cruising speed as described in section 5 on the SHM function. The driver should also be able adjust the warning thresholds on the rear-end collision warning and the blind spot warning as described in the sections on the rear-end collision and blind spot warning functions. The driver also receives highway traffic information through the vehicle and may use it for speed/headway selection or route planning. The driver also monitors the vehicle's operational status and backs up failed vehicle functions. If the SHM system fails the driver takes over manual control of the vehicle in the fall-back mode.

The driver interface must show the driver that the automated functions are operating correctly or that the function has failed. This interface should be simple to understand and use. No adjustments should be required that take the driver's attention away from the driving task.

Head-up displays may be particularly useful for visual indicators of the system operation. For example a heads-up display could show the measured distance to the vehicle ahead as well as the minimum time headway and target speed. Other technologies include liquid crystal displays (LCDs), light-emitting diodes (LEDs), computer generated voice messages, sudden vehicle accelerations/decelerations (jerk), tactile feedback, and directional audio displays.<sup>(64,83,116)</sup>

The interface may be continuous or only active when a driver action is needed. The best design of the driver interface for human factors is an open issue and involves selecting a display technology, type and duration of warnings, user control types and locations, etc. User acceptance is crucial to user interface design.<sup>(54,67)</sup> The systems in each ERSC must be easy to use and understand and easy to adjust without the driver taking his eyes off the roadway for more than 2 seconds.<sup>(73)</sup>

# Roadway/Vehicle Speed & Headway Commands

The roadway measures the average speed and traffic density, assesses environmental conditions, and calculates vehicle target speeds and minimum safe headways. The vehicles should follow the target speeds using the speed and headway maintenance function. The vehicles should respond to headway recommendations for longer headways but not lower than those requested by the driver. The vehicles may also display relevant traffic information from the roadway to

the driver for route planning purposes. This section covers roadway sensing systems, algorithms for smoothing traffic flow, and roadway to vehicle communications.

### Roadway Sensing Systems

The roadway needs to sense the average traffic density and speed over sections of the roadway. This data can be measured by sensors that are part of the infrastructure or the vehicles can transmit their measured speed and headway to the roadway with a vehicle-roadway communication system. The vehicle may also transmit its operational status and identification to the roadway for check-in or incident management. The roadway also needs to measure roadway weather data to choose a safe target speed for the environmental conditions.<sup>(57)</sup>

Proximity sensors such as inductive loops count the number of vehicles that pass over them in a given time.<sup>(112)</sup> The data is collected over sections of roadway to monitor traffic flow.<sup>(112)</sup> The sensors can be hardwired to a roadway traffic control center or they can transmit their data electronically to the traffic control center. These sensors may require changes in the infrastructure and roadbed. These sensors have been in use on freeways for at least a decade and many urban areas already have them in place for traffic monitoring and incident management.<sup>(112)</sup>

Overhead infrared sensors have also been tested for traffic monitoring.<sup>(80,48)</sup> These sensors can measure vehicle speed, density, and vehicle type. This sensor type may allow the roadway to choose minimum headways based on the mix of vehicle capabilities. For example if there are many trucks in the dedicated lane at certain times of the day, then the roadway may lower the target speed and increase the minimum time headway.

The vehicle may send its current speed, headway to the roadway. This information could be used by the roadway to calculate average vehicle speeds and densities for traffic flow control. The vehicle may also communicate its operational status and identification to the roadway for check-in purposes or for incident management. Vehicles equipped with sensors for road-tire friction can report on road conditions on different sections of roadway. Types of communication systems are discussed later in this section.

The condition of the roadway surface also affects the safe headway in between vehicles. Figure 23 shows the time headway for different road-tire friction coefficients. Weather conditions can affect the friction coefficient.<sup>(5)</sup> If the follower vehicle decelerates slower than the vehicle in front then the minimum time headway increases. This figure assumes that both vehicles are on the same surface, for example it's raining and the roadway is equally wet for both vehicles. Localized patches of ice or puddles can cause large changes in the friction coefficient. This leads to fast changes in  $\mathcal{E}A_m$  and the minimum time headway.



Figure 23. As the road-tire friction coefficient decreases (the road gets more slippery), the minimum safe time headway increases.  $A = A_{lm} - A_{fm}$  is the deceleration difference between the vehicles. A positive value means that the lead car has a larger deceleration than the follower vehicle.

Roadway weather conditions can give large changes in time headway. This may require roadway weather information systems (RWIS) that measure and estimate quantities such as water on the roadway, the possibility of ice patches, and snow.<sup>(57)</sup> The roadway may send this weather data to the drivers to help them choose a safe headway for the conditions. The RWIS described by Kelley only measures weather data at or near the sensor location. The system then interpolates conditions between sensors.

## Roadway Traffic Control Algorithms

The roadway can increase the average velocity of the vehicles in an automated lane by smoothing traffic flow.<sup>(10,19)</sup> The roadway receives traffic flow data from sections of roadway in the form of speed and headway data. The roadway then sends target velocities to the vehicles in the automated lane. This lets the roadway compensate for variations in the traffic flow due to accidents or heavy traffic. Smoother flow can increase the traffic flow rate. The roadway also sends minimum time headways to the vehicles based on environmental conditions and vehicle capabilities. Constant velocity adjustments for each section would soon tire the driver. This means that the roadway must be able to set the speed of each vehicle in a section automatically

with an electronic message. If the roadway sends automatic commands to the vehicle then vehicle compliance and overall roadway performance improves.

Traffic control experiments have been performed as part of the Prometheus project in Europe.<sup>(10)</sup> The traffic flow harmonization project in Gothenburg equipped a 3.5 km stretch of highway with short-range VRC beacons. These beacons transmitted data on speed limits, the status of traffic signals, and data on green lights. The vehicles could be operated in manual and automatic modes. The Aspen track project instrumented a 35 km long track of rural roads and motorways with short-range VRC beacons. The beacons transmitted data on speed limits, road curvature, recommended speeds on curves, and warnings for school areas and pedestrian crossings. Driver reaction to these systems was positive and vehicle safety was improved.

Other work by Chien et al "homogenizes" or smoothes traffic flow along a stretch of roadway.<sup>(19)</sup> The roadway measures the average speed and traffic density along each segment of the roadway. The roadway sends velocity commands to the vehicles on each section. This controller tries to keep traffic flow uniform over a length of roadway. This system can improve the average traffic flow rate along the highway and the traffic flow returns to a uniform density faster with control than without it. This type of roadway controller can give large benefits in terms of flow rates with very low technical risk and effort.

## Vehicle to Roadway (VRC) Communications

The roadway sends speed data to the cars to smooth the traffic density. The vehicles may also send their measured velocity and headway to the roadway. This system may also be used for check-in. This allows the roadway to monitor the operational status of the vehicles and ensure that they are correctly operating in the automatic mode. These messages only need to be sent once per section.

VRC systems can be mounted on the side of the roadway, on overpasses, or embedded in the roadway.<sup>(85)</sup> These systems may use automated toll collection technology with variable registers or a system that covers a wider range.<sup>(22,53)</sup> Appendix A covers the operation of these systems. One-way systems from the roadway to vehicles include using subcarriers on existing FM or TV station signals to send data to vehicles .<sup>(16,33)</sup>

This message does not need a high data rate since it will not change very often. Each field will need 6-8 bits to give a resolution of 1 meter for the gap limit and 1 mile per hour for the velocity limit. Since the message goes to all cars a destination ID code is not needed. The source ID code indicates that the message is from the roadway. Error detection coding looks for bit errors in the message. Retransmission requests are unnecessary since we can rely on repeating the message. This message should be repeated at most one time per second. The bit error rate can also be high since the message changes slowly and rarely. Repetition of messages can compensate for the high error rate.

The VRC may also send messages from the vehicles to the roadway. Each vehicle sends its speed and the distance to the preceding vehicle (6-8 bits each) The vehicles may also send a unique identifying code (28 bits) to the roadway for check-in. The operational status of the

automated system can be sent as well, this could take 8-16 bits depending on the amount of detail required by the roadway. Other variables such as estimates of the road-tire friction coefficient can take 4 or more bits. This gives a data length of around 70 bits.

A broadcast system with a range of 100 meters for a single automated lane needs to communicate with 0.3 to 0.75 vehicles per second. Analysis by Polydoros et al shows that the current Hughes VRC system and automatic toll collection systems can meet the data

transmission requirements for VRC in ERSC  $1.^{(85)}$  This system scales up to the next ERSC that requires two way VRC for check-in and traffic management.

#### Issues and Risks

#### 1 Dedicated Lane

An important issue for AHS is how to get a separate lane for automated vehicles. The driving public rebels when a lane is taken away from active use.<sup>(99)</sup> The solution for most highway departments has been to add a new lane on the shoulder or median strip, but most urban highways do not have any more space to spare for the sole use of automated vehicles. This infrastructure change may have to wait until enough vehicles have speed and headway maintenance systems and traffic flow benefits can be easily seen.

#### 2 Legal/Liability

In this ERSC the roadway sends recommended target speeds directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the vehicles are going too fast. The government may not wish to deploy such a system until it has been shown to be reliable and fail-safe.

#### 3 Sensors

Roadway weather sensors only cover small regions of roadway. These sensors could easily miss patches of ice or puddles on the roadway. The roadway may recommend a headway that is not large enough. Research needs to continue into these sensors to improve their capabilities.

#### 4 Privacy

This ERSC gives the vehicles the option to "Check-in" with the roadway. This may leave a record of when and where a vehicle traveled. This raises user privacy concerns about governmental use of this data.

#### 5 Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

# Fall-back Mode

In case of failures in the SHM or the RECW functions, the vehicle returns to manual brake and throttle control. The driver is responsible for the vehicle's operation and for driving it out of the dedicated lane. It may be possible to continue operations in the dedicated lane if the RECW fails. The vehicle need only fall back to a larger headway since human drivers may have a longer reaction time without a warning system. However this option may cause "role confusion" and slow driver response if the driver is used to waiting for a warning before braking.<sup>(90)</sup>

If the blind spot warning system fails, the vehicle informs the driver and continues operations in the dedicated lane. Failure of this function does not impact the vehicle's normal operations in the dedicated lane. The driver is still responsible for making lane changes into and out of the

dedicated lane. The announcement of warning system failure may be repeated when the driver prepares to exit the dedicated lane to remind him/her to look in the blind spot.

# Reliability Requirement Analysis

In ERSC 1, the automation technologies used in the AHS are mainly in the longitudinal control. The vehicle can perform speed and headway maintenance (SHM) using throttle and brake control. Automatic throttle control and brake control are limited within certain ranges so that the vehicle acceleration and deceleration will not cause driver discomfort. The driver is responsible for supervising the operation of the on-board SHM system and overriding it during emergencies. In particular, the driver is responsible for the rear-end collision avoidance. The driver performs the rear-end collision avoidance by applying hard braking to override the SHM system and/or steering maneuver. The driver's rear-end collision avoidance function is assisted by the vehicle's rear-end collision warning (RECW) system. The RECW system can receive braking data from the preceding vehicle to improve its detection accuracy.

The roadway sends target speed commands and time headway recommendations to improve the efficiency of the dedicated lane and the safety of vehicle following operation. The roadway calculates the target speeds and desired headways along sections of the dedicated lane according to the traffic and environmental conditions. The SHM system responds to the roadway target speed commands smoothly as long as the headway is larger than the one set by the driver. The RECW system responds to the roadway headway recommendations by adjusting the warning threshold setting. The safety of the longitudinal control depends on the reliability of the roadway speed/headway commands, the on-board SHM and RECW system, and the driver.

The driver is responsible for all vehicle lateral control functions. The driver is assisted by the on-board blind spot warning (BSW) system for lane changing and merging maneuvers during entry/exit of the dedicated lane. The BSW system gives warnings to the driver if a vehicle in the blind zone is detected. Once in the dedicated lane, the driver is fully responsible for steering to keep the vehicle in the lane. The driver's reliability determines the safety of the vehicle lateral control.

# Speed and Headway Maintenance

In ERSC 1, the on-board speed and headway maintenance (SHM) system assists the driver in vehicle longitudinal control. It maintains the speed and/or time headway chosen by the driver. The SHM system responses to the roadway target speed command if the resulting headway is not smaller than the one selected by the driver. The SHM system controls the throttle and brake. For the concern of riding comfort, only soft braking is applied by the SHM system. The driver can override the SHM system by pressing the throttle or brake pedal. Figure 24 shows a functional block diagram of the SHM system and its interface with the driver and roadway.



Figure 24. The functional block diagram of the SHM system.

The SHM system is composed of a headway/speed sensor, a controller (computer with control algorithms), a throttle actuator and a brake actuator. The headway/speed sensor measures the headway and the relative speed with respect to the preceding vehicle. The controller receives the speed/headway settings from the driver. The controller uses the data from the headway/speed sensor to generate throttle and brake commands for safe cruising or vehicle following. The controller can also smoothly respond to the roadway's target speed command when the vehicle has a headway larger than the one set by the driver. The throttle actuator takes the controller's throttle command to produce the desired throttle angle. The brake actuator uses the controller's brake command to generate the required force.

Safe longitudinal control depends on both the SHM system reliability and the driver's performance during emergencies. A rear-end collision may happen if the driver can not react to the danger such as abrupt braking by the preceding vehicle or a stationary vehicle in the lane. In addition, a SHM system failure may lead to a rear-end collision if the driver can not recover vehicle control in time. Drivers in the United States average a rear-end collision about every 50 years .<sup>(27)</sup> The SHM system needs to be designed in such a way that the operation of the combined SHM-driver system improves the manual driving safety standard significantly.

# Reliability Requirement Estimation

Using the techniques developed in reliability engineering<sup>(11,65)</sup>, one can estimate the required reliability of the SHM system, based on the safety standard in today's manual driving. Under normal operating conditions, the SHM-driver system can operate without rear end collisions if (1) the SHM system operates without failures; or (2) the SHM system fails but the driver successfully takes over the longitudinal control from the SHM system. Since the drivers are always responsible for supervising the vehicle's longitudinal control even when the SHM system is operational, one can consider the SHM-driver system as an active parallel system in the reliability analysis. In terms of reliability block diagram, the SHM-driver system can be represented by the parallel connected system as shown in figure 25.



Figure 25. The reliability functional block diagram of the SHM-driver system.

It is reasonable to assume that the driver failure and SHM failure are two independent events. To simplify the analysis, we characterize the reliability properties of SHM system and driver with constant failure rate models. In this case, if we denote  $\lambda_S$ ,  $R_S(t)$ , MTTF<sub>S</sub>,  $\lambda_D$ ,  $R_D(t)$  and MTTF<sub>D</sub> as the failure rate, reliability function and mean time to failure of the SHM system and the driver respectively, we have

$$R_{s}(t) = e^{-s^{t}}, \quad R_{D}(t) = e^{-D^{t}}$$

and

$$MTTF_{s} = \frac{1}{s}, MTTF_{D} = \frac{1}{D}$$

Using the reliability model in figure 25, we obtain the reliability function for the SHM-driver system

$$R_{SD}(t) = e^{-s^{t}} + e^{-b^{t}} - e^{-(s^{t}-b^{t})t}$$

The mean time to failure for the SHM-driver system is

$$MTTF_{SD} = \frac{1}{s} + \frac{1}{p} - \frac{1}{s + p}$$

With the requirement that MTTF<sub>SD</sub> be much greater than 50 years, the above equation can be used to analyze the required SHM system failure rate  $\lambda_S$ .

Suppose that the driver's reliability in longitudinal control is not degraded while he/she is interfacing with the SHM system, i.e., we have  $MTTF_D = 50$  years and  $\lambda_D = 0.02$  year<sup>-1</sup>. Since

$$MTTF_{SD} - 50 = \frac{0.02}{S(S + 0.02)} > 0$$

we see that the reliability of SHM always produces a net improvement in the reliability of the overall SHM-driver system, if the driver's reliability is not degraded.

The SHM system is not available to the general public and experimental data about human factors in SHM-driver operation is limited. It is not clear that how the SHM system affects the

driver's reliability. For drivers who are suspicious about the SHM system's reliability, they will tend to be more alert. These drivers will have higher reliability than the average in the SHM-driver operation. However, it is possible that some drivers are confident with the SHM system and become over reliant on the system. In this case, it is likely that driver's reliability will be degraded while interfacing with the SHM system.

For safety consideration, it is necessary to analyze how the driver's reliability degradation affects the reliability requirements of the SHM system. The case that the driver has a higher failure rate (or shorter mean time to failure) while interfacing with the SHM system should be considered. We require that

### $MTTF_{SD} > MTTF_a$

where  $MTTF_a >> 50$  years. This requirement implies

$$\frac{2}{D} + \frac{S(S + D)}{S(S + D)} > MTTF_a$$

The above relationship allows one to evaluate the required SHM system reliability as a function of the driver's reliability. For a given driver's degraded reliability (with the value of failure rate  $\lambda_D$ ), the SHM system is required to have reliability with  $\lambda_s$  stays below the curves in figure 26. The three curves are generated based on the assumption that the SHM systems average one hour operation daily and that the public will demand the SHM-driver system to be one, two or three times reliable as today's manual longitudinal control. These curves are plotted with the driver's reliability degradation up to 50%.

# Raytheon

# Task D



Figure 26. The required SHM system failure rates very as the driver's reliability degrades.

## Reliability Functional Requirement

The SHM system is not designed for longitudinal collision avoidance. In ERSC 1, the driver has to be hands on the steering wheel for lateral control, monitor surrounding traffic, and supervise the operation of the SHM system. The driver has to take over longitudinal control during emergencies. Depending on the design, the SHM system may be disengaged temporarily or totally disabled when the driver's override takes place. The driver is essentially still driving the vehicle, and can be considered as a back-up to the SHM system during emergencies. Safe longitudinal control can be achieved by improving the man-machine interface between the driver and the SHM system. When the driver overrides the SHM system by pressing the throttle pedal to accelerate, the SHM system has to deactivate the brake control during the time interval in which the override takes place. When the driver presses the brake pedal, it is potentially an emergency. In this case, it is desirable to deactivate both throttle and brake control.

Since the driver is the ultimate back-up in the longitudinal control, it is imperative that the driver can always override the SHM system successfully when he/she does it. The vehicle can use redundant methods to guarantee safe transition of longitudinal control from the SHM system to the driver. To further ensure safety the redundant methods should be independent. This requirement can significantly reduce the probability of simultaneous failures in both methods

caused by a single component/point failure. With the above discussion, it is reasonable to conclude that the essential reliability functional requirement for the SHM system design is:

"The SHM system should provide redundant, independent means to return the vehicle longitudinal control to the driver when he/she overrides the SHM system."

# System Level Design Requirements

The above requirement can be fulfilled by implementing two independent, redundant methods (e.g., hardware and software based) to detect the driver's override action and to deactivate the throttle/brake actuators. It does not impose any requirement of major components (sensor, controller, and actuators) redundancy in the design of the SHM system.

Detailed design requirements for the SHM system can be obtained by exercising the failure mode and effects analysis (FMEA) Without doing the complete FMEA, one can still come up with important system level design requirements by simply examining the potential system level failure modes from the system functional block diagram figure 24 These potential major failure modes include

- 1. Headway/speed sensor failure
- 2. Controller failure
- 3. Actuator failure
- 4. Improper response to driver's override
- 5. Roadway interface failure

The SHM system should minimize the probability of above failure modes to ensure the fail-safe operation. If a potential failure is detected, the SHM system should be disabled and the driver should be notified to regain control. Some important system level design requirements can be derived by analyzing the above listed potential failure modes.

# Potential Failure Mode 1: Headway/Speed Sensor failure

- 1.1 Headway/Speed Sensor computes incorrect headway or closing rate data. Design requirements:
  - 1.1.1 Headway/Speed Sensor needs to have an internal independent check of the reasonableness of the computed headway and closing rate data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed. The System Failure Response disables the SHM system and warns the driver to take over longitudinal control.
  - 1.1.2 Controller should incorporate an independent check of the reasonableness of the computed data from Headway/Speed Sensor, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

- 1.2 Data from Headway/Speed Sensor is not communicated to Controller, or is corrupted in transmission through interface. Design requirements:
  - 1.2.1 Controller should verify that Headway/Speed Sensors are operating in a timely manner by verifying that new data is received from Headway/Speed Sensors every to be determined (tbd) time interval. If an error is detected, the System Failure Response should be executed.

# Potential Failure Mode 2: Controller failure

- 2.1 Controller computes incorrect throttle or brake command. Design requirements:
  - 2.1.1 Controller should incorporate an independent check of the reasonableness of the computed throttle/brake command before it is sent to the throttle/brake actuator, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
  - 2.1.2 Throttle Actuator and Brake Actuator should incorporate an independent check of the reasonableness of the throttle/brake command computed by Controller, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
  - 2.1.3 Controller should incorporate circuitry, such as a watchdog timer, to detect failure and to ensure the processor functions in a timely manner. If a failure is detected, the circuitry should deactivate Controller, and the System Failure Response should be executed.
- 2.2 Controller's command is not communicated to Throttle/Brake Actuator at appropriate time or is corrupted in transmission through interface. Design requirements:
  - 2.2.1 Throttle/Brake Actuator should verify that it is receiving updated command from Controller in a timely manner. If an error is detected, the System Failure Response should then be executed.

## Potential Failure Mode 3: Actuator failure

- 3.1 Brake Actuator does not generate correct brake force for the given brake commands. Design requirements:
  - 3.1.1 Brake Actuator should incorporate built in tests to be performed on the vehicle startup and during headway maintenance operation to verify that brake force can be produced and is within the allowable range. If an error is detected, the System Failure Response should be executed.
- 3.2 Throttle Actuator produces incorrect throttle angle.

# Design requirements:

3.2.1 Throttle Actuator should incorporate built in tests to be performed on the vehicle startup and during headway maintenance operation to verify that throttle angle can be produced and is within the allowable range. If an error is detected, the System Failure response should be executed.

Potential Failure Mode 4: Improper response to driver's override

- 4.1 Brake Actuator is not disengaged temporarily in response to the movement of the accelerator pedal by the driver. Design requirements:
  - 4.1.1 The SHM system should incorporate independent, redundant devices for detecting the driver's pressing of the accelerator pedal and means to detect failure of these detection devices.
  - 4.1.2 The SHM system should provide for independent, redundant means to deactivate Brake Actuator.
- 4.2 Controller and Throttle/Brake Actuators are not disabled (or disengaged temporarily) when the driver presses the brake pedal. Design requirements:
  - 4.2.1 The SHM system should incorporate independent, redundant devices for detecting the driver's pressing of the brake pedal and means to detect failure of these detection devices.
  - 4.2.2 The SHM system should provide for independent, redundant means to deactivate Throttle Actuator.
  - 4.2.3 The SHM system should provide for independent, redundant means to deactivate Brake Actuator.

## Potential Failure Mode 5: Roadway Interface failure

- 5.1 The target speed commands are not received by the SHM system or are corrupted in transmission. Design requirements:
  - 5.1.1 The SHM system needs to verify that the target speed commands are received in a timely manner. If an error is detected, the SHM system should warn the driver of the missing speed commands and smoothly switch the operation to the desired speed set by the driver.
  - 5.1.2 The SHM system should incorporate an independent check of the reasonableness of the message received from the roadway, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the SHM system

should warn the driver of the error, and smoothly switch the operation to the desired speed set by the driver.

## Key findings:

1. The driver can be regarded as a back-up to the SHM system.

2. The reliability functional requirement for the SHM system is to reliably return the control to the driver when the driver overrides the SHM system.

#### Issues and Risks:

1. The driver may become over-reliant on the SHM system and degrade his/her reliability/performance during emergencies.

2. The vehicle's automatic response to the roadway's target speed command imposes safety concerns in ERSC1. For example, the SHM system headway/speed sensor fails while the controller is responding to a target speed command by accelerating the vehicle. In this case, the driver may perceive that the SHM system is properly following the roadway's command and fails to sense a rear-end collision danger. The driver's reliability as a back-up to the SHM system may be seriously degraded since the SHM operation puts the driver into a situation that he/she can not handle safely. This potential failure mode has serious reliability implications.

3. If the driver can not be regarded as a back-up to the SHM system, a fail-safe SHM system design will require considerable amount of redundant components. This will increase the cost and difficulties in deploying the SHM system in ERSC 1.

4. The above safety concerns can be relieved if we do not allow the vehicle to respond to the roadway's speed commands automatically. In this case, the driver can take the target speed commands from the roadway and input it to the SHM system. However, the effectiveness of the roadway traffic flow control will be much affected since the driver may not always adjust the speed exactly the same way as the roadway demands.

# Rear-End Collision Warning

The rear-end collision warning (RECW) system improves the safety of vehicle longitudinal control by warning the driver of potential rear-end collisions. The RECW system measures the vehicle's and the lead vehicle's dynamics including headway, velocities, braking actions and braking capabilities. It uses these data to evaluate the time to collision (TTC). If the computed TTC is larger than the time headway, the RECW system sends warning to the driver. The driver reacts to the warning by overriding the SHM system to avoid rear-end collisions. The vehicle's braking capabilities depend on the road-tire friction that is affected significantly by the road surface condition. The roadway senses the road surface condition to estimate a headway for collision free vehicle following. The RECW system adopts the headway recommended by the roadway as TTC if it is longer than the existing TTC computed by the RECA system. Figure 27 shows the interaction between the RECW system and the driver.



Figure 27. The functional block diagram of the RECW system.

The RECW system uses a headway/speed sensor and vehicle-to-vehicle communication to collect the dynamic data required for computing TTC. The headway and closing rate data measured by the headway/speed sensor is essential for the RECW system operation. A malfunction in the headway/speed sensor will result in the RECW system failure and cause non-detections or false alarms. The brake data provided by vehicle-to-vehicle communication can be used to improved the accuracy of the predicted TTC. The RECW system degrades its reliability, i.e., may have lower detection rates and higher false alarm rates, if the processor does not correctly receive the preceding vehicle's brake data from the vehicle-to-vehicle communication. Incorrect headway recommendations received from the roadway will result in improper TTC. This will also cause non-detections or false alarms.

## Warning Levels

It has been recommended that at least two levels of warnings be used in the RECW system,<sup>(64,34)</sup> an imminent collision warning and a cautionary collision warning. The imminent collision warning informs the driver that it is necessary to take evasive action to avoid rear-end collision. If the imminent collision warning can not be sensed by the driver promptly, the driver may not be able to react to the danger in time. The RECW system is therefore required to provide redundant imminent collision warnings. Since the driver may become less attentive to a specific type of signals, it is desirable that the redundant imminent collision warning signals be different in modality, e.g., visual and audio.

# Reliability Functional Requirement

The RECW system is not designed to replace part of the driver's longitudinal control function. It is to increase safety by effectively alerting the driver when there is a rear-end collision danger. The effectiveness of the RECW system is affected by many factors including the detection rates, the false/nuisance alarm rates, types of warning signals, and the driver's reaction to different warning signals, etc.

In ERSC 1, the driver is interfacing with SHM and RECW systems for longitudinal control. The driver is feet-off the throttle and brake pedals except during emergencies. It is possible that some drivers become over relaxed and rely on the RECW system for sensing the rear-end collision danger. For these drivers, high detection rates of the RECW system will be vital and a non-detection may contribute to a rear-end collision. The design of the RECW system thus is required to minimize the possibility of non-detection.

Nuisance alarms may irritate the driver. By allowing the driver to adjust the warning sensitivity within a limited range, the RECW system can reduce its nuisance alarm rates and still accommodate the driver's skill. False alarms can be caused by noise, environmental conditions, internal system malfunctions, etc. Frequent false alarms may be disturbing to the driver, and delay his/her reaction to the danger. The driver may even reject the false alarms by turning off the RECW system. A sudden false imminent collision warning may make the driver panic and even endanger his/her vehicle control.

The above discussion suggests that the reliability functional requirement for the RECW system be:

"The RECW system should have high detection rates and low false alarm rates."

## System Level Design Requirements

To achieve this requirement, the RECW system needs to have self-diagnostic element to detect potential internal failures. If a failure is detected, the driver should be notified and the RECW system may have to be deactivated. More specific system level design requirements can be identified by analyzing major potential failure modes and causes based on the system framework shown in figure 28.



Figure 28. The RECW system framework for reliability design.

Potential Failure Mode 1: Headway/Speed Sensor failure

- 1.1 Headway/Speed Sensor computes incorrect headway or closing rate data. Design requirements:
  - 1.1.1 Headway/Speed Sensor needs to have an internal independent check of the reasonableness of the computed headway and closing rate data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed. The System Failure Response warns the driver of the potential failure and asks him/her to turn off the RECW system.
  - 1.1.2 Processor should incorporate an independent check of the reasonableness of the computed data from Headway/Speed Sensor, based on the physical attainability and

valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

- 1.2 Data from Headway/Speed Sensor is not communicated to Processor, or is corrupted in transmission through interface. Design requirements:
  - 1.2.1 Processor should verify that Headway/Speed Sensor is operating in a timely manner by verifying that new data is received from Headway Sensor/Speed every to be determined (tbd) time interval. If an error is detected, the System Failure Response should be executed.
  - 1.2.2 Same as for 1.1.2.

# Potential Failure Mode 2: Braking data reception failure

- 2.1 Braking data of the preceding vehicle is not communicated to Processor, or is corrupted during transmission. Design requirements:
  - 2.1.1 Processor should verify that it is receiving preceding vehicle's braking data in a timely manner by verifying that new data is received every tbd time interval. If an error is detected, Processor should notify the driver of the failure and take it into account to modify the estimation of TTC.

## Potential Failure Mode 3: Processor failure

- 3.1 Processor computes incorrect time to collision. Design requirements:
  - 3.1.1 Processor should incorporate an independent check of the reasonableness of the computed time to collision, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- 3.2 Processor command is not communicated to Warning Interface at appropriate time or is corrupted in transmission through interface with Processor. Design requirements:
  - 3.2.1 Warning Interface should verify that it is receiving updated command from Processor in a timely manner. If an error is detected, the System Failure Response should then be executed.

## Potential Failure Mode 4: Warning Interface failure

4.1 Warning Interface fails to give warnings or generate incorrect level/modality of warnings, after receiving a warning command from Processor. Design requirements: 4.1.1 Warning Interface should incorporate built in tests to be performed on the vehicle startup and during headway maintenance operation to verify that proper warning signals can be generated. If an error is detected, the System Failure Response should be executed.

### Key findings

1. Two levels of warning may be required. Imminent collision warning may require redundancy with different warning modalities.

2. The reliability functional requirements for the RECW system are high detection rates and low false alarm rates.

#### Issues and Risks

1. The driver may become over-reliant on the RECW system and degrade his/her reliability/performance in rearend collision avoidance.

2. The driver may experience warning overload and warning confusion problems while interfacing with other vehicle and roadway functions.

# Blind Spot Warning

The blind spot warning (BSW) system helps the driver change lanes or merge safely. It helps the driver detect other vehicles in the blind spots on both sides of the vehicle. The BSW system gives the driver a warning if a vehicle in the blind zone is detected. Figure 29 shows the BSW system interfacing with the driver.



Figure 29. The functional block diagram of the BSW system.

The BSW system will be most effective if the warning is sent before the vehicle's lane changing motion starts. In this case, the driver will probably cancel or delay his plan of change lanes. However, it is possible that other vehicles may suddenly move into the blind spot after the driver has started turning the steering wheel. In this case, even if the BSW system sends out the warning in time, the driver may not have enough time to react to the danger. The BSW system is not designed to replace the driver's detection function in lane changes or merges. It is mainly to improve the driver's reliability.

# Warning Signals

It is suggested that at least two levels of warning be provided in the BSW system, i.e., imminent collision warning and cautionary collision warning .<sup>(98,64)</sup> To ensure that the imminent collision warning can always be perceived by the driver, it should contain redundant signals with different modalities such as auditory and visual.

The warning signal generated by the BSW system needs to be distinct from those by other function so that the probability of driver confusion is minimized. It is also important that the driver can always perceive the warning message as soon as it is sent out by the BSW system. Visual or audio warnings can be used to warn the driver. If visual warning is used, it needs to be able to guide the driver's attention to the danger in blind spot. "Sound icons" may provide effective audio warnings.<sup>(105)</sup> It simulates the sounds that might be heard in real life. With sound icons, the BSW system can sound a horn from the rear right/left hand side of the vehicle.

## Reliability Functional Requirements

In ERSC 1, the driver is still in charge of the entire lane change and merge process. As shown in figure 29, the BSW system senses the traffic in blind spot and processes the data to determine if a warning should be delivered to the driver. Since the BSW system does not take over control from the driver, a BSW system failure is not likely to cause immediate catastrophe. However, the safety in changing lanes and merging can be improved by well-designed man-machine interface. The BSW system should sense the object in the blind spot, deliver a timely warning, and use effective warning signals. Design of reliable BSW systems should be focused on both the component/hardware reliability and the human factors engineering.

The BSW system should not deliver warnings not too often so that the driver will not be irritated. Warnings should not be too late so that the driver has enough time to react to the collision threat. Since drivers have different levels of driving skill and reaction time, it is desirable to allow drivers to select warning thresholds.<sup>(26)</sup> Such a design can allow the driver to minimize nuisance alarms without compromising his/her reaction time.

Since the driver may become over-reliant on the BSW system, a missed detection could lead to a lateral collision. Frequent false detections are not desirable since they may make the driver ignore the warning or even turn off the BSW system. The reliability functional requirement of the BSW system is thus proposed as:

"The BSW system should have high detection rates and low false alarm rates."

## System Level Design Requirements

Missed detections and false alarms can occur due to BSW system internal failures or environmental causes. High detection and false alarm rates will require reliable components (sensor, processor and warning interface, etc.) and self-diagnosis elements implemented in the BSW system. More specific design requirements at system level can be identified by analyzing major potential failure modes and causes based on the system framework shown in figure 30.





Potential Failure Mode 1: Blind Spot Sensor failure

- Blind Spot Sensor computes incorrect range or direction of the threat.

Design requirements:

- Blind Spot Sensor needs to have an internal independent check of the reasonableness of the computed range and direction data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed. The System Failure Response warns the driver of the potential failure and asks him/her to turn off the BSW system.
- Processor should incorporate an independent check of the reasonableness of the computed data from Blind Spot Sensor, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- Data from Blind Spot Sensor is not communicated to Processor, or is corrupted in transmission through interface.

Design requirements:

Processor should verify that Blind Spot Sensor is operating in a timely manner by verifying that new data is received from Blind Spot Sensor every tbd time interval. If an error is detected, the System Failure Response should be executed.

Same as for above.

#### Potential Failure Mode 2: Processor failure

 Processor command is not communicated to Warning Interface at appropriate time or is corrupted in transmission through interface with Processor. Design requirements: Warning Interface should verify that it is receiving updated command from Processor in a timely manner. If an error is detected, the System Failure Response should then be executed.

# Potential Failure Mode 3: Warning Interface failure

- Warning Interface fails to warn or generate incorrect level/modality of warnings after receiving a warning command from Processor. Design requirements:
  - Warning Interface should incorporate built in tests to be performed on the vehicle startup and during headway maintenance operation to verify that proper warning signals can be generated. If an error is detected, a System Failure response should be executed.

# Key findings

1. Two levels of warning may be required. Imminent collision warning may require redundancy with different warning modalities.

2. The reliability functional requirements for the BSW system are high detection rates and low false alarm rates. Issues and risks:

1. If the warning signal of the BSW system is to be activated only by the turn signal or steering wheel motion, there might not be enough time for the driver to react to the danger. This is because some drivers may perform changing lanes without turning on the turn signal. Warnings may be too late if the driver has already started turning the steering wheel. If the warning signal is to be generated without activation from the driver, the warnings created by the adjacent lanes traffics moving into and out of the blind zone will be irritating to the driver.

2. The effectiveness of the blind spot warning may be discounted since the driver has to deal with the potential warning/information overload problem while traveling in the AHS.

# Roadway to Vehicle Speed and Headway Commands

The roadway sends target speed commands to the on-board SHM system to smooth the traffic flow in the dedicated lane. The roadway also sends headway recommendations to the on-board RECW system to enhance the safety in longitudinal control. The roadway has traffic flow sensors to measure the average traffic speed/density and the environmental conditions. The roadway traffic controller uses this data to evaluate target speeds and minimum time headways

along sections of the dedicated lane for efficient and safe traveling.<sup>(19,10)</sup> The speed and headway commands are sent to the vehicle through the roadway-to-vehicle communication. Figure 31 shows a block diagram of the roadway to vehicle speed and headway command function.



Figure 31. A block diagram of the roadway to vehicle speed/highway command function.

The on-board RECW system adopts the headway recommended by the roadway as TTC if it is larger than the one adjusted by the driver. The SHM system follows the roadway target speed command if the resulting headway is not smaller than the one selected by the driver.

# Reliability Functional Requirement

Since the headway used by the SHM system for longitudinal control is constrained by the one chosen by the driver, incorrect speed commands from the roadway introduce no immediate danger to the SHM operation if the driver properly sets the headway based on the traffic and environmental conditions. However, if the driver relies on the roadway to determine the speed and headway used by the SHM system and RECW system, (i.e., the driver chooses the maximum allowable speed and minimum allowable headway,) the incorrect target speed command and minimum time headway recommendation will affect vehicle safety in longitudinal control. This can happen when the vehicle enters a roadway section that has slippery roadway surface. Without increasing the headway and/or decreasing the speed, the driver may have difficulties stopping the vehicle safety during emergencies. Incorrect commands from the roadway may affect the safety and efficiency of traffic flow. The above discussion suggests that the reliability functional requirement for the roadway to vehicle speed/headway commands be:

"The traffic flow sensors should be accurate and the target speed commands and minimum time headway recommendations should have low error rates."

# System Level Design Requirements

To achieve this requirement, the roadway needs to incorporate reliable sensors to measure traffic flow condition along the roadway.<sup>(112)</sup>, and to detect the roadway surface condition.<sup>(57)</sup> The roadway traffic controller needs to have efficient and reliable algorithms to process the collected information and generate target speed commands and minimum time headway recommendations for smoothing the traffic flow over sections of the dedicated lane. The roadway-to-vehicle communication should provide communication protocols that allow reliable data transmission from the roadway to the vehicle.

More specific system level design requirements can be identified by analyzing major potential failure modes and causes, based on the system framework shown in figure 31.

Potential Failure Mode 1: Traffic Speed/Density Sensors failure

 Traffic Speed/Density Sensors at some roadway section(s) fail to provide correct traffic speed/density.

Design requirements:

- The sensors should have an internal independent check of the reasonableness of the computed traffic speed/density data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, Traffic Controller should be notified and should use the validated data provided by the sensors from the neighboring roadway sections to estimate.
- Traffic Controller should incorporate an independent check of the reasonableness of the average speed/density data provided by the sensors, based on the physical attainability and valid known ranges of the system variables. If an error is detected, Traffic Controller should use the validated data provided by the sensors from the neighboring roadway sections to estimate.
- Speed/density data is not communicated to Traffic Controller, or is corrupted in transmission through interface.

Design requirements:

Traffic Controller should verify that all Traffic Speed/Density Sensors are operating in a timely manner by verifying that new data is received every to be determined (tbd) time interval. If an error is detected, Traffic Controller should use the validated data provided by the sensors from the neighboring roadway sections to estimate.

#### Potential Failure Mode 2: Environmental Condition Sensors failure

 Environmental Condition Sensors at some roadway section fail to provide correct roadway surface condition information.

Design requirements:

- The sensors should have an internal independent check of the reasonableness of the estimated data of road surface condition, visibility, etc., based on the physical attainability and valid known ranges of the system parameters. If an error is detected, Traffic Controller should be notified and should use the validated data provided by the sensors from the neighboring roadway sections to estimate.
- Traffic Controller should incorporate an independent check of the reasonableness of the data provided by Environmental condition Sensors, based on the physical attainability and valid known ranges of the system parameters. If an error is detected, Traffic Controller should use the validated data provided by the sensors from the neighboring roadway sections to estimate.

- Data from some Environmental Condition Sensors is not communicated to Traffic Controller, or is corrupted in transmission through interface. Design requirements:
  - Traffic Controller should verify that all Environmental Condition Sensors are operating in a timely manner by verifying that new data is received every tbd time interval. If an error is detected, Traffic Controller should use the validated data provided by the sensors from the neighboring roadway sections to estimate.
  - Traffic Controller should incorporate an independent check of the reasonableness of the data provided by Environmental condition Sensors, based on the physical attainability and valid known ranges of the system parameters. If an error is detected, Traffic Controller should use the validated data provided by the sensors from the neighboring roadway sections to estimate.

## Potential Failure Mode 3: Traffic Controller failure

 Traffic Controller computes incorrect target speed commands and/or minimum time headway recommendations.

Design requirements:

- Traffic Controller should incorporate an independent check of the reasonableness of the computed target speed commands and minimum time headway recommendations before they are sent to Roadway-to-Vehicle Communication, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the vehicle/driver should be notified and the roadway to vehicle speed/headway commands should be canceled.
- Roadway-to-Vehicle Communication should incorporate an independent check of the reasonableness of the commands/recommendations provided by Traffic Controller, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the vehicle/driver should be notified and the roadway to vehicle speed/headway commands should be canceled.
- Traffic Controller commands are not communicated to Roadway-to-Vehicle Communication at appropriate time or are corrupted in transmission through interface.

Design requirements:

Roadway-to-Vehicle Communication should verify that it is receiving updated commands from Traffic Controller in a timely manner. If an error is detected, the vehicle/driver should be notified and the roadway to vehicle speed/headway commands should be canceled.

Potential Failure Mode 4: Roadway-to-Vehicle Communication failure

- The target speed commands and minimum time headway recommendations are not communicated to the vehicle or are corrupted in transmission.

Design requirements:

- The vehicle should verify that the target speed commands and minimum time headway recommendations are received in a timely manner. If an error is detected, the vehicle should warn the driver of the failure.
- The target speed commands and minimum time headway recommendation should be sent frequently enough to minimize the effects of no communication.
- Roadway-to-Vehicle Communication should provide communication protocol that contains error detection and correction codes to minimize the probability of incorrect message reception by the vehicle.
- The vehicle should incorporate an independent check of the reasonableness of the message received from the roadway, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the vehicle should warn the driver and the roadway to vehicle speed/headway commands should be canceled.

#### Key Findings:

1. Incorrect speed commands from the roadway introduce no immediate danger to the vehicle/driver. However, incorrect commands from the roadway may affect the safety and efficiency of traffic flow.

2. The reliability functional requirement for the roadway to vehicle speed/headway commands is to have low error rates.

#### Issues and Risks:

1. The roadway may be liable for the accidents in the dedicated lane since it sends speed commands and headway recommendations to the vehicle.

2. Safer and more efficient traffic control can be achieved by installing more sensors to collect more accurate traffic and environmental data. This may substantially increase the cost of deployment.

# ERSC 1 Driver Tasks and Workload

The driver drives the car to the automated lane. The driver turns on the turn signal and the blind spot warning system. The driver enters the lane when a safe gap appears in the traffic flow. The driver steers the car into the lane and turns on the SHMS and rear-end collision warning systems. The driver may set a minimum time headway for the vehicle. The vehicle takes over the throttle and brakes when the car is in the lane or as the car enters the lane. The driver steers the car while in the automated lane. The driver supervises the vehicle control of the brake and throttle. The driver workload in ERSC 1 is very similar to that of today's driving.

The driver turns on the turn signal and blind spot warning system to leave the automated lane. The driver then resumes control of the throttle and changes lanes out of the automated lane. The driver turns off the communications links when he leaves the automated lane.

The driver performs all emergency stops and steering maneuvers. The driver can change the threshold of the rear-end collision warning since drivers have different reaction times and tolerances for risks and false alarms. The vehicle may begin soft braking, but the driver must complete the stop. The driver can override the vehicle SHMS at any time to avoid an accident or hazardous situation.

The vehicle's rear-end collision warning system and blind spot detector give warnings to the driver. The driver then reacts to the warnings. Driver response time varies with conditions such as age, fatigue, medical condition, and substance use.

### Task Analysis-Normal Driving

In normal operations, the vehicle controls the throttle and soft braking. The driver steers the vehicle in the lane and sets the speed and time headway. The roadway sends speed commands electronically to the vehicle and the driver. The task analysis in tables 11 and 12 below shows the driver cues for these two tasks.

The most common problem is that the driver may set the headway too short or too long. The cost of this error could be turbulence in the traffic stream, a rear-end collision, or reduced highway capacity. Tests on vehicles with gap control show that headway may need to be adjustable by the driver.<sup>(26)</sup> Most drivers dislike "cut-ins" on crowded freeways, so they use short headways to prevent them.

After extended driving, the driver may accidentally turn the speed and headway maintenance system off. The speed and headway maintenance system needs to clearly indicate that it is on or off. The SHMS may require standardization to make this system less confusing to the driver. If the SHM function fails, then the driver returns to manual driving

Task	Monitor Vehicle Operation and Steer Vehicle in Normal Driving	
General	Task similar to manual driving	
Initiating Cue	Enter Dedicated Lane and Turn on SHM-Set time headway or cruising speed	
Feedback	Visual, auditory, and kinesthetic cues of accelerating, braking, slowing, and	
	steering	
Task Standards	Duration is the same as manual driving	
Task Conditions	Weather, road conditions, or traffic	
Skills Required	Reaction times appropriate to speed, following distance, and roadway conditions.	
_	Perceptual judgment of vehicle kinematics	
Potential Errors	Set to wrong speed or unsafe headway	
	Driver inattention causes vehicle to leave road	
	Turn off headway maintenance system accidentally	
	Vehicle fails to maintain speed or proper headway	
Causes of Error	Fails to perceive or misinterprets system feedback	
	Poor judgment of distance and closing rate or speed and road curvature	
	Hit wrong button or brake pedal	
	Sensor or actuator failure on vehicle	
<b>Consequences of Error</b>	Disrupt traffic flow and lower capacity	
	Rear-end collision	
	Rear-end collision by vehicle in back	
	Lane departure collision	
<b>Recovery Points</b>	Minimum safe headway setting on vehicle to avoid tail-gating	
	Rear-end collision warning	
	Vehicle warns driver that headway maintenance function is not working or is	
	failing or has been turned off	
	Blind spot warning	
Individual Differences	Education	
	Age	
	Driving Skill	
Criticality	Essential	

# Table 11. Driver task description for normal ERSC 1 driving

Task	Adjust Vehicle Cruising Speed or Headway In Normal Driving	
General	Task may be similar to setting a radio in a vehicle or turning on the headlights	
Initiating Cue	Driver enters dedicated lane	
	Driver receives signal from roadway to change speed or headway	
	Driver feels uncomfortable with headway or speed	
Feedback	Dial or indicator that gives chosen speed and headway	
<b>Control Functions</b>	Depend on system design, must be easy to use	
Task Standards	Change speed or headway while steering vehicle in dedicated lane	
Task Conditions	Weather, road conditions, suggestions from traffic control center	
Skills Required	Understand signs in or out of vehicle (literacy)	
	Perceptual judgment of vehicle kinematics	
Potential Errors	Change to wrong speed or headway	
	Ignore request for speed change from traffic control center (option)	
	Turn off vehicle speed and headway maintenance system accidentally	
	Vehicle fails to maintain speed or proper headway	
	Vehicle does not receive target speed and minimum headway from roadway	
	(option)	
Causes of Error	Fails to perceive or misinterprets the request for a headway change	
	Driver hits wrong button or brake pedal	
	Sensor or actuator failure on vehicle	
	Communication link failure between vehicle and roadway	
<b>Consequences of Error</b>	Disrupt traffic flow–lower capacity	
	Rear-end collision	
	Rear-end collision by vehicle in back	
<b>Recovery Points</b>	Minimum allowable headway setting on vehicle to avoid tail-gating or unsafe	
	headway	
	Rear-end collision warning	
	Vehicle warns driver that headway maintenance function is not working or is	
	tailing or has been turned off	
Individual Differences	Education	
	Age	
Criticality	Critical	

### Table 12 Driver task description for speed and headway adjustments while driving.

# Task Analysis-Rear-End Collision Warning

Human factors studies suggest that collision avoidance systems need multiple levels of warning.<sup>(64,98)</sup> In the first level the collision warning system shows the driver that it is operating with a heads-up display or other indicator that shows the distance to the car ahead. In the second level the collision warning system warns of a possible collision with a visual warning such as changing the color or intensity of the display. In the third level the collision warning tells the driver to brake with combined auditory and visual warnings. These levels could correspond to no threat, a distance warning and a conservative time to collision based warning set by the driver. Table 13 lists the driver's tasks for stopping distance and collision avoidance warnings.

Most rear-end collisions are caused by driver's failure to detect objects or other vehicles or by misjudgments of the vehicle's movements.<sup>(60)</sup> Collision avoidance systems(CAS) warn the driver of impending collisions. These systems should reduce accident frequency.<sup>(59)</sup> The

driver must decide whether to brake and how much to brake. This leads to the human factors problem of what criterion to use for driver warning.

### Task Analysis-Blind Spot Warning

The blind spot warning looks for vehicles in the blind spot of the driver when the turn signal is on. If the sensor detects a vehicle then it warns the driver. This sensor backs up the driver when he changes lanes. Table 14 below shows driver-blind spot warning system operation. The blind spot detector may lead to driver carelessness in lane changing. The blind spot warning system should ideally warn the driver before the lane change begins to avoid a strong swerve to avoid an accident. The blind spot warning system may warn the driver in the middle of a lane change since the detector does not look for potential collisions such as a faster car approaching on the side. This could cause an accident if the driver does not respond quickly enough.

Good traffic engineering designs use the "one task at a time" rule to decrease driver response time. The blind spot sensor warning must give the direction of the problem early enough for the driver to evade the crash. These conditions violate the idea of one task at a time and may cause problems.

Task	Perform Hard or Emergency Braking	
General	Task similar to manual driving	
Initiating Cue	Warning from rear-end collision warning system to emergency brake	
	Driver detects: preceding vehicle braking	
	Vehicle ahead stopped or object in roadway	
	Vehicle cuts in and brakes or slows	
Feedback	Visual, auditory, and kinesthetic cues of braking and slowing	
Control Functions	Perception-response time (PRT) studies have found that drivers respond to	
	unexpected hazards by avoidance steering as well as braking	
Task Standards	PRT studies have shown for a wide population segment (young and old) PRT's	
	on the average of about 1.5 sec are required to respond to an unexpected hazard,	
	most respond under 2.0 sec $(64)$ However it is not certain that these data are	
	directly applicable to the car-following scenario	
Task Conditions	Weather traffic conditions, position of following vehicle	
Skills Required	Reaction times appropriate to speed following distance and roadway conditions	
Shins Required	Percentual judgment of vehicle kinematics	
Potential Errors	Respond too slowly to emergency	
	Not brake hard enough	
	Brake too often or too much	
Causes of Error	Fails to perceive or misinterprets the indication of an emergency	
	Poor judgment of distance and closing rate	
	Rear-end collision system does not detect emergency braking situation	
	Rear-end collision system has a high false alarm rate	
<b>Consequences of Error</b>	Rear-end collision with potentially severe results for the system	
	Disrupt of traffic flow and lower capacity	
	Loss of confidence by the driver and public in the system	
	In the case of false alarms the driver may defeat or ignore it	
<b>Recovery Points</b>	Driver sets longer headway or lower warning threshold for slow reactions	
	Improve vehicle braking capabilities	
	Warn driver sooner of a "cut-in"	
	Redundant sensors to improve detection	
Individual Differences	PRTs for older drivers have not shown them to be less alert or significantly	
	slower in responding to unexpected hazards. As noted above, it is not clear that	
	this pertains to the car following scenario and especially under a variety of	
	highway conditions	
Criticality	Very Critical	

# Table 13 Driver task description for hard or emergency braking.
Task	Respond to Blind Spot Warning
General	Task similar to that in manual driving
Initiating Cue	Warning from vehicle that vehicle is in blind spot
	Driver turns on turn signal to enter or leave dedicated lane
	Vehicle detects change in steering angle
Feedback	Visual, auditory, and kinesthetic cues of steering and braking
<b>Control Functions</b>	Perception-response time (PRT) studies have found that drivers respond to
	unexpected hazards by avoidance steering as well as braking
Task Standards	Steer back into lane when warned (~1 second)
	Look in blind spot
Task Conditions	Type and timing of warning
	Environmental conditions, traffic speed and density
Skills Required	Reaction times appropriate to speed, lane change speed, and roadway conditions.
	Perceptual judgment of vehicle kinematics
Potential Errors	Overconfidence in system-driver does not look
	Driver does not use turn signals
	Driver does not respond to the warning or responds too late
	Sensor false alarms (No vehicle in blind spot, system warns driver anyway)
	Sensor does not detect another vehicle in blind spot
Causes of Error	Fails to perceive or misinterprets the indication of an emergency
	Driver delay in response to warning
	Blind spot sensor warning threshold too high or low, malfunction
<b>Consequences of Error</b>	Lane change/Merge collision with potentially severe results for the system
	(Approximately 4% of crashes and 0.5% of fatalities are lane change/merge
	crashes. Some 95% of these crashes are angle/sideswipe crashes <sup>(113)</sup> )
	Disrupt of traffic flow and lower capacity
	Loss of confidence by the driver and public in the system
	In the case of false alarms the driver may defeat or ignore it
<b>Recovery Points</b>	Vehicle reminds driver to look before lane change
	Multiple warnings from detector
Individual Differences	Accident statistics, citations and self report data indicate that older drivers have
	difficulty in merging, changing lanes, and exiting maneuvers. Data suggests that
	persons over 65 may be over represented in lane change/merge crashes <sup>(113)</sup>
	Size of vehicle's blind spot
Criticality	Critical

## Table 14 Driver task description for blind spot warning.

#### Task Analysis-Fall-back mode

The vehicle warns the driver when the SHM or the RECW functions fail. The driver then takes over manual control other vehicle and exits at the next exit. Table 15 shows the task analysis.

#### Table 15 Driver task description for fall-back mode to manual driving.

Task	Resume manual control of vehicle
General	Task similar to manual driving
Initiating Cue	Warning from vehicle that SHM or RECW has failed or is degraded
Feedback	Visual, auditory, and kinesthetic cues of braking and slowing
<b>Control Functions</b>	Perception-response time (PRT) studies have found that drivers respond to unexpected
	hazards by avoidance steering as well as braking
Task Standards	PRT studies have shown for a wide population segment (young and old) PRT's on the
	average of about 1.5 sec are required to respond to an unexpected hazard, most respond
	under 2.0 sec $^{(64)}$ . However it is not certain that these data are directly applicable to
	the car-following scenario. Driver must take over vehicle throttle and brake control in a
	specified time.
Task Conditions	Weather, traffic conditions, position of following vehicle
Skills Required	Reaction times appropriate to speed, following distance, and roadway conditions.
	Perceptual judgment of vehicle kinematics
Potential Errors	Respond too slowly to transfer of control from vehicle to driver
	Vehicle slows
	Not brake hard enough
	Brake too often or too much
Causes of Error	Fails to perceive or misinterprets the indication of a failure and transfer of control
	Driver does not take over throttle control
	Driver waits for warning before braking
	Poor judgment of distance and closing rate
	Vehicle does not detect SHM or RECW failure
Consequences of	Rear-end collision with potentially severe results for the system
Error	Disrupt of traffic flow and lower capacity
	Loss of confidence by the driver and public in the system
	In the case of false alarms the driver may defeat or ignore it
Recovery Points	Driver sets longer headway or lower warning threshold for slow reactions
	Improve venicle braking capabilities
	Warn driver sooner of a "cut-in"
	Redundant sensors to improve detection Poliobility in SHM and PECW functions
Individual	DPTs for older drivers have not shown them to be less elect or significantly slower in
Difforences	responding to unexpected bezords. As noted above, it is not clear that this pertains to
Differences	the car following scenario and especially under a variety of highway conditions
Criticality	Very Critical
Criticality	Very Critical

#### Issues and Risks

#### 1. Driver confidence in warning

If the driver has too much confidence in the collision warning systems then she may not pay enough attention during lane changes or while following another vehicle. This could lead to accidents if the warning system does not detect an object or the system fails. Driver lack of confidence could cause the driver to disable or ignore the warning system. Research needs to be done on how many false alarms and detection failures are acceptable.

2. Enabling/Disabling Automatic Systems

This ERSC includes a speed and headway maintenance system, rear-end and blind spot collision warnings, and communication systems. Research needs to be done on the instrumentation needed to control these different systems. The question of whether all the automatic systems need to be controlled with one switch or many needs to be answered.

#### 3. Multiple Warnings

The blind spot and rear-end collision warnings may occur simultaneously. The vehicle must prioritize the alarms so that the driver only has to work with one task at a time.

Key Results

## Traffic Flow Control

In this section, we study the potential AHS capacity improvement due to roadway controllers that smooth traffic. Traffic tie-ups occur daily on traffic networks due to accidents and congestion. This degrades the performance of the highway system in terms of safety and average travel times. Roadway authorities have installed traffic detectors and communication links to get real-time information to central traffic control centers .<sup>(112)</sup> To date this data has only been used in the form of real-time signs that display traffic information. Studies by Chien et al. show that a more active approach can help smooth traffic after incidents .<sup>(17,19)</sup>

One of the major factors that contributes to congestion on today's freeway is disturbances or inhomogeneities of the traffic streams. Examples of these inhomogeneities are speed differences between consecutive vehicles in one lane, speed differences between lanes, and flow rate differences between lanes. In an inhomogeneous traffic stream, when the traffic volume approaches the maximum capacity, drivers are forced to drive close together and compete for the available space. Hence, gaps are immediately filled up. Shock waves may occur that originate in a chain of vehicles closely following each other at high speed and competing for available gaps. In this situation small disturbances are amplified. Traffic flow instability may occur and lead to a standstill or congestion.

If the roadway sends speed commands to the vehicles on sections of the roadway, the traffic flow recovers from incidents faster and the traffic resumes full speed in a shorter amount of time. This controller sends speed commands so that the density distribution of the dedicated lane tracks a desired or optimal density distribution based on overall traffic considerations. If the traffic density distribution along the dedicated lane could be smoothed, then the formation of the congestion due to the inhomogeneities of the traffic streams can be avoided and the time delays can be reduced. In case of incidents, the traffic density distribution could be controlled to minimize the effects of the incident on traffic flow. Thus, if the traffic density distribution could be re-shaped to a uniform profile (in case of normal traffic conditions) or a suitable one (in case of incident), the congestion can be reduced and the traffic flow rate can be increased.

To reshape the traffic density distribution along the dedicated lane, one approach is to provide proper target velocity commands to the vehicles in the dedicated lane. In this section, we propose control strategies to provide target velocity commands. We show that if the vehicles in the dedicated lane follow the recommended target speed, the traffic density distribution along the dedicated lane converges to a desired density profile. The benefit of this approach is demonstrated by simulations.

# Task D

By definition, the traffic density k is the number of vehicles occupying a unit length (No. of vehicle/km), the spaces mean speed v is the instantaneous average speed of vehicles in a length increment (km/hour) and the traffic volume q is the number of vehicles passing a specific location in a unit time (No. of vehicles /hour). As a direct consequence of the continuum variables' definitions, we have the following equation

q = k v

which is called the fundamental equation. The traffic density controller models a freeway system with a single dedicated lane that is subdivided into N sections with lengths  $L_i$ ,  $(i = 1, \downarrow, N)$  each having at most one on-ramp and one off-ramp as shown in figure 32.



Figure 32. A freeway system subdivided into sections.

At time *t*,  $k_i(t)$  is the number of vehicles in the freeway section *i* at time *t* divided by the length  $L_i$  of the section;  $v_i(t)$  is the mean speed of vehicles on the freeway section *i* at time *t*;  $q_i(t)$  is the number of vehicles leaving sections *i* per unit time; and  $r_i(t)$  and  $s_i(t)$  are on-ramp and off-ramp traffic flow rates of section *i* respectively.

Consider a dedicated lane as shown in figure 32. In general, the higher the desired traffic density, the higher the achievable input traffic flow rate. However, the upper bound of the traffic density is limited by the inter-vehicle safety distance adopted by the vehicles in the dedicated lane.

Assume a constant time headway safety policy is adopted by the vehicles in the dedicated lane

S=h v

S is the safety distance and h is the time headway. For steady-state uniform traffic flow, it is known that the traffic density is inverse proportion to the inter-vehicle safety distance. Hence, for fixed speed, the traffic density is also inversely proportional to the time headway. Therefore, the smaller the time headway h, the higher the traffic density that can be achieved and higher input traffic flow rates can be achieved.

Dynamic Traffic Flow Control

Due to variations in traffic flow such as entering and exiting (the dedicated lane) maneuvers, the traffic stream is seldom in the steady-state. Therefore, a dynamic control strategy should be employed to constantly provide dynamic target speed commands for the vehicles in the dedicated lane.

We propose a dynamic control strategy based on a traffic model proposed by Karaaslan, Varaiya and Walrand<sup>(56)</sup>. Appendix B gives the details of the proposed control strategy and this traffic model. This control strategy continuously provides speed commands to the vehicles in the dedicated lane. The traffic density distribution along the dedicated lane could be controlled to a uniform density or any other desired density profile.

#### Simulation Study

Consider a long segment of dedicated lane that is divided into 12 sections as shown in figure 32. The length of each section is 500 meters. Assume that an accident occurs at section 8 and interrupts traffic. In this situation, vehicles accumulate upstream of section 8. The accident is removed quickly. Due to the direct impact of the accident, the traffic density at section 8 is high after the accident has been removed. And the traffic density at section 6, 7 increases to an intermediate-high value due to the indirect impact of the accident. In addition, since section 8 is blocked by the accident, no vehicle can enter sections 9 - 12. The traffic density of these sections is very low. The initial traffic conditions after the accident is shown below.

Section	1	2	3	4	5	6	7	8	9	10	11	12
Initial Density (Veh/km/lane)	30	30	30	30	30	40	70	80	10	10	10	10
Initial Velocity (km/hour)	10	10	10	10	10	7.5	4.3	3.7	30	30	30	30

In ERSC 1, the traffic flow entering section 1 is in a moderate value. The traffic flow entering section 1 is shown in Figure 33. Since the accident was removed quickly, its impact to the traffic is not very severe. The figures below compare the effects of control with the case of no control. All the simulations use the traffic model described above. The first case shows a situation where an accident occurs in section 8 when there is a moderate flow rate of traffic onto the highway.



Figure 33. Traffic flow entering section 1 during case with a low input traffic flow.

Figure 34(a) shows the evolution of the traffic density without control. The disturbance in section 8 propagates back down the highway as it slowly dies out. The disturbance in traffic density is still very strong in section 1 more than 30 minutes after the original accident is cleared. Figure 34(b) shows the evolution of the traffic density when the controller is operating. The disturbance in section 8 propagates back down the highway but dies out quickly. This lets the highway quickly return to a constant flow rate. Figure 35(a) shows the velocities for each section of the roadway over time without control. Figure 36(a) shows the velocities for each section of the roadway sections. Section 8 is the section where the accident originally occurred. The traffic flow is very jerky and inhomogeneous as the accident effects propagate. Figure 36(b) shows the evolution of the traffic flow rate for the roadway sections. The traffic flow rate for the roadway sections the traffic flow rate for the roadway section of the traffic flow rate for the roadway sections. The traffic flow rate for the roadway sections the traffic flow rate for the roadway sections. The traffic flow rate for the roadway sections of the traffic flow rate for the roadway sections. The traffic flow rate for the roadway sections of the traffic flow rate for the roadway sections of the traffic flow rate for the roadway sections. The traffic flow quickly returns to its original rate. Since the traffic flow rate quickly smoothes out the highway can handle more vehicles smoothly.



Figure 34. (a) Evolution of traffic density when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8. (b) Evolution of traffic density when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8.





**(b)** 

Figure 35. (a) Evolution of traffic velocity when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8 causing changes of speed back to section 1. (b) Evolution of traffic velocity when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8 and the vehicles resume a constant speed.



**(b)** 

Figure 36. (a) The traffic flow evolution with respect to time with no roadway control. It takes more than 30 minutes (2000 s) for the traffic flow disturbances to subside. (b) The traffic flow evolution with respect to time for the roadway with roadway control. The traffic flow quickly smoothes out to the steady state velocity.

### Severe Congestion and High Entering Traffic Flow Rate

In this example, the accident at section 8 occurs during rush hour and it takes longer to be removed. Due to the direct impact of the accident, the traffic density at section 8 is high after the accident has been removed. And the traffic density at section 6, 7 increases to an intermediate-high value due to the indirect impact of the accident. In addition, since section 8 is blocked by the accident, no vehicle can enter sections 9-12. The traffic density of these sections is very low. The initial traffic conditions after the accident are shown below.

Section	1	2	3	4	5	6	7	8	9	10	11	12
Initial Density (Veh/km/lane)	30	30	30	30	30	70	80	97	10	10	10	10
Initial Velocity (km/hour)	10	10	10	10	10	4.3	3.7	3.0	30	30	30	30

In ERSC 1, the traffic flow entering section 1 is a higher value like in rush hour. The traffic flow entering section 1 converges to 2000 vehicles per hour. Since the accident took a long time to remove, its impact to the traffic is severe. The figures below compare the effects of control with the case of no control. All the simulations use the traffic model described in appendix B. The first case shows a situation where an accident occurs in section 8 when there is a moderate flow rate of traffic onto the highway.



Figure 37. The traffic flow evolution with respect to time for the roadway with roadway control. The traffic flow quickly smoothes out to the steady state velocity.

Figure 37 shows the evolution of the traffic flow for the case with control. Figure 38(a) shows the evolution of the traffic density without control. The disturbance in section 8 propagates back down the highway and causes serious congestion. The disturbance in traffic density is still very strong in section 1 more than 30 minutes after the original accident is cleared. Figure 38(b) shows the evolution of the traffic density when the controller is operating. The disturbance in section 8 propagates back down the highway but dies out quickly. This lets the highway quickly return to a constant flow rate. Figure 39(a) shows the velocities for each section of the roadway over time without control. Figure 39(b) shows the velocities for each section of the roadway over time with control.



**(b)** 

Figure 38. (a) Evolution of traffic density when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8. (b) Evolution of traffic density when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8.

Raytheon



**(b)** 

Figure 39. (a) Evolution of traffic velocity when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8 causing changes of speed back to section 1. (b) Evolution of traffic velocity when the roadway controller is present. The disturbance quickly dies out after the accident is removed from section 8 and the vehicles resume a constant speed.

Raytheon

## Safety

A rear-end collision warning system needs to be able to accurately predict the likelihood of a collision. If a collision is likely, then the system warns the driver to respond to the threat. Analysis of minimum time headways between vehicles suggests that small reductions in driver reaction times can dramatically decrease the minimum safe headway. Reductions in the reaction time may be achieved by warning the driver. Many accidents are due to failures in human information processing.<sup>(59)</sup> In many cases the other vehicle was clearly visible. This again suggests that a rear-end collision warning system may increase driver safety.

Most (95%) of lane change/merge crashes are angle or sideswipe collisions .<sup>(113)</sup> The driver of the merging car either did not see the other vehicle or misjudged the distance between the vehicles. These crashes usually happen in dry, clear, daylight conditions.<sup>(113)</sup> The driver "did not see" the other vehicles until too late. The blind spot or "lateral encroachment" warning system on the vehicle detects vehicles in adjacent lanes that are traveling in the same direction.<sup>(59)</sup> The blind spot detection warning system has a sensor and a system that warns the driver.

## Issues and Risks

1 Controller Algorithms

The controller algorithms need to smoothly adjust the brake and throttle to avoid sudden accelerations when the vehicle ahead leaves the lane. Driver tests indicate that drivers feel unsafe when the controller suddenly accelerates.<sup>(26)</sup> The SHM controller needs to use alternative algorithms that limit accelerations.<sup>(51,25)</sup>

## 2 Electromagnetic Interference

Full deployment of ERSC 1 means that most vehicles will have active sensors such as radar. When multiple radar operate in a small area at similar frequencies the radar may interfere with one another. This interference could lead to shorter sensor ranges and poor performance.

## 3 Driver Interface

Research needs to be done on the driver interface with the SHM, RECW, and blind spot warning functions. Each function may have separate controls for enable/disable and adjustments. It may be too much of a burden on the driver to expect him/her to change lanes and start the automated systems immediately. The driver may need to preset the target speed, minimum headway, and warning thresholds when he/she starts the vehicle.

## 4 Sensor Capabilities

The RECW and blind spot warning functions may increase safety and reduce certain types of collisions by decreasing driver reaction time. But there are sensor and human factor issues that still must be resolved. The sensors must detect small vehicles or debris, such as motorcycles with few nuisance alarms.

## 5 Warning system design

First the designers need to decide when the warning system should be turned on: always on, only when turn signal on, or turned on when steering column shows lane change. The warning

thresholds for emergencies and driver awareness must also be set. The system must avoid nuisance alarms, yet detect vehicles accurately. The types of alarms and warnings must also be researched. The emergency alarm needs to be directional to get fast driver response. ERSC 1 introduces multiple types of driver warnings for blind spot and rear-end collision avoidance. Research needs to be done on the interaction and prioritization of these warnings for the human driver.

### 6 Dedicated Lane

The biggest social issue for AHS is how to get a separate lane for automated vehicles. The driving public rebels when a lane is taken away from active use.<sup>(99)</sup> The solution for most highway departments has been to add a new lane on the shoulder or median strip, but most urban highways do not have any more space to spare for the sole use of automated vehicles. This infrastructure change may have to wait until enough vehicles have speed and headway maintenance systems and traffic flow benefits can be easily seen.

### 7 Legal/Liability

In this ERSC the roadway sends recommended target speeds and minimum headways directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the minimum recommended headway is too short.<sup>(102)</sup> The government may not wish to deploy such a system until it has been shown to be reliable and fail-safe.

### 8 Privacy

This ERSC gives the vehicles the option to "Check-in" with the roadway. This may leave a record of when and where a vehicle traveled. This raises user privacy concerns about governmental use of this data.

### 9 Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

## Key Findings

1 The SHM function does not have to be very accurate in this ERSC (5-10%). The human driver takes over all hard braking and collision avoidance. Human drivers have relatively slow perception and reaction times. This suggests that SHM systems at this level may come from the intelligent cruise control systems planned by several companies .<sup>(74,26)</sup> The next ERSC requires a more accurate SHM function. The components for a more accurate SHM function may need to be designed with AHS requirements in mind instead of taking commercial technology off the shelf.

2 There are significant safety and performance benefits even with low levels of automation. Simple systems like the roadway velocity controller can give large increases in capacity and traffic flow volumes. This system uses technology available today in traffic control centers and toll roads.

3 The reliability functional requirement for the warning systems is that they have high detection rates and low false alarm rates. Vehicle to vehicle communications systems can reduce

the number of false alarms in the collision warning systems without reducing the probability of detection.

4 The driver can serve as a back up to the automated systems at this level. If either the SHM or the RECW system fails, then the driver takes over.

5 There are significant human factors issues for warnings and the driver interface that must be researched for even low levels of automation. The warning systems will require multiple levels of warnings.

6 The reliability functional requirement for the SHM function is that it must safely return control to the driver in case of override.

# Raytheon

#### SECTION 6: ERSC 2 ANALYSIS

#### Description of ERSC 2

In ERSC 2, the roadway provides a dedicated lane with the same vehicle accessibility as in ERSC 1. The roadway assists in keeping non-fit vehicles out of the dedicated lane. The roadway receives speed and headway data from the vehicles. It computes the desired target speed at each section of the lane for smoother traffic flow. It sends minimum time headway information to the vehicles based on environmental conditions. It uses the operational status and identification of the vehicle for check-in purposes and for roadway emergency response in case of disabled vehicles.

The vehicle has all the capabilities as in ERSC 1, with the rear-end collision warning being upgraded to rear-end collision avoidance. The SHM function and rear-end collision avoidance allow the driver complete "feet-off" driving in the dedicated lane. The vehicle provides steering assist to smooth the driver's steering by compensating for roadway disturbances, wind gusts, and small driver steering errors. The vehicle also has a lane departure warning to warn the driver of large deviations from the center of the lane. The vehicle sends its speed and headway, operational status and identification to the roadway. It sends its braking capabilities and acceleration/deceleration intentions to the vehicle behind. It receives the braking capabilities and acceleration/deceleration intentions of the vehicle in front. As in ERSC 1 the vehicle has on-board diagnostics that are used to detect malfunctions and alert the driver. The vehicle provides a fall-back mode to ERSC 1 in case the rear-end collision avoidance or speed and headway maintenance functions fail.

The motivation behind ERSC 2 is to

- increase capacity by using smaller headways mad e possible by the rear-end collision avoidance function.
- improve safety and driver comfort by introducing lane departure warnings and steering assist
- smooth traffic flow with roadway specified vehicle target speeds.

Function	Requirement	Driver/Vehicle/Roadway Tasks
Speed and Headway Maintenance	- Maintain the selected cruising speed when no vehicle is ahead. - Calculate and maintain a headway for collision-free vehicle following.	<ul> <li>Respond to driver commands for setting the minimum headway and maintain at least that minimum headway</li> <li>Respond to driver commands for enabling and disabling the SHM function safely</li> <li>Respond to roadway commands for following target speeds in a smooth manner</li> <li>Respond to roadway commands for increasing the minimum headway in a smooth manner</li> <li>Compute a safe minimum headway based on vehicle braking characteristics and those of the preceding vehicle, and environmental conditions</li> </ul>
Rear-end Collision Avoidance	The rear-end collision avoidance function should avoid rear-end collisions due to moving or stationary obstacles in the lane under all road and environmental conditions.	<ul> <li>Detect any obstacle in the lane ahead in time to stop or avoid the obstacle.</li> <li>Actuator response should be fast and free of failures</li> <li>Low rate of false or nuisance alarms</li> <li>Deceleration and jerk may reach maximum possible values in emergency stop</li> <li>Communicate braking capabilities, decelerations, and intentions between vehicles to reduce false alarms and increase safety</li> </ul>
Blind spot warning (described in Section 5)	Warn the driver of a potential lane change/merge collision due to moving or stationary obstacles in the vehicle's blind spot	Same as in ERSC 1.
Lane Departure Warning	Warn the driver when the vehicle is in danger of leaving the lane	<ul> <li>Sense the vehicle's position in the lan and estimate its course</li> <li>Senses roadway preview information</li> <li>Sends driver cautionary and imminen- threat warnings of potential lane departures</li> <li>Have a high detection rate and a low rate of false alarms</li> </ul>
Steering Assist	Works in series with driver to compensate for disturbances due to wind gusts and road surface conditions	<ul> <li>Sense vehicle's yaw rate and slip angle to compensate for high frequency disturbances</li> <li>Increase control bandwidth and make vehicle easier to drive.</li> </ul>

Table 16. E	ERSC 2 functions and	driver/veh	icle/roadway	performance
	requiren	nents.		

Driver Interface	<ul> <li>Adjust thresholds on warnings</li> <li>Enable/Disable automatic functions</li> <li>Adjust headway</li> <li>Steer vehicle with help from steering assist</li> </ul>	<ul> <li>Driver should be able to:</li> <li>adjust headway</li> <li>adjust warning thresholds</li> <li>enable/disable automatic functions</li> <li>steer vehicle</li> <li>Interface should be simple to understand and use</li> <li>No adjustments should be required that take driver's attention away from driving</li> </ul>
Roadway/Vehicle speed & headway commands	Senses vehicle average speed and density and environmental conditions then sends commands directly to vehicle to smooth traffic flow	<ul> <li>Roadway measures the average speed and traffic density, and assesses environmental conditions, and calculates vehicle target speeds and minimum safe headways</li> <li>Vehicles may display relevant traffic information to driver</li> <li>Roadway monitors vehicle status at check-in</li> </ul>
Fall-back Mode to ERSC 1	Driver should be warned of system failures or degradation in time to recover from dangerous situations	<ul> <li>Sense status of different vehicle control channels</li> <li>Safely transfer control to the driver if SHM function or rear-end collision avoidance fails</li> </ul>

Performance Requirements

#### Speed and Headway Maintenance

#### Functional Requirement

The speed and headway maintenance (SHM) function should maintain the selected cruising speed when no vehicle is ahead. It should calculate and maintain a minimum headway for collision-free vehicle following. The SHM must smoothly switch between vehicle following mode and constant speed cruising. The SHM function in ERSC 2 performs the same tasks as in ERSC 1, but they have to be more accurate if smaller headways are used for higher capacity.

The vehicle receives and responds to target speed commands and headway recommendations sent by the roadway in the same way as in ERSC 1. The braking capabilities and acceleration/deceleration intentions of the vehicle in front as well as the headway recommendations sent by the roadway are used by the vehicle to calculate the minimum time headway for collision-free vehicle following. The SHM function chooses a headway that is larger than the calculated one based on some tolerance for additional safety. Since the driver is not considered to be a back up in a rear-end collision avoidance situation the calculated headway does not take into account the driver reaction time and is therefore smaller than the one used for collision warning in ERSC 1. The vehicle responds to driver commands for changing the headway. It also enables/disables the SHM and rear-end collision avoidance functions upon driver commands. If the driver disables the rear-end collision avoidance

function the vehicle initiates a smooth transition sequence that does not put the driver in a dangerous situation. This transition may require a larger headway since human drivers have a longer reaction time compared to an automatic rear-end collision avoidance system.

The SHM function operates by measuring the relative speed and spacing, then sending commands to the throttle and brake actuators. Figure 40 shows the structure of the SHM function. This section will discuss the sensor and actuation requirements, control algorithms, selection of a safe headway, and driver/roadway interface for the SHM function.



Figure 40. Block diagram of the speed and headway maintenance function. The sensors measure the relative speed and spacing of the vehicles, then the controller sends commands to the brake and throttle actuators so that the vehicle maintains the target speed or minimum headway.

#### Sensor Requirements

To maintain a target velocity the vehicle must be able to measure its own speed. The sensor must measure speeds up to 160 km/h (100 mph) or any speed that may be used in the dedicated lane. The sensor needs to be accurate within about 3 km/h (1.8 mph) since the SHM function needs an accurate velocity measurement to select a safe headway for vehicle following.

To maintain a constant time headway the vehicle must be able to measure the distance to the target vehicle ahead. The vehicle must also estimate or measure the closing rate between vehicles. Figure 41 shows the sensor region for headway maintenance. The sensor region shows the sensor requirement for a vehicle with a rear-end collision avoidance system that must detect obstacles as soon as they become a threat.



Figure 41 Sensor requirements for range and closing rate detection for the speed and headway maintenance system.

There are several requirements that determine the maximum sensor range. The sensor needs to work for the collision avoidance and the fall-back mode to ERSC 1 functions as well. The sensor range needs to be the maximum value for these requirements. <sup>(29)</sup> The first requirement is that the vehicle must be able to safely return control to the human driver if the driver decides to turn off or override the SHM function and return to manual driving. The California Driver's Handbook states that 3-4 seconds is a safe time headway for human drivers. If we assume a speed of 60 mph (26 meters/s) this gives a distance of 78-104 meters. Therefore the vehicle range sensor should work over a distance of at least 80-105 meters. The second requirement is that the system be able to stop when there is debris or another vehicle stopped in the roadway ahead. This scenario is known as the "brick wall" scenario. Rear-end collisions with a stationary vehicle outnumber rear-end collisions where the lead vehicle is moving by 2:1.<sup>(60)</sup> Figure 42 shows some range curves for the brick wall scenario for different velocities and a detection time of 0.5 seconds for the rear-end collision avoidance system. For example, at 60 mph we get a range of 80 meters for a vehicle with a maximum braking capability of -0.55 g. Table 17 gives the parameters used in this calculation. This calculation does not take human reaction times into account since fully automated longitudinal control is achieved by the SHM and rear-end collision avoidance functions.

For the vehicle to operate in the "fall-back" mode, the sensor needs to meet the range requirements for ERSC 1 as well. In the fall-back mode the vehicle returns to ERSC 1 if one of the control channels for the rear-end collision avoidance function fails. This implies a range of 120 meters as calculated in section 5.





Figure 42. The maximum distance sensor range versus the maximum vehicle deceleration for the "brick wall" scenario at two different velocities. The maximum vehicle deceleration is the vehicle capability on the roadway at a given time. It reflects the road-tire friction coefficient and vehicle's capabilities.

The sensor must cover the lane so it does not miss the target vehicle. But it must also minimize interference from adjacent lanes which means that the sensor beam cannot spread out too much or it must use multiple beams to reject interference. The sensor must be able to track the vehicle around highway curves without losing the target. Reliable sensing may require roadway curvature preview data to track vehicles around curves.

The vehicle must also detect or estimate the closing rate between itself and the target vehicle. This may be done directly with a pulse-doppler radar or it may be estimated by taking the derivative of the distance measurement over time. The sensor accuracy needs to be within 5% of the true value

The most common type of distance sensor measures the time-of-flight from the time that a pulse is sent out to the time that the pulse returns. These time-of-flight sensors can measure the range, relative speed, and angular position of the target vehicle. These sensors are commercially available for automotive use today. <sup>(12; 33; 76)</sup> Other sensors use triangulation techniques such as stereo to measure the distance to the vehicle ahead. These sensors include vision-based systems and radar. (68; 86)

#### Control Algorithms

The vehicle must respond quickly and smoothly to commands from the driver and roadway. The controller must also limit the jerk and accelerations in normal operation. Humans find longitudinal accelerations above 0.17 g and jerks above 0.2 g/s uncomfortable at low speeds.<sup>(15)</sup> At high speeds the deceleration should be kept below 0.1g with jerk as low as possible. Soft braking should be limited to around 0.2 g with low jerk.<sup>(15)</sup>

The controller must switch smoothly between the vehicle following and cruise modes and avoid sudden accelerations when the preceding vehicle exits the dedicated lanes. Fancher et al showed that abrupt accelerations by the automated vehicle made drivers feel uncomfortable and unsafe.<sup>(26)</sup> The controller must also switch smoothly between the throttle and brake without high frequency oscillations.

The controller calculates a safe minimum headway for vehicle following. This calculation uses the deceleration values of the vehicle itself and the preceding vehicle, environmental conditions measured by the vehicle and roadway, and the velocity of the vehicles.

#### Actuator Requirements

The brake and throttle actuators must respond quickly to commands from the controller. Faster actuators may give better control. If the delay is too long then vehicle following may be unstable if there are many vehicles in a row following one another. <sup>(50)</sup> Delays less than or equal to 0.1 seconds should be sufficient for ERSC 2. The actuators must also be accurate. The actuator delays should be taken into account in the calculation of the minimum safe headway.

#### Safe Headway Selection

The vehicle calculates a safe headway between vehicles. This calculation requires data on the braking capabilities of both vehicles, estimates of the road-tire friction coefficients for both vehicles and measurements of the vehicles' velocities. <sup>(101)</sup> Each vehicle can assess its own braking capabilities with built-in tests that track braking. The "check-in" procedures may also asses how well a vehicle can brake before the vehicle enters the highway with a built-intest function that monitors the vehicle braking history. <sup>(108)</sup> The preceding vehicle can send its estimated braking capabilities to the vehicle behind it. This will help the vehicles choose a safe following distance. Without this data from the preceding vehicle or if the estimate appears incorrect, the vehicle chooses a worst case braking condition for the preceding vehicle such as A<sub>lm</sub>=1 g to calculate the headway.

Table 17 gives the variables used in the minimum time headway calculations for ERSC 2. Unless noted these values are used for all the figures. Figure 43 shows the minimum time headway for different maximum deceleration values of the follower car. When the follower vehicle decelerates faster than the lead vehicle the minimum time headway decreases. When the follower vehicle decelerates slower than the lead vehicle the minimum time headway

Table 17 Variable values used in the simulations in SHM section for ERSC 2.					
Variable	Description	Value			
Name					
V <sub>l</sub> , V <sub>f</sub>	Initial velocity of the lead and following vehicles	100 km/h			
J <sub>lmax</sub> , J <sub>fmax</sub>	Maximum jerk of the lead and following vehicles	72 m/s <sup>3</sup>			
mlmax, mfmax	Maximum road-tire friction coefficient of the lead and	1,1			
	following vehicles				
A <sub>lm</sub> ,A <sub>fm</sub>	Maximum deceleration of the lead and following vehicles	0.85 g,0.8 g			
	respectively				
a <sub>fac</sub>	Acceleration under vehicle following situation	0.05 g			
a <sub>fauto</sub>	Acceleration value for soft braking	0.20 g			
J <sub>fc</sub>	Jerk value for soft braking	20 m/s <sup>3</sup>			
tb	Braking actuation delay	0.1 s			
T(1)	Detection delay of the automated vehicle headway system	0.25s			
tfc	Time when vehicle begins hard or emergency braking	0.35 s			

increases. This suggests that proper minimum time headways should be set as if the lead vehicle has a high maximum possible acceleration.



Figure 43. Minimum safe time headway with respect to differences in the maximum deceleration between vehicles  $\pounds A_m = A_{lm} - A_{fm}$ . A negative acceleration value means that the follower car can brake faster than the lead car. ERSC 1 is the headway when the driver is responsible for any emergency braking situations. ERSC 2 shows the minimum headway when the vehicle has a rear-end collision avoidance function.

## Raytheon



Figure 44. Minimum safe time headway with respect to differences in the initial velocity of the vehicles  $\mathcal{E}V_m = V_{fm} - V_{lm}$ . A negative velocity difference means that the follower car is going faster than the lead car. ERSC 1 is the headway when the driver is responsible for any emergency braking situations with a 1.5 second human reaction time. ERSC 2 shows the minimum headway when the vehicle has a rear-end collision avoidance function.

Figure 44 shows the minimum time headway for different initial velocity values of the follower car. When the follower vehicle is going faster than the lead vehicle the safe time headway increases. When the follower vehicle decelerates slower than the lead vehicle the safe time headway decreases. This shows the needed gaps for cars changing lanes in and out of the dedicated lane from the manual lanes. If there is a large speed difference between lanes, vehicles will need to find large gaps to merge into the automated lane.

The condition of the roadway surface also affects the safe headway in between vehicles. Figure 45 shows the time headway for different road-tire friction coefficients. Weather conditions and roadway surface can affect the friction coefficient. <sup>(5,41)</sup> The antilock braking system on board the vehicle helps achieve the maximum deceleration without sliding the tires. This maximum deceleration is limited by the tire-road friction coefficient  $\mu_x$ . The road-tire

**Page 136** 

friction coefficient is weather and speed dependent. This coefficient may be estimated by the vehicle over time.  $^{(41)}$ 

If the follower vehicle decelerates slower than the vehicle in front then the minimum time headway increases. This figure assumes that both vehicles are on the same surface. For example it's raining and the roadway is equally wet for both vehicles. Localized patches of ice or puddles can cause large changes in the friction coefficient. This leads to fast changes in  $\mathcal{E}A_m$  and the minimum time headway.





Figure 45. As the road-tire friction coefficient decreases (the road gets more slippery), the minimum safe time headway increases.  $\mathcal{E} A_m = A_{lm} - A_{fm}$  is the deceleration difference between the vehicles in g. A positive value means that the lead car has a larger deceleration than the follower vehicle.

Figure 46 shows the minimum time headway for different initial velocities. Faster velocities lead to slightly larger minimum time headways. This parameter has a small effect on the final calculation of a safe headway. This implies that the velocity measurement accuracy may not be an important design requirement.

# Raytheon





## Driver/Roadway Interface

The vehicle responds to driver commands for switching the SHM function on and off, selecting the minimum headway, and overriding the SHM function. It also responds to roadway speed commands, headway recommendations, and displays traffic status information received from the roadway to the driver.

The driver turns the SHM function on and off. The SHM function may be turned on before the driver enters the dedicated lane or once the driver has successfully merged into the lane. Tests on a vehicle with one experimental SHM system show that most drivers do not feel safe using this automated system while merging.  $^{(26)}$  This switch needs to be standardized and clearly marked so that the driver can begin automated operations quickly and easily. If the driver turns off the SHM function, the vehicle first increases the headway between itself and the vehicle in front since the human driver has a slower reaction time than the automated rearend collision avoidance system. Then it transfers brake and throttle control to the driver. The check-out function regulates this transition.  $^{(108)}$ 

The vehicle responds smoothly to target speed commands from the roadway when the vehicle is in cruise mode and it is safe. The vehicle uses the time headway recommendations from the roadway and driver as lower bounds for safe headway selection.

The SHM function must clearly inform the driver of its goals and intentions so that the driver knows that it is working properly. This means that driver needs to know whether the vehicle is in following or cruise mode. The driver also needs to know the target speed and headway. This will help the driver monitor the vehicle's performance and status.

#### Issues and Risks

#### 1 Function Accuracy

The SHM function has to be very accurate in this ERSC. The components for a more accurate SHM function need to be designed with AHS requirements in mind instead of taking commercial technology off the shelf.

#### 2 Controller Algorithms

The controller algorithms need to smoothly adjust the brake and throttle to avoid sudden accelerations when the vehicle ahead leaves the lane. Driver tests indicate that drivers feel unsafe when the controller suddenly accelerates.(26) The controller needs to use algorithms that limit accelerations.(51,25)

#### 3 Electromagnetic Interference

Full deployment of these ERSCs means that most vehicles will have active sensors such as radar. When multiple radar operate in a small area at similar frequencies the radar may interfere with one another. This interference could lead to shorter sensor ranges and poor performance.

#### 4 Driver Interface

Research needs to be done on the driver interface to the SHM function. This device may have separate controls for enable/disable and minimum time headway. It may be too much of a burden on the driver to expect him or her to change lanes and start the SHM immediately. This function may require the driver to preset the target speed and minimum headways when he/she starts the vehicle. Human factors studies with pilots show that touch pads and buttons distract the pilot and may cause accidents.(8)

#### Rear End Collision Avoidance

The rear-end collision avoidance function should avoid rear-end collisions due to moving or stationary obstacles in the lane ahead under all road and environmental conditions. The rearend collision avoidance (RECA) function needs to analyze threats fast enough to take action and override the SHM function when emergency braking is needed. The RECA also needs a low rate of false or nuisance alarms. Vehicle-to-vehicle communication of braking capabilities, decelerations, and intentions are useful in reducing false alarms. Figure 47 shows the block diagram of the rear-end collision avoidance system.

Figure 48 shows some different rear-end collision avoidance scenarios for this function. Rear end collision can be avoided by emergency braking and/or lateral maneuvers. In this ERSC,

we consider rear end collision avoidance achieved by automatic emergency braking only. Lateral maneuvers are performed only by the driver in this ERSC.

The system detects the headway and the preceding vehicle's velocity/acceleration and uses these data to determine the required deceleration. If the vehicle decides that the soft braking of the SHM function can not provide enough brake force, then the rear end collision avoidance system will brake more to stop the vehicle. The driver's comfort will be affected and driver's steering control will be crucial for keeping the vehicle in the lane when harder braking is needed. The vehicle needs to warn the driver of the braking action to be taken.

The setting of the headway threshold for emergency braking should be such that the vehicle can brake safely without collision. The headway threshold is a function of the road-tire friction coefficients, the vehicle speed, and the vehicle braking capabilities. A conservative threshold setting will improve longitudinal control safety. It will also reduce the highway capacity.



Figure 47. Block diagram of the rear-end collision avoidance system. The sensor measures the distance to obstacles. The controller assesses collision potential. This controller overrides the SHM function if an emergency stop is required.



Figure 48. Different collision avoidance scenarios for the rear-end collision avoidance function. (a) The vehicle ahead begins an emergency stopping maneuver. (b) Another vehicle is stopped in the lane ahead. (c) The preceding vehicle performs an evasive maneuver by steering out of the lane to avoid a vehicle or debris ahead. (d) Another vehicle cuts-in ahead.

#### Sensor

The rear-end collision avoidance system uses a velocity sensor, a ranging sensor and a closing rate sensor to recognize potential collisions. These inputs allow the vehicle to estimate the time to collision that is the gap distance over the closing rate, the time headway, and the safe stopping distance. Sensors that estimate the roadway coefficient of friction may also improve the performance of a RECA system. <sup>(41)</sup>

The second requirement is that the system be able to stop when there is debris or another vehicle stopped in the roadway ahead. This scenario is known as the "brick wall" scenario. Rear-end collisions with a stationary vehicle outnumber rear-end collisions where the lead vehicle is moving by 2:1.  $^{(60)}$  Figure 42 shows some range curves for the brick wall scenario for different velocities and a detection time of 0.5 seconds for the rear-end collision avoidance system. For example, at 100 km/h we get a range of 80 m for a vehicle with a maximum braking capability of -0.55 g. Table 17 gives the parameters used in this calculation. Figure 49 compares the stopping distances for ERSC 1 and ERSC 2 for the brick wall scenario. Since the driver's reaction time is eliminated in ERSC 2 maximum sensor range for safe stops is smaller.



Maximum Deceleration, Afm (g)

Figure 49. The maximum sensor range with respect to the maximum deceleration of the vehicle at 60 mph for ERSCs 1 and 2 (SHM without and with rear-end collision avoidance) These values come from the "brick wall" scenario when the lead vehicle is stationary or there is debris on the highway.

The sensor must also detect vehicles that cut-in in front of the automated vehicle. A single beam sensor does not see the cut-in until it enters the beam as shown in figure 11. The speed of detection depends on the area covered by the sensor, the lateral velocity of the vehicle that is cutting-in, and the distance of the cut-in from the vehicle. Most lane changes can take about 2-6 seconds. <sup>(21)</sup>

# Task D

Since this system may not allow the driver to override the brake and throttle controllers, the sensors must detect all dangerous situations that the driver can see. Human factors studies on function allocation recommend that drivers should be able to override automated functions since a human driver may detect danger that the sensor does not see. (70) This implies that the vehicle must use a vision-based sensor to see what the driver sees. The problem lies in scene interpretation. Humans are superb at interpreting general scenes and identifying objects such as cardboard boxes or a dog loose on the highway. Computers require large databases to perform the same task. Most vision-based sensors track road lane lines or look for simple structures in the scene. (76,68) Other systems use "optical flow" to track objects.

There are no computer vision systems available today that can interpret a general scene as well as human driver. Current systems look for expected objects such as other vehicles, signs, and lane lines. Humans can see animals and debris.

#### Control Algorithm

The warning algorithm for rear-end collision avoidance is based on the time-to-collision (TTC) or minimum headway between the lead and following vehicles.



Figure 50. Changes in the detection time for lane change/merge collisions have a large effect on the potential accident severity.

The collision avoidance system must warn the driver when the vehicle ahead is braking without false alarms. High rates of false alarms may cause the driver to ignore the warnings, second guess the system, or turn off the system. This problem has been observed in the

Traffic alert and Collision Avoidance System (TCAS) for airplanes. <sup>(96)</sup> This system recommends and coordinates actions for aircraft. False alarms cause hesitation and second-guessing by the pilots.

The follower vehicle must detect that the lead vehicle is starting to brake hard. This can be done either with the detection of a change in the relative velocity or acceleration between the vehicles or with the reception of an emergency transmission from the leading vehicle (a sort of electronic brake light). A deployed system may require both measurements for an emergency warning to the driver.

#### Vehicle to Vehicle Communications

The vehicle needs to estimate the time-to-collision (TTC) or minimum headway as described above. When the car ahead begins to brake with a deceleration,  $a_{bmax}$ , the TTC decreases. However the measurements for the distance and closing rate between the vehicle itself and the target vehicle can be quite noisy and inaccurate. The RECA function must distinguish between measurement noise in the controller and a true braking maneuver. A communications system lets the vehicle know the intentions of the preceding vehicle. The lack of this information may cause unsafe maneuvers or excessive braking. Appendix A gives the derivation of this relationship between detection time, false alarms, and detections.

Figure 51 shows the relationship for  $P_D$  and  $P_{FA}$  with respect to d. These parameters are explained in more detail in the appendix to the lateral and longitudinal control section. The  $= \infty$  corresponds to the case where the receiver always decides no curves change with acceleration. = 0 corresponds to the case where the receiver always decides that acceleration is present. In this case the cost for a missed detection is very high since the driver may not have an immediate override capability. The cost of a false alarm is lower since a false alarm only results in excess braking and lowered capacity. This gives a < 1.This shows that as the time available to detect a deceleration ahead increases for constant noise and accelerations, the performance of the detector improves. This may require a d on the order of 4 or higher. For example if the maximum deceleration of the lead vehicle is 8  $m/s^2$  and the noise deviation is 1-2 m/s for the sensor and controller noise then t > 0.4 s gives d=5. This means that we require at least a half second after the vehicle ahead has started braking to make a good decision. Note that the brake actuator delay (0.1 s) increases the total time delay to around 0.5 seconds before the RECA begins braking.



Figure 51. The probability of correctly deciding that the vehicle ahead is decelerating depends on the space between the two signals d. As d increases the detection-false alarm trade-off improves.

Vehicle-to-vehicle communications sends braking capabilities, velocity and acceleration data directly between vehicles. The probability of false alarm is low. As the lead vehicle brakes it sends its brake data directly to the vehicle behind. This braking data only needs to say that the vehicle in front is braking hard to communicate intent. This communications system can have a very short delay on the order of 0.05 to 0.1 seconds.

Each vehicle sends its deceleration to the car behind it. This lets the car behind anticipate any braking before changes in the distance or velocity can be measured. This data may be in the form of brake line pressure or an accelerometer measurement. Figure 52 shows how accident severity is affected by the deceleration capabilities of the vehicles.


Figure 52. The deceleration difference between the vehicles  $\mathcal{E} A_m = A_{fm} - A_{lm}$  has a large effect on accident severity. As  $\mathcal{E} A_m$  increases the potential accident severity increases.

Other forces that decelerate a vehicle are the aerodynamic drag coefficient, the friction between the car and the roadway, and the slope of the roadway. The aerodynamic drag stays constant over the life of the vehicle. It may be compensated for by calibrating each car model. The frictional force between the car and the road depends on the roadway, the tire condition, and the mass of the vehicle and passengers. The road-tire friction coefficient may be estimated by measurements from an anti-lock braking system sensors. <sup>(40)</sup> This

measurement coupled with an on-line estimate of the vehicle's maximum deceleration can give the vehicle's estimated braking capability.

Data Contents	Source	Number of Bits
Road-tire friction coefficient	Estimated by vehicle	4-6
Maximum deceleration capability	Set by manufacturer or estimated by vehicle	4
Braking Intentions (Brake line	Measurement of brake line pressure or RECA intent	4
pressure)		
Velocity	Measured by vehicle	8
Acceleration	Measured by vehicle	4
	Total bits	22

Table 18. Data used in the message for vehicle to vehicle communications.

Table 18 shows the fields needed in message passed between vehicles. This data may require on-line calibration to ensure that false braking signals are not sent out. The message between vehicles needs a length of around 65 bits in the worst case. This gives 16 bits for synchronization. <sup>(100)</sup> 8 bits each for the source and destination IDs if local addressing is used. The data field is 22 bits. 16 bits of error correction can help detect transmission errors. <sup>(100)</sup> If directional systems such as infrared (IR) or radio are used then the source and destination fields may not be necessary. <sup>(6)</sup>

The communications system must transmit over a range of at least 40 meters in a single lane. The signal delay is 0.05 seconds. The transmission rate is 20\* M bits per second with a message length of M bits. The worst case transmission rate is 1100 bits per second. The transmission delays should be less than 20 ms since the brake data is time-critical. If the decision time is 0.1 seconds, the RECA receives at least two messages before beginning an emergency stop. The probability of message error ( $P_m$ ) should be very low. Research at California PATH has produced IR systems that meet this data rate and accuracy requirement.<sup>(31)</sup> The tested system can transmit at 19.2 Kbits/second over a range of 30-40 meters.<sup>(31)</sup> As the distance between vehicles decreases the allowable data rate increases.

# Driver Interface

The rear-end collision avoidance function needs to warn the driver that the vehicle is braking. This assures the driver that the system is working correctly. The warnings end when the vehicle takes action to end the threat.

The warning should be visual and located within 15-30 ° above the driver's line-of-sight for ordinary driving. <sup>(64)</sup> Visual signals are more detectable if they are near the center of the line-of-sight. The warning needs to give the driver informational feedback for headway maintenance and vehicle performance. <sup>(37)</sup> A headway display could provide this type of feedback with a heads-up display. <sup>(116,83)</sup> This type of display could also assist the driver in selecting a safe headway.

Issues and Risks

1 Sensor Capabilities

The RECA system may increase safety and reduce certain types of rear-end collisions by decreasing driver reaction time. But there are sensor and human factor issues that still must be resolved. The sensors must detect all vehicles or debris that a human can see with few nuisance alarms. A video system may be able to see small vehicles better during daylight. The video system sees what the driver sees so it may be more natural for human drivers to use. The performance of vision based systems may degrade at night and in rainy weather plus they are more expensive and complicated than radars or ultrasonic systems.

#### 2 Collision avoidance system design

First the designers need to decide when the RECA function should be turned on: always on, only when the SHM function is on, or on only at higher speeds. The warning thresholds for emergencies and driver awareness must also be set. The system must avoid nuisance alarms, yet detect vehicles accurately. The types of alarms and warnings must also be researched. The emergency alarm needs to get consistent and fast driver responses.

#### 3 Multiple Warnings

ERSC 2 introduces multiple types of driver warnings for blind spot, lane departures, and rear-end collision avoidance. Research needs to be done on the interaction and prioritization of these warnings for the human driver.

#### 4 Driver Override

This system may not allow the driver to override the brake controller immediately, without first going through check-out. This design may keep the driver safer by keeping the RECA active. On the other hand, it may interfere with the driver's ability to perform lateral collision avoidance. This area needs research to determine the best configuration.

# Lane Departure Warning

The lane departure warning helps the driver safely follow the lane. When the system senses a lane departure danger, it sends a signal to warn the driver. This system consists of a lane edge detector, a warning algorithm, and a warning signal generator. Figure 53 shows the structure of lane departure warning system. The lane departure warning helps prevent single-vehicle roadway departure crashes. These crashes cause 37.4% of all fatalities, more than any



Figure 53. Block diagram of the lane departure warning system. The lane position sensors give the vehicle's position in the lane. The warning system calculates the time-to-lane-crossing and warns the driver if necessary.

other type of crash.  $^{(114)}$  SVRD crashes are 20.8% of all police-reported crashes.  $^{(114)}$  The lane departure warning may help prevent the crashes that occur when the driver loses control and drives off the road. But many of these crashes occur when the driver is trying to avoid a rear-end collision. Kinematics analyses of single vehicle roadway departure crashes indicate that for most crashes, the warning would come too late and the driver would not have time to respond  $^{(72)}$ 

### Sensor Requirements

The sensor needs to detect the vehicle's position and heading. The lane edge detector will require some form of lane reference aid provided by the roadway. This aid may be as simple as today's lane lines and "Bott's dots" or it could mean magnetic markers or wires embedded in the roadway. The sensing technologies can be vision-based (4,49,68), magnetic-based (95,43), or radar. (23) For accurate prediction of lane departures the vehicle may need roadway preview information that gives data on changes in roadway curvature. The system must also sense vehicle's turning angle and its turn signal status.

Vehicles with vision-based sensing will need an on-board forward looking camera or an infrared laser scanning system to sense the lane lines. An on-board real-time image processing unit then processes the sensed data and determines the lateral position. The lane reference aid can be as simple as stripes that show the lane edges. This sensing technique requires more instrumentation on the vehicle. Also, the lane reference aids can be easily maintained with existing systems. This sensing system suffers from the same limitations as human drivers at night or in bad weather .

Magnetic sensing techniques use sensors that sense magnets in the roadway. The induced voltage from the coils indicates the direction and lateral displacement from the lane center. The installation and maintenance of such lane reference aids on the roadway may be costly. Magnetic sensors may work better in snowy or rainy conditions. These sensors can encode preview data by changing the polarization of the magnetic field. The vehicle receives a binary code that can give data on the course of the roadway.

Radar-based sensing tracks a vehicle's lateral position with respect to reflectors mounted along the edge or center of the lane. The on-board transceivers emits laser beams to the reflectors mounted on the lane reference aid and receives the reflected signals. The on-board data processing unit computes the lateral deviation based on the sensed data. These systems may use reflective particles in the paint of the lane lines on the roadway. The main disadvantage in the radar sensing is that it provides little preview information. Without adequate preview information, the vehicle can not improve its tracking performance especially when it travels at high speeds.

The roadway may also send preview information to the vehicle with a vehicle to roadway communication system.  $^{(10,53)}$  This data tells the vehicle what to expect ahead. This information may be easily updated to reflect variable road conditions such as detours.

# Warning

The warning threshold should take into account the lane width, driver reaction time and, if possible, the rate of vehicle lateral deviation. The threshold should be designed to minimize the nuisance alarms while allowing the driver as much time as needed to correct his steering. Since reaction times vary among drivers, it might be necessary to let the driver select the warning threshold within some pre-specified range.

This system must estimate the time-to-lane-crossing (TLC). The TLC is the time necessary for the vehicle to reach either edge of the lane. This quantity may be based on a predictor

model that assumes a fixed steering strategy during the time span of the prediction. <sup>(36)</sup> The estimate takes into account the initial lateral lane position  $y_0$ , the heading angle  $\psi_0$ , the vehicle speed V, the steering angle, and the roadway width. Figure 54 shows these parameters. Preview information can improve the estimate of lane departure time. Studies show that drivers need curvature information for distances proportional to the radius of curvature.



Figure 54. Variables used in the calculation of the time to lane crossing.

The lane departure warning system may be active whenever the vehicle ignition is on and the vehicle is going forward. <sup>(64)</sup> The system could also be active only if the driver is in the dedicated lane with the steering assist function turned on. The turn signal or turning on the blind spot warning would cancel the lane departure warning. The lane departure warnings may also be triggered by a change in the vehicle's steering angle that signals a lane change. This warning would probably occur too late to avoid an accident.

The lane departure warning system should have at least two levels of warning: an imminent crash avoidance warning and a cautionary warning. <sup>(98,64)</sup> A cautionary warning tells the driver that the vehicle has a short time until it leaves the lane. This warning may operate when the vehicle is moving forward or only whenever the automated systems have been turned on by the driver. An imminent crash warning alerts the driver to an imminent lane departure situation that exists when the vehicle will soon leave the lane or has already left it. A turn signal or large changes in the vehicle's steering angle can indicate a course change for the vehicle and may cancel the warning. The imminent lane departure warnings should turn off when the driver responds to the warning by correcting the steering angle.

The warning should give information on which side of the lane the vehicle is departing. Directional audio systems or a haptic warning can tell the driver which side of the lane the vehicle is approaching. Three-dimensional audio systems can help a driver recognize the angle of a threat faster. <sup>(9)</sup>

# Driver Interface

The lane departure warning system provides levels of warning to the driver and provides a detector sensitivity adjustment for the driver. The different warning levels allow the vehicle to warn the driver without intrusive nuisance alarms. The detector sensitivity adjustment lets the driver adjust the range of the sensor to reduce sensitivity and false alarm rates. <sup>(64)</sup>

The cautionary warning should be visual and located within 15 ° above the line-of-sight of the driver.<sup>(64)</sup> Visual signals are more detectable if they are near the center of the line-of-sight. The warning needs to give the direction and degree of threat without startling or annoying the driver. The imminent lane departure warning should use two modalities of warnings. Lerner et al suggests a combination of auditory and visual warnings. <sup>(64)</sup> An auditory warning can help the driver determine the direction of the imminent collision with a directional signal. <sup>(9)</sup> This system could also use tactile warnings that use the steering wheel to give cues to the driver. Haptic warnings should only be used when they directly relate to the driver's needed action.<sup>(64)</sup> The system provides artificial feel feedback to the driver through the steering wheel. If necessary, the driver could override the lane departure warning and steering assist system by turning the wheel with extra force.

The lane departure warning system may have an adjustable threshold. This lets the driver reduce the sensor range to avoid excessive false alarms in which the driver perceives frequent hazards. For example, on curvy highways the sensors may be triggered as the vehicle drives safely around curves.

# Issues and Risks

### 1 Sensor Capabilities

The lane departure sensor system may increase safety and reduce certain types of lane change/merge crashes. But there are sensor and human factor issues that still must be resolved. The amount of preview data needed for this system needs to be studied.

# 2 Warning system design

First the designers need to decide when the system should be turned on: always on, only when turn signal on, or turned on when steering column shows lane change. The warning thresholds for emergencies and driver awareness must also be set. The system must avoid nuisance alarms, yet detect vehicles accurately. The types of alarms and warnings must also be researched. The emergency alarm needs to be directional to get fast driver response.

# 3 Multiple Warnings

ERSC 2 introduces multiple types of driver warnings for blind spot and rear-end collision avoidance. Research needs to be done on the interaction and prioritization of these warnings for the human driver.

Stability Augmentation System

Flight control systems have used stability augmentation systems for decades to help improve flight quality. A vehicle's steering control system may also contain a steering assist system to improve the driver's performance in lane keeping. <sup>(61)</sup> The driver uses the steering wheel to keep the vehicle in the lane and the steering assist system helps reject high frequency disturbances. The driver-vehicle system with steering assist thus improves its safety and performance when cruising at high speed.



Figure 55. Block diagram of the steering assist system. The steering assist system acts as an inner loop to correct vehicle steering in the presence of disturbances such as wind gusts.

For a driver-vehicle system without a steering assist system the driver generates steering corrections based on his/her perceived heading and lane position errors. When the vehicle is traveling at high speed, the lateral control system has a high bandwidth and it is sensitive to external disturbances. The steering assist system serves as an inner loop to the human-controlled (outer) loop of the driver/vehicle system as shown in figure 55. The steering assist system consists of sensors, a computer with control algorithms, and a steering actuator. The computer generates steering correction commands to the actuator based on the data provided by the sensors. The actuator then carries out the high frequency steering actions to compensate the effects of the undesired external disturbances.

# Sensors

The sensors collect the data on the vehicle's lateral dynamics (yaw rate, lateral acceleration and sideslip angle, etc.) A gyro built into the steering assist system can measure the yaw rate and lateral acceleration.  $^{(2,32)}$  However, there are currently no sensors that measure sideslip angle and yaw angle accurately. Figure 56 shows the parameters that need to be measured for a steering assist system.



Figure 56. The sensors for the steering assist system measure the heading deviation  $\psi$  and the side slip angle  $\beta$ .

# Control Algorithms

In the driver-vehicle system, the driver closes the steering control loop.  $^{(14)}$  The driver's steering action contains a high frequency random response mode to keep the steering system in a dynamic mode, and a low frequency response mode that is related to the perceived lateral dynamic information. The driver's high frequency random response does not help lane keeping. There is a driver delay or lag between the time that a steering error occurs and the time that a low frequency steering correction is taken. The lag is made up of the time needed to decide what corrective action should be taken and the time necessary to initiate that action. The lag time varies as driving continues and is different for each driver. A value of 0.5 second<sup>(14)</sup> approximates the driver's average time delay. The average driver reaction time to strong wind gusts ranges from 0.3 seconds to 0.6 seconds. <sup>(115)</sup> This reaction time depends

on the rise time and amplitude of the wind gust.

The steering assist system can reduce the driver's lagging effect on vehicle steering. The steering assist system senses lateral dynamics faster than the driver. If the sensors measure this data quickly enough, the steering assist system's control actions can compensate for high frequency lateral disturbances. This gives a stability augmented vehicle with higher stability margin, lower bandwidth, and better ride quality. Studies have shown that pilots prefer planes with natural frequencies of  $\omega \le 1$  radians/second and a damping ratio > 0.5. Aircraft that violate these guidelines are generally considered fatiguing to fly and are undesirable. <sup>(32)</sup> During high speed lane keeping, the driver may be subject to constant lateral and vertical vibrations. The average lateral acceleration should be less than 0.06 g for steady cruising comfort. Other studies indicate that the dominant natural frequency of the lateral motion should not exceed 1.2 Hertz (7.5 radian/sec) for buses. <sup>(2)</sup> In addition, for ride comfort, the maximum instantaneous lateral acceleration should be within the limit of 0.4 g.<sup>(3)</sup>

# Actuators

An important factor that affects the steering performance of the driver-vehicle system is how fast the steering assist system generates the required steering correction. A steering assist system can improve the performance with a small delay relative to driver response. Studies show that driver reaction time to sudden wind gusts ranges from 0.3 to 0.6 seconds. <sup>(115)</sup>

The control effort generated by the steering assist system actuators needs to limited to small changes so that a system malfunction will not endanger the driver's steering control. This can be done by limiting the actuator output within certain range. Since the driver still controls the steering wheel, the driver's lateral control degradation will be minimal.

### Driver Interface

The driver interface for this system should be transparent to the driver. This system should turn on automatically when the driver is at cruising speed or when the driver turns on the SHM function. The driver can override the steering assist function at any time by changing the steering angle of the vehicle and saturating the actuator.

### Issues and Risks

1 Driver Override

The driver needs to be able to easily override this function to get the vehicle back into the center of the lane or to avoid a collision.

2 Interactions with Lane Departure Warnings

This function could work with the lane departure warning by providing inputs such as tactile steering "hints" that encourage the driver to return to the middle of the lane. This system requires different types of steering actuators that work in parallel with the driver to provide independent control inputs.

#### 3. Faster Speeds

The steering assist function may allow drivers to comfortably go faster in a dedicated lane. This may cause unsafe conditions if this function fails.

### Driver Interface

The driver should be able to override the speed & headway maintenance function and adjust the headway as described in the section describing the SHM function. The driver should also be able to adjust the warning thresholds on the lane departure warning and the blind spot warning. The driver also receives highway traffic information through the vehicle and may use it for speed/headway selection or route planning. The driver may also receive preview information from the roadway. This data may tell the driver about sharp curves up ahead or other roadway conditions. The driver also monitors the vehicle's operational status and backs up failed vehicle functions. If the SHM system fails the driver takes over manual control of the vehicle in the fall-back mode.

The driver interface must show the driver that the automated functions are operating correctly or that a function has failed. This interface should be simple to understand and use. No adjustments should be required that take the driver's attention away from the driving task. Input systems like touch pads or buttons may be distracting and may require too much driver attention.<sup>(8)</sup> The driver must be able to tell what the vehicle is doing at all times and why it is doing it. Studies show that drivers cooperate more when they know the reasons for an action.<sup>(10)</sup>

Head-up displays may be particularly useful for visual indicators of the system operation. For example a heads-up display could show the measured distance to the vehicle ahead as well as the minimum time headway and target speed. Other technologies include liquid crystal displays (LCDs), light-emitting diodes (LEDs), computer generated voice messages, sudden vehicle accelerations/decelerations (jerk), tactile feedback, and directional audio displays. <sup>(64,8,9,116)</sup>

The interface may be continuous or only active when a driver action is needed. The best design of the driver interface for human factors is an open issue and involves selecting a display technology, type and duration of warnings, user control types and locations, etc. User acceptance is crucial to user interface design.  $^{(54,67)}$  The systems in each ERSC must be easy to use and understand and easy to adjust without the driver taking his/her eyes off the roadway for more than 1-2 seconds.  $^{(73)}$ 

# Roadway/Vehicle Speed & Headway Commands

The roadway measures the average speed and traffic density, assesses environmental conditions, and calculates vehicle target speeds and minimum safe headways. The vehicles follow the target speeds set by the roadway using the speed and headway maintenance function. The vehicles should respond to headway recommendations for longer headways due to changes in environmental conditions such as slick roads. The vehicles use the environmental data from the roadway to adjust their time headways.

with the roadway by sending their identification, automated function status, and location. <sup>(108)</sup> The vehicles may also display relevant traffic information from the roadway to the driver for route planning purposes. This section covers roadway sensing systems and roadway-vehicle communications.

# Roadway Sensing Systems

The roadway needs to sense the average traffic density over sections of the roadway. <sup>(58)</sup> This data can be measured by sensors that are part of the infrastructure or the vehicles can transmit their measured speed and headway to the roadway with a vehicle-roadway communication system. The vehicle may also transmit its operational status and identification to the roadway for check-in or incident management. The roadway also needs to measure roadway weather data to select a safe target speed for the environmental conditions. <sup>(57)</sup>

# Vehicle to Roadway Communications (VRC)

The roadway sends speed data to the cars to smooth traffic densities and manages incidents. The vehicles may also send their measured velocity and headway to the roadway. This system may also be used for check-in. This allows the roadway to monitor the operational status of the vehicles and ensure that they are correctly operating in the automatic mode. These messages only need to be sent once per section. The vehicles will also check-in with the roadway to ensure that they are working correctly. Each vehicle sends status data on all of its automated functions to the roadway. In ERSC 2 these functions are: speed and headway maintenance, rear-end collision avoidance, blind spot warning, lane departure warning, steering assist, VRC systems, and critical engine functions. <sup>(108)</sup>

VRC systems can be mounted on the side of the roadway, on overpasses, or embedded in the roadway.<sup>(85)</sup> These systems may use automated toll collection technology with variable registers or a system that covers a wider range.<sup>(22,53)</sup> Appendix A covers the operation of these systems.

Target speed messages from the roadway to the vehicles do not need a high data rate since it will not change very often. Each field will need 6-8 bits to give a resolution of 1 meter for the gap limit and 1 mile per hour for the velocity limit. Since the message goes to all cars a destination ID code is not needed. The source ID code indicates that the message is from the roadway. Error detection coding looks for bit errors in the message. Retransmission requests are unnecessary since we can rely on repeating the message. This message should be repeated at most one time per second. The bit error rate can also be high since the message changes slowly and rarely. Repetition of messages can compensate for the high error rate. If the roadway sends preview data to the vehicles, the messages may need to be longer in curvy regions of the roadway.

The VRC may also send messages from the vehicles to the roadway. Each vehicle sends its speed and the distance to the preceding vehicle (6-8 bits each). The vehicles may also send a unique identifying code (28 bits) to the roadway for check-in. The operational status of the automated system can be sent as well, this could take 8-16 bits depending on the amount of detail required by the roadway. Other variables such as estimates of the road-tire friction coefficient can take 4 or more bits. Table 19 shows the possible variables to be sent to the roadway for check-in. <sup>(108)</sup> This gives a data length of around 90 bits.

Function	Components	Number of Bi		
Speed & Headway Maintenance	Sensors, Actuators, Controller	1-6		
Rear-end Collision Avoidance	Sensors, Actuators, Controller, Vehicle-Vehicle Communications System	1-8		
Blind Spot Warning	Sensor, warning	1-2		
Vehicle-Roadway Communication System	Transmitter, receiver, variable register	1-2		
Lane Departure Warning	Sensor, warning	1-2		
Steering Assist	Sensors, Actuators, Controller	1-4		
Vehicle Identification Code		28		
Other Vehicle Functions	Critical fluids level and pressures, brake conditions, tire pressure, engine temperature, headlights	12-16		

Table 19. Potential status variables to be sent to the roadway by the vehicles for check-in.

A broadcast system with a range of 100 meters for a single automated lane must communicate with 0.3 to 0.75 vehicles per second. Analysis by Polydoros et al shows that the current Hughes VRC system and automatic toll collection systems can meet the data transmission requirements for VRC in ERSC 2. (85)

### Issues and Risks

1 Dedicated Lane

An issue for AHS is how to get a separate lane for automated vehicles. The driving

public rebels when a lane is taken away from active use. <sup>(99)</sup> The solution for most highway departments has been to add a new lane on the shoulder or median strip, but most urban highways do not have any more space to spare for the sole use of automated vehicles. This infrastructure change may have to wait until enough vehicles have speed and headway maintenance systems and traffic flow benefits can be easily seen.

### 2 Legal/Liability

In this ERSC the roadway sends recommended target speeds and minimum headways directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the minimum recommended headway is too short. The government may not wish to deploy such a system until it has been shown to be reliable and fail-safe.

### 3 Sensors

Roadway weather sensors only cover small regions of roadway. These sensors could easily miss patches of ice or puddles on the roadway. The roadway may then choose an unsafe headway. Research needs to continue into these sensors to improve their capabilities.

### 4 Privacy

The vehicles "Check-in" with the roadway. This may leave a record of when and where a vehicle traveled. This raises user privacy concerns about governmental use of this data.

# 5 Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

# Fall-back Mode

The system needs to degrade gracefully if a failure occurs. Many failures such as blind spot warnings may not cause the vehicle to be non-operational. Vehicles with different capabilities can coexist in the dedicated lane without incident.

In case of failure in a single control channel of the rear-end collision avoidance system, the vehicle can operate in ERSC 1 control mode with the SHM and rear-end collision warning functions since these systems do not require parallel paths for reliability. The vehicle increases the following distance or headway to a safe headway for ERSC 1 and the driver resumes control of the emergency or hard-braking function with the help of the rear-end collision warning. However, according to Riley decoupling warning and avoidance modes

# Task D

can cause "role confusion" for the driver.  $^{(90)}$  The driver may also be uncomfortable dealing with a sudden change in responsibility. The driver may also be uncomfortable taking over control at higher speeds or if there is reduced visibility.  $^{(90)}$  System design may require that the driver can force the vehicle not to transfer control to the driver until the vehicle is ready to leave the dedicated lane.  $^{(90)}$  Meisner et al believe that such transfers between automated and manual functions need to be handled jointly by the driver and the vehicle.  $^{(70)}$ 

If the lane departure warning or the steering assist functions fail the vehicle informs the driver. Failure of these functions does not require a transfer of control to the driver so they do not require the vehicle to return to ERSC 1 control. Loss of the steering assist function may make the driver more uncomfortable at higher speeds and increase driver fatigue, but it does not require a return to ERSC 1.

### Issues and Risks

1 Control Transfer

Any control transfers between the vehicle and the driver need to show which one is in charge at any given time. Abrupt changes need to be avoided unless they are initiated by the driver.

### 2 Mixed Modes

Vehicles with different capabilities require diverse headways and different speeds.

### Reliability Requirement Analysis

In ERSC 2, the vehicle longitudinal control is fully automated with the introduction of the vehicle rear-end collision avoidance function. The on-board speed/headway maintenance (SHM) and rear-end collision avoidance (RECA) system allows the driver complete feet-off driving. The SHM and RECA system takes into account the roadway target speed commands and headway recommendations, and the braking data from the preceding vehicle to determine the desired cruising speed and safe headway for collision free vehicle following.

The driver is not part of the longitudinal control loop and is not allowed to override the SHM and RECA system directly. The driver's reliability has no contribution to the overall reliability of the vehicle longitudinal control. The reliability of the SHM and RECA system is crucial for safe longitudinal control. It needs to be highly reliable if it is to accepted by the general public.

The driver is still responsible for all lateral control functions but receives more assistance from the vehicle and the roadway. The driver's lane changing and merging maneuvers are assisted by the blind spot warning (BSW) system. While in the dedicated lane, the driver's lane keeping control is helped by the steering assist (SA) system and the lane departure warning (LDW) system. The SA system compensates for high frequency lateral disturbances while the vehicle is cruising in the lane. I helps the vehicle follow a straight line. The driver can override the SA system by simply turning the steering wheel hard. The LDW system gives warnings to the driver if it detects the vehicle is departing from the lane. The LDW system may require roadway lane reference aids to help measure lateral deviation from the lane center. In ERSC 2, all the automation technologies related to lateral control are only to assist the driver. All lateral maneuvers are initiated and performed by the driver. The driver's reliability dictates the reliability of lateral control. The on-board BSW, SA, and LDW systems are to improve the driver's lateral control reliability.

# Speed/Headway Maintenance and Rear-End Collision Avoidance

In ERSC 2, the speed and headway maintenance (SHM) system together with rear-end collision avoidance (RECA) system allows the driver complete feet-off driving. The SHM system controls throttle and soft braking to maintain a desired speed and/or headway during cruising and normal vehicle following. The RECA system uses hard braking to avoid rear-end collisions during emergencies, such as sudden hard braking by the preceding vehicle.

Due to the close interaction and coupling between the SHM and RECA systems in vehicle longitudinal control, it is desirable to analyze the reliability requirements of these two systems together. We will consider the SHM and RECA as one combined system. Since the SHM system has all the major components (headway/speed sensor, controller, brake actuator, etc.) that are needed to implement the RECA system, it is reasonable that the RECA system is embedded into the SHM system in the design .



Figure 57. The functional block diagram of the SHM and RECA system.

The major components required to implement the SHM and RECA system are a headway/speed sensor, a controller, a brake actuator, and a throttle actuator, as shown in figure 57. The headway/speed sensor measures the headway and the preceding vehicle's speed. The brake data receiver receives the preceding vehicle's braking capabilities and acceleration/deceleration intentions. The roadway command receiver receives the target speed and minimum time headway commands, road surface conditions, etc. The controller uses the vehicle's braking capabilities, those of the preceding vehicle, and speed command and headway recommendation provided by the roadway to determine a desired speed for cruising and a safe headway for vehicle longitudinal control. The controller generates throttle/brake commands to the throttle/brake actuators for collision free vehicle following.

The received preceding vehicle's braking intention can be used to predict the necessity of hard braking more accurately.

# Required Reliability Estimation

The headway used by the controller does not take into account the driver reaction time since the vehicle handles the rear-end collision avoidance. The driver is not expected to take over longitudinal control safely if the SHM and RECA system fails. The driver can not be considered as a back-up to the SHM and RECA system. The SHM and RECA system has to be so reliable that an internal system malfunction will not lead to a rear-end collision. Drivers in the U.S. average one rear-end collision about every 50 years. <sup>(27)</sup> This suggests that the SHM and RECA system will not be accepted by the general public if its mean time to failure is not much greater than 50 years.

### Reliability Functional Requirement

Since the driver is not considered to be a back-up to the SHM and RECA system, a rear-end collision may happen if the SHM and RECA system fails. A single component/point failure can lead to a rear-end collision unless the SHM and RECA system is designed to be fail-safe. The reliability functional requirement for the SHM and RECA system is thus proposed as:

"Under no circumstances should a single component/point failure let the vehicle crash into any moving or stationary object in the lane, and there should be no common failure mode."

To fulfill this requirement, the SHM and RECA system needs to be able to detect potential internal failures. If an internal failure occurs, the system must be able to continue its function until the driver takes over control safely.

# Required Redundancy

Safe longitudinal control relies on the accurate output from the subsystems of headway/speed sensor, controller, and throttle/brake actuators. The fact that these subsystems are connected in series implies that the whole system will fail if any one of these subsystems fails. To achieve the proposed reliability functional requirement, component redundancy will be needed to improve the reliability of these subsystems. Figure 58 shows a design framework that can potentially fulfill the reliability functional requirement.



Figure 58. The SHM and RECA system framework for reliability design.

The proposed design framework has two active control channels. Each control channel has a headway/speed sensor, a controller, a throttle actuator, and a brake actuator. The controller of each channel receives roadway target speed commands and minimum time headway recommendations, and receives braking data from the preceding vehicle. Each control channel is capable of performing the SHM and RECA functions alone. If a component/point failure causes one control channel to fail, the other channel continues its operation. This gives the driver enough time to successfully take over control from the non-fail-safe system. A third headway/speed sensor may be used to improve the accuracy of headway sensing and to help identify any failed headway/speed sensor. A supervisory controller is used to help detect any failed component and to deactivate the failed control channel.

If a potential internal failure causes one control channel to be deactivated, the SHM and RECA system is not in the fail-safe condition. Any additional component/point failure may cause a rear-end collision. For safety reasons, the SHM and RECA system should warn the driver of the detected potential failure and ask him/her to back up the rear-end collision avoidance function. The system framework as in figure 58 allows the vehicle longitudinal control to fall back to ERSC 1.

When one control channel is deactivated due to a potential component/point failure, the SHM and RECA system may need a period of transition time to have the other channel operate effectively. Since the vehicle may be traveling at high speeds, a performance degradation such as this may have safety implications. To reduce the transition time required for fully activating or deactivating a control channel, it is desirable that the two redundant control

# Task D

channels be kept active all the time as long as the SHM and RECA system is in operation. In this case, one can activate or deactivate a control channel by quickly adjusting its control gain. Feedback design can be used to efficiently adapt the control gains based on the operational status.

The "no common failure modes" requirement avoids the situation when redundant components fail simultaneously due to the same failure cause. This implies that the redundancies in the proposed framework need to be independent, i.e., redundant components have to operate with different technologies and/or be supported by independent hardware/software. For example, the three redundant headway/speed sensors are one radar based<sup>(75,109)</sup>, one laser based<sup>(103,118,92)</sup>, and one vision based.<sup>(91,109,78)</sup> Furthermore, the three headway/speed sensors have to be powered independently so that a power source failure will not cause two of them to failed simultaneously.

# System Level Design Requirements

Without doing the complete FMEA, one can still identify the following major system failure modes based on the system design framework proposed in figure 58.

- 1. Headway/speed sensor failure
- 2. Controller failure
- 3. Actuator failure
- 4. Incorrect speed/headway commands received from roadway
- 5. Incorrect braking data received from the preceding vehicle

These potential failure modes impose important system level design requirements that are presented in the following.

Potential Failure Mode 1: Headway/Speed Sensor failure

- 1.1 Headway/Speed Sensor 1 or 2 computes incorrect headway or closing rate data. Design requirements:
  - 1.1.1 Headway/Speed Sensor 1 and 2 need to have an internal independent check of the reasonableness of the computed headway and closing rate data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should then be executed. The System Failure Response disables the control channel that connects the failed component, warns the driver of the failure, and asks driver to take over longitudinal control.
  - 1.1.2 Controller 1 and 2 and Supervisory Controller should verify that each headway/speed sensor is operating in a timely manner by verifying that new data is received from each headway/speed sensor every to be determined (tbd) time interval. If an error is detected, the System Failure Response should be executed.
- 1.2 The data values generated by Headway/Speed Sensor 1 and 2 are substantially different. Design requirements:

- 1.2.1 Supervisory Controller should sense the discrepancy and use the data from Headway/Speed Sensor 3 to determine the failed headway/speed sensor. The System Failure Response should then be executed.
- 1.3 Headway/Speed Sensor 3 computes incorrect headway or closing rate data. Design requirements:
  - 1.3.1 Headway/Speed Sensor 3 needs to have an internal independent check of the reasonableness of the computed headway and closing rate data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
  - 1.3.2 Supervisory Controller should compare the data computed by Headway/Speed Sensor 3 with those computed by Headway/Speed Sensor 1 and 2 to check if Headway/Speed Sensor 3 is working properly. If an error is detected, the System Failure Response should be executed.
  - 1.3.3 Supervisory Controller should verify that Headway/Speed Sensor 3 is operating in a timely manner by verifying that new data is received from Headway/Speed Sensor 3 every tbd time interval. If an error is detected, the System Failure Response should be executed.
- 1.4 Data from Headway/Speed Sensor 1 or 2 is not communicated to the controllers, or is corrupted in transmission through interface. Design requirements:
  - 1.4.1 Each controller should incorporate an independent check of the reasonableness of the computed headway and closing rate data from Headway/Speed Sensor 1 and 2, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
  - 1.4.2 Same as for 1.1.2.
- 1.5 Data from Headway/Speed Sensor 3 is not communicated to Supervisory Controller, or is corrupted in transmission through interface. Design requirements:
  - 1.5.1 Supervisory Controller should incorporate an independent check of the reasonableness of the computed headway and closing rate data from Headway/Speed Sensor 3, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
  - 1.5.2 Same as for 1.3.2.

Potential Failure Mode 2: Controller failure

2.1 Controller 1 or 2 computes incorrect throttle or brake command.

Design requirements:

- 2.1.1 Each of Controller 1 and 2 should incorporate an independent check of the reasonableness of the computed throttle or brake command before it is sent to the throttle or brake actuator, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- 2.1.2 Each of Throttle Actuator 1 and 2 should incorporate an independent check of the reasonableness of the throttle command computed by the associated controller, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- 2.1.3 Each of Brake Actuator 1 and 2 sh ould incorporate an independent check of the reasonableness of the brake command computed by the associated controller after it is received, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- 2.2 Throttle (or brake) commands computed by Controller 1 and 2 are substantially different.Design requirements:
  - 2.2.1 Supervisory Controller should sense the discrepancy in the throttle (or brake) commands computed by Controller 1 and 2, and determine the controller that fails. The System Failure Response should then be executed.
- 2.3 Supervisory Controller fails. Design requirements:
  - 2.3.1 Supervisory Controller should incorporate circuitry, such as a watchdog timer, to detect failure and to ensure that the processor functions in a timely manner. If a failure is detected, the circuitry should deactivate Supervisory Controller, and the System Failure Response should be executed.
- 2.4 Supervisory Controller computes incorrect throttle or brake command. Design requirements:
  - 2.4.1 Supervisory Controller should compare the computed throttle and brake commands with those computed by Controller 1 and 2 to determine if Supervisory Controller is working properly. If an error is detected, the System Failure Response should be executed.
- 2.5 Throttle/brake commands from Controller 1 or 2 is not communicated to Throttle/Brake Actuator 1 or 2 at appropriate time, or is corrupted in transmission through interface. Design requirements:

- 2.5.1 Each of Throttle/Brake Actuator 1 and 2 should verify that it is receiving updated throttle/brake commands from the associated controller in a timely manner. If an error is detected, the System Failure Response should be executed.
- 2.6 Supervisory Controller's command data is not communicated to controllers and throttle/brake actuators at appropriate time, or is corrupted in transmission through interface.

Design requirements:

2.6.1 Each of the two controllers and th rottle/brake actuators should verify that it is receiving updated commands from Supervisory Controller in a timely manner. If an error is detected, the System Failure Response should be executed.

Potential Failure Mode 3: Actuator failure

- 3.1 Throttle Actuator 1 or 2 produces incorrect throttle angle for a given throttle command.Design requirements:
  - 3.1.1 Each throttle actuator should incorporate built in tests to be performed on the vehicle startup and during speed/headway maintenance operation to verify that proper throttle angle can be produced and is within the allowable range. If an error is detected, the System Failure Response should be executed.
- 3.2 Throttle Actuator 1 or 2 can not be deactivated when the associated control channel is determined to be deactivated. Design requirements:
  - 3.2.1 The SHM and RECA system should provide for two independent, redundant means of deactivating the throttle actuators.
- 3.3 Brake Actuator 1 or 2 generates incorrect brake force for a given brake command. Design Requirements:
  - 3.3.1 Brake Actuator 1 and 2 should incorporate built in tests to be performed on the vehicle startup and during speed/headway maintenance operation to verify that proper brake force can be produced. If an error is detected, System Failure Response should be executed.
- 3.4 Brake Actuator 1 or 2 can not be deactivated when the associated control channel is determined to be deactivated. Design Requirements:
  - 3.4.1 The SHM and RECA system should provide for two independent, redund ant means of deactivating the brake actuators.

Potential Failure Mode 4: Incorrect target speed commands or minimum time headway recommendations received from the roadway

- 4.1 The roadway target speed commands or minimum time headway recommendations are not communicated to Roadway Command Receiver, or are corrupted during transmission.
  - 4.1.1 The data/commands sent from the roadway should contain error detection and error correction codes to improve the accuracy of the data received by Roadway Command Receiver.
  - 4.1.2 Supervisory Controller should verify that target speed commands and minimum time headway recommendations are received in a timely manner. If updated data is not received for tbd consecutive times, Supervisory Controller should re-evaluate a new lower target speed and/or a more conservative safe time headway for speed and headway maintenance and rear-end collision avoidance, based on the known roadway surface characteristics and traffic conditions. Supervisory Controller should notify the driver the speed and headway adjustment actions taken by the controllers, and may have to ask the driver to take over the rear-end collision avoidance function.
  - 4.1.3 Supervisory Controller should incorporate an independent check of the reasonableness of the received target speed commands and minimum time headway recommendations from the roadway, based on the physical attainability and known ranges of the system variables. If unreasonable data is received over tbd consecutive times, the SHM and RECA system should respond as described in 4.1.2.

Potential Failure Mode 5: Incorrect braking data received from the preceding vehicle

- 5.1 The braking data from the preceding vehicle is not communicated to Brake Data Receiver or is corrupted during transmission. Design Requirements:
  - 5.1.1 The communicated braking data should contain error detection and correction codes to improve the accuracy of the data received by Brake Data Receiver.
  - 5.1.2 Supervisory Controller should verify that, during headway maintenance operation, braking data from the preceding vehicle is received in a timely manner. If updated braking data is not received for tbd consecutive times, controllers should evaluate a new safe time headway for headway maintenance and rear-end collision avoidance, based on the known braking characteristics of the preceding vehicle. Supervisory Controller should notify the driver of the headway adjustment action taken by the controllers, and may have to ask the driver to take over rear-end collision avoidance.
  - 5.1.3 Supervisory Controller should incorporate an independent check of the reasonableness of the received braking data from the preceding vehicle, based on the physical attainability and known ranges of the system variables. If unreasonable braking data is received over tbd consecutive times, the SHM and RECA system should respond as described in 5.1.2.

# Key Findings

1. The driver can not be considered as a back-up to the SHM and RECA system. A fail-safe SHM and RECA system design will have considerable structure complexity and component redundancies.

2. The reliability functional requirement for the SHM and RECA system is: "A single component or point failure is not allowed to cause a rear-end collision, and there should be no common failure modes."

3. The redundant control channels in the proposed design framework need to be always kept active to ensure safe and smooth deactivation of a control channel.

# Issues and Risks

1. There could be rear-end collisions that can not be avoided by the SHM and REC A system. For example, if the preceding vehicle abruptly change lanes to avoid crashing into a close, stationary obstacle in the lane, the SHM and RECA system may not have enough distance for safe braking. In this case, lateral control will be required for evasive maneuver. The driver in this case is positioned in a situation that he/she may not be able to handle since he/she may not have enough time to avoid collision by using lateral control. This suggests that the RECA function probably should be presented in ERSC 4 together with the vehicle's lateral collision avoidance function, instead of in ERSC 2.

2. The redundancy requirement in the design of the SHM and RECA system will substantially increase the cost for deployment. Potential system failures raise serious liability concerns.

3. Complete feet-off driving may further relax the driver and affect his/her performance in lateral collision avoidance.

Blind Spot Warning

The analysis of the reliability requirement is the same as in ERSC 1.

# Steering Assist

The steering assist (SA) system smoothes the driver's steering control by compensating the high frequency lateral disturbances. When the vehicle is traveling at high speeds in the dedicated lane, the vehicle lateral dynamics has high bandwidth and is sensitive to external disturbances such as wind gust or uneven road surface. The SA system generates high frequency but small scale steering commands to compensate the disturbances and to reduce the bandwidth of the vehicle lateral dynamics. The SA system thus improves the safety and ride comfort by augmenting the steering control stability and smoothing the driver's steering.



Figure 59. The block diagram of the steering assist system interfacing with the driver.

The SA system consists of a lateral dynamics sensor, controller (computer with control algorithms), and a steering actuator. The sensor collect the vehicle's lateral dynamics

information, such as yaw rate, lateral acceleration and sideslip angle  $^{(61)}$ , etc. The computer generates steering correction commands to the actuator based on the data provided by the lateral dynamics sensor. The actuator then carries out the high frequency steering actions to compensate the effects of the external disturbances. Figure 59 shows a block diagram for the SA system interfacing with the driver.

The SA system is similar to the stability augmentation system (SAS) that has been used in the flight control for decades. The steering actuator is connected to the steering shaft in series so that the driver's high frequency random steering actions do not cause the vehicle to wiggle in the lane. The steering actuator should be saturated if it receives large steering commands from the controller or large counteracting steering force from the driver. This allows the driver to steer the vehicle to following a curve or to change lanes by applying extra force at the steering wheel.

# Reliability Functional Requirement

The SA system only assist the driver in lateral control. The driver is hands on the steering wheel and steers the vehicle. A sudden loss of the steering assist is not likely to endanger of the driver's lateral control. However, if the SA system generate large steering force when it fails, the vehicle may depart from the lane before the driver can react to the danger. The reliability functional requirement for the SA system is thus proposed as:

"The SA system should provide for independent, redundant means to constrain the automatically generated steering force within safe bounds."

The independent, redundant means of constraining steering force are required to protect the system from any common failure mode. This requirement does not impose redundancy requirement for major components in the design. However, extra hardware/software will be needed to detect internal malfunction, to constrain or saturate the steering force, and to warn the driver of the potential failures.

### System Level Design Requirements

Based on the above discussion, one can identify the following major system failure modes:

- 1. Lateral dynamics sensor failure
- 2. Controller failure
- 3. Steering actuator failure

These potential failure modes impose important system level design requirements that are presented in the following.

Potential Failure Mode 1: Lateral Dynamics Sensor failure

1.1 Sensor computes incorrect lateral dynamics data. Design requirements:

- 1.1.1 Lateral Dynamics Sensor needs to have an internal independent check of the reasonableness of the computed lateral dynamics data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the driver should be warned and the SA system should be deactivated.
- 1.1.2 Controller should verify that Lateral Dynamics Sensor is operating in a timely manner by verifying that new data is received every tbd time interval. If an error is detected, the driver should be warned and the SA system should be deactivated.
- 1.2 Data from Lateral Dynamics Sensor is not communicated to Controllers, or is corrupted in transmission through interface. Design requirements:
  - 1.2.1 Controller should incorporate an independent check of the reasonableness of the received data from Lateral Dynamics Sensor, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the driver should be warned and the SA system should be deactivated.
  - 1.2.2 Same as for 1.1.2.

Potential Failure Mode 2: Controller failure

- 2.1 Controller computes incorrect steering command. Design requirements:
  - 2.1.1 Controller should incorporate an independent check of the reasonableness of the computed steering correction command before it is sent to Steering Actuator, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the driver should be warned and the SA system should be deactivated.
  - 2.1.2 Steering Actuator should incorporate an independent c heck of the reasonableness of the steering command computed by Controller after it is received, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the driver should be warned and the SA system should be deactivated.

2.2 Steering command from Controller is not communicated to Steering Actuator at appropriate time, or is corrupted in transmission through interface.Design requirements:

2.2.1 Steering Actuator should verify that it is receiving upda ted steering commands from Controller in a timely manner. If an error is detected, the driver should be warned and the SA system should be deactivated.

Potential Failure Mode 3: Incorrect steering force generated from Steering Actuator.

- 3.1 Steering Actuator produces incorrect steering force for a given steering command. Design requirements:
  - 3.1.1 Steering Actuator should incorporate built in tests to be performed on the vehicle startup and during steering assist operation to verify that proper steering force can be produced and is within the allowable range. If an error is detected, the driver should be warned and the SA system should be deactivated.
- 3.2 Excessive steering force generated from Steering Actuator. Design requirements:
  - 3.2.1 The SA system should provide for independent, redundant means to constrain the steering force from the Steering Actuator within safe ranges.

# Key Findings

1. Loss of steering assist will not impose immediate danger to the driver's lane keeping control since the driver is hands on the steering wheel.

2. The proposed reliability functional requirement of the SA system is to provide independent and redundant means to constrain the steering force generated by the SA system.

### Issues and Risks

1. The SA system may need the information from lane reference aid provided by the roadway to improve the accuracy of the measured lateral dynamics.

2. The driver needs to apply extra steering force to override the SA system. This may degrade the driver's performance in lateral collision avoidance.

# Lane Departure Warning

The lane departure warning (LDW) system improves the safety of the driver's lane keeping control. The LDW system warns the driver of lane departure danger if it detects that the vehicle lateral displacement exceeds a certain threshold. The roadway provides lane reference aid to assist the LDW system in measuring the lateral deviation. The driver has to respond to the warning by applying steering correction to avoid lane departure. Figure 60 shows the interaction between the LDW system and the driver.



Figure 60. The LDW system interfaces with the driver and roadway.

The on-board LDW system is composed of a lateral deviation sensor, a processor and a warning interface. The lateral deviation sensor measures the vehicle's lateral displacement with the help from the lane reference aid. The processor uses the data from the sensor to compute time-to-line crossing (TLC). <sup>(36)</sup> The processor compares the TLC with the driver reaction time to determine if a warning should be issued. The warning interface takes the command from the processor and sends the warning signal to the driver.

# Reliability Functional Requirement

Since the driver is hands on the steering wheel for steering control, lane keeping depends on the man-machine interactions between the driver and the LDW system. The LDW system fails if it does not detect a lane departing danger or if it generates a false alarm. A LDW system with low detection rates will not benefit driver safety/reliability in lane keeping. Frequent false alarms may irritate the driver or cause the driver to turn off the LDW system. To improve the driver's reliability in lane keeping, the reliability functional requirement for the LDW system is proposed as:

"The LDW system should have high detection rates and low false alarm rates."

To reduce false alarms and nuisance alarms, it may be necessary to allow the driver to adjust the sensitivity of warning. The driver set the warning threshold based on his/her driving skill and environmental conditions. Two levels of warning, cautionary warning and imminent warning<sup>(64)</sup>, may be required for effective interface with the driver. To ensure that the driver can perceive imminent warnings, imminent warnings should contain signals with redundant, independent modalities, such as auditory and visual.

# System Level Design Requirements

To achieve fail-safe design, the LDW system will need to have reliable hardware/software with self-diagnosis element to detect internal failures. The lane reference aid needs to be able to effectively support the lane position detection technology used by the LDW system. Further system level design requirements can be identified by analyzing major potential failure modes and causes, based on the system framework in figure 61.



Figure 61. The LDW system framework for reliability design.

Potential Failure Mode 1: Lateral Deviation Sensor failure

- 1.1 Lateral Deviation Sensor measures incorrect lateral displacement data. Design requirements:
  - 1.1.1 Lateral Deviation Sensor needs to have an internal independent check of the reasonableness of the computed lateral displacement data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed. The System Failure Response warns the driver of the potential failure and asks him/her to turn off the LDW system.
  - 1.1.2 Processor should incorporate an independent check of the reasonableness of the measured data from Lateral Deviation Sensor, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- 1.2 Data from Lateral Deviation Sensor is not communicated to Processor, or is corrupted in transmission through interface. Design requirements:
  - 1.2.1 Processor should verify that Lateral Deviation Sensor is operating in a timely manner by verifying that new data is received from Lateral Deviation Sensor every tbd time interval. If an error is detected, the System Failure Response should be executed.
  - 1.2.2 Same as for 1.1.2.
- 1.3 The roadway provides incorrect lane reference data. Design requirements:
  - 1.3.1 The roadway should provide for reliable lane reference aid to support the LDW function. The roadway should have an internal check on the accuracy of the lane reference data. If correct lane reference data can not be provided, drivers on the dedicated lane should be notified to deactivate the LDW system.

- 1.3.2 Processor should incorporate technologies that can detect potential failures in the lane reference aid. If an error is detected, the System Failure Response should be executed.
- 1.3.3 Same as for 1.1.2.

Potential Failure Mode 2: Processor failure

- 2.1 Processor computes incorrect time-to-line crossing. Design requirements:
  - 2.1.1 Processor should incorporate an independent check of reasonableness of the computed time-to-line crossing, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- 2.2 Processor command is not communicated to Warning Interface at appropriate time or is corrupted in transmission through interface. Design requirements:
  - 2.2.1 Warning Interface should verify that it is receiving updated command from Processor in a timely manner. If an error is detected, the System Failure Response should then be executed.

Potential Failure Mode 3: Warning Interface failure

- 3.1 Warning Interface fails to give warnings or generate incorrect level/modality of warnings, after receiving a warning command from Processor. Design requirements:
  - 3.1.1 Warning Interface should incorporate built in tests to be performed on the vehicle startup and during driver's lane keeping operation to verify that proper warning signals can be generated. If an error is detected, the System Failure Response should be executed.

# Key findings

1. Two levels of warning may be required. Imminent collision warning may require redundancy with different warning modalities.

2. The reliability functional requirements for the LDW system are high detection rates and low false alarm rates.

### Issues and Risks

1. The driver may become over-reliant on the LDW system and degrade his/her reliability/performance in lateral collision avoidance.

2. The driver may experience warning overload and warning confusion problems while interfacing with other vehicle and roadway functions.

Roadway to Vehicle Speed and Headway Commands

The analysis of the reliability requirement is the same as in ERSC 1.

ERSC 2 Driver Tasks and Workload

This section discusses the drivers' tasks in this ERSC. It covers normal driving in the dedicated lane when all of the automated functions work perfectly. Then it covers the fall-back mode in which the vehicle switches into ERSC 1 with speed and headway maintenance and rear-end collision warning if the rear-end collision avoidance system fails or degrades. Finally we discuss the lane departure warnings and the steering assist function and their interactions with the drivers.

The driver drives the car to the automated lane. The driver turns on the vehicle's turn signal and blind spot warning system. The driver enters the lane when a safe gap appears in the traffic flow. The driver steers the car into the lane and turns on the SHM function and rearend collision avoidance system. The vehicle takes over the throttle and brake when the car is in the lane or as the vehicle enters the lane. The driver steers the vehicle while in the automated lane with the help of the steering assist function and the lane departure warning. The driver workload in ERSC 2 is very similar to that of today's driving except that the vehicle has full longitudinal control while the vehicle is in the dedicated lane.

The driver turns on the turn signal and blind spot warning system to leave the automated lane. This initiates check-out procedures. Check-out returns the throttle and brake control to the driver and turns off the lane departure warning and the steering assist function. The vehicle increases the headway before transferring throttle control to the driver since human drivers have slower reaction times that the automated rear-end collision avoidance system. The driver then resumes control of the throttle and changes lanes out of the automated lane. The driver turns off the communications links when he leaves the automated lane.

### Task Analysis-Normal Driving

In normal operations, the vehicle controls the throttle and brake. The driver steers the vehicle in the lane with the aid of the steering assist function. The roadway sends target speed commands and minimum headway information electronically to the vehicle. This data helps the vehicle compute a safe headway. The task analysis table below shows the driver cues for this task.

Task	Monitor Vehicle Operation and Steer Vehicle in Normal Driving
General	Task similar to manual driving
Initiating Cue	Enter Dedicated Lane and turn on speed and headway maintenance and the rear-end collision avoidance function–The driver sets time headway and initial speed
Feedback	Visual, auditory, and kinesthetic cues of accelerating, braking, slowing, and steering
Task Standards	Duration the same as manual driving for steering
Task Conditions	Weather, road conditions, or traffic
Skills Required	Reaction times appropriate to speed and roadway conditions. Perceptional judgment of vehicle kinematics
Potential Errors	Driver inattention causes vehicle to leave road Turn off speed and headway maintenance system accidentally Vehicle fails to maintain speed or proper headway
	Vehicle selects unsafe headway Roadway sends incomplete or wrong speed command
Causes of Error	Fails to perceive or misinterprets system feedback Poor judgment of speed and road curvature Hit wrong button or brake pedal Sensor or actuator failure on vehicle
Consequences of Error	Disrupt traffic flow and lower capacity Rear-end collision with preceding vehicle Rear-end collision by vehicle in back Lane departure collision
Recovery Points	Minimum safe headway setting on vehicle to avoid collisions Rear-end collision warning Vehicle warns driver that headway maintenance function is not working or is failing or has been turned off; prevent easy override of SHM and RECA functions Blind spot or lane departure warning
Individual Differences	Education Age Driving Skill
Criticality	Essential

Table 20.	Driver	task	description	for	ERSC	2	driving
1 4010 20.	211101	<i>cabi</i>	a courrent in	101	LICOU	-	

# Task Analysis–Fall-back Mode

In the fall-back mode, the rear-end collision avoidance system suffers a partial failure when one control channel fails, but the other one remains intact. If the failure is such that the speed and headway maintenance system can still function then the vehicle switches to ERSC 1 operations in which the driver is responsible for emergency or hard braking. The vehicle warns the driver that he is now responsible for rear-end collision avoidance and increases the headway to a safe distance for manual collision avoidance.

If the warnings fail, the vehicle notifies the driver and the driver performs the task without the aid of warnings.

Task	Fall-back mode to ERSC 1					
General	Task may be similar to manual driving					
Initiating Cue	Vehicle warns driver to resume responsibility for braking					
_	Driver receives warning signal for braking from vehicle					
	Vehicle increases headway for to allow for manual braking					
Feedback	Visual, auditory, and kinesthetic cues of accelerating, braking, slowing, and					
	steering					
Task Standards	Duration the same as manual driving, transition to manual driving must be					
	gradual, vehicle keeps deceleration low while increasing time headway					
Task Conditions	Weather, road conditions, suggestions from traffic control center					
Skills Required	Reaction times appropriate to speed, following distance and roadway					
_	conditions.					
	Perceptual judgment of vehicle kinematics					
Potential Errors	Driver inattention causes vehicle to have a rear-end collision					
	Driver overrides speed and headway maintenance system accidentally					
	Driver does not respond in time to rear-end collision warning					
	Vehicle fails to maintain speed or proper headway					
	Vehicle selects wrong speed or unsafe headway					
	Vehicle does not warn driver adequately of a control transfer					
Causes of Error	Fails to perceive or misinterprets system feedback					
	Poor judgment of speed and following distance					
	Hit wrong button or brake pedal					
	Sensor or actuator failure on vehicle					
	Role confusion					
Consequences of Error	Disrupt traffic flow and lower capacity					
	Rear-end collision					
	Rear-end collision by vehicle in back					
Recovery Points	Minimum safe headway setting on vehicle to avoid tail-gating					
	Rear-end collision warning					
	Vehicle warns driver that headway maintenance function is not working or is					
	failing or has been turned off					
	Vehicle requires active response from driver before a control transfer					
	Driver interface shows headway and driver control status clearly					
Individual Differences	Older or less experienced drivers may be uncomfortable taking over braking					
	control at high speeds or in reduced visibility					
Criticality	Essential					

Table 21 Driver resumes control of rear-end collision avoidance function.

# Task Analysis-Rear-End Collision Avoidance

Drivers avoid collisions by using both the brake and the throttle. <sup>(66)</sup> In most cases the driver only uses the brake. The driver may not be able to override the rear-end collision avoidance system for braking. This may limit the driver's options for lateral and rear-end collision avoidance if they can only control the steering.

Human factors studies suggest that collision avoidance systems need multiple levels of

warning.<sup>(64)</sup> In the first level the collision warning system shows the driver that it is operating with a head up display or other indicator that shows the distance to the car ahead. In the second level the collision warning system warns of a possible collision with a visual

warning such as changing the color or intensity of the display. In the third level the collision warning tells the driver to brake with combined auditory and visual warnings. These levels could correspond to no threat, a distance warning and a conservative time to collision based warning set by the driver. The driver needs to control the vehicle's steering during hard braking. This requires the vehicle to warn the driver. Table 22 shows the driver task description for this task. Rear-end collision avoidance systems may also help decrease the number of single vehicle roadway departure crashes since many of these crashes occur when the driver tries to avoid a rear-end collision. <sup>(114)</sup>

Task	Steer while vehicle brakes to avoid a collision
General	Task similar to manual driving
Initiating Cue	Warning from vehicle that emergency braking occurs
Feedback	Visual, auditory, and kinesthetic cues of braking and slowing
Control Functions	Perception-response time (PRT) studies have found that drivers respond to
	unexpected hazards by avoidance steering as well as braking, driver needs to
T 1 0 1 1	control steering
Task Standards	Driver must maintain control of vehicle with steering alone
Task Conditions	Weather, traffic conditions, position of following vehicle
Skills Required	Reaction times appropriate to speed, following distance, and roadway
	conditions.
	Perceptional judgment of vehicle kinematics
Potential Errors	Respond too slowly to emergency
	Vehicle does not brake smoothly
	Driver loses control of vehicle steering
	Driver ignores vehicle's braking warning
Causes of Error	Fails to perceive or misinterprets the indication of an emergency
	Poor judgment of steering
	Rear-end collision system does not detect emergency braking situation
	Rear-end collision system has a high false alarm rate
	Driver cannot avoid collision due to no override of RECA
Consequences of Error	Lateral collision with potentially severe results for the system
	Lose control of the vehicle
	Disrupt of traffic flow and lower capacity
	Loss of confidence by the driver and public in the system
	In the case of false alarms the driver may defeat or ignore it
Recovery Points	Check-in of brakes and tires
	Improve vehicle braking capabilities
	Redundant sensors to improve detection
	Lane departure warning and steering assist help stabilize vehicle
	Vehicle to vehicle communications
Individual Differences	PRTs for older drivers have not shown them to be less alert or significantly
	slower in responding to unexpected hazards.
Criticality	Very Critical

Table 22 Driver task description for hard or emergency braking.

Task Analysis-Lane Departure Warning

The lane departure warning watches for possible lane departures. If the sensor detects a lane departure then it warns the driver. This sensor backs up the driver when he steers. Table 23 shows driver-lane departure warning system operation. The lane departure warning system should ideally warn the driver before the lane departure begins to avoid a strong swerve to avoid an accident. This could cause an accident if the driver does not respond quickly enough.

The lane departure warning helps prevent single-vehicle roadway departure crashes. These crashes cause 37.4% of all fatalities, more than any other type of crash.  $^{(114)}$  SVRD crashes are 20.8% of all police-reported crashes.  $^{(114)}$  The lane departure warning may help prevent the crashes that occur when the driver loses control and drives off the road. But many of these crashes occur when the driver is trying to avoid a rear-end collision. The combination of a rear-end collision avoidance system and the lane departure warning may reduce the number and severity of these crashes.  $^{(114)}$ 

Task	Respond to Lane Departure Warning
General	Task similar to that in manual driving
Initiating Cue	Warning from vehicle that vehicle may leave lane
	Vehicle detects change in steering angle
Feedback	Visual, auditory, and kinesthetic cues of steering
Control Functions	Perception-response time (PRT) studies have found that drivers respond to
	unexpected hazards by avoidance steering as well as braking
Task Standards	Steer back into lane center when warned
Task Conditions	Weather, traffic, roadway condition, position of vehicle in the lane
Skills Required	Reaction times appropriate to speed and roadway conditions.
	Perceptional judgment of vehicle kinematics
Potential Errors	Overconfidence in system-driver does not look
	Driver does not respond to the warning or responds too late
	Sensor false alarms (Vehicle in center of lane, system warns driver anyway)
	Sensor does not detect lane departure
Causes of Error	Fails to perceive or misinterprets the indication of an emergency
	Driver delay in response to warning
	Lane position sensor warning threshold too high or low, malfunction
Consequences of Error	Lane change/Merge collision with potentially severe results for the system
	Single vehicle departs roadway
	Loss of confidence by the driver and public in the system
	In the case of false alarms the driver may defeat or ignore warning
Recovery Points	Vehicle steering assist applies resistance or correction to steering wheel if la
	departure sensed (option)
	Multiple warnings from detector
Individual Differences	Accident statistics, citations and self report data indicate that older drivers h
	difficulty in merging, changing lanes, and exiting maneuvers (115)
	Younger male drivers (<25 years) are most likely to be involved in single
	vehicle roadway departure crashes due too drowsiness and excessive
	$speed^{(114)}$
Criticality	Critical

Table 23 Driver task description for lane departure warning.

# Issues and Risks

1. Driver confidence in warning

If the driver has too much confidence in the collision warning systems then she may not pay enough attention during lane changes or while following another vehicle. This could lead to accidents if the warning system does not detect an object or the system fails. Driver lack of confidence in the warning could cause the driver to disable or ignore the warning system. Research needs to be done on how many false alarms and detection failures are acceptable.

# 2. Enabling/Disabling Automatic Systems

This ERSC includes a speed and headway maintenance system, lane departure and blind spot warnings, and communication systems. Research needs to be done on the instrumentation needed to control these different systems. The question of whether all the automatic systems need to be controlled with one switch or many needs to be answered.

# 3. Multiple Warnings

The blind spot and lane departure warnings may occur simultaneously. The vehicle must prioritize the alarms so that the driver only has to work with one task at a time.

# 4. Driver override of braking

Even with multiple sensors the vehicle may not see and recognize everything that a human driver sees. This is especially true of unexpected objects such as animals or debris on the roadway. It may be necessary for the driver to override the rear-end collision avoidance system on the vehicle to avoid these objects. Research needs to be done on whether this driver override should be allowed to occur.

Key Results

Benefits

# Capacity

Vehicles with rear-end collision avoidance systems can safely use shorter following distances than manually driven vehicles since an automatic system has a smaller reaction time than a human driver. Figure 62 shows the maximum steady-state highway capacity for vehicles travelling 25 m/s. Rear-end collision avoidance can increase the maximum highway capacity.



Figure 62. Maximum steady state flow rates for different velocities and time headways.

# Traffic Flow Control

Consider a long segment of dedicated lane that is divided into 12 sections as shown in figure 32. The length of each section is 500 meters. Assume that an accident occurs at section 8 and interrupts traffic. In this situation, vehicles accumulate upstream of section 8. The accident is removed quickly. Due to the direct impact of the accident, the traffic density at section 8 is high after the accident has been removed. And the traffic density at section 6, 7 increases to an intermediate-high value due to the indirect impact of the accident. In addition, since section 8 is blocked by the accident, no vehicle can enter sections 9 - 12. The traffic density of these sections is very low. The initial traffic conditions after the accident are shown below.

Section	1	2	3	4	5	6	7	8	9	10	11	12
Initial Density (Veh/km/lane)	30	30	30	30	30	40	70	80	10	10	10	10
Initial Velocity (km/hour)	10	10	10	10	10	7.5	4.3	3.7	30	30	30	30

# Task D

ERSC 2 increases traffic flow rate by increasing traffic density. Therefore in this case, the traffic flow entering section 1 has a high value. The traffic flow entering section 1 is shown in Figure 63. Since the accident was removed quickly, its impact on the traffic is not very



Figure 63. Traffic flow entering section 1 during the simulation.

severe.

The figures below compare the effects of control with the case of no control for high traffic flow. All the simulations use the traffic model described in section 5 and appendix C. The example shows a situation where an accident occurs in section 8 when there is a high flow rate of traffic onto the highway.

Figure 64 shows the evolution of the traffic density when there is no control. The disturbance in section 8 propagates back down the highway and causes a severe traffic jam in earlier sections. The disturbance in traffic density brings traffic to a standstill since the lane has more vehicles than it can handle. Section 8 is the section where the accident originally occurred. The traffic flow is very jerky and inhomogeneous as the accident effects propagate.


Figure 64. Evolution of traffic density when no roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8.

Figure 65 shows the evolution of the traffic density when the controller is operating. The disturbance in section 8 propagates back down the highway but dies out quickly. This lets the highway quickly return to a constant flow rate. The roadway without control stayed jammed up and it could not handle the increased traffic. Figure 66(a) shows the velocities for each section of the roadway over time when the controller is not operating. Figure 66(b) shows the velocities for each section of the roadway over time when the controller is operating. The velocity returns to a constant speed. Figure 67 shows the evolution of the traffic flow rate for sections 8 and 3. Section 8 is the section where the accident originally occurred. The traffic flow quickly returns to its original rate. Since the traffic flow rate quickly smoothes out the highway can handle more vehicles smoothly.



Figure 65. Evolution of traffic density when the roadway controller is present. The disturbance propagates back down the roadway after the accident is removed from section 8.

When many vehicles enter the congested area, the congestion phenomenon does not gradually disappear. Instead, the congestion is propagated upstream until a traffic jam occurs. Since the simulation model is not validated to represent the traffic evolution after a traffic jam. We stop our simulation when the traffic jam is achieved. However, based on daily freeway experience, it will take a very long time for the traffic recovering to a normal condition from a traffic jam.

The accident in the dedicated lane caused a traffic jam for highway systems without control. However, the congestion phenomenon around section 8 is damped out quickly and the steady state is quickly reached for highway system with roadway control.



Figure 66. (a) Evolution of traffic velocity when there is no roadway controller.(b) Evolution of traffic velocity when the roadway controller is present. The disturbance quickly dies out and the vehicles stay at a uniform speed.



Figure 67. The traffic flow evolution with respect to time for the roadway with roadway control. The traffic flow quickly smoothes out to the steady state velocity.

### Issues and Risks

### 1. Driver Confidence

The driver may not trust the collision avoidance system to handle emergency braking itself. This may cause the driver to choose larger headways than the automated system requires. This reduces the capacity of the automated lane. The driver is not a truly parallel system to the SHM function with rear-end collision avoidance since the driver has a much slower reaction time. Extra redundancy may be needed to make the collision avoidance system as reliable as a human driver.

### 2. Liability Issues

Rear end collision avoidance leads to new liability issues. If a vehicle collides with the vehicle in front, then who is at fault? The driver may set the time headway too small or the collision avoidance system may not work properly.

### 3 Driver Attention to Driving Task

As the vehicle assumes more of the driver tasks, the driver may pay less attention to the other vehicles or the highway. This increases driver reaction time to emergencies requiring lateral collision avoidance. The type of feedback that the driver needs from the system needs to be studied.

4 Rear-end Collision Avoidance Sensors

The sensors for rear-end collision avoidance must recognize hazardous situations quickly and accurately. More research needs to done to reduce the number of false alarms and improve the detection performance of these systems.

### 5 Driver Override of Braking

The driver may need to override the automated braking system if he/she sees an object that the collision avoidance sensors do not. However, this could lead to dangerous situations and a loss of vehicular control. This issue needs further study.

### Key Findings

1 The driver can not act as a back-up for an automated rear-end collision avoidance system. There is not enough time for the driver to react to dangerous situations. The reliability requirement for the rear-end collision avoidance system is: "Under no circumstances should a single component/point failure let the vehicle crash into any moving or stationary object in the lane, and there should be no common failure mode." One possible design leads to multiple control paths and sensors to ensure system reliability. This increases the cost and complexity of automated rear-end collision avoidance system.

2 Driver warnings from the lane departure warning system and the blind spot warning system must be prioritized to avoid driver confusion if multiple warnings occur. Great care must taken in ERSCs that combine automated systems with manual driving to make sure that the driver interface is simple to operate and understand.

3 Traffic control systems can smooth traffic and respond to incidents in the dedicated lane. These controllers prevent high density traffic from slowing to a stop after a minor incident. These controllers may prove to be particularly important for the high density traffic that results from automated vehicles with rear-end collision avoidance systems and automated longitudinal control.

# Section 7: ERSC 3 Analysis

### **Description of ERSC 3**

In ERSC 3 the roadway provides and maintains a dedicated lane with a similar accessibility to vehicles as in ERSC 1, 2. The roadway provides lane reference aids that support the lane keeping function of the vehicle as well as lane preview information for smooth and accurate lane following.

The vehicle has all the capabilities as in ERSC 2. The blind spot warning evolves into a lateral collision warning. The vehicle is fully responsible for vehicle-following in the longitudinal direction and rear-end collision avoidance. The vehicle tracks the center of the dedicated lane without any support from the driver. It sends its speed, headway and operational status to the roadway. It responds to target speed commands and traffic information received from the roadway in a similar manner as in ERSC 2. The vehicle notifies the driver in case of malfunctions and responds to commands for switching between the manual and automated modes. The vehicle also coordinates its maneuvers with other vehicles when entering and exiting the dedicated lane. The vehicle warns the driver of potential lateral collisions in time for the driver to respond to the warnings.

The vehicle alerts the driver when it is time to assume manual control and transfers control to the driver in a smooth way that is compatible with driver's skills and reaction times. The vehicle uses its on-board diagnostics to notify the driver of the fitness of the vehicle to operate on the dedicated lane. In case of malfunctions or failures in some of the automated vehicle functions the vehicle has a fall-back mode that allows the vehicle to operate as in ERSC 2 or ERSC 1 or the manual mode.

The driver is responsible for merging the vehicle into the dedicated lane with the aid of lateral collision warning and maneuver coordination between the vehicles and roadway. The driver switches the automated mode on and off. Since driving is hands-off and feet-off, the driver has no responsibility during normal vehicle operation in the dedicated lane apart from deciding the vehicle's route and the end of the trip. The point at which the driver releases manual control could be on the dedicated ramp or in the dedicated lane depending on the entry configuration.

The motivations behind ERSC 3 are:

- to smooth traffic flow and increase capacity by using the roadway commands and smaller headways
- to improve safety by introducing a lateral collision warning.

ERSC 3 allows us to focus on the analysis of both the fully automated longitudinal control and lane keeping functions, i.e., feet-off/hands-off operation without the complexity of automatic lane changing. Table 24 summarizes the functions of the driver, vehicle, and roadway in this ERSC.

Raytheon

Function	Requirement	Driver/Vehicle/Roadway Tasks
Automatic Lane Keeping	<ul> <li>Senses the position and course of the vehicle in the lane and generates the appropriate commands for the steering actuator</li> <li>Keeps vehicle in the center of the lane around curves and on straight-a-ways</li> </ul>	<ul> <li>Sense position of vehicle in the lane with sufficient accuracy</li> <li>Sense or receive preview information about roadway geometry</li> <li>Keep vehicle in the center of the roadway</li> <li>High reliability required since the driver cannot serve as a back-up in case of failure</li> <li>Transfer of lane keeping function to driver should be done gradually in a smooth way</li> <li>Interact with rear-end collision function to avoid spinning or rollover during stopping emergencies around curves</li> </ul>
Lateral collision warning	Warn the driver of potential lateral collisions due to moving or stationary obstacles around the vehicle	<ul> <li>Detect objects on both sides of vehicle</li> <li>High detection rate and a low rate of false or nuisance alarms</li> <li>Warn driver in time to avoid collision</li> <li>Driver enables warning</li> <li>Respond to driver command to adjust warning thresholds</li> </ul>
Maneuver Coordination	Coordinate lane change/merge maneuvers for entry and exit into the dedicated lane	<ul> <li>Roadway sends commands to increase headway to help vehicles merge into the dedicated lane</li> <li>Vehicles in dedicated lane respond to headway commands from roadway</li> </ul>
Driver Interface	<ul> <li>Adjust thresholds on warnings</li> <li>Enable/Disable automatic functions</li> <li>Adjust headway</li> <li>Show driver what functions are operable at any given time</li> </ul>	<ul> <li>Driver should be able to:</li> <li>» adjust headway</li> <li>» adjust warning thresholds</li> <li>» enable/disable automatic functions</li> <li>Interface should be simple to understand and use</li> <li>No adjustments should be required that take driver's attention away from driving</li> </ul>
Fall-back Mode to ERSC 2	Driver should be warned of system failures or degradation in time to recover from dangerous situations	<ul> <li>Sense status of different vehicle control channels</li> <li>Safely transfer control of steering to the driver if lane keeping function fails or degrades</li> </ul>
Speed and Headway Maintenance (See Section 6-Performance Requirements)	<ul> <li>Maintain the selected cruising speed when no vehicle is ahead.</li> <li>Calculate and maintain a headway for collision-free vehicle following.</li> </ul>	Same as in ERSC 2.

ſ	Table 24.	ERSC 3 functions a	and driver/vehicle/roadway	performance requirements.
---	-----------	--------------------	----------------------------	---------------------------

Rear-end Collision Avoidance (See Section 6-Performance Requirements)	The rear-end collision avoidance function should avoid rear-end collisions due to moving or stationary obstacles in the lane under all road and environmental conditions.	Same as in ERSC 2.
Roadway/Vehicle speed & headway commands	Senses vehicle average speed and density and environmental conditions then sends commands directly to vehicle to smooth traffic	Same as in ERSC 2.
Requirements)	flow; Performs check-in and vehicle status monitoring	

# **Performance Requirements**

# **Automatic Lane Keeping**

The automatic lane keeping function keeps the vehicle in the center of the lane on normal highway geometries. The vehicle should accurately follow a desired path and provide satisfactory ride comfort over a wide range of speeds, roadway conditions, and disturbing forces. The automatic lane keeping function controls the vehicle's steering smoothly and without oscillations. The automatic lane keeping function should also prevent the vehicle from rolling over or going out of control when the rear-end collision avoidance system needs to brake on curved roads.

The driver checks-in and then enters the automated lane. The point at which the driver releases manual control could be on the dedicated ramp or in the dedicated lane depending on the entry configuration. Once the vehicle has entered the lane the driver starts the automated systems for lane keeping and speed and headway maintenance. The vehicle smoothly takes over control from the driver. The lane keeping function keeps the vehicle in the center of the lane and smoothly navigates curves. The driver is responsible for merging the vehicle into the dedicated lane with the aid of lateral collision warning and maneuver coordination between the vehicles and roadway.

The automatic lane keeping system requires sensors, a computer with control algorithms, and steering actuators. The sensors measure the vehicle's dynamics and the forward roadway geometry. Lane reference aids  $(^{30})$  help the vehicle sensors measure roadway geometry information. The computer generates control signals for vehicle steering as shown in figure 68.



Figure 68. Block diagram for the automatic lane keeping function.

### Sensors

The vehicle sensors must detect the vehicle's position in the lane and the curvature of the roadway. The sensors must also measure or estimated the vehicle's yaw rate and slip angle for smooth response. Section 7 on the vehicle lane departure warning discusses sensors for detecting the vehicle's position in the lane.

Drivers use preview information to track the roadway. The driver extracts the information to align the vehicle and to anticipate future actions. Several studies cited in McLean and Hoffman show that human drivers need a preview time of around 3 seconds at most highway speeds.<sup>(69)</sup> This number corresponds to the "effective" preview time. Distances beyond this limit do not improve system performance. The effective preview time depends on the control bandwidth and reaction time <sup>(62)</sup>:

$$T_p \approx -0.5 + \frac{4}{BW} \approx \frac{3}{BW}$$

 $T_p$  is the preview time. BW is the bandwidth of the controller. Automatic controllers with steering assist will need less preview information if they have larger controller bandwidths than a human driver.

Preview information is important for the performance of the hands off lane keeping control. <sup>(84,4)</sup> Automatic steering with preview information feedback can tolerate the actuators with lower bandwidth than the one designed without preview. <sup>(62)</sup> However, more preview means more complex data processing and delays for the sensors.

The required accuracy's of the lane position and preview sensors depends on the lane and vehicle width. Since the goal is to keep the vehicle within a few inches of the center of the lane at all times the sensors must at least detect the vehicle's position within a few inches.

There are several combinations of sensors that can meet these requirements: magnetic markers, vision-based sensors, radar systems, navigation data from the roadway, and global positioning data.

### 1. Magnetic-Marker-Based

This sensing technique requires a lane reference aid with high degree of roadway instrumentation. In such a lane reference aid, magnetic markers are installed along the center of the lane with about 1 meter spacing. Each marker is with 2.5 cm diameter and 10 cm long, and can provide a 25 cm radius magnetic field when measured at 12 cm above the road surface. <sup>(95)</sup> The road geometry information can be stored in an on-board database or in the markers as binary codes. When the vehicle is traveling in the lane, the on-board magnetic reference sensor measures the magnetic field to determine the lateral deviation and direction. The sensor can also read the lane geometry information from the markers or the on-board database.

The installation and maintenance of such a lane reference aid can be more costly than the other sensing techniques. However, the vehicle will require less hardware to accommodate the lane reference aid. Experimental results indicate that such technique allows the vehicle to track the center of line with small tracking errors (< 15 cm)<sup>(84)</sup> at slow speeds. This lane reference aid tends to be robust to the change of weather conditions.

### 2. Vision-Based

Vision-based sensing uses cameras to detect a vehicle's position on the road. <sup>(86,4)</sup> Visionbased sensing needs more instrumentation on the vehicle. The vehicle will be equipped with a forward looking camera to sense the vehicle's lateral deviation and roadway geometry. The image signals are sent to an on-board real-time image processing unit that processes the image data. The processor extracts roadway parameters by finding the edges of the roadway and fitting these edges to a roadway model. The camera needs to cover wide angles and large distances to provide the required preview information.

Vision based sensing requires few infrastructure changes. However, it requires more instrumentation on the vehicle. The lane reference aid can be as simple as standard lane markers (stripes) on the roadway to indicate the lane. The lane markers on the lane surface need to give clear contrast for the video camera. Experiments <sup>(4)</sup> indicate that standard white markers 9 feet long, 4 inches wide, with 15 feet spacing on two edges of the 12 feet wide freshly sealed asphalt lane can provide excellent contrast. The lane lines can be easily furnished and maintained. However, vision systems might not be effective in some weather conditions or at night.

# 3. Radar-Based

Radar-based sensing tracks the vehicle's lateral position with respect to reflector(s) mounted along the road lane. The reflector can be aluminum foil glued to the plywood wall surface.<sup>(23)</sup> Experiments <sup>(23)</sup> show that a vehicle can stay within  $\pm 2.5$  cm of the desired lateral position when the vehicle is traveling at 14 m/s.

The reflectors can also be corner cubes (103) installed along the road lane. The vehicle has transceiver(s) to emit and receive laser beam signals. When the vehicle passes through the

corner cubes, the vehicles recognize the corner cubes by detecting the reflected laser beam. The vehicle then measures the moved distances and estimates the lateral position and heading relative to the corner cubes based on the triangulation principle. The estimated information is then used as the initial values in the iterated calculation for final position and heading determination. Experimental results <sup>(103)</sup> show that tracking errors within millimeters can be achieved at speed 20 km/h.

The main disadvantage in the radar sensing is that it provides little preview information. Without adequate preview information, the vehicle may not improve its tracking performance especially when it travels at high speeds.

### 4. Navigation Data from Roadway

The roadway could also send data to the vehicles about roadway curvature. <sup>(33)</sup> This approach uses beacons at fixed spots to tell the vehicles about the roadway ahead. The vehicle then uses information on the location of the beacon and dead-reckoning for preview information. <sup>(55)</sup> Dead-reckoning systems use wheel odometers accelerometers to measure distance and gyroscopes to measure the vehicle's heading. This approach has already been tested in the Prometheus project. <sup>(10)</sup>

# 5. Global Positioning System

The global positioning system (GPS) provides absolute position data free from error accumulation.<sup>(55)</sup> However GPS accuracy is degraded by tall buildings, tunnels, and large trees that are common in large cities. The vehicle can then use on-board maps and dead-reckoning to calculate the preview data. <sup>(55)</sup> The absolute position accuracy of GPS can then

provide feedback and calibration signals to correct dead-reckoning errors. These systems can be integrated with algorithms that switch between systems depending on which has better conditions for operation or a filter can combine all sensor information to get the best

estimate.<sup>(55)</sup> This method suffers from time delays which may make it impractical for realtime control applications.

# Control Algorithms

The control algorithms need to keep the vehicle in the center of the lane. The accuracy depends on the lane width, the vehicle length and width, and the roadway design parameters such as curvature. The controller must not oscillate in position. The controller should also restrict the amount of lateral acceleration to less than 0.08 g and the jerk to less than 0.17 g/s for driver comfort. <sup>(15)</sup> Figure 69 shows the roadway geometry and required accuracy's for an automated lane keeping system. <sup>(82)</sup>



(a) (b) Figure 69. Roadway geometry for the lane width calculations. (a) shows these calculations for lane width on a straight section of roadway. (b) shows the turning path for a vehicle on a curve.

One way to increase highway capacity without building more highways is to increase the number of lanes on the same right-of-way. This will require narrower lanes. In California most freeway lanes are 12 feet (3.7 m) wide with a minimum radius of curvature of 300 meters. Automatic lane keeping is another approach to allow vehicles to cruise on narrowed lanes at freeway speeds. The lane width can be made smaller by taking advantage of the precise lane tracking performance of automated lane keeping. Table 25 gives some parameters for the design vehicles of different types.

Design Vehicle Type	Height (Feet)	Width (Feet)	Length (Feet)	Front Overhang (Feet)	Rear Overhang (Feet)
Passenger Car	4.25	7	19	3	5
Single Unit Truck	13.5	8.5	30	4	6
Single Unit Bus	13.5	8.5	40	7	8
Semi-Trailer Truck	13.5	8.5	50	4	6
Large Semi-Trailer	13.5	8.5	55	3	2
Full-Trailer Truck	13.5	8.5	65	2	3
Recreational Vehicles		8	30	4	8

Table 25. Design vehicle parameters for highway design.
---

There is a trade-off between the lane width and safety. The vehicle may not cross the lane boundaries while in lane keeping mode. Therefore, the required minimum lane width will depend upon the width of the vehicle and the performance of the automatic lane keeping. Vehicles with wider bodies (e.g., trucks) will need wider lanes. The lane has to be wide enough to accommodate the lateral deviation error caused by the lane keeping system.

In the U.S., 12 feet (3.7 meters) wide lanes are used on more than 97% of the total length of all interstate highways and more than 82% of the total length of other principal arterial

highways.<sup>(1)</sup> Drivers (of trucks and passenger vehicles) in U.S. are already used to driving on highways with standard 12 feet lane width. Most passenger vehicles have width less than 7 feet. If the automatic lane keeping can maintain tracking error within 2-3 inches (5-7.5 cm) bound, 8 feet lanes can be used for passenger vehicles. The width for most trucks is less than 8.5 feet. If the tracking error can always be kept within 6 inches (15 cm), 10 foot lanes can be used for trucks. Some preliminary research/experimental results <sup>(84)</sup>, have demonstrated that such high accuracy lane tracking is possible. More reduction of the lane width is possible if we can further restrict or regulate the width of the automated vehicles. The roadway width for curves needs to be wider to allow for the track width of the outside wheels as shown in figure 70 a. The track width changes with the radius of curvature. The minimum radius of curvature for highways in California is around 300 m. Human drivers require extra width on curves. The amount of extra width depends on the design speed of the highway .  $^{(82)}$  Even for narrowed lanes, the rules of highway geometric design still apply. Curve widening will be necessary since the vehicle's front and rear wheels do not track exactly the same trajectories on a curved roadway. This is particularly true for combination-unit trucks. The required width increment will depend on the roadway curvature. Up to a 3 foot per lane increment may be necessary for trucks on sharp freeway curves.





**(b)** 

Reduction of lane width implies the reduction of the required real estate, construction and maintenance costs. For ERSC's with multiple dedicated lanes, capacity gains are also possible. For examples, 3 standard 12 feet lanes can be converted into 2 narrowed lanes restricted for automated passenger vehicles and 2 narrowed lanes for any automated vehicles. This leads to a potential 33% increase of traffic capacity. The benefits will be significant especially in the urban areas where the land is expensive and traffics get congested often.

The automatic lane keeping has to be so reliable that the benefits of lane width reduction will not be offset by the increase of lateral collision rates. The automated check-in procedure needs to make sure that the vehicle passes the more restrictive fitness test to operate in the narrowed lane. That will insure the vehicle's desired lateral control performance. The required infrastructure instrumentation to support the lane keeping also needs to be checked frequently to ensure proper interaction with on-board lateral control sensor(s).

#### Actuators

This system will use hydraulic or electronic steering components. Electronic steering may increase the energy efficiency, driver comfort and stability of the steering system. <sup>(94)</sup> The actuator should have a small delay and respond accurately to the commands from the driver. The steering actuator needs to be reliable and fail-safe since the driver will not be able to regain control if the steering fails suddenly. <sup>(24)</sup>

#### Driver Interface

The transfer of control from the driver to the automatic lane keeping system depends on the entry/exit design of the highway. If there are dedicated on and off ramps for the dedicated lane then the vehicle may take over lateral control on the on-ramp. Other entry/exit configurations require the driver to steer the vehicle into the automated lane during entry and to resume steering control before exit takes place, the automated lane should be kept at standard lane width near entry and exit points. It is desirable to gradually reduce the lane width from some point near an entry point and gradually increase the lane width from some point where the driver's resumption control begins.

The driver may not feel safe or comfortable when the vehicle is cruising at high speed in narrowed lanes. The driver's reaction to narrow lanes is a human factors issue and requires more investigation. The driver may be uncomfortable taking over control of the vehicle in the fall-back mode in narrow lanes. The potential damage caused by a lateral collision in the narrowed lanes may be more serious than in the case of standard lanes. All these concerns can only be alleviated by improving the reliability of automated control functions.

If the driver is allowed to override the system for lateral collision avoidance, the driver may have difficulties in maintaining the vehicle in the lane. In the case that the on-board lane keeping system fails, it will be difficult for the driver to take over lateral control in a narrowed lane. This implies that the driver is given less reaction time to plot and execute steering corrections in emergency.

#### Issues and Risks

1 Preview Data- The vehicle requires preview data for smooth steering and lateral control. The vehicle may depend on the roadway to help get this data. The system reliability requirements may require two independent sources of data for good results.

2 Weather concerns- The controller needs to safely steer the vehicle during all types of weather. Many of the lane reference aids do not work well when the weather is snowy or visibility is limited.

3 Safe driver override - The driver may need to override the automatic functions if he/she sees an obstacle that the vehicle does not. The override must not put the driver in an unsafe position.

4 Reduced lane width - Automatic control may allow smaller lane widths. However, if the vehicle needs to fallback to a manual control mode this change may not be practical.

# Lateral Collision Warning

# Task D

The lateral collision warning system warns the driver of lateral collisions. Most lateral collisions are of the lane change/merge variety with a smaller number of lane departure collisions. Most (95%) of lane change/merge crashes are angle or sideswipe collisions. <sup>(113)</sup> The driver of the merging car either did not see the other vehicle or misjudged the distance between the vehicles. These crashes usually happen in dry, clear, daylight conditions. <sup>(113)</sup> The driver "did not see" the other vehicles until too late. The lateral collision warning system on the vehicle detects vehicles in adjacent lanes that are traveling in the same direction. <sup>(59)</sup> Figure 71 shows the different types of lane change crashes. <sup>(21)</sup> Most collisions are of the proximity type, only around 10% of crashes are of the fast approach variety. <sup>(21)</sup>



Figure 71. Lane change crash subtypes and variations. <sup>(21)</sup> SV is the subject vehicle. POV is the Principal Other Vehicle. (a) Types of proximity crashes. Both vehicles have a small velocity differential. (b) Fast approach crashes. There is a large velocity difference between the vehicles that rapidly closes the longitudinal gap between the vehicles.



Figure 72. Block diagram of the lateral collision warning system. Sensors detect possible obstacles. Then the warning algorithm calculates the time to collision and warns the driver of potential collisions.



Figure 73. The shaded area shows the region of coverage for a blind spot sensor on the left side of the vehicle.

This type of lateral collision warning system helps avoid lateral crashes. <sup>(21)</sup> The lateral collision warning system has a sensor and a system that warns the driver. Figure 72 shows the block diagram of the lateral collision warning system. The communications system helps the vehicles signal intent, a sort of electronic turn signal. This type of signal can come from the turn signal or other indicators of the driver's intent to change lanes such as lane position, eye movements, or steering wheel angles.

# <u>Sensor</u>

Since lane change/merge accidents occur with equal frequency on both sides of the vehicle

there should be lateral collision detection sensors on both sides of the vehicle. <sup>(64)</sup> Figure 73 shows the left-side lateral collision sensor zone for a vehicle. These sensors should increase safety as vehicles enter and exit the automated lane. The sensor needs to detect vehicles accurately with few unnecessary warnings or false alarms. Too many warnings will cause the driver to turn off the system or ignore its warnings.

The lateral collision detector should cover the adjacent lane to the side and behind the vehicle. The detection zone should cover the full length of the vehicle on both sides. <sup>(21)</sup> The sensor should reliably detect small vehicles such as motorcycles in all likely lane positions. <sup>(64)</sup> The detection capability might also extend into adjacent lanes to avoid collisions with vehicles going out of control. However this type of collision is rare and it may happen too quickly for the driver to respond. The driver may not be able to avoid it since drivers have been found to have an average reaction time of 0.82 seconds to swerving emergencies. <sup>(21)</sup> The reaction time may be longer for automated vehicles when the driver is "hands-off and feet-off." We use 0.82 seconds as a lower bound for the collision avoidance warning.

Currently available lateral collision sensor systems use ultrasound, radar and video systems.<sup>(76)</sup> Ultrasonic systems transmit ultrasound pulses to detect objects and measure

their distance. Radar systems use frequency modulation and range-gating to find the range and relative velocity of objects. Video systems can show the driver a shot of the blind spot or they may add in computer vision technology to search for objects. These systems can be improved if each vehicle signals its "intent" to leave or enter the lane.

Ultrasonic sensors detect objects and their range with pulses of ultrasound. The sensor estimates the pulse time-of-flight. Sensor accuracy depends on local air properties, such as humidity and temperature. These sensors are short range  $(0.3-10 \text{ m})^{-(76)}$  with accurate range measurements. These sensors have too short of a range to be useful for fast approach lateral collisions.

Radar sensors measure the range and relative velocity of objects. The radar estimates the pulse time-of-flight with range gates that look for return pulses at different time ranges. Pulse doppler radars measure the relative velocity between vehicles. This lets the radar tell which vehicles are approaching and which are dropping back. Radar sensors have a longer range than do ultrasound sensors (15-45 m), but they are not as accurate.

Video systems can show the driver an image of what is happening around them. This system has image processing that searches for vehicles. These systems use the continuity between image frames to track vehicles and estimate their distance. These systems have problems with one vehicle blocking or obscuring another.

The lateral collision warning system may also use a combination of these sensors to get full coverage around the vehicle and to track different types of targets. The driver may have a

driver situation display that shows the driver's vehicle as well as surrounding vehicles. <sup>(21)</sup> This system could actively warn the driver or leave all judgments on maneuver safety to the vehicle. The vehicles may also use a cooperative communication system to help track other vehicles. In this system each vehicle would broadcast its position and heading to the surrounding vehicles.

### Warning

The lateral collision warning system may be active whenever the vehicle ignition is on and the vehicle is going forward or whenever the driver turns it on. <sup>(64)</sup> This system may have many nuisance alarms caused by slow traffic or parked cars. Nuisance alarms occur when the sensor detects an object that is not there or is not a threat. The system could also be active only if the driver indicates an intent to change course, such as with a turn signal. This system relies on the driver to signal course changes. The lateral collision warnings may also be triggered by a change in the vehicle's steering angle that signals a lane change. This warning would probably occur too late to avoid an accident.

The lateral collision warning system should have at least two levels of warning: an imminent crash avoidance warning and a cautionary warning.  $^{(98,64)}$  A cautionary warning tells the driver that there is a potential collision. This warning may operate when the vehicle is moving forward or only when the automated systems have been turned on by the driver. An imminent lateral collision warning alerts the driver to an imminent crash avoidance situation that exists when an object is sensed and there is an indication that the vehicle's path may cause a collision with that object.  $^{(64)}$ 

The imminent lateral collision warnings should turn off when the hazard disappears or the driver responds to the warning. The cautionary warning stays on until there is no vehicle in the sensor field. The warning should give information on the location and speed of the threat without causing the driver to look away from the roadway. Directional audio systems can tell the driver the location of the threat and help them recognize the threat more quickly. <sup>(9)</sup>

The lateral collision detector may turn on only when the driver turns on the turn signal to change lanes or it could be on the entire time the vehicle is in the automated lane. But if the lateral collision sensor was on continuously, it would sound warnings every time the vehicle passed another vehicle. Another option is have the warnings sound only when the turn signal is on.

### Driver Interface

The lateral collision warning system provides levels of warning to the driver and provides a detector sensitivity adjustment for the driver. The different warning levels allow the vehicle to warn the driver without intrusive nuisance alarms. The detector sensitivity adjustment lets the driver adjust the range of the sensor to reduce sensitivity and false alarm rates. <sup>(64)</sup>

The cautionary warning should be visual and located within 15  $^{\circ}$  above the line-of-sight of the side view mirror. <sup>(64)</sup> Visual signals are more detectable if they are near the center of the line-of-sight. The imminent lateral collision warning should use two types of warnings.

Lerner et al suggests a combination of auditory and visual warnings. <sup>(64)</sup> The auditory warning can help the driver determine the direction of the imminent collision with a directional signal. This system has been tested on pilots in simulators. A directional audio signal helps a pilot locate a potential problem 1.5-2 seconds faster than a non-directional warning.<sup>(9)</sup>

A more sophisticated system is the driver situation display that renders the driver's own vehicle and the vehicles around it. <sup>(21)</sup> This display gives the driver information on whether it is safe to perform a lane change and where potential trouble might lurk. The sensors would provide vehicle location, velocity, and closing rate data. This display could give the driver the situation awareness data needed to guide the driver's actions.

The lateral collision warning system may have an adjustable threshold. This lets the driver reduce the sensor range to avoid excessive false alarms in which the driver perceives frequent hazards. For example, on multiple-lane highways the sensors may be triggered by objects across the adjacent lane or by vehicles two lanes away.

### Issues and Risks

### 1 Sensor Capabilities

The lateral collision sensor system may increase safety and reduce certain types of lane change/merge crashes. But there are sensor and human factor issues that still must be resolved. The sensors must detect small vehicles, such as motorcycles with few nuisance

alarms. A video system may be able to track small vehicles better during daylight. The video system sees what the driver sees so it may be more natural for human drivers to use. The performance of vision based systems may degrade at night and in rainy weather plus it is more expensive and complicated than radars of ultrasonic systems.

### 2 Warning system desig n

First the designers need to decide when the system should be turned on: always on, only when turn signal on, or turned on when steering column shows lane change. The warning thresholds for emergencies and driver awareness must also be set. The system must avoid nuisance alarms, yet detect vehicles accurately. The types of alarms and warnings must also be researched. The emergency alarm needs to be directional to get fast driver response.

### 3 Multiple Warnings

ERSC 3 introduces multiple types of driver warnings for lateral collisions. Research needs to be done on the interaction and prioritization of these warnings for the human driver. For lateral collision avoidance the driver may have very little time to react to any warnings let alone try to figure out what to do.

# **Maneuver** Coordination

This ERSC introduces limited maneuver coordination to help the vehicles merge into the dedicated lane provided by the roadway. The roadway has sensors that monitor traffic backup for entry onto the dedicated lane. If there are many vehicles waiting to enter the dedicated then the roadway can recommend that the vehicles increase their gaps to allow the drivers to merge onto the highway. This system is similar to today's ramp metering systems. Merge maneuvers may also be simplified by using "rules of the road" that specify which vehicle must give way when a merge occurs. Figure 74 shows how this system would work.

This maneuver coordination tries to make merging easier by increasing gaps between vehicles. However it is still the driver's responsibility to perform the lane entry maneuver. This means that the roadway control system cannot assign each vehicle a slot in traffic. The roadway may assist the driver by sending data on traffic gaps, but it is still the driver's responsibility.



Figure 74. Geometry and operation for entry/exit maneuver coordination. The roadway adjusts the speed and headway of the vehicles to allow for easy entrance in segment  $S_{i+I}$ . (a) Geometry for designated entry/exit points. (b) Geometry for dedicated ramps.

#### Roadway Sensing Systems

The roadway needs to sense the average traffic density over sections of the roadway and the amount of traffic entering the dedicated lane. This data can be measured by sensors that are part of the infrastructure. Proximity sensors such as inductive loops count the number of vehicles that pass over them in a given time period. <sup>(112)</sup> The data is collected over sections of roadway to monitor traffic flow. <sup>(112)</sup> The sensors can be hardwired to a roadway traffic control center or they can transmit their data electronically to the traffic control center. These sensors may require changes in the infrastructure and roadbed. These sensors have been in use on freeways for at least a decade and many urban areas already have them in place for traffic monitoring and incident management. <sup>(112)</sup>

Overhead infrared sensors have also been tested for traffic monitoring. <sup>(80,48)</sup> These sensors can measure vehicle speed, density, and vehicle type. This sensor type may allow the roadway to choose minimum headways based on the mix of vehicle capabilities.

### Communications Requirements

The vehicle needs to check-in with the roadway. The roadway needs to communicate with vehicles on different sections of roadway to adjust their gaps and speed to aid merging. This system can use a roadway-vehicle communication system similar to that described in ERSCs 1 and 2.

### Driver Interface

The driver needs to know when gaps in the traffic may be approaching the merge point. The roadway could send a message to the vehicle in the form of in-vehicle signing or a ramp metering system. This would help the driver merge.

### Issues and Risks

1 Driver compliance- Maneuver coordination depends on driver coordin ation with the commands from the roadway. If the drivers ignore the inputs from the roadway the vehicle could crash.

2 Entry/Exit Configurations - This type of maneuver coordination with probably require a dedicated ramp so that entering vehicles can match speeds with the vehicles on the roadway.

# Driver/Vehicle/Roadway Interface

The driver should be able to override the speed & headway maintenance function and adjust the headway as described in Section 7 on the SHM function. The driver should also be able adjust the warning thresholds on the lane departure warning and the lateral collision warning as described in sections 6 and 7 on the lane departure warning and lateral collision warning functions. The driver also receives highway traffic information through the vehicle and may use it for route planning. The driver may also receive preview information from the roadway. This data may tell the driver about sharp curves up ahead or other roadway conditions. The driver also monitors the vehicle's operational status and backs up failed vehicle functions if possible. If the automated lane keeping system fails the driver takes over manual steering of the vehicle in the fall-back mode.

The driver interface must show the driver that the automated functions are operating correctly or that a function has failed. This interface should be simple to understand and use. No adjustments should be required that take the driver's attention away from the driving task. Input systems like touch pads or buttons may be distracting and may require too much driver attention.<sup>(8)</sup> The driver must be able to tell what the vehicle is doing at all times and why it is doing it. This shows the driver that the vehicle is operating correctly. Studies show that drivers cooperate more when they know the reasons for an action.<sup>(10)</sup>

Head-up displays may be particularly useful for visual indicators of the system operation. For example a heads-up display could show the measured distance to the vehicle ahead as well as the minimum time headway and target speed. Other technologies include liquid crystal displays (LCDs), light-emitting diodes (LEDs), computer generated voice messages, sudden vehicle accelerations/decelerations (jerk), tactile feedback, and directional audio displays.<sup>(64,8,9)</sup>

The interface may be continuous or only active when a driver action is needed. The best design of the driver interface for human factors is an open issue and involves selecting a display technology, type and duration of warnings, user control types and locations, etc. User acceptance is crucial to user interface design. The systems in each ERSC must be easy to use and understand and easy to adjust without the driver taking his eyes off the roadway for more than 1-2 seconds . <sup>(73)</sup>

### Issues and Risks

1 Role Confusion

The driver always needs to know what he/she is responsible for. It may not be a good idea to allow a vehicle to have different modes of operation.

- 2 Driver Training
- 3 Complexity

The system must be easy to use and understand.

# Fall-back Mode

The system needs to degrade gracefully if a failure occurs. Many failures such as lateral collision warnings may not cause the vehicle to be non-operational. Vehicles with different capabilities may coexist in the dedicated lane without incident.

In case of failure in a single control channel of the automatic lane-keeping system, the vehicle can operate in ERSC 2 control mode with the SHM and rear-end collision avoidance functions. The vehicle may decrease its speed to a safe value for ERSC 2 and the driver resumes control of the steering function with the help of the steering assist system and the lane departure warning. However, according to Riley decoupling warning and avoidance modes can cause "role confusion" for the driver. <sup>(90)</sup> The driver may also be uncomfortable dealing with a sudden change in responsibility especially during hands-off, feet-off driving. The driver may also be uncomfortable taking over steering control at higher speeds or if there is reduced visibility. <sup>(90)</sup> Safe system design may require that the driver can force the vehicle not to transfer control until the vehicle is ready to leave the dedicated lane. <sup>(90)</sup> Meisner et al believe that such transfers between automated and manual functions need to be handled

jointly by the driver and the vehicle. <sup>(70)</sup> The vehicle should not transfer control to the driver in dangerous situations. The driver should not be forced into an uncomfortable situation.

If the lateral collision warning or the maneuver coordination functions fail the vehicle informs the driver. Failure of these functions does not require a transfer of control to the driver so they do not require the vehicle to return to ERSC 2 control.

### Vehicle to Roadway Communications

The roadway measures the average speed and traffic density, assesses environmental conditions, and calculates vehicle target speeds and minimum safe headways. The vehicles follow the target speeds set by the roadway using the speed and headway maintenance function. The vehicles should respond to headway recommendations for longer headways due to changes in environmental conditions such as slick roads. The vehicles use the environmental data from the roadway to adjust their time headways. The vehicles check-in with the roadway by sending their identification, automated function status, and location. <sup>(108)</sup>

route planning purposes. This section covers roadway-vehicle communications for check-in in ERSC 3.

### Vehicle to Roadway Communications (VRC)

The roadway sends speed data to the cars to smooth traffic densities, manage incidents, and aid in maneuver coordination. The vehicles may also send their measured velocity and headway to the roadway. The VRC system will be used for check-in. This allows the roadway to monitor the operational status of the vehicles and ensure that they are correctly operating in the automatic mode. These messages only need to be sent once per section. Each vehicle sends status data on all of its automated functions to the roadway. In ERSC 3 these functions are: speed and headway maintenance, rear-end collision avoidance, lateral

collision warning, automated lane keeping, VRC systems, and critical engine functions. <sup>(108)</sup>

VRC systems can be mounted on the side of the roadway, on overpasses, or embedded in the roadway.<sup>(85)</sup> These systems may use automated toll collection technology with variable registers or a system that covers a wider range. <sup>(22,53)</sup> Appendix A covers the operation of these systems.

The VRC will send messages from the vehicles to the roadway. Each vehicle sends its speed and the distance to the preceding vehicle (6-8 bits each). The vehicles also send a unique identifying code (at least 28 bits) to the roadway for check-in. The operational status of the automated system can be sent as well, this could take 8-16 bits depending on the amount of detail required by the roadway. Other variables such as estimates of the road-tire friction coefficient can take 4 or more bits. Table 26 shows the possible variables to be sent to the roadway for check-in. <sup>(108, page 25)</sup> This gives a data length of around 90 bits. The overhead for synchronization and error detection and correction gives a message length of around 200 bits.

Function	Components	Number of Bits
Speed & Headway Maintenance	Sensors, Actuators, Controller	1-6
Rear-end Collision Avoidance	Sensors, Actuators, Controller, Vehicle-Vehicle Communications System	1-8
Lateral Collision Warning	Sensor, warning	1-2
Vehicle-Roadway Communication System	Transmitter, receiver, variable register	1-2
Automatic lane keeping	Sensors, Actuators, Controller	1-6
Vehicle Identification Code	Communication code	28
Other Vehicle Functions	Critical fluids level and pressures, brake conditions, tire pressure, engine temperature, headlights	12-16

Fable 36	Determinal status		he cout to	the meed-meet	has the	for for	ale ale in
i able 20.	Potential status	variables to	) de sent lo	lne roadway	DV LNe	venicies for	спеск-іп.
					~		

A broadcast system with a range of 100 meters for a single automated lane must communicate with 0.3 to 0.75 vehicles per second. Analysis by Polydoros et al shows that the current Hughes VRC system and automatic toll collection systems can meet the data transmission requirements for VRC in ERSC 3. <sup>(85)</sup>

#### Issues and Risks

1 Dedicated Lane

An issue for AHS is how to get a separate lane for automated vehicles. The driving public rebels

when a lane is taken away from active use.<sup>(99)</sup> The solution for most highway departments has been to add a new lane on the shoulder or median strip, but most urban highways do not have any more space to spare for the sole use of automated vehicles. This infrastructure change may have to wait until enough vehicles have speed and headway maintenance systems and traffic flow benefits can be easily seen.

#### 2 Legal/Liability

In this ERSC the roadway sends recommended target speeds and minimum headways directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the minimum recommended headway is too short. The government may not wish to deploy such a system until it has been shown to be reliable and fail-safe.

3 Sensors

Roadway weather sensors only cover small regions of roadway. These sensors could easily miss patches of ice or puddles on the roadway. The roadway may then choose an unsafe headway. Research needs to continue into these sensors to improve their capabilities.

4 Privacy

The vehicles "Check-in" with the roadway. This may leave a record of when and where a vehicle traveled. This raises user privacy concerns about governmental use of this data.

5 Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

### **Reliability Requirement Analysis**

In additions to the vehicle functions in ERSC 2, the vehicle can perform automatic lane keeping (ALK) and lateral collision warning (LCW) in ERSC 3. The on-board ALK system together with the speed/headway maintenance and rear-end collision avoidance (SHM/RECA) system allows the driver feet-off and hands-off when the vehicle is in the dedicated lane. The driver is assisted by the LCW system to carry out lateral collision avoidance. To improve the safety during the entry/exit of the dedicated lane, the roadway may assist the maneuver coordination near the entry/exit points.

The ALK system is designed in such a way that it can be overridden by the driver for lateral collision avoidance maneuvers. However, while the vehicle is following the lane and cruising at high speed, it is not guaranteed that the driver can safely take over a failed ALK system. The driver can not be regarded as a back-up to the ALK system. This observation leads to serious reliability requirements for a fail-safe ALK system design. The structure complexity and component redundancy required to implement a fail-safe design will be enormous.

The operation of automatic lane keeping requires lane reference aids from the roadway. The lane reference aids help the ALK system to obtain the necessary lateral dynamics information for lane keeping control. The lane reference aids need to be implemented based on the sensing technologies that are used by the ALK system. The reliability of lane reference aids affects the safety and availability of the ALK operation in the dedicated lane.

The LCW system warns the driver of the potential lateral collision threats. It assists the driver in performing lateral collision avoidance. A no-detections failure is likely to lead to a lateral collision especially when the driver becomes too relaxed in hands-off and feet-off driving. False alarms may affect the effectiveness of the LCW system. Improving detection rates and reducing false alarm rates are the LCW system's major reliability concern.

The roadway can help the entry/exit maneuver coordination by sending slower speed and/or larger headway commands to the vehicles near the entry/exit points. The roadway has to detect the vehicles that intend to enter and get off the lane, and broadcast the speed/headway commands. Reliable communication between the roadway and vehicles is essential for safe maneuver coordination.

### Speed/Headway Maintenance and Rear-End Collision Avoidance

The reliability analysis is mainly the same as in ERSC 2 with additional reliability requirements discussed below.

In ERSC 3, the SHM and RECA system may take advantage of the automatic lane keeping and adopt a higher desired cruising speed and/or a smaller headway. If the vehicle can not perform lane keeping, the speed and headway commands from the roadway may be too extreme for the driver to keep the vehicle in the dedicated lane. In this case, the operation of the SHM and RECA system falls back to ERSC 2, i.e., the driver has to steer the vehicle. The speed and headway used by the SHM and RECA systems may be too extreme and have to be re-evaluated. Since the SHM and RECA system determines the safe cruising speed and vehicle following headway, the controller of the SHM and RECA system needs to know the operation status of the automatic lane keeping (ALK) system. Figure 75 shows the functional block diagram of the SHM and RECA system in ERSC 3 that detects the operational status of the ALK system.



Figure 75. The functional block diagram of the SHM and RECA system in ERSC 3.

Since the impacts of losing automatic lane keeping while the vehicle is at high speed may be severe, the SHM and RECA system may require redundant, independent methods for monitoring the deactivation of the ALK system.

# Automatic Lane Keeping

The automatic lane keeping (ALK) system keeps the vehicle in the lane without the driver's steering control. The ALK system is designed in such way that it can be overridden by the driver. The driver overrides the ALK system by turning the steering wheel hard in case of emergencies.

The major components required to implement an ALK system are a lateral dynamics sensor, a controller (a computer with control algorithms), and a steering actuator. The lateral dynamics sensor collects data, such as lateral deviation, lateral speed/acceleration, yaw rate, preview information, etc. The lateral dynamic sensor is supported by the lane reference aids provided by the roadway for data collection. The controller uses the data provided by the lateral dynamics sensor to compute steering commands. The steering actuator takes the steering commands and generates steering force to keep the vehicle in the lane. Figure 76 shows a block diagram of the ALK system.



#### Figure 76. The reliability block diagram of the automatic lane keeping system.

### Reliability Requirement Estimation

The ALK system together with SHM and RECA system allows the driver complete handsoff and feet-off driving in the dedicated lane. Since the vehicle is cruising at high speed, the driver will have difficulties in recovering the lateral control if the ALK system fails. The vehicle can depart from the lane before the driver can successfully apply steering correction. Consequently, the driver can not be regarded as an emergency back up when the ALK system fails.

The ALK system is acceptable only if it is much more reliable than manual lane keeping control. The lane departure accident rates in today's manual driving can be used to estimate the required ALK system reliability. The expected number of lane departure collisions during the average vehicle life is 0.0865. <sup>(114)</sup> The average vehicle life is about 13 years. That gives us an estimate that a vehicle experiences one lane departure collision about every 150 years. The ALK system should have mean time to failure much more than 150 years if it is to be accepted by the general public.

### Reliability Functional Requirement

The driver can not be regarded as a back-up to the ALK system since he/she can be hands off the steering wheel. The ALK system has to be fail-safe enough so that an internal component failure or malfunction will not affect the safety in lane keeping. Since the driver is not expected to recover the ALK system failures, any failure that can potentially cause vehicle lane departure is not allowed. The reliability functional requirement for the ALK system is thus proposed as

"Under no circumstances, should a single component/point failure let the vehicle drift or depart from the lane, and there should be no common failure mode."

### Required Redundancy

Safe lane keeping control depends on the reliability of sensor, controller and steering actuator. The fact that these major components are connected in series, as shown in figure 76, indicates that a single component/point failure may lead to break down of the ALK system. Since the driver is not expected to help recover lane keeping control in emergency and a single component failure is not allowed to cause catastrophe, redundancy will be required in



designing the ALK system. A proposed ALK system design framework that has the potential of satisfying the proposed reliability functional requirement is shown in figure 77.

#### Figure 77. The block diagram of the proposed automated lane keeping design.

The proposed system framework has two redundant active control channels. Each control channel contains a lateral dynamics sensor, a controller and a steering actuator. Each control channel is capable of keeping the vehicle in the dedicated lane. If a component/point failure causes one control channel to fail, the other channel can still operate until the driver safely back up the less reliable ALK system. A third lateral dynamics sensor is used to improve the accuracy of the sensing and to help identify any failed sensor. A supervisory controller is used to help detect any failed component and to deactivate the failed control channel. The probability that any additional component/point fails while the driver is taking over lateral control is negligible. Safe lateral control transition from the ALK system to the driver thus can be obtained.

When one control channel is deactivated due to a potential component/point failure, the ALK system may need a period of transition time to have the other channel operate effectively. Since the vehicle may be traveling at high speeds, a lateral control degradation like this may have safety implications. To reduce the transition time required for fully activating or deactivating a control channel, it is desirable that the two redundant control channels be kept active all the time as long as the ALK system is in operation. In this case, one can activate or deactivate a control channel by quickly adjusting its control gain. Feedback design can be used to efficiently adapt the control gains based on the operational status.

When one control channel is deactivated due to a potential component/point failure, the ALK system is not in the fail-safe condition. Any additional component/point failure may cause the vehicle depart from the lane. In this case, the ALK system should warn the driver of the detected potential failure and ask him/her to take over lateral control. The ALK system should also notify the SHM and RECA system of the action taken so that the SHM and RECA system design framework allows the vehicle lane keeping control to fall back to steering assist and lane departure warning in ERSC 2.

It is unsafe that the driver takes over rear-end collision avoidance and/or speed/headway maintenance while the vehicle is performing lane keeping. The ALK system should detect the deactivation of the SHM or RECA system operation. If the SHM or RECA system is deactivated, the ALK system should ask the driver to take over lane keeping control.

The reliability functional requirement does not allow the existence of any common failure mode. A common failure mode can cause simultaneous failures in redundant components and destroy the merits of redundancy. This requirement demands that the redundancies in the proposed framework be independent. The redundant sensors, controllers and actuators have to operate with different technologies, and be supported by independent hardware/software. For example, the three redundant lateral dynamic sensors use three different technologies such as laser scanning of lane markers, vision based sensing, and magnetic marker based sensing. Furthermore, they are powered independently so that a power down will not cause two of them to fail simultaneously.

#### System Level Design Requirements

In addition to the redundancy discussed above, more detailed design requirements can be obtained by analyzing ,major potential system failure modes. Simply using the system framework in figure 77, one can easily identify the following major potential failure modes:

- 1. Lateral dynamics sensor failure
- 2. Controller failure
- 3. Steering actuator failure
- 4. Incorrect lane reference data received from the roadway lane reference aids

The ALK system design should minimize the probability of the above major potential failure modes to improve its fail-safe operation. The following important system level design requirements are to counteract the major failure modes.

Potential Failure Mode 1: Lateral Dynamics Sensor failure

- 1.1 Lateral Dynamics Sensor 1 or 2 computes incorrect lateral dynamics or lane reference data. Design requirements:
  - 1.1.1 Lateral Dynamics Sensor 1 and 2 need to have an internal independent check of the reasonableness of the computed data, based on the physical

attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed. The System Failure Response disables the channel that connects the failed component, warns the driver of the potential failure, and asks him/her to take over lane keeping control.

- 1.1.2 Controller 1 and 2 and Supervisory Controller should verify that the lateral dynamics sensors are operating in a timely manner by verifying that new data is received from the lane sensor every to be determined (tbd) time interval. If an error is detected, the System Failure Response should be executed.
- 1.2 The data values provided by Lateral Dynamics Sensor 1 and 2 are substantially different. Design requirements:
  - 1.2.1 Supervisory Controller should sense the discrepancy and use the data from Lateral Dynamics Sensor 3 to determine the sensor that fails. The System Failure Response should then be executed.
- 1.3 Lateral Dynamics Sensor 3 computes incorrect lateral dynamics or lane reference data. Design requirements:
  - 1.3.1 Lateral Dynamics Sensor 3 needs to have an internal independent check of the reasonableness of the computed data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, Supervisory Controller should be notified, and the driver should be warned of the failure and be asked to take over steering control.
  - 1.3.2 Supervisory Controller should compare the data computed by Lateral Dynamics Sensor 3 with that computed by Sensor 1 and 2 to check if Lateral Dynamics Sensor 3 is working properly. If an error is detected, Supervisory Controller should be notified, and the driver should be warned of the failure and be asked to take over steering control.
  - 1.3.3 Supervisory Controller should verify that the processor of Lateral Dynamics Sensor 3 is operating in a timely manner by verifying that new data is received from the lane sensor every tbd time interval. If an error is detected, Supervisory Controller should be notified, and the driver should be warned of the failure and be asked to take over steering control.
- 1.4 Data from Lateral Dynamics Sensor 1 or 2 is not communicated to the controllers, or is corrupted in transmission through interface. Design requirements:
  - 1.4.1 Each controller should incorporate an independent check of the reasonableness of the computed data from Lateral Dynamics Sensor 1 and 2, based on the physical attainability and valid known ranges of the

system variables. If an error is detected, the System Failure Response should be executed.

- 1.4.2 Same as for 1.1.2.
- 1.5 Data from Lateral Dynamics Sensor 3 is not communicated to Supervisory Controller, or is corrupted in transmission through interface. Design requirements:
  - 1.5.1 Supervisory Controller should incorporate an independent check of the reasonableness of the computed data from Lateral Dynamics Sensor 3, based on the physical attainability and valid known ranges of the system variables. If an error is detected, Supervisory Controller should be notified, and the driver should be warned of the failure and be asked to take over steering control.
  - 1.5.2 Same as for 1.3.2.

Potential Failure Mode 2: Controller failure

- 2.1 Controller 1 or 2 computes incorrect steering commands. Design requirements:
  - 2.1.1 Each of Controller 1 and 2 should incorporate an independent chec k of the reasonableness of the computed steering commands before it is sent to the steering actuator, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
  - 2.1.2 Each of Actuator 1 and 2 should incorporate an independent check of the reasonableness of the received steering commands computed by the associated controller, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
- 2.2 Steering commands computed by Controller 1 and 2 are substantially different. Design requirements:
  - 2.2.1 Supervisory Controller should sense the discrepancy and determine the controller that fails. The System Failure Response should then be executed.
- 2.3 Supervisory Controller fails. Design requirements:
  - 2.3.1 Supervisory Controller should incorporate circuitry, such as a watchdog timer, to detect failures and to ensure the processor functions in a timely manner. If a failure is detected, the circuitry should deactivate

Supervisory Controller, send a signal to warn the driver and to ask him/her to take over steering control.

- 2.4 Supervisory Controller computes incorrect steering command. Design requirements:
  - 2.4.1 Supervisory Controller should compare the computed steering commands with those computed by Controller 1 and 2 to determine if it is working properly. If an error is detected, the driver should be warned of the failure and be asked to take over steering control.
- 2.5 Steering command is not communicated to Actuator 1 or 2 at appropriate time or is corrupted in transmission through interface with Controller 1 or 2. Design requirements:
  - 2.5.1 Each of Actuator 1 and 2 should verify that it is receiving updated steering command from the controller in a timely manner. If an error is detected, the System Failure Response should be executed.
- 2.6 Supervisory Controller's command data is not communicated to controllers and actuators at appropriate time, or is corrupted in transmission through interface. Design requirements:
  - 2.6.1 Each of the two controllers and two actuators should verify that it is receiving updated command from Supervisory Controller in a timely manner. If an error is detected, the driver should be warned of the failure and be asked to take over steering control.

Potential Failure Mode 3: Steering Actuator failure

- 3.1 Actuator 1 or 2 does not generate correct steering force for a given steering command.Design requirements:
  - 3.1.1 Each actuator should incorporate built in tests to be performed on the vehicle startup and during lane keeping operation to verify that steering force can be produced and is within the allowable range. If an error is detected, the System Failure Response should be executed.
  - 3.1.2 The ALK system should provide for independent, redundant means to constrain the steering force generated from each steering actuator with safe ranges.
- 3.2 Actuator 1 or 2 fails to be deactivated when the associated control channel is determined to be deactivated. Design requirements:
  - 3.2.1 The ALK system should provide for two independent, redundant means of deactivating each steering actuator.

Potential Failure Mode 4: Incorrect lane reference data received from the roadway lane reference aid(s)

- 4.1 Some lane reference aids does not provide lane reference information to the onboard lateral dynamics sensor. Design requirements:
  - 4.1.1 Each lateral dynamics sensor should incorporate reliable means to detect potential failures in the associated lane reference aid. If an error is detected, the System Failure Response should be executed.
- 4.2 Lateral Dynamics Sensors fail. Design requirements:
  - 4.2.1 Same as for the design requirements in 1.1.1, 1.1.2, 1.2.1, 1.3.1, 1.3.2, 1.3.3, 1.4.1, and 1.5.1.

### Key Findings

1. The driver can not be considered as a back-up to the ALK system. A fail-safe ALK system design will require considerable structure complexity and component redundancies.

2. The proposed reliability functional requirement for the ALK system is: "A single component or point failure is not allowed to cause the vehicle to depart from the lane, and there should be no common failure modes."

3. The redundant control channels in the proposed ALK system design framework need to be always kept active to ensure safe and smooth deactivation of a control channel.

### Issues and Risks

1. Some lateral dynamics sensing technologies, such as vision based, may be sensitive to the environmental conditions. Due to the strict reliability requirement, the automatic lane keeping operation may be easily interrupted by poor weather. The reliability degradation in automatic longitudinal control may also affect the availability of the ALK function.

2. The considerable amount of components redundancies and roadway lane reference aids may substantially increase the cost of deploying the ALK function.

3. Since the driver may easily become overly relaxed in hands-off and feet-off driving, he/she may have difficulties in overriding the ALK system to perform lateral collision avoidance.

4. The operation of lane keeping requires the assistance of lane reference aids to provide lateral dynamics information. The reliability of the roadway lane reference aids is crucial for the availability of the vehicle ALK function.
# Lateral Collision Warning

The lateral collision warning (LCW) system improves the safety of lateral control by warning the driver of potential lateral collisions. It helps the driver detect collision danger coming from both sides of the vehicle. The LCW system can provide warnings when the vehicle is changing lanes or merging, and when the vehicle is performing lane keeping in the dedicated lane.

The LCW system measures the distance, direction and lateral closing rate for vehicles in adjacent lanes. It uses these data to evaluate the lateral time to collision (TTC) and determine the direction of the collision threat. If the computed lateral TTC is smaller than the threshold setting, which takes into account the driver's steering reaction time, the LCW system sends warning signal to the driver.



Figure 78. The LCW system interfacing with the driver.

The driver reacts to the warning by overriding the automatic lane keeping system to execute lateral collision avoidance maneuver. Figure 78 shows the interaction between the LCW system and the driver.

# Reliability Functional Requirement

The LCW system is designed to improve the safety in the driver's steering control. It is not to replace the driver's lateral collision avoidance function. The effectiveness of the LCW system depends on the detection rates, the false/nuisance alarm rates, types of warning signals, and the driver's reaction to different warning modalities, etc.

In ERSC 3, the driver is interfacing with the ALK system and the SHM/RECA system. The driver is feet-off and hands-off in the dedicated lane. The driver can be over-relaxed and can not sense the lateral collision danger in time. A non-detection may lead to a lateral collision. The LCW system needs to have high detection rates.

Frequent false alarms may be disturbing to the driver, and may delay his/her reaction to the real collision danger. The driver may even reject the false alarms by turning off the LCW system. A sudden false collision warning may make the driver panic and even endanger the vehicle lateral control. Frequent nuisance alarms may irritate the driver. The nuisance alarm

rates can be reduced by allowing the driver to adjust the warning threshold within a safe range to accommodate his/her driving skill.

The above discussion suggests that the reliability functional requirement for the LCW system be:

"The LCW system should have high detection rates and low false alarm rates."

#### System Level Design Requirements

The LCW system is composed of lateral sensors, a processor and a warning interface as shown in Figure 79. The lateral sensors detect vehicles in all lateral directions and from both adjacent lanes, and measure the necessary dynamic data required by the processor for computing TCC. The processor compares the computed lateral TTC with the warning threshold and determines the severity of the danger. The warning interface takes warning commands from the processor and generates warning signals to the driver. The warning signals include the information about the direction of the collision threat.

It has been recommended that at least two levels of warnings be used in the LCW system. <sup>(64)</sup>, an imminent collision warning and a cautionary collision warning. The imminent collision warning informs the driver that it is necessary to take evasive action to avoid a lateral collision. If the imminent collision warning can not be sensed by the driver promptly, the driver may not be able to react to the danger in time. The LCW system is therefore required to provide redundant imminent collision warning signals. Since the driver may become less attentive to a specific type of signals, it is desirable that the redundant imminent collision warnings be different in modality, e.g., visual and audio.

To achieve the proposed requirement, the LCW system needs to have self-diagnosis element to detect potential internal failures. If a failure is detected, the driver should be notified and the LCW system may have to be deactivated. More specific design requirements at system level can be identified by analyzing major potential failure modes and causes based on the system framework shown in figure 79.





- 1. Lateral Sensor failure
  - 1.1 Lateral Sensors compute incorrect lateral TTC or direction of the threat. Design requirements:
    - 1.1.1 Lateral Sensors need to have an internal independent check of the reasonableness of the computed data such as lateral distance, closing rate and direction of danger, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed. The System Failure Response warns the drive of the potential failure and asks him/her to turn off the LCW system.
    - 1.1.2 Processor should incorporate an independent check of the reasonableness of the computed data from Lateral Sensors, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.
  - 1.2 Data from Lateral Sensors is not communicated to Processor, or is corrupted in transmission through interface. Design requirements:
    - 1.2.1 Processor should verify that Lateral Sensors are operating in a timely manner by verifying that new data is received from Lateral Sensors every tbd (to be determined) time interval. If an error is detected, the System Failure Response should be executed.
    - 1.2.2 Same as for 1.1.2.
- 2. Processor failure
  - 2.1 Processor command is not communicated to Warning Interface at appropriate time or is corrupted in transmission through interface with Processor. Design requirements:

2.1.1 Warning Interface should verify that it is receiving updated warning commands from Processor in a timely manner. If an error is detected, the System Failure Response should then be executed.

## 3. Warning Interface failure

- 3.1 Warning Interface fails to warn or generate incorrect level/modality of warnings after receiving a warning command from Processor. Design requirements:
  - 3.1.1 Warning Interface should incorporate built in tests to be performed on the vehicle startup and during headway maintenance operation to verify that proper warning signals can be generated. If an error is detected, the System Failure response should be executed.

#### Key findings

1. Two levels of warning may be required. Imminent collision warning may require redundancy with different warning modalities.

2. The reliability functional requirements for the LCW system are high detection rates and low false alarm rates.

#### Issues and Risks

1. The driver may become over-reliant on the LCW system and degrade his/her reliability/performance in lateral collision avoidance.

2. The driver may experience warning overload and warning confusion problems while interfacing with other vehicle and roadway functions.

# **Roadway to Vehicle Speed and Headway Commands**

The reliability analysis is mainly the same as in ERSC 1 with additional reliability requirements discussed below.

The roadway may send higher speed commands and/or shorter headway recommendations to the vehicle since the driver does not have to steer the vehicle in the dedicated lane in ERSC 3. A lane reference aid failure may cause the disabling of the on-board ALK system. Without adjusting the speed/headway, the driver may have difficulties taking over lane keeping control. The roadway therefore has to re-evaluate the safe speed commands and minimum headway recommendations for the sections of the dedicated lane that are with failed lane reference aid(s) Figure 80 shows the functional block diagram of the roadway to vehicle speed/headway command function that takes into account the operational status of the lane reference aids.



Figure 80. The functional block diagram of the roadway to vehicle speed/headway command function.

# **Roadway Maneuver Coordination**

The roadway assists the vehicles safely getting into and out of the dedicated lane by maneuver coordination near the entry/exit points.

At an entry point, the roadway detects the vehicles that intend to enter the dedicated lane. If a vehicle ready to enter is detected, the roadway sends commands to the vehicles near the entry point in the dedicated lane to slow down and to increase the headway. This ease the merging or lane changing maneuver for the entering vehicle due to larger gaps and slower traffic speeds. The roadway also detects the exit intention of vehicles. To improve the safety in exit maneuver, the roadway can broadcast slower speed and larger headway commands to the vehicles in the highway sections near the exit point. Figure 81 shows the functional block diagram of the roadway coordination



Figure 81. The functional block diagram of the roadway maneuver coordination.

For roadway with continuous entry/exit points, the traffic speed and density in the manual (or transition) lane adjacent to the dedicated lane are also important for entry and exit coordination. The roadway has to use the traffic information in the adjacent manual/transition lane to determine the desired speed and headway for the dedicated lane near entry/exit points.

In ERSC 3, the AHS may take advantages of the rear-end collision avoidance and lane keeping, and adopt an operation strategy with higher speed and smaller headway. It will be dangerous if the driver attempts to enter or exit the dedicated lane without a safe gap at lower speed being created. The roadway traffic controller thus has to have reliable means to detect the entry/exit requests from vehicles. Maneuver coordination commands must be sent through reliable communication protocols to the vehicles in the dedicated lane. The roadway can use the traffic speed/density sensors to make sure that maneuver coordination commands are executed. The roadway may permit the vehicle's entry/exit maneuver only if a safe enough merging or lane changing condition can be created.

#### Key Findings

1. The reliability of the roadway maneuver coordination depends heavily on the reliability of the roadway-to-vehicle communication.

#### Issues and Risks

1. Since the roadway involves in the dedicated lane entry and exit coordination, the roadway may be liable for accidents if it fails to send out proper speed/headway commands to coordinate the entry/exit maneuvers. The driver may still blame the roadway for an entry/exit accident even it is caused by the his/her improper merging/demerging maneuvers.

# **Roadway Lane Reference Aids**

In ERSC 3, the roadway provides multiple lane reference aids to support the vehicle's lane keeping function. The lane reference aids are designed to provide lane reference information to the on-board automatic lane keeping (ALK) system. Independent, redundant lateral dynamics sensors are required to implement a fail-safe ALK system. Depending on the sensing technologies used by the lateral dynamics sensors, various types of lane reference aids will be needed. Figure 82 shows the functional block diagram of the roadway lane reference aids.



Figure 82. The functional block diagram of the roadway lane reference aids.

Based on the ALK system framework discussed earlier, if a lane reference aid fails to provide correct lane reference data to the associated lateral dynamics sensor, it may degrade the lane keeping performance or lead to the disabling of the ALK system. The lane reference aids thus have to be reliable enough to ensure efficient and safe lane keeping operation.

For magnetic-marker-based sensing, the reliability of the lane reference aid can be improved by installing markers with higher density along the dedicated lane and frequently checking the intensity of magnetic field generated by each magnetic marker. For vision-based sensing, the roadway can improve the reliability of the lane reference aid by providing lane markers with high contrast under various weather conditions. Redundant road lane reflectors may be used to improve the reliability of the radar-based lane reference aid.

The roadway needs to have reliable self-diagnostic procedure to detect any lane reference aid failure along any section of the highway. If a failure is detected, the roadway should warn the vehicle/driver of the danger through the roadway-to-vehicle communication. In this case, the AHS operation falls back to ERSC 2. It is also desirable that the data provided by the lane reference aids be designed in such a way that the lateral dynamics sensors of the ALK system can detect possible failures in the lane reference aids.

# Key Findings

1. The roadway lane reference aids are to support the sensors of the ALK system. Since the proposed ALK system design will have redundant and independent lateral dynamics sensors, the roadway lane reference aids will have to be redundant and independent.

#### Issues and Risks

1. The roadway may be liable for the accidents caused by the failure of the lane reference aids.

2. The required redundancy in the lane reference aids may substantially increase the cost of the roadway.

# **Driver Tasks and Workload**

This section discusses the drivers' tasks in this ERSC. It covers normal driving in the dedicated lane when all of the automated functions work perfectly. Then it covers the fall-back mode in which the vehicle switches into ERSC 2 with speed and headway maintenance and rear-end collision avoidance if the automatic lane keeping system fails or degrades. Finally we discuss the lateral collision warning and the maneuver coordination function and their interactions with the drivers.

The driver drives the car to the automated lane. The driver turns on the vehicle's turn signal and lateral collision warning system. The driver enters the lane when a safe gap appears in the traffic flow. The driver steers the car into the lane and turns on the SHM function, rearend collision avoidance system and the automatic lane keeping function. The vehicle takes over the throttle, steering, and brake when the car is in the lane or as the vehicle enters the lane. The driver supervises the vehicle control of the steering, brake, and throttle. The driver workload in ERSC 3 is very light once the vehicle enters the dedicated lane.

The driver turns on the turn signal to leave the automated lane. This initiates check-out procedures. Check-out returns the steering, brake, and throttle control to the driver. The driver then changes lanes out of the automated lane. The driver turns off the communications links when he leaves the automated lane.

# Task Analysis-Normal Driving

In normal operations, the vehicle controls the steering, throttle, and brake. The driver performs lane changes into and out of the dedicated lane. The driver also avoids lateral collisions. The roadway sends target speed commands and minimum headway information electronically to the vehicle. This data helps the vehicle compute a safe headway. The roadway also sends traffic data to the driver to help him/her select a route. The task analysis in table 27 shows the driver cues for this task.

Task	Monitor Vehicle Operation		
General	Task similar to being a passenger in the vehicle		
Initiating Cue	Enter Dedicated Lane and turn on speed and headway maintenance, the rear-end		
	collision avoidance, and the automatic lane-keeping functions-The driver sets		
	time headway and initial speed		
Feedback	Visual, auditory, and kinesthetic cues of accelerating, braking, slowing, and		
	steering		
Task Standards	Perform lane change into dedicated lane within 2-6 seconds, transfer control to		
	vehicle		
Task Conditions	Weather, road conditions, or traffic		
Skills Required	Reaction times appropriate to speed and roadway conditions.		
	Perceptional judgment of vehicle kinematics for lateral collisions		
Potential Errors	Turn off or interfere with automatic lane-keeping system accidentally		
	Turn off speed and headway maintenance system accidentally		
	Vehicle fails to maintain speed or proper headway		
	Vehicle selects unsafe headway		
	Vehicle leaves roadway		
	Roadway sends incomplete or wrong speed command to vehicle		
Causes of Error	Fails to perceive or misinterprets system feedback, believes system has		
	malfunctioned		
	Poor judgment of speed and road curvature		
	Hit wrong button and overrides function		
	Sensor, controller, or actuator failure on vehicle		
	Roadway controller malfunctions		
Consequences of Error	Disrupt traffic flow and lower capacity		
	Collision with surrounding vehicles		
	Rear-end collision by vehicle in back		
	Lane departure collision		
<b>Recovery Points</b>	Minimum safe headway setting on vehicle to avoid collisions		
	Collision warnings		
	Vehicle warns driver that headway maintenance function is not working or is		
	failing or has been turned off; prevent easy override of SHM, ALK, and RECA		
	functions		
T 11 1 1 D 00	Lateral collision or lane departure warning		
Individual Differences	Education		
	Age		
	Essential		

#### Table 27 Driver task description for ERSC 3 driving

# Task Analysis-Fall-back Mode

In the fall-back mode, the automatic lane-keeping system suffers a partial failure when one control channel fails, but the other one remains intact. If the steering assist or lane departure warning functions can still operate then the vehicle switches to ERSC 2 operations in which the driver is responsible for steering as described in Section 6. The vehicle warns the driver that he is now responsible for lane-keeping and slows the vehicle to a safe speed for manual steering. The vehicle should then exit the dedicated lane if it cannot safely maintain the speeds commanded by the roadway. If the warnings fail, the vehicle notifies the driver and the driver performs the task without the aid of warnings. Table 28 shows the task analysis for the lane keeping function.

Task	Fall-back mode to ERSC 2			
General	Task may be similar to manual driving with cruise control			
Initiating Cue	Vehicle warns driver to resume responsibility for lane-keeping			
	Driver receives warning signal for lane departure from vehicle			
	Vehicle may decrease speed for to allow for manual steering			
Feedback	Visual, auditory, and kinesthetic cues for steering			
Task Standards	Duration the same as manual driving, transition to manual driving must be			
	gradual, vehicle keeps deceleration low while decreasing speed			
Task Conditions	Weather, road conditions, suggestions from traffic control center			
Skills Required	Reaction times appropriate to speed and roadway conditions.			
	Perceptual judgment of vehicle kinematics			
Potential Errors	Driver inattention causes vehicle to leave lane			
	Driver overrides lane-keeping system accidentally			
	Driver does not respond in time to lane departure collision warning			
	Vehicle fails to maintain speed or proper headway			
	Vehicle selects unsafe speed			
	Vehicle does not warn driver adequately of a control transfer			
Causes of Error	Fails to perceive or misinterprets system feedback			
	Poor judgment of speed and following distance			
	Hits wrong button			
	Sensor or actuator failure on vehicle			
	Role confusion by driver-responsibilities are not clear			
<b>Consequences of Error</b>	Disrupt traffic flow and lower capacity			
	Lane departure or lane change/merge collision			
	Driver confusion			
<b>Recovery Points</b>	Maximum speed limit setting on vehicle			
	Lane departure collision warning			
	Vehicle warns driver that lane-keeping function is not working or is failing or has			
	been turned off			
	Vehicle requires active response from driver before a control transfer			
	Driver interface shows driver control status clearly			
Individual Differences	Drivers may be uncomfortable taking over steering control at high speeds or in			
	reduced visibility <sup>(90)</sup>			
Criticality	Essential			

#### Table 28 Driver resumes control of lane-keeping function.

# Task Analysis-Lateral Collision Warning

The lateral collision warning watches for potential lateral collisions. If the sensor detects a lateral collision then it warns the driver. This system aids the driver during lane changes, merges, and normal driving. Table 29 shows driver-lateral collision warning system operation. The lateral collision warning system should ideally warn the driver fast enough for the driver to plan and execute evasive steering actions. This warning system may just give a warning or it may also suggest course changes to avoid the accident.

Drivers avoid collisions with a combination of brakes and steering. <sup>(82)</sup> The less time the driver has to respond to a problem the more likely he is to use only the brakes. This affects the design of the speed and headway maintenance system and the driver override. If the driver is responsible for lateral collision avoidance, then he or she must be allowed to use all the tools available (e.g. brakes and steering) to avoid an accident.

# Task Analysis-Maneuver Coordination

The maneuver coordination system helps vehicle merge into the dedicated lane by commanding larger gaps between vehicles approaching a designated entry/exit point. The driver can then merge into the dedicated lane safely. This system may require the roadway or vehicle to send a signal to the driver to let him know that a gap large enough for entry into the dedicated lane is approaching. The vehicle may also coordinate suggested lateral collision avoidance maneuvers. This arrangement is similar to the TCAS system for commercial aircraft.<sup>(96)</sup> This system may require special training for drivers to ensure that they respond correctly. Table 30 shows the task analysis for the maneuver coordination function.

Task	Respond to Lateral Collision Warning		
General	Task similar to that in manual driving		
Initiating Cue	Warning from vehicle that a lateral collision may occur		
	Driver turns on turn signal to enter or leave dedicated lane		
	Vehicle detects change in steering angle		
Feedback	Visual, auditory, and kinesthetic cues of steering and braking		
Control Functions	Perception-response time (PRT) studies have found that drivers respond to		
	unexpected hazards by avoidance steering as well as braking		
Task Standards	Avoid the collision when worned by steering $(-1.4 \text{ second } (21))$		
	Look in direction of lateral collision		
Task Conditions	Type and timing of warning		
Task Conditions	Environmental conditions, traffic speed and density		
Skills Dogwirod	Provide times appropriate to speed land change speed and readway conditions		
Skiis Kequiteu	Percentual judgment of vehicle kinematics		
Potontial Errors	Overconfidence in system driver dees not look in lane changes		
1 otentiar Errors	Driver does not use turn signals		
	Driver does not respond to the warning or responds too late		
	Sensor false alarms (No potential collision system warns driver anyway)		
	Sensor does not detect notential collision		
	Vehicle recommends incorrect maneuver		
Causes of Error	Fails to perceive or misinterprets the indication of an emergency		
	Driver delay in response to warning		
	Lateral collision sensor warning threshold too high or low, malfunction		
	Warning controller misinterprets the data and the risk		
<b>Consequences of Error</b>	Lane change/Merge collision with potentially severe results for the system		
-	(Approximately 4% of crashes and 0.5% of fatalities are lane change/merge		
	crashes Some 95% of these crashes are angle/sideswipe crashes $(113)$		
	Disrupt traffic flow and lower canacity		
	Loss of confidence by the driver and public in the system		
	In the case of false alarms the driver may defeat or ignore it as pilots and		
	(96)		
<b>D</b>	controllers have done in TCAS system		
Recovery Points	Vehicle reminds driver to look before lane change		
	Multiple warnings from detector		
I 1' ' I I D'66	Venicle suggests a collision avoidance course to the driver		
Individual Differences	Accident statistics, citations and self report data indicate that older drivers have		
	unificative in merging, changing lanes, and exiting maneuvers. Data suggests that		
	persons over 65 may be over represented in lane change/merge crashes. <sup>(113)</sup>		
	Size of vehicle's blind spot		
Criticality	Essential		

# Table 29 Driver task description for lateral collision warning.

# Table 30 Driver task description for maneuver coordination.

Task	Respond to maneuver coordination signals from roadway or vehicle		
General	Task similar to that in manual driving for merges		
Initiating Cue	Cue from vehicle that a gap in traffic is approaching		
	Driver turns on turn signal to enter dedicated lane		
	Vehicle detects change in steering angle		
Feedback	Visual, auditory, and kinesthetic cues of steering and accelerating		
<b>Control Functions</b>	Drivers must be able to merge safely		

Task Standards	Safely merge into traffic		
	Accelerate as directed by vehicle or roadway		
Task Conditions	Type and timing of warning		
	Environmental conditions, traffic speed and density		
Skills Required	Reaction times appropriate to speed, lane change speed, and roadway conditions.		
	Perceptual judgment of vehicle kinematics		
Potential Errors	Overconfidence in system-driver does not look in lane changes		
	Driver does not use turn signals to signal merge		
	Driver does not respond to the cue or responds too late		
	Maneuver coordination does not open a gap		
	Maneuver coordination opens large gaps		
	Vehicle recommends incorrect maneuver		
Causes of Error	Fails to perceive or misinterprets the indication of an emergency		
	Driver delay in response may cause unsafe lane change		
	Warning controller misinterprets the data and the risk		
<b>Consequences of Error</b>	Lane change/Merge collision with potentially severe results for the system		
	(Approximately 4% of crashes and 0.5% of fatalities are lane change/merge		
	crashes. Some 95% of these crashes are angle/sideswipe crashes <sup>(113)</sup> )		
	Disrupt traffic flow and lower capacity		
	Loss of confidence by the driver and public in the system		
	In the case of false alarms the driver may defeat or ignore it $(96)$		
<b>Recovery Points</b>	Vehicle reminds driver to look before lane change		
	Multiple cues from vehicle and roadway		
	Vehicle suggests an acceleration profile to the driver for merging		
	Vehicle suggests an avoidance path to driver		
Individual Differences	Accident statistics, citations and self report data indicate that older drivers have		
	difficulty in merging, changing lanes, and exiting maneuvers. Data suggests that		
	persons over 65 may be over represented in lane change/merge crashes <sup>(113)</sup>		
Criticality	Essential		

#### Issues and Risks

# 1. Driver confidence in warning

If the driver has too much confidence in the collision warning systems then she may not pay enough attention during lane changes or while following another vehicle. This could lead to accidents if the warning system does not detect an object or the system fails. Driver lack of confidence in the warning could cause the driver to disable or ignore the warning system. Research needs to be done on how many false alarms and detection failures are acceptable.

#### 2. Enabling/Disabling Automatic Systems

This ERSC includes a speed and headway maintenance system, lane keeping, lateral collision warnings, and communication systems. Research needs to be done on the instrumentation needed to control these different systems. The question of whether or not all of the automatic systems need to be controlled with one switch or many needs to be answered.

#### 3. Multiple Warnings

The lateral collision warnings may occur simultaneously. The vehicle must prioritize the alarms so that the driver only has to work with one task at a time.

4. Driver override of braking

Even with multiple sensors the vehicle may not see and recognize everything that a human driver sees. This is especially true of unexpected objects such as animals or debris on the roadway. It may be necessary for the driver to override the rear-end collision avoidance system on the vehicle to avoid these objects. The driver is also responsible for lateral collision avoidance. The lack of a braking override may put the driver in an unsafe position. The existence of such an override may also put the driver in an unsafe position if the driver brakes unnecessarily. Research needs to be done on whether or not this driver override should be allowed to occur.

# **Key Results**

# Benefits

1. Automatic lane keeping will reduce the collisions caused by vehicle lane departures. It is helpful to the driver in particular for long trip driving.

2. Lateral collision warning improves the safety of the changing lanes and merging. It also helps the driver for lateral collision avoidance.

3. Dedicated lane capacity can be improved by adopting higher speed and/or smaller headway since the driver is hands-off and feet-off in normal driving.

4. Roadway maneuver coordination improves the safety for entry and exit. It may also help reduce the effect of traffic flow disturbance caused by vehicles entering and leaving the lane.

#### **Issues and Risks**

1. A potential ALK system design to fulfill the reliability functional requirement will need two redundant, independent steering control channels. This substantially increases the cost of automated vehicles in ERSC 3 due to the design complexity and required redundant components.

2. The operation of lane keeping control requires preview information. The on-board automatic lane keeping (ALK) system obtains it with the help from roadway lane reference aids. The sensing technologies used by the ALK system determine the types of lane reference aids. It might not be easy for some types of lane reference aids, such as vision based, to reliably provide preview information under various environmental conditions.

3. The sensors of the automatic lane keeping (ALK) system interface with roadway lane reference aids to obtains lateral dynamics information. The reliability functional requirement implies that redundant, independent control channels are needed for the ALK system design. The lateral dynamics sensors will be based on independent sensing technologies. Since a lane reference aid needs to be made compatible with a lateral dynamics sensors, there will be multiple lane reference aids. This may substantially increase the cost of the roadway infrastructure.

4. Lateral collision avoidance may require combined lateral and longitudinal control. In ERSC 3, the driver can override the automatic lane keeping system but not the longitudinal control. The effectiveness of the driver's lateral collision avoidance, using steering control only, is yet to be verified.

# Key Findings

1. The on-board-automatic lane keeping system will require redundant active control channels for fail-safe design.

2. It may be impractical to have driver perform lateral collision avoidance while giving him/her only steering override authority.

3. AHS with narrow lanes has the potential of increasing traffic capacity and real estate saving. However, the automatic lane keeping and lateral collision warning will be difficult to implement on a narrow dedicated lane.

4. A malfunction in the ALK system will result in the vehicle operation falling back from ERSC 3 to ERSC 2. However, a malfunction in the SHM and RECA system will cause the operation to fall back from ERSC 3 to ERSC 1 or 0.

Raytheon

Section 8: ERSC 4 Analysis

# **Description of ERSC 4**

In ERSC 4 the vehicle is fully automated. Automatic lane changing and lateral collision avoidance capabilities are added to the vehicles of ERSC 3. The roadway provides and maintains dedicated lanes and provides emergency vehicle response. It also provides lane reference aids for lane keeping and lane changing. The dedicated lanes could be segregated and accessed from dedicated ramps or they could be next to manual lanes and accessed through transition lanes depending on the particular entry/exit configuration. As in ERSC 2 and 3 the roadway receives vehicle status information, assists in the check-in and merging process and sends target speed, headway commands and traffic information to the vehicles.

The additional capabilities of automatic lane-changing and lateral collision avoidance allow the vehicle to drive automatically when the vehicle is in the dedicated lanes. Each vehicle coordinates its maneuvers, such as lane changes, merges, and speed changes, with the vehicles around it. The vehicles use cooperative communications and their own sensor measurements for this coordination. Each vehicle broadcasts its position, heading, and intent to the vehicles around it. These communications may be routed through the roadway so that the roadway can monitor the highway status and coordinate communication traffic and protocols. The vehicle plans, coordinates, and performs all collision avoidance maneuvers. The vehicle must have no single point or common mode fault that could cause the automated lateral and longitudinal control to fail. The vehicle uses its onboard diagnostics to check the fitness of the vehicle for operating in the automated mode and notifies the driver accordingly.

The vehicle plans its route according to the trip destination entered by the driver and the traffic information receives from the roadway. The vehicle may display the planned route and the estimated travel time to the driver. The vehicle alerts the driver and assists him/her in resuming manual control at the end of the trip. It has a contingency plan if the driver cannot resume manual control at the end of the trip or during certain emergencies.

The vehicle has a fall-back mode that allows the driver to briefly take over some of the automated functions in case of vehicle malfunctions. The purpose of this mode is to allow the vehicle to leave the dedicated lanes on the highway if a malfunction occurs. This mode must be designed so that no single point fault will cause the vehicle to be unable to transfer control to the driver.

The driver enters the destination and manually drives the vehicle to the entry point. If the vehicle passes a check-in test, then the driver switches on the automatic mode. The vehicle assumes control of all vehicle functions. Once the automated mode is on the driver becomes a passenger until he/she requests a switch back to a lower ERSC or to manual mode. The vehicle initiates a check-out procedure and the driver gradually regains full manual control in a transition lane or at the exit ramp depending on the exit configuration. When malfunctions occur and the fall-back mode of the vehicle becomes active the driver is responsible for operating the vehicle as in lower ERSCs.

The motivation behind ERSC 4 is to analyze the benefits in terms of driver comfort, safety and capacity that could be achieved by full vehicle automation. ERSC 4 allows us to focus on the analysis of the vehicle fully automated functions as they interact with each other to perform the

various driving and emergency tasks without relying too much on the roadway infrastructure. Table 31 shows the functions for ERSC 4.

Function	Requirement	Driver/Vehicle/Roadway Tasks
Automatic Lane Changing and Keeping	<ul> <li>Smoothly and safely change lanes</li> <li>Keeps vehicle in the center of the lane around curves and on straight-a-ways</li> </ul>	<ul> <li>Senses the position and course of the vehicle in the lane and generates the appropriate commands for the steering actuator</li> <li>Sense or receive preview information about roadway geometry and number of lanes</li> <li>Keep vehicle in the center of the lane</li> <li>High reliability required since the driver cannot serve as a back-up in case of failure</li> <li>Transfer of lane keeping/changing function to driver should be done gradually in a smooth way</li> <li>Interact with speed and headway maintenance function</li> </ul>
Collision Avoidance	Avoid collisions due to moving or stationary obstacles around the vehicle	<ul> <li>Detect objects on both sides of vehicle</li> <li>High detection rate and a low rate of false or nuisance alarms</li> <li>Plan evasive actions to avoid collision</li> <li>Coordinate actions with surrounding vehicles</li> </ul>
Navigation/Route Selection	Select a route to reach the driver's destination	<ul> <li>Receive traffic data from roadway information systems</li> <li>Dynamically select lanes and route for trip</li> </ul>
Maneuver Coordination	Coordinate lane change, merge, speed change and collision avoidance maneuvers in the dedicated lanes	<ul> <li>Roadway sends commands for maneuver protocols</li> <li>Vehicles send their heading, position, and intention to the vehicles around them</li> <li>Vehicles send out or respond to emergency signals</li> </ul>
Driver Interface	<ul> <li>Show driver route and allow driver to change route and destination</li> <li>Show driver what functions are operable at any given time</li> </ul>	<ul> <li>Driver should be able to:</li> <li>adjust headway</li> <li>enable/disable automatic functions</li> <li>Interface should be simple to</li> <li>understand and use</li> <li>No adjustments should be required that</li> <li>take driver's attention away from driving</li> </ul>

Table 31.	ERSC 4 functions	and driver/vehicle/roadway	performance requirements.
-----------	------------------	----------------------------	---------------------------

Fall-back Mode to ERSC 2 or 3	Driver should be warned of system failures or degradation in time to recover from dangerous situations	<ul> <li>Sense status of different vehicle control channels</li> <li>Safely transfer control of steering to the driver if lane keeping function fails or degrades</li> </ul>
Speed and Headway Maintenance	<ul> <li>Maintain the selected cruising speed when no vehicle is ahead.</li> <li>Calculate and maintain a headway for collision-free vehicle following.</li> </ul>	Same as in ERSC 2. Coordinate with lateral control functions for safe lane changes and collision avoidance
Roadway/Vehicle speed & headway commands	Senses vehicle average speed and density and environmental conditions then sends commands directly to vehicle to smooth traffic flow; Performs check-in and vehicle status monitoring	<ul> <li>Roadway commands different speeds for each dedicated lane.</li> </ul>

# **Performance Requirements**

# **Automatic Lane Changing**

The automatic lane changing function moves the vehicle from the center of one lane to the center of an adjacent lane on normal highway geometries if it is safe. The vehicle must first check if the lane change is safe and signal its intention to change lanes. When there is an appropriate gap between vehicles in the adjacent lane then the vehicle selects a trajectory and performs the lane change. The vehicle should accurately follow a desired path and provide satisfactory ride comfort over a wide range of speeds, roadway conditions, and disturbing forces. The automatic lane changing function controls the vehicle's steering smoothly and without oscillations. The automatic lane changing function should also prevent the vehicle from rolling over or going out of control when the rear-end collision avoidance system needs to brake on curved roads. The lane changing function works with the vehicle's throttle and brake controller to keep safe gaps between vehicles.

The driver checks-in and then enters the automated lane. The point at which the driver releases manual control could be on the dedicated ramp or in the dedicated lane depending on the entry configuration. Once the vehicle has entered the automated highway the driver starts the automated systems for steering and speed and headway maintenance. The vehicle smoothly takes over control from the driver. The lane changing function allows the vehicle to change lanes. The vehicle merges itself into the dedicated lane with the aid of maneuver coordination between the vehicles and roadway.

The automatic lane changing system requires sensors, a computer with control algorithms, and steering, brake, and throttle actuators. The navigation system chooses when the vehicle should change lanes. The communications system allows the vehicle to coordinate their actions and avoid accidents. The sensors measure the vehicle's dynamics and the forward roadway geometry. Lane reference aids. <sup>(30)</sup> help the vehicle sensors measure roadway geometry information. The computer generates control signals for vehicle steering as shown in figure 83.



Figure 83 Block diagram for the automatic lane changing system. Sensors and communication systems give information about the surrounding vehicles and their intent. The vehicle navigation system tells the vehicle when it should change lanes to following the vehicle's course. The roadway gives the vehicle information on the number of lanes available for traffic and their status.

#### Sensors

The vehicle must sense vehicles in the lane that they wish to change into. The vehicle must also sense roadway geometry such as curvature and the number of lanes at their current position and during the duration of the lane change. Figure 84 shows the sensor range for the automatic lane changing system.



Figure 84. Sensor range for the automatic lane changing system for a lane change to the left side. The vehicle must detect all vehicles that could affect its lane change.

The vehicle sensors must detect the vehicle's position in the lane and the curvature of the roadway. The sensors must also measure or estimated the vehicle's yaw rate and slip angle for smooth response. Section 6 on vehicle control discusses sensors for detecting the vehicle's position in a lane.

# Task D

Drivers use preview information to track the roadway. The driver extracts the information to align the vehicle and to anticipate future actions.. Several studies cited in McLean and Hoffman show that human drivers need a preview time of around 3 seconds at most highway speeds. <sup>(69)</sup> This number corresponds to the "effective" preview time. Distances beyond this limit do not improve system performance. The effective preview time depends on the control bandwidth and reaction time <sup>(62)</sup>:

$$T_p \approx -0.5 + \frac{4}{BW} \approx \frac{3}{BW}$$

 $T_p$  is the preview time. BW is the bandwidth of the controller. Automatic controllers with steering assist will need less preview information if they have larger controller bandwidths than a human driver. The vehicle must have preview data for the entire time that it takes to change lanes. This could mean that the vehicle needs 6-8 seconds of preview data.

Preview information is important for safe lane changing. <sup>(84,4)</sup> Automatic steering with preview information feedback can tolerate the actuators with lower bandwidth than the one designed without preview. <sup>(62)</sup> However, more preview means more complex data processing and delays for the sensors.

The accuracy of the lane position and preview sensors depends on the lane and vehicle width. Since the goal is to keep the vehicle within a few inches of the center of the lane at all times the sensors must at least detect the vehicle's position within a few inches. There are several combinations of sensors that can meet these requirements: magnetic markers, vision-based sensors, radar systems, navigation data from the roadway, and global positioning data.

# 1. Magnetic Markers

Magnetic-markers sensing requires a lane reference aid with high degree of roadway instrumentation. The road geometry information can be stored in an on-board database or in the markers as binary codes. When the vehicle is traveling in the lane, the on-board magnetic reference sensor measures the magnetic field to determine the lateral deviation and direction. The sensor can also read the lane geometry information from the markers or the on-board database. The sensor described above requires the vehicle to loose track of the sensor for part of the lane change. This lack of exact position information may make it difficult to perform collision avoidance or emergency maneuvers during a lane change. Magnetic markers may restrict the vehicle to only changing lanes when the magnetic markers lay out a path for the vehicle or an extra row of magnetic markers may be required between lanes.

# 2. Vision-Based

Vision-based sensing uses cameras to detect a vehicle's position on the road. <sup>(86,4)</sup> Vision-based sensing needs more instrumentation on the vehicle. The vehicle will be equipped with a forward looking camera to sense the vehicle's lateral deviation and roadway geometry. The image signals are sent to a on-board real-time image processing unit that processes the image data. The processor extracts roadway parameters by finding the edges of the roadway and fitting these edges to a roadway mode. The camera needs to cover wide angles and large distances to provide the required preview information. Vision-based systems can track the vehicle's position through the entire lane change. The vehicle is never "blind" at any part of the lane change.

Vision based sensing requires few infrastructure changes. However, it requires more instrumentation on the vehicle. The lane reference aid can be as simple as standard lane markers (stripes) on the roadway to indicate the lane. The lane markers on the lane surface needs to give clear contrast for the

video camera. The lane lines can be easily furnished and maintained. However, vision systems might not be effective in some weather conditions or at night.

# 3. Radar-Based

Radar-based sensing tracks the vehicle's lateral position with respect to reflector(s) mounted along the road lane. Radar-based sensing only works for vehicles in the side lanes of a freeway if side wall reflectors are used. The reflectors can also be corner cubes (103) installed along the road lane. The vehicle has transceiver(s) to emit and receive laser beam signals. When the vehicle passes through the corner cubes, the vehicle recognize the corner cubes by detecting the reflected laser beam. The vehicle then measures the moved distances and estimates the lateral position and heading relative to

the corner cubes based on the triangulation principle. The estimated information is then used as the initial values in the iterated calculation for final position and heading determination. Experimental

results <sup>(103)</sup> show that tracking errors within millimeters can be achieved at speed 20 km/h.

The main disadvantage in the radar sensing is that it provides little preview information. Without adequate preview information, the vehicle may not improve its tracking performance especially when it travels at high speeds.

# 4. Navigation Data from Roadway

The roadway could also send data to the vehicles about roadway curvature. <sup>(33)</sup> This approach uses beacons at fixed spots to tell the vehicles about the roadway ahead. The vehicle then uses

information on the location of the beacon and dead-reckoning for preview information. <sup>(55)</sup> Dead-reckoning systems use wheel odometers accelerometers to measure distance and gyroscopes to measure the vehicle's heading. This approach has already been tested in the Prometheus project. <sup>(10)</sup>

# 5. Global Positioning System

The global positioning system (GPS) provides absolute position data free from error

accumulation.<sup>(55)</sup> However GPS accuracy is degraded by tall buildings, tunnels, and large trees that are common in large cities. The vehicle can then use on-board maps and dead-reckoning to calculate the preview data.<sup>(55)</sup> The absolute position accuracy of GPS can then provide feedback and calibration signals to correct dead-reckoning errors. These systems can be integrated with algorithms that switch between systems depending on which has better conditions for operation or a filter can combine all sensor information to get the best estimate .<sup>(55)</sup> This method suffers from time delays which may make it impractical for real-time control applications.

# **Controllers**

The control algorithms need to assess the safety of a lane change and move the vehicle from the center of one lane to the center of an adjacent lane. The lane change controller needs to control the steering actuators and send inputs to the speed and headway maintenance system. The controller accuracy depends on the lane width, the vehicle length and width, and the roadway design parameters such as curvature. The controller must not oscillate in position. The controller should also restrict the amount of lateral acceleration to less than 0.08 g and the jerk to less than 0.17 g/s for driver comfort.<sup>(15)</sup>

A more difficult task is to assess the safety of a lane change. Most systems use an intelligent system approach to decide when to change lanes.  $(^{81,79})$  Fuzzy logic systems use commonsense rules in an IF-THEN form to determine when and how a vehicle should change lanes. Fuzzy systems can

generalize between similar situations. Expert systems have also been applied in simulation to this problem.

#### Actuators

This system will use hydraulic or electronic steering components. Electronic steering may increase the energy efficiency, driver comfort and stability of the steering system. <sup>(94)</sup> The actuator should have a small delay and respond accurately to the commands from the driver. The steering actuator needs to be reliable and fail-safe since the driver will not be able to regain control if the steering fails suddenly. <sup>(24)</sup>

#### Lane Change Coordination

Lane changes may be performed more efficiently if the vehicle coordinates its maneuvers with the vehicles around it. Studies done by the California PATH program investigate using a communication link to coordinate a lane change. Before making a lane change the vehicle sends out a signal of its intent to change lanes. This signal serves as an electronic turn signal. Any vehicles that are affected by this maneuver must acknowledge the lane change and grant permission. If acknowledgments are not received then the intent signal is repeated. The analysis in Streisand and Walrand only assumed that one vehicle needs to acknowledge the maneuver and in effect promise to maintain its current course for a given period of time. Realistically the vehicle needs responses from all the vehicles in its sensor range. However this could cause problems in coordinating the maneuver if there are problems of signal collision. In some vehicle following situations the vehicles may need to request cooperative actions from the other vehicles. This could include asking the vehicle to speed up or slow down so the maneuvering vehicle can change lanes. But asking other vehicles to change speeds affects traffic flow and possibly causes traffic jams.

Lane change coordination requires a sophisticated communications system that coordinates the timing and synchronization of messages. Figure 85 shows the message structure for a lane change.

Lane Change	Request to Change Lanes				Signal to Begin
Vehicle 1		Veh. 1			Lane Change
Vehicle 2		dgement	Veh 2		
Vehicle 3			Ack.	Veh 3 Ack	Ī
venicie 3				Ven 5 Aek	<b>&gt;</b>
					Time

Figure 85. Message timeline for lane change coordination messages. The lane changing vehicle (LCV) sends out a request to change lanes. The surrounding vehicles send back acknowledgments. The vehicle then begins the lane change.

Other possibilities are lane change protocols over a section of roadway. For instance one section of roadway may only allow lane changes to the left, while the next section could allow only right lane changes.

#### Issues and Risks

1 Lane Change Coordination

Coordinating sensors and communications will be a very difficult task. This will require very sophisticated algorithms to coordinate the lane changes.

2 Automated lateral and longitudinal control

Lane changes require a combination of lateral and longitudinal control. Some work has been done combining these modes. Under normal driving conditions, lateral and longitudinal control is decoupled. Studies need to be done for lane changing to verify that this holds true for most lane changing conditions.

# Lateral Collision Avoidance

The collision avoidance system combines lateral and longitudinal control to avoid collisions due to moving or stationary obstacles around the vehicle. This system must be designed so that no single point faults or common mode faults will permit the collision avoidance system to fail. The collision avoidance system must also send out an emergency signal when it must take evasive actions or a control system fails. Figure 86 shows the block diagram of the collision avoidance controller.



Figure 86. Block diagram for the collision avoidance controller. This controller uses lateral and longitudinal control to avoid collisions.

# Sensors

Collision avoidance requires multiple sensors that use different technology to avoid single point failures. These sensors must detect vehicles all around the primary vehicle as shown in figure 87. This system uses preview information from the roadway to predict the path of the other vehicles. The collision avoidance system sensors and communications system allows the vehicle to create a model of the highway and vehicle's behavior. This allows it to predict hazardous situations and avoid collisions.



Figure 87. The shaded area shows the sensor range for the primary vehicle.

Possible sensors include radar, ultrasound, vision-based and infrared systems. These sensor have been discussed in the section on lateral collision warnings. Communications can also help the vehicles to plan coordinated actions for collision avoidance.

# Controllers

The collision avoidance system sends commands to the brake, throttle and steering actuators. Drivers use all three of these systems to avoid accidents. <sup>(66)</sup> The controllers must keep the vehicle under control while making the evasive action. There are no limits on the vehicle's lateral and longitudinal acceleration due to driver comfort. The collision avoidance system for the aircraft control system uses a Time-to-collision (TTC) system to assess danger. When the TTC is less than a given threshold then the planes take evasive actions.

The collision avoidance system could have an expert system structure that creates a "mental" model of the vehicles and the roadway. The system then uses fuzzy rules or an artificial intelligence system to select the best evasive action. This system is used in many path planning robots. The problem with this design is that these models require a lot of computation and sensor data to create and they are cumbersome to work with. They are also not very flexible to sudden changes in situations and uncertain data.

An alternative approach is the "subsumption architecture" proposed by Rodney Brooks [Brooks, 1989]. This architecture uses local measurement data to make decisions. It does not explicitly create a detailed world model. This approach has created many simple robots that perform tasks such as giving tours and finding soda cans and are quite robust. However it is not known how well these techniques will scale up to tasks such as driving.

#### Actuators

The actuators for the brake, throttle and steering must be reliable, accurate and fast. These actuators will probably be a combination of hydraulic and electronic similar to today's airplanes. The speed of response depends on how quickly the vehicle detects a dangerous situation.

#### Emergency Collision Avoidance

If the vehicle needs to take emergency action due to failures in the automated system, then it sends out an emergency signal to the vehicles around it. The vehicle may sense or have a problem and need

to warn other vehicles. Emergency messages must be transmitted quickly and reliably. This signal gives the vehicle's position, heading, and intentions.

Emergency signals identify what vehicle sent the message. The destination address may be a selective broadcast address specifying the affected section of the highway. <sup>(100)</sup> Emergency messages have high priority depending on the type of emergency. The data field of the message contains essential emergency information such as the location, nature (such as flat tire, debris in roadway, or sudden loss of control), and suggested actions to be taken. Research at California PATH indicates that this message may be up to 1 kilobyte long. <sup>(100)</sup>

The emergency signal system must be extremely reliable so that no single point or common mode faults will cause the system to malfunction. The emergency system requires an emergency channel for emergency use only. This system may need to be tested regularly to be sure that each vehicle is working properly and can respond quickly to an emergency message.

#### Issues and Risks

1 Emergency Dynamics

Very little modeling work has been done to characterize vehicle dynamics under extreme conditions. This work needs to be extended.

2 Coordination

Collision avoidance is most effective when done early before an emergency exists. The collision avoidance system must coordinate actions in ways similar to the TCAS system to keep vehicles safe.

3 Path Planning

All of the work on path planning has used robots in fairly static environments. Research needs to be done on real-time systems in a dynamic environment.

# **Route Selection and Navigation**

The driver selects the destination at the check-in point or at the beginning of the trip. The vehicle then chooses a route based on traffic information, road conditions, and driver preference. Route selection will be a dynamic process so the route may changes as conditions change. The driver has the final say in the route selection. The vehicle decides what lanes to use for the trip and when to change lanes as it navigates on the highway.

#### Route Selection

The vehicle selects a route based on traffic conditions, distance and roadway status. The driver can then approve or disapprove the route. If the driver disapproves the route then the vehicle or driver selects an alternative route. The vehicle or driver can change the route at any time during the trip if traffic or personal conditions warrant it.

# Navigation

The roadway could also send data to the vehicles about roadway curvature. <sup>(33)</sup> This approach uses beacons at fixed spots to tell the vehicles about the roadway ahead. The vehicle then uses

# Task D

information on the location of the beacon and dead-reckoning for preview information. <sup>(55)</sup> Dead-reckoning systems use wheel odometers accelerometers to measure distance and gyroscopes to measure the vehicle's heading. This approach has already been tested in the Prometheus project. <sup>(10)</sup>

The global positioning system (GPS) provides absolute position data free from error accumulation.<sup>(55)</sup> However GPS accuracy is degraded by tall buildings, tunnels, and large trees that are common in large cities. The vehicle can then use on-board maps and dead-reckoning to calculate the preview data.<sup>(55)</sup> The absolute position accuracy of GPS can then provide feedback and calibration signals to correct dead-reckoning errors. These systems can be integrated with algorithms that switch between systems depending on which has better conditions for operation or a filter can combine all sensor information to get the best estimate.<sup>(55)</sup> This method suffers from time delays which may make it impractical for real-time control applications.

#### Issues and Risks

1 Information Sources

Traffic and road status data may be available from public or private sources. The driver may choose to obtain information from a private source that is more detailed than public data.

2 Lemming Effect

If all drivers get data from the same source then most drivers may end up taking the same route to avoid traffic tie-ups. This could create severe problems on the roadway. Studies need to be done on what type of information is available to drivers and vehicles and how they respond to it.

# **Maneuver** Coordination

The vehicles coordinate their maneuvers with a communications system to make the maneuvers safer and easier. Work at PATH suggests that there are three basic types of maneuvers needed for platoons on automated highways: lane change, platoon split, and platoon merge. <sup>(111)</sup> Since we are looking at a collision-free system without platooning these basic maneuvers can be restated as lane change, maintain gap, maintain a constant speed, and change speed up or down. These basic maneuvers will allow a vehicle to move on the automated lanes. Maneuver coordination will require communications between vehicles. Figure 88 shows the block diagram for the maneuver coordination system. This system uses sensors to figure out which vehicles need to be communicated with. The surrounding vehicles need to know the position, intentions and heading of the vehicle.



Figure 88 Block diagram for the maneuver coordination system.

#### Sensors

The sensors help the vehicle determine which vehicles must be informed of different maneuvers. The sensor range is the same as that shown in figure 87 for collision avoidance. Single-lane maneuvers such as speed up, brake or change headway only require information from the vehicles in the same lane.

# **Communications**

Table 32 summarizes the basic message types for maneuver coordination. The information needed column show the contents or data for the message. The type of communication column lists the requirements for the communications system. Possible requirements include whether the message must be acknowledged by other vehicles, the priority of the message, and whether the message is directed at any particular vehicle.

Maneuver Name	Information Needed	Type of communication
Lane Change	Direction and duration of lane change Acceleration Position Course and Heading	One to many Message directed to certain vehicles Acknowledgments needed from surrounding vehicles Medium priority
Collision Avoidance	Type and nature of emergency What Action is being taken Acceleration Position Course and heading	One to many Message sent to all vehicles in vicinity Acknowledgments needed from surrounding vehicles High priority message
Braking	Degree of braking Road-tire coefficient	One to one Message sent to vehicle behind only
Change speed	Acceleration Position Course and Heading	One to one Message sent to vehicle behind only
Change headway	Acceleration Position Course and Heading	One to one Message sent to vehicle behind only
Status	Acceleration Position Course and Heading	One to many Unacknowledged Low priority

Table 32.	Types of	communications	needed for	maneuver	coordination.
	J 1				

# **Controllers**

The controller must synchronize the messages for all of the vehicles.

#### Maneuver Protocols

The message frequency may be minimized by the use of protocols for maneuvers. These protocols control which vehicle has the right of way and which one must maneuver. These protocols give a fixed set of rules for lane changes and collision avoidance.

#### Issues and Risks

The needed information and the types of communication may have to be re-evaluated to ensure that maneuver coordination controller can always follow the protocols for safe maneuvers.

# Driver/Vehicle/Roadway Interface

The driver should be able to override the automated lateral and longitudinal control systems if a control channel fails. However the driver must go through a check-out procedure first to verify that he/she is ready to resume control of the vehicle. The driver also receives highway traffic information through the vehicle and may use it for route planning. The driver may also receive preview information from the roadway. This data may tell the driver about sharp curves up ahead or other roadway conditions. The driver also monitors the vehicle's operational status and backs up failed

vehicle functions if possible. If the automated lane keeping system fails the driver takes over manual steering of the vehicle in the fall-back mode.

The driver interface must show the driver that the automated functions are operating correctly or that a function has failed. This interface should be simple to understand and use. No adjustments should be required that take the driver's attention away from the driving task. Input systems like touch pads

or buttons may be distracting and may require too much driver attention.  $^{(8)}$  The driver must be able to tell what the vehicle is doing at all times and why it is doing it. This shows the driver that the vehicle is operating correctly. Studies show that drivers cooperate more when they know the reasons for an action.  $^{(10)}$ 

Head-up displays may be particularly useful for visual indicators of the system operation. For example a heads-up display could show the measured distance to the vehicle ahead as well as the minimum time headway and target speed. Other technologies include liquid crystal displays (LCDs), light-emitting diodes (LEDs), computer generated voice messages, sudden vehicle

accelerations/decelerations (jerk), tactile feedback, and directional audio displays. (64,8,9)

The interface may be continuous or only active when a driver action is needed. The best design of the driver interface for human factors is an open issue and involves selecting a display technology, type and duration of warnings, user control types and locations, etc. User acceptance is crucial to user interface design.

#### Issues and Risks

1 Role Confusion

The driver needs to know what he/she is responsible for at all times. It may not be a good idea to allow a vehicle to have different modes of operation.

2 Driver Training

The driver may have to be trained in order to effectively interface with the roadway and the vehicle under various operation conditions.

# Fall-back Mode

The system needs to degrade gracefully if a failure occurs. Many failures such as communications systems or isolated sensors may not cause the vehicle to be non-operational. Vehicles with different capabilities may coexist in the dedicated lanes without incident for short periods of time.

In case of failure in a single control channel of the automatic lane-changing or lateral collision avoidance system, the vehicle can operate in ERSC 3 control mode with the SHM, ALK, and rearend collision avoidance functions. The vehicle may decrease its speed to a safe value for ERSC 2 and the driver resumes control of the steering function with the help of the steering assist system and the lane departure warning. However, according to Riley decoupling warning and avoidance modes can cause "role confusion" for the driver. <sup>(90)</sup> The driver may also be uncomfortable dealing with a sudden change in responsibility especially during hands-off, feet-off driving. The driver may also be uncomfortable taking over steering control at higher speeds or if there is reduced visibility. <sup>(90)</sup> Safe system design may require that the driver can force the vehicle not to transfer control until the vehicle is ready to leave the dedicated lane.  $^{(90)}$  Meisner et al believe that such transfers between automated and manual functions need to be handled jointly by the driver and the vehicle.  $^{(70)}$  The vehicle should not transfer control to the driver in dangerous situations.. The driver should not be forced into an uncomfortable situation by the vehicle.

If the maneuver coordination functions fail the vehicle informs the driver. Failure of these functions may not require a transfer of control to the driver so they do not require the vehicle to return to ERSC 3 control.

If the longitudinal controller fails the vehicle should send out an emergency signal and exit the dedicated lanes as soon as possible. While multiple paths of degradation are technologically possible they are not desirable from a human factors point of view since it may require extensive driver training and practice to handle potential failures safely.

# **Roadway/Vehicle Speed and Headway Commands**

The roadway measures the average speed and traffic density, assesses environmental conditions, and calculates vehicle target speeds and minimum safe headways for the dedicated lanes. The vehicles follow the target speeds for their lane set by the roadway using the speed and headway maintenance function. The vehicles should respond to headway recommendations for longer headways due to changes in environmental conditions such as slick roads. The vehicles use the environmental data from the roadway to adjust their time headways. The vehicles check-in with the roadway by sending their identification, automated function status, and location. <sup>(108)</sup> The vehicles may also display relevant traffic information from the roadway to the driver for route planning purposes. This section covers roadway sensing systems and roadway-vehicle communications.

#### Roadway Sensing Systems

The roadway needs to sense the average traffic density over sections of the roadway in each lane. <sup>(58)</sup> This data can be measured by sensors that are part of the infrastructure or the vehicles can transmit their measured speed and headway to the roadway with a vehicle-roadway communication system for each lane. The vehicle may also transmit its operational status and identification to the roadway for check-in or incident management. The roadway also needs to measure roadway weather data to select a safe target speed for the environmental conditions . <sup>(57)</sup> Section 5 in the ERSC 1 analysis gives more details on these sensors.

#### Vehicle to Roadway Communications (VRC)

The roadway sends speed data to the cars in each lane to smooth traffic densities and manage incidents. The roadway needs to avoid large speed differences between lanes for safe transitions. The vehicles may also send their measured velocity and headway to the roadway. This system may also be used for check-in. This allows the roadway to monitor the operational status of the vehicles and ensure that they are correctly operating in the automatic mode. These messages only need to be sent once per section. There needs to be separate messages for each lane. The vehicles will also check-in with the roadway to ensure that they are working correctly. Each vehicle sends status data on all of its automated functions to the roadway. In ERSC 4 these functions are: combined lateral

and longitudinal control, collision avoidance, communication systems, and critical engine functions (108)

VRC systems can be mounted on the side of the roadway, on overpasses, or embedded in the roadway.<sup>(85)</sup> These systems may use automated toll collection technology with variable registers or a system that covers a wider range.<sup>(22,53)</sup> Appendix A covers the operation of these systems. A separate system may be required for each lane.

Target speed messages from the roadway to the vehicles do not need a high data rate since it will not change very often. Each field will need 6-8 bits to give a resolution of 1 meter for the gap limit and 1 mile per hour for the velocity limit. Since the message goes to all cars in a given lane a destination ID code must tell which lane the message is intended for. The source ID code indicates that the message is from the roadway. Error detection coding looks for bit errors in the message. Retransmission requests are unnecessary since we can rely on repeating the message. This message should be repeated at most one time per second. The bit error rate can also be high since the message changes slowly and rarely. Repetition of messages can compensate for the high error rate. If the roadway sends preview data to the vehicles, the messages may need to be longer in curvy regions of the roadway.

The VRC may also send messages from the vehicles to the roadway. Each vehicle sends its speed and the distance to the preceding vehicle (6-8 bits each). The vehicles may also send a unique identifying code (28 bits) to the roadway for check-in. The vehicle must also send its lane position to the roadway. The operational status of the automated system can be sent as well, this could take 8-16 bits depending on the amount of detail required by the roadway. Other variables such as estimates of the road-tire friction coefficient can take 4 or more bits. Table 33 shows the possible variables to be sent to the roadway for check-in. (108) This gives a data length of around 95 bits.

Function	Components	Number of Bits
Speed & Headway Maintenance	Sensors Actuators Controller	1-6
Speed & Headway Maintenance	Sensors, Actuators, Controller	1-0
Rear-end Collision Avoidance	Sensors, Actuators, Controller, Vehicle-Vehicle	1-8
	Communications System	
Lateral Collision Avoidance	Sensors, Actuators, Controller, Vehicle-Vehicle	1-2
	Communications System	
Vehicle-Roadway Communication	Transmitter, receiver, variable register	1-2
System		
Vehicle Identification Code		28
Other Vehicle Functions	Critical fluids level and pressures, brake conditions,	12-16
	tire pressure, engine temperature, headlights	

Table 33. Potential status variables to be sent to the roadway by the vehicles for check-in.

A broadcast system with a range of 100 meters for multiple automated lane must communicate with 0.3 to 0.75\*NL vehicles per second. NL is the number of dedicated lanes. Analysis by Polydoros et al shows that the current Hughes VRC system and automatic toll collection systems can meet the data transmission requirements for VRC in ERSC 4. <sup>(85)</sup>

#### Issues and Risks

1 Legal/Liability

In this ERSC the roadway sends recommended target speeds and minimum headways directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the minimum recommended headway is too short. The government may not wish to deploy such a system until it has been shown to be reliable and fail-safe.

2 Sensors

Roadway weather sensors only cover small regions of roadway. These sensors could easily miss patches of ice or puddles on the roadway. The roadway may then choose an unsafe headway. Research needs to continue into these sensors to improve their capabilities.

3 Privacy

The vehicles "Check-in" with the roadway. This may leave a record of when and where a vehicle traveled. This raises user privacy concerns about governmental use of this data.

4 Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

5 Coordination between lanes

The roadway must coordinate the traffic speed in adjacent lanes to smooth traffic flow. Speed differences of more than 10-20 mph will make lane changes and merges unsafe as well as slowing down traffic in the high speed lanes. This requires a more sophisticated roadway controller than in the previous ERSCs.

# **Reliability Requirement Analysis**

In ERSC 4, the vehicle is heavily instrumented and fully automated. The on-board automatic lateral and longitudinal control (ALLC) system can carry out vehicle functions such as speed and headway maintenance, lane keeping, lane changing, and collision avoidance. Automated vehicles in the multiple dedicated lanes perform maneuver coordination. The ALLC system uses vehicle-to-vehicle communication to help maneuver coordination with neighboring vehicles. The vehicle also has a navigation system to guide the vehicle to the destination specified by the driver.

The ALLC system uses the speed/headway commands from the roadway, maneuver commands from the navigation system, and neighboring vehicles' operating conditions from vehicle-to-vehicle communication to determine its lateral and longitudinal control actions. The vehicle takes over all driving tasks from the driver once the vehicle passes the check-in procedure. When the vehicle is in the dedicated lanes, the driver can not override the ALLC system. The ALLC system needs to be reliable in itself so that all lateral and longitudinal maneuvers can be safely performed.

The roadway sends speed and headway commands to vehicles to smooth traffic flow and to improve safety as described in earlier ERSCs. The roadway provides lane reference aids in multiple dedicated lanes to assist the ALLC system. The roadway may assist the vehicle-to-vehicle communication by relaying the communicated messages between vehicles. The roadway is required to be reliable so that safe and efficient operation of the ALLC system can be guaranteed.

# Automatic Lateral and Longitudinal Control

In ERSC 4, the vehicle is instrumented with the automatic lateral and longitudinal control (ALLC) system to perform cooperative driving in the multiple dedicated lanes. The ALLC system carries out the functions such as speed and headway maintenance, lane keeping, lane changing, and lateral/longitudinal collision avoidance. The ALLC system receives target speed commands and headway recommendations from the roadway, and maneuver commands from the on-board navigation system. It also receives neighboring vehicles' dynamics data and maneuver intentions from vehicle-to-vehicle communication. The vehicle coordinates its lateral and longitudinal maneuvers with other neighboring vehicles.

The ALLC system is mainly composed of the lateral/longitudinal dynamics sensor, the controller (computer with control algorithms), and the actuator. Figure 89 shows a functional block diagram of the ALLC system.



Figure 89. The functional block diagram of the automatic lateral and longitudinal control system.

The longitudinal dynamics sensor detects the relative distances and speeds for the neighboring vehicles in multiple lanes. The lateral dynamics sensor collects data such as lateral deviations and deviation rates of the vehicle and the neighboring vehicles. It also senses multiple lane preview information with the help from the roadway lane reference aids. The controller receives the lateral/longitudinal dynamics data provided by the sensors, and the neighboring vehicles' dynamics data and maneuver intentions from the vehicle-to-vehicle communication. The controller also receives the speed/headway commands from the roadway, and maneuver commands from the navigation system. The controller follows some "right of way" protocols to determine the lateral and longitudinal control actions to be taken. The controller uses the lateral/longitudinal dynamics data provided by the sensors of the determined control actions. The data from sensors allows the controller to generate timely throttle, brake and steering commands to the actuators. The actuators follow the controller commands and carry out control actions.

# Reliability Requirement Estimation

The ALLC system allows the driver complete feet-off and hands-off driving once the driver switches on the automated mode at an AHS entry point. Since the vehicles are in cooperative driving, the driver is not allowed to override the ALLC system. Consequently, the driver's reliability has no contribution at all to the lateral and longitudinal control operation. The driver's improper override actions may even endanger the automated driving.

The reliability of the vehicle control depends fully on the ALLC system reliability. The ALLC system needs to be so reliable that the traffic accident rates in rear-end collisions, lane departure collisions and lane changing collisions, can be significantly improved from today's manual driving.

#### Reliability Functional Requirement

If the ALLC system is not designed to be failed-safe, a single component or point failure will lead to a system failure. Since the driver plays no role in controlling the vehicle once he/she gives vehicle control to the ALLC system, an ALLC system failure will result in catastrophe, even series collisions. The reliability functional requirement for the ALLC system is thus proposed as

"Under no circumstances, should a single component/point failure result in a lateral or longitudinal collision, and there should be no common failure mode."

#### Safe Lateral and Longitudinal Control Operation

The ALLC system receives neighboring vehicles' dynamics data and maneuver intentions from the vehicle-to-vehicle communication. This information allows the ALLC system to generate throttle/brake and steering commands for maneuver coordination. A failure in the vehicle-to-vehicle communication may lead to an ALLC system failure. For safe operation, the ALLC system should not rely only on the vehicle-to-vehicle communication for providing the neighboring vehicles' dynamics data. It should take advantages of the on-board sensors and use them to verify the correctness and to improve the accuracy of the data received from the vehicle-to-vehicle communication.

The ALLC system receives maneuver commands from the on-board navigation system and speed/headway commands from the roadway. The ALLC system communicates with other vehicles for maneuver coordination. Furthermore, the ALLC system needs to perform collision avoidance maneuvers if its sensors detect potential collision danger. All the above mentioned can affect the vehicle lateral and longitudinal control. For safety concern, the ALLC system controller should properly set up the priority of the external commands and sequence the maneuvers to be executed. It is essential that all the maneuvers be performed under the condition that collision avoidance can be guaranteed.

# Required Redundancy

Safe lateral and longitudinal control depends on the reliability of sensor, controller and actuator. Since these major components are connected in series, as shown in figure 89, a single component/point failure will lead to an ALLC system failure. The proposed reliability functional requirement does not allow a single component/point failure to be catastrophic. For fail-safe design, redundancy will be required to implement the ALLC system. A proposed ALLC system design framework that can potentially satisfy the proposed reliability functional requirement is shown in figure 90.



Figure 90. The ALLC system framework for reliability design.

The reliability functional requirement do not allow the existence of any common failure mode, since a common failure mode can cause simultaneous failures in redundant components. Common failure modes can destroy the merits of redundancy. This requirement demands that the redundancies in the proposed framework be independent. The redundant sensors, controllers and actuators have to operate with different technologies, and be supported by independent hardware/software.

The proposed system framework has two redundant control channels. Each control channel contains a lateral dynamics sensor, a longitudinal dynamics sensor, a controller, a throttle actuator, a brake actuator and a steering actuator. Each control channel is capable of performing lateral and longitudinal control functions. If a component/point failure causes one control channel to fail, the other channel can still operate. In this case, the ALLC system is not fail-safe and has to notify the driver of the failure and automatically start the check-out procedure. A third lateral dynamics sensor and a third longitudinal dynamics sensor are used to improve the accuracy of the sensed data and to help identify any failed sensors in the two control channels. A supervisory controller is used to help detect any failed component and to deactivate the failed control channel. Since the probability that any additional component/point fails while the vehicle is checking out of the AHS is negligible, the reliability functional requirements thus can be satisfied.

When the ALLC system is not fail-safe, any additional component/point failure may cause collisions. In this case, the ALLC system needs to be able to guide the vehicle out of the AHS. The driver takes over vehicle control after the vehicle leaves the dedicated lane. Consequently, a non-fail-safe ALLC system does not allow the vehicle longitudinal and lateral control to fall back to the previous ERSCs.

# Important System Level Design Requirements

When one control channel is deactivated due to a potential component/point failure, the ALLC system may need a period of transition time in order to have the other channel operate effectively. Since the vehicle may be traveling at high speeds, a performance degradation like this may have safety implications. To reduce the transition time required for fully activating or deactivating a control channel, it is desirable that the two redundant control channels be kept active all the time as long as the ALLC system is in operation. In this case, one can activate or deactivate a control channel by quickly adjusting its control gain. Feedback design can be used to efficiently adapt the control gains based on the operational status.

In addition to the above discussed, more design requirements are needed to implement a fail-safe ALLC system. The detailed design requirements can be derived by exercising the failure mode and effect analysis (FMEA). Complete FMEA may require more specific and detailed system design framework. Simply examining the proposed system design framework in figure 90, one can easily identify the following important potential failure modes that may affect the safe operation of the ALLC system:

- 1. Longitudinal Dynamics Sensor failure
- 2. Lateral Dynamics Sensor failure
- 3. Controller failure
- 4. Throttle Actuator failure
- 5. Brake Actuator failure
- 6. Steering Actuator failure
- 7. Incorrect speed/headway commands received from the roadway
- 8. Incorrect lane reference data provided by the lan e reference aids
- 9. Incorrect information received form the vehicle-to-vehicle communication

The following proposed design requirements attempt to protect the longitudinal and lateral control functions from the above potential failure modes. In particular, once a potential failure mode is detected, the ALLC system needs to perform the System Failure Response. In the response, the ALLC system disables the control channel in which the failed component or fail point is. It notifies the driver of the detected failure and broadcast its operational status to the roadway and to the neighboring vehicles. It then initiates the check-out procedure by asking the navigation system to deliver maneuver commands for guiding the vehicle out of the dedicated lanes.

Potential Failure Mode 1: Longitudinal Dynamics Sensor failure

1.1 Longitudinal Dynamics Sensor 1 or 2 computes incorrect headways or closing rates data. Design requirements:

1.1.1 Longitudinal Dynamics Sensor 1 and 2 need to have an internal independent check of the reasonableness of the computed headways and closing rates data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

1.1.2 Controller 1 and 2 and Supervisory Controller should verify that each Longitudinal Dynamics Sensor is operating in a timely manner by verifying that new data is received from each sensor every to be determined (tbd) time interval. If an error is detected, the System Failure Response should be executed.
1.2 The data values generated by Longitudinal Dynamics Sensor 1 and 2 are substantially different. Design requirements:

1.2.1 Supervisory Controller should sense the discrepancy and use the data from Longitudinal Dynamics Sensor 3 to determine the failed sensor. The System Failure Response should then be executed.

1.3 Longitudinal Dynamics Sensor 3 computes incorrect headways or closing rates data. Design requirements:

1.3.1 Longitudinal Dynamics Sensor 3 needs to have an internal independent check of the reasonableness of the computed headways and closing rates data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

1.3.2 Supervisory Controller should compare the data computed by Longitudinal Dynamics Sensor 3 with those computed by Longitudinal Dynamics Sensor 1 and 2 to check if Longitudinal Dynamics Sensor 3 is working properly. If an error is detected, the System Failure Response should be executed.

1.3.3 Supervisory Controller should verify that Longitudinal Dynamics Sensor 3 is operating in a timely manner by verifying that new data is received from Longitudinal Dynamics Sensor 3 every tbd time interval. If an error is detected, the System Failure Response should be executed.

1.4 Data from Longitudinal Dynamics Sensor 1 or 2 is not communicated to the controllers, or is corrupted in transmission through interface.

Design requirements:

1.4.1 Each controller should incorporate an independent check of reasonableness of the computed headways and closing rates data from Longitudinal Dynamics Sensor 1 and 2, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

1.4.2 Same as for 1.1.2.

1.5 Data from Longitudinal Dynamics Sensor 3 is not communicated to Supervisory Controller, or is corrupted in transmission through interface.

Design requirements:

1.5.1 Supervisory Controller should incorporate an independent check of the reasonableness of the computed headway and closing rate data from Longitudinal Dynamics Sensor 3, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

1.5.2 Same as for 1.3.2.

Potential Failure Mode 2: Lateral Dynamics Sensor failure

2.1 Lateral Dynamics Sensor 1 or 2 computes incorrect lateral dynamics or lane reference data.

Design requirements:

2.1.1 Lateral Dynamics Sensor 1 and 2 need to have an internal independent check of the reasonableness of the computed data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

2.1.2 Controller 1 and 2 and Supervisory Controller should verify that the lateral dynamics sensors are operating in a timely manner by verifying that new data is received from each sensor every to be determined (tbd) time interval. If an error is detected, the System Failure Response should be executed.

2.2 The data values provided by Lateral Dynamics Sensor 1 and 2 are substantially different. Design requirements:

2.2.1 Supervisory Controller should sense the discrepancy and use the data from Lateral Dynamics Sensor 3 to determine the sensor that fails. The System Failure Response should then be executed.

2.3 Lateral Dynamics Sensor 3 computes incorrect lateral dynamics or lane reference data. Design requirements:

2.3.1 Lateral Dynamics Sensor 3 needs to have an internal independent check of the reasonableness of the computed data, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

2.3.2 Supervisory Controller should compare the data computed by Lateral Dynamics Sensor 3 with that computed by Sensor 1 and 2 to check if Lateral Dynamics Sensor 3 is working properly. If an error is detected, the System Failure Response should be executed.

2.3.3 Supervisory Controller should verify that the processor of Lateral Dynamics Sensor 3 is operating in a timely manner by verifying that new data is received every tbd time interval. If an error is detected, the System Failure Response should be executed.

2.4 Data from Lateral Dynamics Sensor 1 or 2 is not communicated to the controllers, or is corrupted in transmission through interface.

Design requirements:

2.4.1 Each controller should incorporate an ind ependent check of reasonableness of the computed data from Lateral Dynamics Sensor 1 and 2, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

2.4.2 Same as for 2.1.2.

2.5 Data from Lateral Dynamics Sensor 3 is not communicated to Supervisory Controller, or is corrupted in transmission through interface.

Design requirements:

2.5.1 Supervisory Controller should incorporate an independent check of the reasonableness of the computed data from Lateral Dynamics Sensor 3, based on the physical

attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

2.5.2 Same as for 2.3.2.

Potential Failure Mode 3: Controller failure

3.1 Controller 1 or 2 computes incorrect throttle or brake or steering commands. Design requirements:

3.1.1 Each of Controller 1 and 2 should incorporate an independent check of the reasonableness of the computed throttle, brake and steering commands before they are sent to the actuators, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

3.1.2 Each of the throttle actuators, brake actuators, and the steering actuators should incorporate an independent check of reasonableness of the received commands computed by the associated controller, based on the physical attainability and valid known ranges of the system variables. If an error is detected, the System Failure Response should be executed.

3.2 Throttle (or brake/steering) commands computed by Controller 1 and 2 are substantially different.

Design requirements:

3.2.1 Supervisory Controller should sense the discrepancy and determine the controller that fails. The System Failure Response should then be executed.

3.3 Supervisory Controller fails. Design requirements:

3.3.1 Supervisory Controller should incorporate circuitry, such as a watchdog time r, to detect failure and to ensure the processor functions in a timely manner. If a failure is detected, the System Failure Response should be executed.

3.4 Supervisory Controller computes incorrect throttle or brake or steering command. Design requirements:

3.4.1 Supervisory Controller should compare the computed commands with those computed by Controller 1 and 2 to determine if it is working properly. If an error is detected, the System Failure Response should be executed.

3.5 Control commands are not communicated to actuators at appropriate time or is corrupted in transmission through interface with Controller 1 or 2.

Design requirements:

3.5.1 Each of the actuators should verify that it is receiving updated commands from the controller in a timely manner. If an error is detected, the System Failure Response should be executed.

3.6 Supervisory Controller's command data is not communicated to controllers and actuators at appropriate time, or is corrupted in transmission through interface.

Design requirements:

3.6.1 Each of the controllers and actuators should verify that it is receiving updated commands from Supervisory Controller in a timely manner. If an error is detected, the System Failure Response should be executed.

Potential Failure Mode 4: Throttle Actuator failure

4.1 Throttle Actuator 1 or 2 produces incorrect throttle angle for a given throttle command. Design requirements:

4.1.1 Each throttle actuator should incorporate built in tests to be performed on the vehicle startup and during speed/headway maintenance operation to verify that proper throttle angle can be produced and is within the allowable range. If an error is detected, the System Failure Response should be executed.

4.2 Throttle Actuator 1 or 2 can not be deactivated when the associated control channel is determined to be deactivated.

Design requirements:

4.2.1 The ALLC system should provide for two independent, redundant means of deactivating the throttle actuators.

Potential Failure Mode 5: Brake Actuator failure

5.1 Brake Actuator 1 or 2 generates incorrect brake force for a given brake command. Design Requirements:

5.1.1 Brake Actuator 1 and 2 should incorporate built in tests to be performed on the vehicle startup and during speed/headway maintenance operation to verify that proper brake force can be produced. If an error is detected, the System Failure Response should be executed.

5.2 Brake Actuator 1 or 2 can not be deactivated when the associated control channel is determined to be deactivated.

Design Requirements:

5.2.1 The ALLC system should provide for two independent, redundant means of deactivating the brake actuators.

Potential Failure Mode 6: Steering Actuator failure

6.1 Steering Actuator 1 or 2 does not generate correct steering force for a given steering command. Design requirements:

6.1.1 Each of the steering actuators should incorporate built in tests to be performed on the vehicle startup and during lane keeping operation to verify that steering force can be produced and is within the allowable range. If an error is detected, the System Failure Response should be executed.

6.1.2 The ALLC system should provide for independent, redundant means to constrain the steering force generated from each steering actuator with safe ranges.

6.2 Actuator 1 or 2 fails to be deactivated when the associated control channel is determined to be deactivated.

Design requirements:

6.2.1 The ALLC system should provide for two independent, redundant means of deactivating each steering actuator.

Potential Failure Mode 7: Incorrect target speed commands or minimum time headway recommendations received from the roadway

7.1 The roadway target speed commands or minimum time headway recommendations is not communicated to the ALLC system controllers, or is corrupted during transmission. Design requirements:

7.1.1 The data/commands sent from the roadway should contain error detection and error correction codes to improve the accuracy of the data received by the ALLC system controller.

7.1.2 Supervisory Controller should verify that target speed commands and minimum time headway recommendations are received in a timely manner. If updated data is not received for tbd consecutive times, Supervisory Controller should re-evaluate a new lower target speed and/or a more conservative safe time headway for speed and headway maintenance and rear-end collision avoidance, based on the known roadway surface characteristics and traffic conditions. Supervisory Controller should also notify the driver the speed and headway adjustment actions taken by the controllers.

7.1.3 Supervisory Controller should incorporate an independent check of the reasonableness of the received target speed commands and minimum time headway recommendations from the roadway, based on the physical attainability and known ranges of the system variables. If unreasonable data is received over tbd consecutive times, the ALLC system should respond as described in 7.1.2.

- 7.2 Roadway-to-vehicle communication fails. Design requirements:
  - 7.2.1 Same as for 7.1.2 and 7.1.3.

Potential Failure Mode 8: Incorrect lane reference data received from the roadway lane reference aid(s)

8.1 Some lane reference aid does not provide lane reference information to the associated on-board lateral dynamics sensor(s).

Design requirements:

8.1.1 Each lateral dynamics sensor should incorporate reliable means to detect potential failures in the associated lane reference aid. If an error is detected, the System Failure Response should be executed.

8.2 Some Lateral Dynamics Sensor fails. Design requirements:

8.2.1 Same as for the design requirements in 2.1.1, 2.1.2, 2.2.1, 2.3.1, 2.3.2, 2.3.3, 2.4.1, and 2.5.1.

Potential Failure Mode 9: Incorrect vehicle dynamics information received from the vehicle-to-vehicle communication

9.1 The dynamics data and maneuver intentions from the neighboring vehicles are not communicated to the ALLC controllers or are corrupted during transmission.

Design Requirements:

9.1.1 The communicated data should contain error detection and correction codes to improve the accuracy of the data received by the ALLC system controllers.

9.1.2 Supervisory Controller should verify that neighboring vehicles' dynamics data and maneuver intentions are received in a timely manner. If updated data is not received for tbd consecutive times, the System Failure Response should be executed.

9.1.3 Supervisory Controller should incorporate an independent check of the reasonableness of the received dynamics data from the neighboring vehicles, based on the physical attainability and known ranges of the system variables. If unreasonable braking data is received over tbd consecutive times, the System Failure Response should be executed.

9.2 The vehicle-to-vehicle communication fails. Design Requirements:

9.2.1 Same as for 9.1.2 and 9.1.3.

#### Key Findings

1. The driver can not be considered as a back-up to the ALK system. A fail-safe ALK system design will require considerable structure complexity and component redundancies.

2. The proposed reliability functional requirement for the ALK system is: "A single component or point failure is not allowed to cause a lateral or longitudinal collision, and there should be no common failure modes."

3. A potential system design to guarantee reliability will require two active control channels with substantial amount of components. The redundant control channels in the proposed ALK system design framework need to be always kept active to ensure safe and smooth deactivation (or full activation) of a control channel.

#### Issues and Risks:

1. The required reliability enormously increases the system complexity and design difficulty. The required instrumentation on both the vehicle and the roadway will be very costly.

2. The roadway may be liable for some accidents since it may assist vehicle maneuver coordination.

#### **Roadway to Vehicle Speed and Headway Commands**

The roadway sends target speed commands and headway recommendations to the vehicles in the multiple dedicated lane. Depending on the traffic conditions in each lane, even in the same roadway section, different lanes may have different speed and headway commands. The reliability requirement analysis for ERSC 1, 2 and 3 applies to each dedicated lane in ERSC 4.

#### **Roadway Lane Reference Aids**

The lane reference aids help the vehicle in lane keeping and lane changing control. They assist the lateral dynamics sensors of the ALLC system in collecting information such as position deviation from the lane center, roadway preview data, lane identification, and number of lanes around the vehicle, etc.

The lane reference aids need to support the lane keeping and lane changing functions in each dedicated lane. If the ALLC system is required to have triple redundancy of independent lateral dynamics sensors, there should be three corresponding lane reference aids for each dedicated lane. Figure 91 shows the functional block diagram of the roadway lane reference aids in ERSC 4.



n = total number of dedicated lanes.

#### Figure 91. The functional block diagram of the lane reference aids in ERSC 4.

A lane reference aid failure will result in non-safe ALLC operation. For safety concern, the vehicles on the lane with a failed lane reference aid will have to leave the lane as soon as possible. The vehicles should also avoid getting into the lane with a failed lane reference aid. If we allow the vehicles to operate in the lane with a failed lane reference aid, the vehicle functions will fall back to ERSC 1 or 2 in this lane. This is not desirable, since such a mixed modes operation in some section of the roadway may affect the safety and efficiency of the overall AHS operation.

For magnetic-marker-based sensing, the reliability of the lane reference aid can be improved by installing markers with higher density along the dedicated lane and frequently checking the intensity of magnetic field generated by each magnetic marker. For vision-based sensing, the roadway can

improve the reliability of the lane reference aid by providing lane markers with excellent contrast under various weather conditions. Redundant road lane reflectors may be used to improve the reliability of the radar-based lane reference aid.

The roadway needs to have reliable self-diagnostic procedure to detect any lane reference aid failure along any section of the highway. If a failure is detected, the roadway should warn the vehicle/driver of the danger through the roadway to vehicle communication. It is also desirable that the data provided by the lane reference aids be designed in such a way that the lateral dynamics sensors can detect possible failures in the lane reference aids.

#### Key Findings:

1. Multiple types of lane reference aids are necessary in each dedicated lane. The design of lane reference aids depends on the sensing technologies used by the sensors of the ALLC system.

#### Issues and Risks:

1. The roadway may be liable for accidents that are contributed by failures in the lane reference aids.

2. If a dedicated lane has lane reference aids failure in some roadway sections, the AHS may have to prohibit vehicles from entering those sections. This will reduce the highway flow rate and cause traffic control problems.

#### **Vehicle Navigation**

The on-board navigation system guides the vehicle to the trip destination entered by the driver. The navigation system plans a route to the destination that minimizes the trip time or/and travel distance. It performs route planning while taking into accounts the traffic information provided by the roadway and the vehicle's operational status (such as the amount of fuel available). The navigation system needs to have frequently updated vehicle position data, including the highway and the lane the vehicle is on, and the relative distance from highway entry/exit points or intersections. The navigation system evaluates the above information with the planned route to generate lateral and longitudinal commands to the ALLC system. Figure 92 shows the functional block diagram of the vehicle navigation system.



Figure 92. The functional block diagram of the lane reference aids in ERSC 4.

The navigation system displays to the driver the location of the vehicle and the path it plans to take. During the trip, the driver can re-enter the trip destination to the navigation system. The driver can also specify a desirable path that can lead the vehicle to the destination. The maneuver commands sent to the ALLC system include lane changing or merging. A command can be as simple as changing lanes from lane 3 to lane 2 between locations A and B, or merging to left at location C. Since the maneuver commands will be processed by the controller of the ALLC system before the maneuver action is carried out, an incorrect maneuver commands will not immediately endanger the vehicle operation. However, vehicles with failed navigation systems may elongate the trip time. The efficiency of AHS will be affected by the vehicles that can not navigate themselves. It is essential that the vehicle with a failed navigation system get off the dedicated lanes as soon as possible.

The ALLC system carries out the check-out procedure with the help of maneuver commands from the navigation system. The navigation system generates the check-out maneuver commands in any one of the following situations:

- 1. The vehicle arrives its destination.
- 2. The driver initiates the check-out procedure.
- 3. The navigation system detects an internal malfunction.
- 4. The ALLC system performs its System Failure Response to leave the dedicated lanes.

Some navigation system internal failures, such as losing the map database, may affect the system's capability of guiding the vehicle accurately. There will be safety and implications if the vehicle can not check out of the AHS when it has to.

With the above discussion, we propose the reliability functional requirement for the navigation system as:

"The navigation system should provide for redundant means to guide the vehicle out of the dedicated lanes."

This requirement can be achieved if the vehicle has redundant methods to accurately evaluate the positions of the vehicle and the next available AHS exit point. Dead-Reckoning and map matching techniques.  $^{(44,52)}$ , global positioning system (GPS).  $^{(55,107)}$ , and roadway-to-vehicle communication can be used to implement a navigation system that can fulfill the above reliability functional requirement.

#### Key Findings:

1. The navigation system is required to be able to guide the vehicle out of the dedicated lanes reliably.

#### Issues and Risks:

1. More research needs to be done in finding methods for locating the vehicle position and for improving the accuracy of the position data. This may require substantial instrumentation on the vehicle and the roadway.

#### **Driver Tasks and Workload**

This section discusses the drivers' tasks in this ERSC. It covers normal driving in the dedicated lane when all of the automated functions work perfectly. In ERSC 4 this consists of starting the automated systems and selecting a destination. Then it covers the fall-back mode in which the vehicle switches to ERSC 3 control with speed and headway maintenance, rear-end collision avoidance, and automatic lane keeping if the coordination or maneuver coordination system fails or degrades. The driver has only the role of a passenger in this ERSC.

The driver drives the car to the automated lane. The driver turns on the vehicle's automated systems for longitudinal and lateral control. The vehicle enters the lane when a safe gap appears in the traffic flow. The vehicle steers into the lane and turns on the SHM function, collision avoidance system and the automatic steering function. The vehicle takes over the throttle, steering, and brake when the car is on the highway or as the vehicle enters the highway.

The vehicle turns on the turn signal to leave the automated lane. This initiates check-out procedures. Check-out returns the steering, brake, and throttle control to the driver. The vehicle then changes lanes out of the automated lane. The driver turns off the communications links when he leaves the automated lanes.

#### Task Analysis-Normal Driving

In normal operations, the vehicle controls the steering, throttle, and brake. The vehicle performs lane changes into and out of the dedicated lane. The vehicle also avoids lateral collisions. The driver selects a destination and turns on the automated systems on the vehicle. The roadway sends target speed commands and minimum headway information electronically to the vehicle. This data helps the vehicle compute a safe headway. The roadway also sends traffic data to the driver to help him/her select a route. The task analysis in table 34 shows the driver cues for this task.

Task	Select destination and turn on automated systems				
General	Task similar to being a passenger in the vehicle				
Initiating Cue	Enter Check-in point and turn on automated systems –The driver sets destination				
Feedback	Situation display that shows vehicle location and status				
Task Standards	Select destination before vehicle enters designated lane				
Task Conditions	Weather, road conditions, or traffic				
Skills Required	Choose destination				
	Correctly turn on automated systems				
Potential Errors	Choose wrong destination				
	Turn on wrong systems				
	Do not turn on proper automated systems				
	Fail Check-in, enter highway anyway				
	Vehicle chooses wrong or inefficient route				
Causes of Error	Fails to perceive or misinterprets system feedback, believes system has				
	malfunctioned				
	Poor judgment of location				
	Hit wrong button and overrides function				
	Sensor, controller, or actuator failure on vehicle				
	Roadway controller malfunctions				

<b>Consequences of Error</b>	Disrupt traffic flow and lower capacity				
-	Collision with surrounding vehicles				
	Disturbance due to sudden route changes				
<b>Recovery Points</b>	Minimum safe headway setting on vehicle to avoid collisions				
	Collision warnings from other vehicles				
	Vehicle warns driver that headway maintenance function is not working or is				
	failing or has been turned off; prevent easy override of SHM, ALK, and RECA				
	functions				
Individual Differences	Education				
	Age				
	Technology background				
Criticality	Essential				

#### Task Analysis-Fall-back Mode

In the fall-back mode, the automatic lane-changing or maneuver coordination system suffers a partial failure when one control channel fails, but the other one remains intact. If the automatic lane keeping functions can still operate then the vehicle switches to ERSC 3 operations in which the driver is responsible for lane changing. The vehicle warns the driver that he is now responsible for lane-changing and slows the vehicle to a safe speed for manual steering. The vehicle should then exit the dedicated lanes if it cannot safely maintain the speeds commanded by the roadway. If the warnings fail, the vehicle notifies the driver and the driver performs the task without the aid of warnings. Table 35 shows the task analysis for the lane changing function.

Task	Fall-back mode to ERSC 3					
General	Task may be similar to manual driving and lane changing with cruise control					
Initiating Cue	Vehicle warns driver to resume responsibility for lane-changing					
	Driver receives warning signal for lane departure from vehicle					
	Vehicle may decrease speed for to allow for manual steering					
Feedback	Visual, auditory, and kinesthetic cues for steering					
Task Standards	Duration the same as manual driving, transition to manual driving must be					
	gradual, vehicle keeps deceleration low while decreasing speed					
Task Conditions	Weather, road conditions, suggestions from traffic control center					
Skills Required	Reaction times appropriate to speed and roadway conditions.					
	Perceptual judgment of vehicle kinematics					
Potential Errors	Driver inattention causes vehicle to collide or leave lane					
	Driver overrides lane-changing system accidentally					
	Driver does not respond in time to lateral collision warning					
	Vehicle fails to maintain speed or proper headway					
	Vehicle selects unsafe speed					
	Vehicle does not warn driver adequately of a control transfer					
Causes of Error	Fails to perceive or misinterprets system feedback					
	Poor judgment of speed and following distance					
	Hits wrong button					
	Sensor or actuator failure on vehicle					
	Role confusion by driver-responsibilities are not clear					
<b>Consequences of Error</b>	Disrupt traffic flow and lower capacity					
	Lane departure or lane change/merge collision					
	Driver confusion					

#### Table 35. Driver resumes control of lane-changing function.

<b>Recovery Points</b>	Maximum speed limit setting on vehicle			
	Lane departure collision warning			
	Vehicle warns driver that lane-changing function is not working or is failing or			
	has been turned off			
	Vehicle requires active response from driver before a control transfer			
	Driver interface shows driver control status clearly			
Individual Differences	Drivers may be uncomfortable taking over steering control at high speeds or in			
	reduced visibility <sup>(90)</sup>			
Criticality	Essential			

The maneuver coordination system helps vehicle merge into the dedicated lanes, change lanes, and avoid collisions. The vehicles can then maneuver safely. The vehicles may also coordinate suggested lateral collision avoidance maneuvers. This arrangement is similar to the TCAS system for commercial aircraft. <sup>(96)</sup> If this system fails then the driver takes over the maneuver coordination task by pre-approving all maneuvers before the vehicle executes them. Table 36 shows the task analysis for the maneuver coordination function.

Task	Driver resumes control of maneuver coordination		
General	Driver approves all maneuvers before execution		
Initiating Cue	Cue from vehicle that a gap in traffic is approaching		
	Driver turns on turn signal to enter dedicated lanes		
	Vehicle detects change in steering angle		
Feedback	Visual, auditory, and kinesthetic cues of steering and accelerating		
Control Functions	Drivers must be able to merge and change lanes safely		
Task Standards	Safely merge into traffic		
	Accelerate as directed by vehicle or roadway		
Task Conditions	Type and timing of warning		
	Environmental conditions, traffic speed and density		
Skills Required	Reaction times appropriate to speed, lane change speed, and roadway conditions.		
	Perceptual judgment of vehicle kinematics		
Potential Errors	Overconfidence in system-driver does not look in lane changes		
	Driver does not use turn signals to signal merge		
	Driver does not respond to the cue or responds too late		
	Maneuver coordination does not open a gap		
	Maneuver coordination opens large gaps		
	Vehicle recommends incorrect maneuver		
Causes of Error	Fails to perceive or misinterprets the indication of an emergency		
	Driver delay in response may cause unsafe lane change		
	Warning controller misinterprets the data and the risk		
Consequences of Error	Lane change/Merge collision with potentially severe results for the system		
	(Approximately 4% of crashes and 0.5% of fatalities are lane change/merge		
	crashes. Some 95% of these crashes are angle/sideswipe crashes $(113)$		
	Disrupt traffic flow and lower capacity		
	Loss of confidence by the driver and public in the system		
	In the case of false alarms the driver may defeat or ignore it $(96)$		
Recovery Points	Vehicle reminds driver to look before lane change		
Recovery rollits	Multiple cues from vehicle and roadway		
	Vehicle suggests an acceleration profile to the driver for merging		
	Vehicle suggests an avoidance path to driver		
Individual Differences	Accident statistics, citations and self report data indicate that older drivers have		
marviadur Differences	difficulty in merging, changing lanes, and exiting maneuvers. Data suggests that		
	persons over 65 may be over represented in lane change/merge crashes <sup>(113)</sup>		
Criticality	Essential		
Childunty	Lisbentiui		

#### Table 36. Driver resumes control of maneuver coordination.

#### Issues and Risks

#### 1. Driver confidence in automated system

If the driver has too much confidence in the automated systems then she may not pay attention if the vehicle needs to fall-back to manual driving. This could lead to accidents if the warning system does not detect an object or the system fails. Driver lack of confidence in the automated system could cause the driver to disable or ignore the system. Research needs to be done on when and how the driver may disable automated functions.

2. Enabling/Disabling Automatic Systems

This ERSC includes a speed and headway maintenance system, lane keeping, lateral collision avoidance, and communication systems. Research needs to be done on the instrumentation needed to control these different systems. The question of whether or not all of the automatic systems need to be controlled with one switch or many needs to be answered.

3 Driver override of automated functions

Even with multiple sensors the vehicle may not see and recognize everything that a human driver sees. This is especially true of unexpected objects such as animals or debris on the roadway. It may be necessary for the driver to override the collision avoidance system on the vehicle to avoid these objects. Research needs to be done on whether or not this driver override should be allowed to occur.

## **Key Results**

## **Key Findings**

1. The driver can not back up the automatic lateral and longitudinal control (ALLC) system if it fails. The reliability functional requirement of the ALLC system is: "Under no circumstances should a single component/point failure result in a lateral or longitudinal collision, and there should be no common failure modes.

2. A potential ALLC system design to achieve the reliability functional requirement will need two active redundant control channels with substantial amount of components.

3. The fully automated vehicles perform maneuver coordination to further reduce accident rates and improve traffic flow rates. This will require complicated vehicle-to-vehicle communication and maneuver protocols.

#### **Issues and Risks**

#### 1. Cost

The required reliability of the ALLC system enormously increases the system complexity and design difficulty. The required instrumentation on the vehicle will be very costly.

#### 2. Legal/Liability

The roadway may be liable for traffic accidents due to its involvement in the vehicle lateral and longitudinal control.

#### 3. Cost

There are multiple dedicated lanes in ERSC 4. Since each lane needs to be equipped with redundant independent lane reference aids, the cost of the roadway infrastructure will be increased considerably.

Raytheon

Section 9: ERSC 5 Analysis

## **Description of ERSC 5**

In ERSC 5, the roadway provides multiple dedicated lanes as described in ERSC 4. The roadway is responsible for maintaining these dedicated lanes and the communications infrastructure.

The roadway receives position, speed, destination, and capabilities data from the vehicles. The roadway senses traffic flow and environmental conditions on the dedicated lanes. The roadway uses this data to provide detailed course commands to the vehicles. These commands tell the vehicle what lane to travel in, when to change lanes and what speed and headway to use. The roadway coordinates these maneuvers to avoid collisions and smooth traffic flow. Based on the indicated trip destination the roadway plans the vehicle's route.

The vehicle has all the capabilities for full lateral and longitudinal control described in ERSC 4. Additionally the vehicle receives detailed course commands from the roadway. The vehicle responds to these commands automatically as long as it judges the maneuver to be safe. The vehicle sends its position, speed, maneuver status, and capabilities to the roadway. The vehicle coordinates each roadway-commanded maneuver with the vehicles around it. The vehicle provides the driver with traffic information and trip status data. It responds to driver commands for changes in destination. In case of malfunctions the vehicle may fall-back to a lower ERSC and notify the roadway and driver appropriately.

The driver manually drives the vehicle to the dedicated lane check-in point. The driver initiates the check-in procedure and releases control to the vehicle and roadway. The driver inputs its trip destination that is used by the vehicle and roadway for route planning. The driver, as in ERSC 4, can request a check-out procedure and resume manual control at the next chosen exit ramp or transition lane.

The motivation behind ERSC 5 is:

- to maximize highway system throughout by permitting full roadway control of all vehicle activities.
- to improve driver comfort and reduce trip time.
- to efficiently detour traffic around traffic incidents and roadwork's.

ERSC 5 allows us to focus on the analysis of a highly coordinated AHS where the roadway plays an active role in the lateral and longitudinal control of the automated vehicles by planning their path and coordinating their maneuvers. Table 37 summarizes the functions of this ERSC.

Function

runction	Requirement	Differ/ Venicie/Roduway Tasks
Automatic Lateral and Longitudinal Control and Maneuver Coordination	<ul> <li>Smoothly and safely change lanes</li> <li>Keeps vehicle in the center of the lane around curves and on straight-a-ways</li> <li>Vehicle performs maneuvers commanded by roadway if they are safe</li> </ul>	<ul> <li>Same as ERSC 4</li> <li>Receive commands from roadway</li> <li>Perform maneuver within the given period of time if safe and inform roadway of status</li> </ul>
Navigation/Route Selection	Select a route to reach the driver's destination	<ul> <li>Dynamically select lanes and route for trip</li> <li>Send destination to roadway</li> <li>Receive route commands from the roadway</li> </ul>
Driver Interface	<ul> <li>Show driver route and allow driver to change route and destination</li> <li>Show driver vehicle status</li> </ul>	<ul> <li>Driver should be able to:</li> <li>» select a destination</li> <li>» initiate check-out</li> <li>Interface should be simple to understand and use</li> </ul>
Fall-back Mode to ERSC 4	Vehicle selects its own route through traffic if communication links to roadway fail	<ul> <li>Sense status of different vehicle control channels</li> <li>Safely transfer control of course selection and navigation to vehicle</li> </ul>
Roadway/Vehicle maneuver commands	Senses vehicle average speed and density and environmental conditions then sends commands directly to vehicle to smooth traffic flow; Performs check-in and vehicle status monitoring	– Roadway commands maneuvers to each vehicle.

Table 37	FRSC 5 functions and	driver/vehicle/roadway	nerformance red	mirements
Table 57.	ENSC 5 functions and	univer/venicle/roadway	perior mance rec	jun emenus.

Driver/Vehicle/Roadway Tasks

Requirement

## Performance Requirements

## Automatic Lateral and Longitudinal Control

The automatic lane changing function moves the vehicle from the center of one lane to the center of an adjacent lane on normal highway geometries if it is safe. The vehicle must first check if the lane change is safe and signal its intention to change lanes. When there is an appropriate gap between vehicles in the adjacent lane then the vehicle selects a trajectory and performs the lane change. The vehicle should accurately follow a desired path and provide satisfactory ride comfort over a wide range of speeds, roadway conditions, and disturbing forces. The automatic lane changing function controls the vehicle's steering smoothly and without oscillations. The automatic lane changing function should also prevent the vehicle from rolling over or going out of control when the rear-end collision avoidance system needs to brake on curved roads. The lane changing function works with the vehicle's throttle and brake controller to keep safe gaps between vehicles.

The driver checks-in and then enters the automated lane. The point at which the driver releases manual control could be on the dedicated ramp or in the dedicated lane depending on the entry configuration. Once the vehicle has entered the automated highway the driver starts the automated systems for steering and speed and headway maintenance. The vehicle smoothly takes over control from the driver. The lane changing function allows the vehicle to change lanes. The vehicle merges itself into the dedicated lane with the aid of maneuver coordination between the vehicles and roadway.

The automatic lane changing system requires sensors, a computer with control algorithms, and steering, brake, and throttle actuators. The navigation system chooses when the vehicle should change lanes. The communications system allows the vehicle to coordinate their actions and avoid accidents. The sensors measure the vehicle's dynamics and the forward roadway geometry. Lane reference aids <sup>(30)</sup> help the vehicle sensors measure roadway geometry information. The computer generates control signals for vehicle steering.

#### Lane Change Coordination

Lane changes may be performed more efficiently if the vehicle coordinates its maneuvers with the vehicles around it. Studies done by the California PATH program investigate using a communication link to coordinate a lane change. Before making a lane change the vehicle sends out a signal of its intent to change lanes. This signal serves as an electronic turn signal. Any vehicles that are affected by this maneuver must acknowledge the lane change and grant permission. If acknowledgments are not received then the intent signal is repeated. The analysis in Streisand and Walrand only assumed that one vehicle needs to acknowledge the maneuver and in effect promise to maintain its current course for a given period of time. Realistically the vehicle needs responses from all the vehicles in its sensor range. However this could cause problems in coordinating the maneuver if there are problems of signal collision. In some vehicle following situations the vehicles may need to request cooperative actions from the other vehicles. This could include asking the vehicle to speed up or slow down so the maneuvering vehicle can change lanes. But asking other vehicles to change speeds affects traffic flow and possibly causes traffic jams.

Lane change coordination requires a sophisticated communications system that coordinates the timing and synchronization of messages. Other possibilities are lane change protocols over a section of roadway. For instance one section of roadway may only allow lane changes to the left, while the next section could allow only right lane changes.

#### Issues and Risks

1 Lane Change Coordination

Coordinating sensors and communications will be a very difficult task. This will require very sophisticated algorithms to coordinate the lane changes.

2 Automated lateral and longitudinal control

Lane changes require a combination of lateral and longitudinal control. Some work has been done combining these modes. Under normal driving conditions, lateral and longitudinal control is decoupled. Studies need to be done for lane changing to verify that this holds true for most lane changing conditions.

#### Emergency Collision Avoidance

The collision avoidance system combines lateral and longitudinal control to avoid collisions due to moving or stationary obstacles around the vehicle. This system must be designed so that no single point faults or common mode faults will permit the collision avoidance system to fail. The collision avoidance system must also send out an emergency signal when it must take evasive actions or a control system fails. Possible sensors include radar, ultrasound, vision-based and infrared systems. These sensor have been discussed in the section on lateral collision warnings. Communications can also help the vehicles to plan coordinated actions for collision avoidance.

If the vehicle needs to take emergency action due to failures in the automated system, then it sends out an emergency signal to the vehicles around it. The vehicle may sense or have a problem and need to warn other vehicles. Emergency messages must be transmitted quickly and reliably. This signal gives the vehicle's position, heading, and intentions.

Emergency signals identify what vehicle sent the message. The destination address may be a selective broadcast address specifying the affected section of the highway. <sup>(100)</sup> Emergency messages have high priority depending on the type of emergency. The data field of the message contains essential emergency information such as the location, nature (such as flat tire, debris in roadway, or sudden loss of control), and suggested actions to be taken. Research at California PATH indicates that this message may be up to 1 kilobyte long. <sup>(100)</sup>

The emergency signal system must be extremely reliable so that no single point or common mode faults will cause the system to malfunction. The emergency system requires an emergency channel for emergency use only. This system may need to be tested regularly to be sure that each vehicle is working properly and can respond quickly to an emergency message.

The vehicles coordinate their maneuvers with a communications system to make the maneuvers safer and easier. Work at PATH suggests that there are three basic types of maneuvers needed for platoons on automated highways: lane change, platoon split, and platoon merge. <sup>(47)</sup> Since we are looking at a collision-free system without platooning these basic maneuvers can be restated as lane change, maintain gap, maintain a constant speed, and change speed up or down. These basic maneuvers will allow a vehicle to move on the automated lanes. Maneuver coordination will require communications between vehicles. Figure 93 shows the block diagram for the maneuver coordination system. This system uses sensors to figure out which vehicles need to be communicated with. The surrounding vehicles need to know the position, intentions and heading of the vehicle.



Figure 93 Block diagram for the maneuver coordination system.

#### Sensors

The sensors help the vehicle determine which vehicles must be informed of different maneuvers. Intra-lane maneuvers such as speed up, brake or change headway only require information from the vehicles in the same lane.

#### **Communications**

Table 38 summarizes the basic message types for maneuver coordination. The information needed column show the contents or data for the message. The type of communication column lists the requirements for the communications system. Possible requirements include whether the message must be acknowledged by other vehicles, the priority of the message, and whether the message is directed at any particular vehicle.

Maneuver Name	Information Needed	Type of communication
Lane Change	Direction and duration of lane change Acceleration Position Course and Heading	One to many Message directed to certain vehicles Acknowledgments needed from surrounding vehicles Medium priority
Collision Avoidance	Type and nature of emergency What Action is being taken Acceleration Position Course and heading	One to many Message sent to all vehicles in vicinity Acknowledgments needed from surrounding vehicles High priority message
Braking	Degree of braking Road-tire coefficient	One to one Message sent to vehicle behind only
Change speed	Acceleration Position Course and Heading	One to one Message sent to vehicle behind only
Change headway	Acceleration Position Course and Heading	One to one Message sent to vehicle behind only
Status	Acceleration Position Course and Heading	One to many Unacknowledged Low priority

Table 38	Types of	communications	needed for	maneuver	coordination
	J 1				

#### **Controllers**

The controller must synchronize the messages for all of the vehicles.

#### Maneuver Protocols

The message frequency may be minimized by the use of protocols for maneuvers. These protocols control which vehicle has the right of way and which one must maneuver. These protocols give a fixed set of rules for lane changes and collision avoidance.

#### Issues and Risks

1 Emergency Dynamics

Very little modeling work has been done to characterize vehicle dynamics under extreme conditions. This work needs to be extended.

2 Coordination

Collision avoidance is most effective when done early before an emergency exists. The collision avoidance system must coordinate actions in ways similar to the TCAS system to keep vehicles safe.

#### 3 Path Planning

All of the work on path planning has used robots in fairly static environments. Research needs to be done on real-time systems in a dynamic environment.

### **Route Selection and Navigation**

The driver selects the destination at the check-in point or at the beginning of the trip. The vehicle broadcasts the destination to the roadway at check-in or if the destination changes at any time during the trip. The roadway then chooses a route based on traffic information, road conditions, and driver preference. Route selection will be a dynamic process so the route may changes as conditions change. The driver(?) has the final say in the route selection. The roadway decides what lanes the vehicle should use for the trip and when it should change lanes as it navigates on the highway.

#### Route Selection

The roadway selects a route based on traffic conditions, distance and roadway status. The driver can then approve or disapprove the route. If the driver disapproves the route then the vehicle or driver selects an alternative route. The driver can change the route at any time during the trip if traffic or personal conditions warrant it.

#### **Navigation**

The roadway could also send data to the vehicles about roadway curvature. <sup>(33)</sup> This approach uses beacons at fixed spots to tell the vehicles about the roadway ahead. The vehicle then uses information on the location of the beacon and dead-reckoning for preview information. <sup>(55)</sup> Dead-reckoning systems use wheel odometers accelerometers to measure distance and gyroscopes to measure the vehicle's heading. This approach has already been tested in the Prometheus project. <sup>(10)</sup>

The global positioning system (GPS) provides absolute position data free from error accumulation.<sup>(55)</sup> However GPS accuracy is degraded by tall buildings, tunnels, and large trees that are common in large cities. The vehicle can then use on-board maps and dead-reckoning to calculate the preview data.<sup>(55)</sup> The absolute position accuracy of GPS can then provide feedback and calibration signals to correct dead-reckoning errors. These systems can be integrated with algorithms that switch between systems depending on which has better conditions for operation or a filter can combine all sensor information to get the best estimate.<sup>(55)</sup> This method suffers from time delays which may make it impractical for real-time control applications.

#### Issues and Risks

#### Information Sources

Traffic and road status data may be available from public or private sources. The driver may choose to obtain information from a private source that is more detailed than public data.

## Driver/Vehicle/Roadway Interface

The driver should be able to override the automated lateral and longitudinal control systems if a control channel fails. However the driver must go through a check-out procedure first to verify that he/she is ready to resume control of the vehicle. The driver also monitors the vehicle's operational status and backs up failed vehicle functions if possible.

The driver interface must show the driver that the automated functions are operating correctly or that a function has failed. This interface should be simple to understand and use. Input systems like touch pads or buttons may be distracting and may require too much driver attention. <sup>(8)</sup> The driver must be able to tell what the vehicle is doing at all times and why it is doing it. This shows the driver that the vehicle is operating correctly. Studies show that drivers cooperate more when they know the reasons for an action. <sup>(10)</sup>

Head-up displays may be particularly useful for visual indicators of the system operation. For example a heads-up display could show the measured distance to the vehicle ahead as well as the minimum time headway and target speed. Other technologies include liquid crystal displays (LCDs), light-emitting diodes (LEDs), computer generated voice messages, sudden vehicle accelerations/decelerations (jerk), tactile feedback, and directional audio displays. <sup>(64,8,9)</sup>

The interface may be continuous or only active when a driver action is needed. The best design of the driver interface for human factors is an open issue and involves selecting a display technology, type and duration of warnings, user control types and locations, etc. User acceptance is crucial to user interface design.

## Fall-back Mode

The system needs to degrade gracefully if a failure occurs. Many failures such as communications systems or isolated sensors may not cause the vehicle to be non-operational. Vehicles with different capabilities may coexist in the dedicated lanes without incident for short periods of time.

In case of failure in a single control channel of the vehicle-roadway communication system, the vehicle can operate in ERSC 4 control mode. The vehicle selects its route and its maneuver timing.

If the maneuver coordination functions fail the vehicle informs the driver. Failure of these functions may not require a transfer of control to the driver so they do not require the vehicle to return to ERSC 3 control.

If the longitudinal controller fails the vehicle should send out an emergency signal and exit the dedicated lanes as soon as possible. While multiple paths of degradation are technologically possible they are not desirable from a human factors point of view since it may require extensive driver training and practice to handle potential failures safely.

## **Roadway/Vehicle Maneuver Commands**

The roadway measures the average speed and traffic density, assesses environmental conditions, and calculates vehicle target speeds and route for the dedicated lanes. The vehicles follow the target speeds for their lane set by the roadway using the speed and headway maintenance function. The vehicles should respond to headway recommendations for longer headways due to changes in environmental conditions such as slick roads. The vehicles use the environmental data from the roadway to adjust their time headways. The vehicles check-in with the roadway by sending their identification, automated function status, and location. <sup>(108)</sup> The vehicles may also display relevant traffic information from the roadway to the driver for route planning purposes. This section covers roadway sensing systems and roadway-vehicle communications.

#### Roadway Sensing Systems

The roadway needs to sense the position, heading, and speed on each vehicle on the highway. This data can be measured by sensors that are part of the infrastructure or the vehicles can transmit their measured speed, position, and headway to the roadway with a vehicle-roadway communication system for each lane. The vehicle may also transmit its operational status and identification to the roadway for check-in or incident management. The roadway also needs to measure roadway weather data to select a safe target speed for the environmental conditions. <sup>(57)</sup> Section 5 on roadway sensors gives more details on these sensors.

#### Vehicle to Roadway Communications (VRC)

The roadway needs to have one-to-one communications with each vehicle on the roadway. The roadway sends lane change, merge, and demerge commands directly to individual vehicles. The roadway sends speed and headway data to the cars in each lane to smooth traffic densities and manage incidents. The roadway needs to avoid large speed differences between lanes for safe transitions. The vehicles may also send their measured velocity, position, and heading to the roadway. This system may also be used for check-in. This allows the roadway to monitor the operational status of the vehicles and ensure that they are correctly operating in the automatic mode. These messages only need to be sent once per section depending how directly the roadway controls maneuvers. There needs to be separate messages for each lane and vehicle. The vehicles will also check-in with the roadway to ensure that they are working correctly. Each vehicle sends status data on all of its automated functions to the roadway. In ERSC 4 these functions are: combined lateral and longitudinal control, collision avoidance, communication systems, and critical engine functions  $^{(108)}$ 

VRC systems can be mounted on the side of the roadway, on overpasses, or embedded in the roadway.<sup>(85)</sup> These systems may use automated toll collection technology with variable registers or a system that covers a wider range. <sup>(22,53)</sup> Appendix A covers the operation of these systems. A separate system may be required for each lane.

Target speed messages from the roadway to the vehicles do not need a high data rate since it will not change very often. Each field will need 6-8 bits to give a resolution of 1 meter for the gap limit and 1 mile per hour for the velocity limit. Since the message goes to all cars in a given lane a destination ID code must tell which lane the message is intended for. The source ID code indicates that the message is from the roadway. Error detection coding looks for bit errors in the message.

Retransmission requests are unnecessary since we can rely on repeating the message. This message should be repeated at most one time per second. The bit error rate can also be high since the message changes slowly and rarely. Repetition of messages can compensate for the high error rate. If the roadway sends preview data to the vehicles, the messages may need to be longer in curvy regions of the roadway.

The VRC may also send messages from the vehicles to the roadway. Each vehicle sends its speed and the distance to the preceding vehicle (6-8 bits each). The vehicles may also send a unique identifying code (28 bits) to the roadway for check-in. The vehicle must also send its lane position to the roadway. The operational status of the automated system can be sent as well, this could take 8-16 bits depending on the amount of detail required by the roadway. Other variables such as estimates of the road-tire friction coefficient can take 4 or more bits. Table 39 shows the possible variables to be sent to the roadway for check-in <sup>(108)</sup> This gives a data length of around 95 bits.

Function	Components	Number of Bits
Longitudinal and Lateral Control	Sensors, Actuators, Controller, Vehicle-Vehicle Communications System	8-?
Vehicle-Roadway Communication System	Transmitter, receiver, variable register	1-2
Vehicle Identification Code		28
Other Vehicle Functions	Critical fluids level and pressures, brake conditions, tire pressure, engine temperature, headlights	12-16

Table 39.	<b>Potential status</b>	variables to b	be sent to t	he roadway	by the	vehicles for	check-in.

A broadcast system with a range of 100 meters for multiple automated lane must communicate with 0.3 to 0.75\*NL vehicles per second. NL is the number of dedicated lanes. Analysis by Polydoros et al shows that the current Hughes VRC system and automatic toll collection systems can meet the data transmission requirements for VRC in ERSC 4. (85)

#### Issues and Risks

1 Legal/Liability

In this ERSC the roadway sends recommended target speeds and minimum headways directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the minimum recommended headway is too short. The government may not wish to deploy such a system until it has been shown to be reliable and fail-safe.

2 Sensors

Roadway weather sensors only cover small regions of roadway. These sensors could easily miss patches of ice or puddles on the roadway. The roadway may then choose an unsafe headway. Research needs to continue into these sensors to improve their capabilities.

3 Privacy

The vehicles "Check-in" with the roadway. This may leave a record of when and where a vehicle traveled. This raises user privacy concerns about governmental use of this data.

4 Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

5 Coordination between lanes

The roadway must coordinate the traffic speed in adjacent lanes to smooth traffic flow. Speed differences of more than 10-20 mph will make lane changes and merges unsafe as well as slowing down traffic in the high speed lanes. This requires a more sophisticated roadway controller than in the previous ERSCs.

### **Reliability Requirement Analysis**

In ERSC 5, the automated vehicle functions are basically the same as in ERSC 4. In addition to the roadway functions in ERSC 4, the roadway can further improve the AHS efficiency by providing route planning and navigation commands to each vehicle in the roadway network. The roadway has better scope of the entire roadway network traffic conditions, thus can come up with strategies for more efficient traffic flow control. The roadway carries out the traffic control strategies by sending navigation commands to vehicles through roadway-to-vehicle communication.

As discussed in ERSC 4, the ALLC system uses maneuver coordination and on-board sensors to ensure the safe execution of navigation commands. This reduces the safety concern when incorrect navigation commands are received from the roadway. However, failures in the roadway navigation can potentially affect the AHS traffic flow if incorrect navigation commands are adopted by vehicles. Potential problems caused by the roadway navigation failure can be dealt with by using the on-board vehicle navigation system to verify the correctness of the roadway navigation commands and to serve as a back-up to the roadway navigation. The reliability of the roadway navigation controller.

#### **Reliability of Roadway to Vehicle Navigation Commands**

In ERSC 5, the roadway can guide the vehicle to its destination by sending navigation commands to vehicles. The roadway receives the speed, position and destination data of all vehicles in the roadway network. The roadway uses the above information and the roadway network geometry to determines an optimal route for each vehicle to get to its destination. The optimal routes for vehicles are computed in such a way that the efficiency of the overall roadway network is maximized. The roadway sends navigation commands to the vehicle so that the vehicle can follow the route planned by the roadway. The roadway updates the planned routes frequently to cope with the roadway network traffic condition.

For safety concern, it is desirable that the roadway navigation commands be sent to the vehicle navigation system instead of the ALLC system. This allows the on-board navigation system to verify the reasonableness of the roadway navigation commands before they are forwarded to the ALLC system for execution. The ALLC system uses vehicle-to-vehicle communication and on-board

sensors to ensure safe execution of the navigation commands. Figure 94 shows the functional block diagram of the roadway navigation system interfacing with vehicles in the roadway network.



igure 94. The functional block diagram of the roadway navigation system.

The roadway navigation system is mainly composed of the roadway-to-vehicle communication and the navigation controller. The roadway-to-vehicle communication receives position, speed and destination data of all vehicle in the network and forwards the information to the navigation controller. The navigation controller computes on-line an optimal route for each vehicle. It also generates navigation commands for vehicles. The navigation commands are sent to vehicles through the roadway-to-vehicle communication.

## Reliability Functional Requirement

A failure in the roadway navigation system may result in sending wrong navigation commands to vehicles. Since the roadway navigation commands will be filtered by the on-board navigation system before they can be executed by the ALLC system, failures in the roadway navigation system impose no immediate danger to vehicles in the network. However, failures in roadway navigation may reduce the efficiency of the roadway network. For the roadway network efficiency, the reliability functional requirement of the roadway navigation system is proposed as

'The navigation commands sent to the vehicles should have low error rates.'

## System Level Design Requirements

To achieve the above reliability functional requirement, the roadway needs to have reliable roadwayto-vehicle communication to accurately collect the operational status of each vehicle and to transmit navigation commands. The roadway navigation controller needs to have efficient and reliable algorithms to process the enormous amount of data and to generate correct navigation commands for each vehicle. Some important system level design requirements can be identified by analyzing major potential system failure modes and their causes, based on the system framework shown in figure 94.

Potential Failure Mode 1: Roadway-to-Vehicle Communication failure

1.1 The position, speed and destination data of some vehicle(s) can not be received by the roadway or is corrupted during reception.

Design requirements:

- The roadway should verify that the reasonable position, speed and destination data from each vehicle is received in a timely manner. If an error is detected, the roadway navigation controller should use the previously available information to estimate. The roadway should also notify the vehicle(s) whose position, speed and destination data can be properly received by the roadway.
- The Roadway-to-Vehicle communication should provide communication protocol that contains error detection and correction codes to minimize the probability of incorrect message reception. The position, speed and destination data should be sent to the roadway frequently enough so as to minimize the effects of no communication.
- 1.2 Navigation commands can not be sent to the vehicles or are corrupted in transmission.

Design requirements:

 The roadway-to-vehicle communication should provide communication protocol that contains error detection and correction codes to minimize the probability of incorrect message transmission. The navigation commands should be sent to vehicles frequently enough so as to minimize the effects of no communication.

Potential Failure Mode 2: Roadway Navigation Controller failure

2.1 Roadway Navigation Controller computes incorrect navigation commands.

Design requirements:

- Navigation Controller should incorporate an independent check of the reasonableness of the computed navigation commands before they are sent to the Roadway-to-Vehicle Communication. If an error is detected, the navigation commands should be canceled and warnings should be sent to vehicles through the Roadway-to-Vehicle Communication.
- The Roadway-to-Vehicle Communication should incorporate an independent check of the reasonableness of the navigation commands provided by the Navigation Controller. If an error is detected, the navigation commands should be canceled and warnings should be sent to vehicles through the Roadway-to-Vehicle Communication.
- 2.2 Navigation commands are not communicated to the Roadway-to-Vehicle Communication at appropriate time or is corrupted in transmission through interface.

Design requirements:

- The Roadway-to-Vehicle Communication should verify that it is receiving updated navigation commands from the Navigation Controller in a timely manner. If an error is detected, the navigation commands should be canceled and warnings should be sent to vehicles through the Roadway-to-Vehicle Communication.

Issues and Risks

1. The roadway navigation needs to communicate on a one-to-one basis with a great number of vehicles in the roadway network. Designing a reliable roadway-to-vehicle communication system to properly transmit and receive huge amount of data will not be trivial.

2. The controller of the roadway navigation system has to process enormous amount of data received from vehicles. The hardware needed to implement the controller will be considerable so as to satisfy the capacity requirement. The software (control algorithms) run by the controller will be very sophisticated. These impose substantial difficulties in designing a reliable controller.

3. The on-board vehicle navigation system achieves user optimality, while the roadway navigation system strives for AHS system optimality. The required reliability makes the roadway navigation very costly. Unless the roadway navigation can reliably improve AHS efficiency to a satisfactory degree from ERSC 4, it will probably not cost effective.

### **Vehicle Navigation**

The reliability requirements of the on-board navigation system in ERSC 5 contain those in ERSC4 and the following additional requirements.

In ERSC 5, each automated vehicle receives navigation commands from the roadway. The roadway navigation commands are to maximize the efficiency of the entire roadway network. However, if the roadway navigation commands are directly forwarded to the ALLC system for execution, failures in the roadway navigation can cause unnecessary interruptions in the network traffic. It is desirable that the on-board navigation system should verify the reasonableness of the roadway navigation commands as shown in figure 95. The roadway navigation commands will be sent to the ALLC system for execution only if they are approved by the on-board navigation system. If a potential roadway navigation failure is detected, the roadway navigation commands should be discarded and replaced by the commands generated by the on-board navigation system.



Figure 95. The functional block diagram of the vehicle navigation system in ERSC 5.

#### **Roadway to Vehicle Speed and Headway Commands**

The reliability requirements of the roadway to vehicle speed/headway commands are the same as in ERSC 4.

#### **Roadway Lane Reference Aids**

The reliability requirements of the roadway lane reference aids are the same as in ERSC 4.

### **Automatic Lateral and Longitudinal Control**

The reliability requirement of the automatic lateral and longitudinal control system is the same as in ERSC 4.

### **Driver Tasks and Workload**

This section discusses the drivers' tasks in ERSC 5. It covers normal driving on the dedicated highway when all of the automated functions work perfectly. In ERSC 5 this consists of starting the automated systems and selecting a destination. This case is identical to ERSC 4. Then it covers the fall-back mode in which the vehicle switches to ERSC 4 control with maneuver coordination and selection performed by the vehicle. The driver has only the role of a passenger in this ERSC.

The driver drives the car to the automated lane. The driver turns on the vehicle's automated systems for longitudinal and lateral control. The vehicle enters the lane when a safe gap appears in the traffic flow. The vehicle steers into the lane and turns on the SHM function, collision avoidance system and the automatic steering function. The vehicle takes over the throttle, steering, and brake when the car is on the highway or as the vehicle enters the highway.

The vehicle turns on the turn signal to leave the automated lane. This initiates check-out procedures. Check-out returns the steering, brake, and throttle control to the driver. The vehicle then changes lanes out of the automated lane. The driver turns off the communications links when he leaves the automated lanes.

#### **Task Analysis–Normal Driving**

In normal operations, the vehicle controls the steering, throttle, and brake. The vehicle performs lane changes into and out of the dedicated lane. The vehicle also avoids lateral collisions. The driver selects a destination and turns on the automated systems on the vehicle. The roadway sends target speed commands and minimum headway information electronically to the vehicle. This data helps the vehicle compute a safe headway. The roadway also sends traffic data to the driver to help him/her select a route. The task analysis in table 40 shows the driver cues for this task.

Task	Select destination and turn on automated systems
General	Task similar to being a passenger in the vehicle
Initiating Cue	Enter Check-in point and turn on automated systems – The driver sets destination
Feedback	Situation display that shows vehicle location and status
Task Standards	Select destination before vehicle enters designated lane
Task Conditions	Weather, road conditions, or traffic
Skills Required	Choose destination
	Correctly turn on automated systems
Potential Errors	Choose wrong destination
	Turn on wrong systems
	Do not turn on proper automated systems
	Fail Check-in, enter highway anyway
	Vehicle chooses wrong or inefficient route
<b>Causes of Error</b>	Fails to perceive or misinterprets system feedback, believes system has
	malfunctioned
	Poor judgment of location
	Hit wrong button and overrides function
	Sensor, controller, or actuator failure on vehicle
	Roadway controller malfunctions
<b>Consequences of Error</b>	Disrupt traffic flow and lower capacity
	Collision with surrounding vehicles
	Disturbance due to sudden route changes
<b>Recovery Points</b>	Minimum safe headway setting on vehicle to avoid collisions
	Collision warnings
	Vehicle warns driver that headway maintenance function is not working or is
	failing or has been turned off; prevent easy override of SHM, ALK, and RECA
	functions
Individual Differences	Education
	Age
	Technology background
Criticality	Essential

#### Table 40. Driver task description for ERSC 5 driving.

#### Task Analysis–Fall-back Mode

In the fall-back mode, the communication between the roadway and the vehicle suffers a partial failure when one control channel fails, but the other one remains intact. The maneuver coordination system helps vehicle merge into the dedicated lanes, change lanes, and avoid collisions. The vehicles can then maneuver safely. The vehicles may also coordinate suggested lateral collision avoidance

maneuvers. This arrangement is similar to the TCAS system for commercial aircraft <sup>(96)</sup> If this system fails then the driver takes over the maneuver coordination task by pre-approving all maneuvers before the vehicle executes them.

#### Issues and Risks

Driver override of automated functions

Even with multiple sensors the vehicle may not see and recognize everything that a human driver sees. This is especially true of unexpected objects such as animals or debris on the roadway. It may be necessary for the driver to override the collision avoidance system on the vehicle to avoid these objects. The lack of an override may put the driver in an unsafe position. The existence of such an override may also put the driver

in an unsafe position if the driver brakes unnecessarily. Research needs to be done on whether or not this driver override should be allowed to occur.

## **Key Results**

### Benefits

The roadway controller proposed in ERSC for vehicle velocity control was local in nature. That controller aimed to improve local traffic conditions on a single stretch of road without coordination with other related systems such as ramp metering. The roadway controller is also isolated from other traffic control systems for route selection and navigation. Freeway network control uses real-time and predicted information to smooth traffic flow throughout the traffic network. This ERSC proposes an integrated controller that makes route and course recommendations and ramp metering decisions to achieve a common goal. The goal of freeway network control can be either user optimality or network optimality  $^{(71)}$  User optimality can be in terms of trip time or fuel expended. Network optimality can be in terms of overall travel time at the cost of some vehicles taking longer paths to meet their goals.

Simulation studies by Messmer and Papageorgiou indicate that there is a high potential for improving traffic flow in freeway networks in terms of travel time reduction and congestion avoidance or limitation  $^{(71,7)}$  More detailed controllers will select lanes and speeds for each vehicle on the highway.

#### **Issues and Risks**

1. Network freeway control needs extensive study and simulation before appropriate controllers can be selected. The system must also offer demonstrable improvements from the driver's existing ERSC 4 route selection system. Otherwise drivers will not use the network controller and will rely directly on their own vehicle to choose their route.

2. Driver overrides would allow individual drivers and vehicles to disregard commands from the roadway and use their own navigation systems. While this may benefit one driver, it may hurt overall traffic flow and cause safety problems if one vehicle makes many lane changes in an attempt to go a certain speed. The effect of having network guided vehicles and independent vehicles on the same traffic network needs to be studied.

3. Some researchers <sup>(111)</sup> have proposed network controllers that send control signals directly to each individual vehicle. This puts stringent performance and reliability constraints on the communications system since the messages are frequent and time-sensitive. System reliability considerations dictate that the communications system must be fail-safe with no single point failures for autonomous vehicles. This could lead to multiple independent communications systems.

4. Guided vehicles with wires or cables provide another option for direct roadway control since vehicle maneuverability is restricted.

## **Key Findings**

Network controllers may improve traffic flow through traffic networks in terms of travel time reduction and congestion avoidance or limitation. The communication system to the vehicles must be highly reliable since incorrect speed or lane change messages could cause congestion or unsafe situations.

## Section 10: Conclusions

## **Summary of Key Findings and Issues**

1. Communication systems and sensors such as radar will be used on vehicles in all stages of AHS deployment. When multiple radar operate in a small area at similar frequencies, the radar on different vehicles may interfere with one another. This interference could lead to shorter sensor ranges and poor performance. Such interference is a safety issue since the sensors or communications systems may not work as expected. This issue may also affect how the highways are built and what separation is required between lanes.

2. Vehicle-to-vehicle communication can reduce the number of false alarms in rear-end collision warning systems without reducing the detection rates. Each vehicle sends its braking data to the vehicle behind it. These communications can serve as a sort of electronic brake light for the vehicle behind. This lets the following vehicle anticipate the braking of the vehicle ahead. This warning can decrease the number of false alarms in the rear-end collision warning system. However, the government will need to set communication frequencies and protocols so that a common communication system can be used on all highways.

3. A roadway traffic controller can decrease traffic congestion even in case of low levels of automation. This controller eases roadway congestion by regulating vehicle speed and traffic density along sections of the roadway. This can smooth traffic flow and decrease backups due to incidents on the roadway. Studies at USC show considerable improvements in congestion reduction and stability of traffic flow by using appropriate roadway traffic controller.

4. Multiple types of driver warnings for lateral/longitudinal collisions will be installed on vehicles in the early stages of AHS deployment. Some examples are rear-end collision and lateral collision warnings. The driver may experience warning overload and warning confusion problems while interacting with other vehicle and roadway functions. More research needs to be done on the interaction and prioritization of these warnings for the driver.

5. Collision avoidance systems use braking and steering to avoid crashing into other vehicles or stationary obstacles in the lane. Under the current system, the vehicle manufacturer may be liable for collisions caused by vehicle failures. This may have an impact on the automobile manufacturer's willingness to produce automated vehicles. It may also affect the design and operation of the automated systems. The legal system may need to limit manufacturer's liability.

6. Collision avoidance requires combined lateral and longitudinal control. Our work proposes an evolutionary system for the development of AHS. In one partial automation scenario the driver can override the automatic lane keeping system but not the longitudinal control for the throttle. The driver may not be able to perform lateral collision avoidance maneuvers using steering control only. This issue needs more human factors studies.

7. Our original evolutionary design split the development of full lateral control into two parts. The first part was automated lane keeping. The second part was lane changing and lateral collision avoidance. These control modes should not be separated since the driver becomes a passenger once the vehicle takes over lane keeping. Human factors studies show that humans do not perform monitoring tasks well.

8. A roadway controller that sends speed commands to vehicles can smooth vehicle speeds and improve traffic flow when incidents occur. The vehicle can respond to the roadway controller commands automatically as proposed in the Prometheus project or the driver can change the vehicle's cruising speed. If the vehicle automatically responds to the speed commands from the roadway, the driver's reliability in detecting rear-end collision danger may be degraded. The driver may be put into a situation that he/she can not handle and may blame the roadway for any rear-end collision. The roadway may be liable for accident caused by incorrect speed commands. However, if the vehicle does not respond directly to roadway speed commands, the effectiveness of the roadway traffic flow control will be reduced since the driver may not follow the roadway commands.

9. At each stage of the AHS evolution, drivers are required to have a basic understanding about the system operation so they can interface with AHS safely and efficiently. Drivers may need to be trained to handle the necessary tasks and workload.

10. Vehicles require a distance and closing rate sensor for speed and headway maintenance or intelligent cruise control. This system can also warn the driver of potential rear-end collisions. The rear-end collision warning system computes the Time-to-Collision and warns the driver of a potential collision in time to avoid the collision (about 1.5 s before the collision). The sensors for this type of system only need to be accurate to within 3-5 km/hr.

11. As the AHS evolves to higher ERSCs, the driver 's functions are gradually replaced by the automatic control systems of vehicle and roadway. These systems needs to be more reliable than human drivers so that the AHS can be accepted by the public. They also need to be fail-safe to guarantee the safety of drivers. These reliability requirements may substantially increase the component redundancy and design complexity.

12. The reliability requirements for automatic systems in each ERSC can be used to assess the required component redundancy and design complexity. This gives us a way to estimate the gaps between ERSCs in terms of technical difficulties, cost and risks.

13. Safety and efficiency benefits in higher ERSCs can only be realized by heavily instrumented vehicles, using roadway lane reference aids and a roadway navigation system. It will take joint efforts of the public, the automobile manufacturers and the government to settle the potential legal and liability problems arising from AHS operation.

14. The communication requirements for AHS increase rapidly with the level of automation. When the vehicles have only longitudinal control, each vehicle needs to communicate only with the vehicle behind and in front. In a fully automated system the vehicles must communicate with the roadway and the surrounding vehicles to coordinate maneuvers. This change requires increases in bandwidth and reliability and more sophisticated communication protocols.

15. The human factors issues are very complex in the ERSCs that mix automated and manual control systems. Humans do not perform well as supervisors of automated systems. Systems that rely on a human driver who is not actively involved in lateral or longitudinal control to respond to infrequent warnings for collision avoidance or other emergencies may not work well. Once the vehicle can perform hands-off, feet-off driving in the dedicated lane, the human factors issues may be less critical to system design.

# ERSC 1 Key Findings

1 The speed and headway maintenance function does not have to be very accurate in this ERSC (5-10%) since it is limited by human perception and reaction times. This suggests that speed and headway maintenance systems at this level may be derived from the intelligent cruise control systems planned by several companies. (74,26) The next ERSC requires a more accurate speed and headway maintenance function. The components for a more accurate speed and headway maintenance function need to be designed with AHS requirements in mind instead of taking commercial technology off the shelf.

2 There are significant safety and performance benefits even with low levels of automation. Simple systems like the roadway velocity controller can give large increases in capacity and traffic flow volumes. This system uses technology available today in traffic control centers and toll roads.

3 Vehicle to vehicle communication systems can reduce the number of false alarms in collision warning systems without reducing the detection rates.

4 The driver can serve as a back up to the automated systems at this level. If the speed and headway maintenance or rear end collision warning systems fails, the driver takes over system performance.

5 There are significant human factors issues associated with warnings and driver interface that must be researched for even low levels of automation. The warning systems will require multiple levels of warnings.

6 The combined speed and headway maintenance-driver system needs to have mean time to failure much more than 50 years. The reliability functional requirement for the speed and headway maintenance system is that it must safely and reliably return control to the driver in case of override.

7 The reliability functional requirement for the warning systems is that they have high detection rates and low false alarm rates.

## ERSC 1 Issues and Risks

1. Controller Algorithms

The controller algorithms need to smoothly adjust the brake and throttle to avoid sudden accelerations when the vehicle ahead leaves the lane. Driver tests indicate that drivers feel unsafe when the controller suddenly accelerates. (26) The speed and headway maintenance controller needs to use algorithms that limit accelerations to make drivers feel comfortable and accept automatic control. (51,25)

2. Electromagnetic Interference

Full deployment of ERSC 1 means that most vehicles will have active sensors such as radar. When multiple radar operate in a small area at similar frequencies the radar may interfere with one another. This interference could lead to shorter sensor ranges and poor performance.
#### 3. Driver Interface

Research needs to be done on the driver interface with the speed and headway maintenance, rear end collision warning, and blind spot warning functions. Each function may have separate controls for enable/disable and adjustments. It may be too much of a burden on the driver to expect him/her to change lanes and enable the automated systems immediately. The driver may need to preset the target speed, minimum headway, and warning thresholds when he/she starts the vehicle.

#### 4. Sensor Capabilities

The rear end collision warning and blind spot warning functions may increase safety and reduce certain types of collisions by decreasing driver reaction time. But there are sensor and human factor issues that still must be resolved. The sensors must detect small vehicles or debris, such as motorcycles with few nuisance alarms.

#### 5. Warning system design

First the designers need to decide when the warning system should be turned on: always on, only when turn signal on, or turned on when steering column shows lane change. The warning thresholds for emergencies and driver awareness must also be set. The system must avoid nuisance alarms, yet detect vehicles accurately. The types of alarms and warnings must also be researched. The emergency alarm needs to be directional to get fast driver response. ERSC 1 introduces multiple types of driver warnings for blind spot and rear-end collision avoidance. Research needs to be done on the interaction and prioritization of these warnings for the human driver.

#### 6. Dedicated Lane

The biggest social issue for AHS is how to get a separate lane for automated vehicles. The driving public rebels when a lane is taken away from active use. <sup>(99)</sup> The solution for most highway departments has been to add a new lane on the shoulder or median strip, but most urban highways do not have any more space to spare for the sole use of automated vehicles. This infrastructure change may have to wait until enough vehicles have speed and headway maintenance systems and traffic flow benefits can be easily seen.

#### 7. Legal/Liability

In this ERSC the roadway sends recommended target speeds and minimum headways directly to the vehicles. This may make state and local governments liable for any accidents that may occur if the minimum recommended headway is too short. (102) The government may not wish to deploy such a system until it has been shown to be reliable and fail-safe.

#### 8. Privacy

This ERSC gives the vehicles the option to "Check-in" with the roadway. This may leave a record of when and where a vehicle traveled. This raises user privacy concerns about governmental use of this data.

#### 9. Communication Protocols

The government will need to set communication frequencies and protocols so that a common communication system can be used on all U.S. highways.

#### 10. Driver Reliability

The driver may become over-reliant on the speed and headway maintenance system and degrade his/her reliability in rear-end collision avoidance.

#### 11. Roadway Speed Commands

The vehicle's automatic response to the roadway's target speed command imposes safety concerns. For example, the speed and headway maintenance system headway/speed sensor fails while the controller is responding to a target speed command by accelerating the vehicle. In this case, the driver may perceive that the speed and headway maintenance system is properly following the roadway's command and fail to sense a rear-end collision danger in time. The driver's reliability as a back-up to the speed and headway maintenance system may be seriously degraded since the speed and headway maintenance operation puts the driver into a situation that he/she can not handle safely. This potential failure mode has serious reliability implications.

#### 12 Warning Overload and Confusion

The driver may experience warning overload and warning confusion problems while interfacing with other vehicle and roadway functions.

#### 13. Activation of Blind Spot Warning

If the warning signal of the BSW system is to be activated only by the turn signal or steering wheel motion, there might not be enough time for the driver to react to the danger. This is because some drivers may perform changing lanes without turning on the turn signal. Warnings may be too late if the driver has already started turning the steering wheel. If the warning signal is to be generated without activation from the driver, the warnings created by the adjacent lanes traffics moving into and out of the blind zone will be irritating to the driver.

# ERSC 2 Key Findings

1. The mean time to failure of the speed and headway maintenance and rear end collision avoidance system needs to be much more than 50 years.

2. The driver can not act as a back-up for an automated rear-end collision avoidance system that uses short time headways. In this case there is not enough time for the driver to react to dangerous situations. The reliability requirement for the rear-end collision avoidance system is: "Under no circumstances should a single component/point failure let the vehicle crash into any moving or stationary object in the lane, and there should be no common failure modes." One possible design leads to multiple control paths and sensors to ensure system reliability. This increases the cost and complexity of automated rear-end collision avoidance systems.

3. The redundant control channels in the p roposed speed and headway maintenance and rear end collision avoidance system framework need to be always kept active. This ensures safe and smooth deactivation of a control channel.

4. Driver warnings from the lane departure warning system and the blind spot warning system must be prioritized to avoid driver confusion if multiple warnings occur. Great care must be taken in ERSCs that combine automated systems with manual driving to make sure that the driver interface is simple to understand and operate.

# ERSC 2 Issues and Risks

1. Driver Confidence

The driver may not trust the collision avoidance system to handle emergency braking itself. This may cause the driver to choose larger headways than the automated system requires. This reduces the capacity of the automated lane. The driver is not a truly parallel system to the speed and headway maintenance function with rear-end collision avoidance since the driver has a much slower reaction time. Extra redundancy may be needed to make the collision avoidance system as reliable as a human driver.

#### 2. Legal/Liability

Rear end collision avoidance gives rise to new liability issues. If a vehicle collides with the vehicle in front, then who is at fault? The driver may set the time headway too small or the collision avoidance system may have not worked properly.

#### 3. Driver Attention to Driving Task

As the vehicle assumes more of the driver tasks, the driver may pay less attention to the other vehicles or the highway. This increases driver reaction time to emergencies requiring lateral collision avoidance. The type of feedback that the driver needs from the system needs to be studied.

#### 4. Rear-end Collision Avoidance Sensors

The sensors for rear-end collision avoidance must recognize hazardous situations quickly and accurately. More research needs to done to reduce the number of false alarms and improve the detection performance of these systems.

#### 5. Driver Override of Braking

The driver may need to override the automated braking system if he/she sees an object that the collision avoidance sensors do not. However, this could lead to dangerous situations and a loss of vehicular control. This issue needs further study.

#### 6. Driver's Reliability in Lateral Control

Complete feet-off driving may make the driver over-reliant on the vehicle and degrade the driver's reliability in lateral collision avoidance.

7. Cost

The required reliability of the speed and headway maintenance and rear end collision avoidance system imposes much complexity in the design. This increases the cost and technical difficulty in deploying this system.

#### 8. Fully automated longitudinal control with manual lateral control

When only the longitudinal control is automated, the speed/headway maintenance and rear-end collision avoidance system may have difficulties in avoiding crashing into stationary obstacles that suddenly appear in the lane. This can happen when, for example, the preceding vehicle abruptly changes lanes to avoid hitting a stationary object in the lane. In this case, the vehicle may not have enough distance to stop the vehicle without a collision. The probability of such collisions can be reduced if we increase the headway used in vehicle following control. However, the capacity of the AHS will be affected by increasing the headway.

#### 9 Warning Overload and Confusion

The driver may experience warning overload and warning confusion problems while interfacing with BSW, LDW and other vehicle and roadway functions.

# ERSC 3 Key Findings

1. The on-board-automatic lane keeping (A LK) system will require redundant active control channels for fail-safe design.

2. It may be impractical to have the driver perform lateral collision avoidance while giving him/her only steering override authority.

3. AHS with narrow lanes has the potential of increasing traffic capacity and real estate saving. However, the automatic lane keeping and lateral collision warning will be difficult to implement on a narrow dedicated lane.

4. A malfunction in the automatic lane keeping system will result in t he vehicle operation falling back from ERSC 3 to ERSC 2. However, a malfunction in the speed and headway maintenance and rear end collision avoidance system will cause the operation to fall back from ERSC 3 to ERSC 1 or 0.

5. Automatic lane keeping will reduce the collisions caused by vehicle lane departures. The most destructive lane departure accidents take place when the driver is fatigued during a long drive. Automatic lane keeping should help reduce this type of accident.

6. Lateral collision warning improves the safety of the changing lanes and merging. It also helps the driver for lateral collision avoidance.

7. Dedicated lane capacity can be improved by adopting higher speed and/or smaller headway since the driver is hands-off and feet-off in normal driving.

8. Roadway maneuver coordination improves the safety for entry and exit. It may also help reduce the effect of traffic flow disturbance caused by vehicles entering and leaving the lane.

9. The mean time to failure of the automatic lane k eeping system needs to be much longer than 150 years.

10. The driver can not back up the automatic lane keeping system if it fails. The reliability functional requirement of the automatic lane keeping system is: "Under no circumstances, should a single point/component failure let the vehicle depart from the lane, and there should be no common failure modes".

11. The operation of lane keeping control requires preview information about the roadway. The on-board automatic lane keeping (ALK) system obtains this data with the help of roadway lane reference aids. The sensing technologies used by the automatic lane keeping system determine the types of lane reference aids. Possible lane reference aids include vision based systems, magnetic nails, radar, and GPS with dead-reckoning. Some types of lane reference aids may not reliably provide preview information under certain environmental conditions such as rain or snow. Reliability studies show that two independent control paths may be necessary for reliable fail-safe lane keeping. Each control path needs to have an independent sensor type.

ERSC 3 Issues and Risks

#### 1. System Complexity

A potential automatic lane keeping system design to fulfill the reliability functional requirement will need two redundant, independent steering control channels. This substantially increases the cost of automated vehicles in ERSC 3 due to the design complexity and required redundant components.

#### 2. Reliability of Lane Reference Aids

The operation of lane keeping control requires preview information. The on-board automatic lane keeping (ALK) system obtains it with the help of roadway lane reference aids. The sensing technologies used by the automatic lane keeping system determine the types of lane reference aids. It might not be easy for some types of lane reference aids, such as vision based, to reliably provide preview information under certain environmental conditions.

#### 3. Cost

The sensors of the automatic lane keeping (automatic lane keeping) system interface with roadway lane reference aids to obtains lateral dynamics information. The reliability functional requirement implies that redundant, independent control channels are needed for the automatic lane keeping system design. The lateral dynamics sensors will be based on independent sensing technologies. Since a lane reference aid needs to be made compatible with a lateral dynamics sensors, there will be multiple lane reference aids. This may substantially increase the cost of the roadway infrastructure.

4. Availability of the automatic lane keeping function

Some lateral dynamics sensing technologies, such as vision based, may be sensitive to the environmental conditions. Due to the strict reliability requirement, the automatic lane keeping operation may be easily interrupted by poor weather. The reliability degradation in automatic longitudinal control may also affect the availability of the automatic lane keeping function.

#### 5. Driver's Lateral Collision Avoidance

Lateral collision avoidance may require combined lateral and longitudinal control. In ERSC 3, the driver can override the automatic lane keeping system but not the longitudinal control. The effectiveness of the driver's lateral collision avoidance, using steering control only, has not yet been verified and requires further study.

#### 6. Legal/Liability

Since the roadway is involved in the dedicated lane entry and exit coordination, the roadway may be liable for accidents if it fails to send out proper speed/headway commands to coordinate the entry/exit maneuvers. The driver may still blame the roadway for an entry/exit accident even it is caused by the his/her improper merging/demerging maneuvers.

#### 7. Legal/Liability of lane reference aids

The roadway may be liable for accidents caused by failures of the lane reference aids.

# ERSC 4 Key Findings

1. The driver will not be able to back up the automatic lateral and longitudinal control (ALLC) system if it fails. The reliability functional requirement of the ALLC system is: "Under no circumstances should a single component/point failure result in a lateral or longitudinal collision, and there should be no common failure modes.

2. A potential ALLC system design to achieve the reliability functional requirement will need two active redundant control channels with substantial doubling of components.

3. The fully automated vehicles perform maneuver coordination to further reduce accident rates and improve traffic flow rates. This will require complicated vehicle-to-vehicle communication and maneuver protocols.

# ERSC 4 Issues and Risks

#### 1. Cost

The required reliability of the ALLC system enormously increases the system cost and design complexity. The required instrumentation on the vehicle will be very costly.

#### 2. Legal/Liability

The roadway may be liable for traffic accidents due to its involvement in the vehicle lateral and longitudinal control, such as sending speed/headway commands, assisting maneuver coordination, and providing lane reference aids, etc.

#### 3. Cost

There are multiple dedicated lanes in ERSC 4. Since each lane needs to be equipped with redundant independent lane reference aids, the cost of the roadway infrastructure will be increased considerably.

#### 4. Reliability of Lane Reference Aids

If a dedicated lane has lane reference aids failure in some roadway sections, the AHS may have to prohibit vehicles from entering those sections. This will reduce the highway flow rate and cause traffic control problems.

# ERSC 5 Key Findings

1. Network controllers may improve traffic flow through traffic net works in terms of travel time reduction and congestion avoidance or limitation. The roadway to vehicle communication system must be highly reliable since incorrect speed or lane change messages could cause congestion or unsafe situations.

2. The roadway controller proposed in earlier ERSCs for vehicle velocity control was local in nature. That controller aimed to improve local traffic conditions on a single stretch of road without coordination with other related systems such as ramp metering. The roadway controller is also isolated from other traffic control systems for route selection and navigation. Freeway network control uses real-time and predicted information to smooth traffic flow throughout the traffic network. ERSC 5 proposes an integrated controller that makes route and course recommendations and ramp metering decisions to achieve a common goal. The goal of freeway network control can be either user optimality or network optimality. <sup>(71)</sup> User optimality can be in terms of trip time or fuel expended. Network optimality can be in terms of overall travel time at the cost of some vehicles taking longer paths to meet their goals.

3. Simulation studies by Messmer and Papageorgiou indicate that there is good potential for improving traffic flow in freeway networks in terms of travel time reduction and congestion

avoidance or limitation. <sup>(71,7)</sup> More detailed controllers will select lanes and speeds for each vehicle on the highway.

# ERSC 5 Issues and Risks

#### 1. Network Traffic Control Effectiveness

Network freeway control needs extensive study and simulation before appropriate controllers can be selected. The system must also offer demonstrable improvements from the driver's existing ERSC 4 route selection system. Otherwise drivers will not use the network controller and will rely directly on their own vehicle to choose their route.

#### 2. Driver's Navigation

Driver overrides would allow individual drivers and vehicles to disregard commands from the roadway and use their own navigation systems. While this may benefit one driver, it may hurt overall traffic flow and cause safety problems if one vehicle makes many lane changes in an attempt to go at a certain speed. The effect of having network guided vehicles and independent vehicles on the same traffic network needs to be studied.

#### 3. Communication

Some researchers <sup>(111)</sup> have proposed network controllers that send control signals directly to each individual vehicle. This puts stringent performance and reliability constraints on the communications system since the messages are frequent and time-sensitive. System reliability considerations dictate that the communications system must be fail-safe with no single point failures for autonomous vehicles. This could lead to multiple independent communications systems.

#### 4. Options

Guided vehicles with wires or cables provide another option for direct roadway control since vehicle maneuverability is restricted. Direct roadway control of free moving vehicles is not practical due to the communications requirements for controller commands. Controller commands need to be transmitted frequently and with little delay. This does not seem to be practical for a roadway communications system.

#### Appendix A: Communication Systems

Automated highway systems will integrate vehicular navigation, route guidance, vehicle location, and control systems. Direct communication between vehicle control systems will help vehicles drive safely at short time headways. Systems for vehicular navigation and vehicle location will allow the roadway to coordinate and smooth traffic flow. Our ERSCs combine these advances in road transport informatics.<sup>(33)</sup> with control algorithms.

Vehicle-roadway communication (VRC) links let the roadway traffic control center send navigation, route and control commands to the vehicles. The vehicle sends status and position data to the roadway. This communication link also supports check-in/check-out procedures on the automated highway. For the low level ERSCs vehicle-roadway communications occur infrequently over sections of roadway. Data can be sent between vehicles and the roadway with either a broadcast media such as radio or an advanced vehicle identification (AVI) system similar to those already used by some automatic toll collection systems. AVI systems uniquely identify vehicles as they pass specific locations. These systems may also include roadside or central computers to analyze or handle the data. AVI systems can be used for check-in and check-out on automated highways, enforcement of dedicated lanes, or congestion based pricing schemes.

Navigation aids provide turn-by-turn route guidance. <sup>(33)</sup> These systems use the Global Positioning System (GPS), proximity beacons, and dead-reckoning to determine the vehicle location. Electronic maps allow the system to select a route based on distance and traffic information.

Vehicle to vehicle communication links permit the direct exchange of information between vehicle control systems. This lets the control system anticipate the actions of other vehicles before change in speed, acceleration, or position can be sensed by on-board sensors. The communication link can also work in parallel with the sensors to speed response and reduce false alarms. Control systems require constant communication with short delays. Maneuver coordination between vehicles requires a two-way link to ensure that messages are received.

AHS applications for control, identification, and navigation have different communications requirements for message size and contents, maximum allowable delay, frequency of transmission, and connectivity (vehicle to vehicle, roadway to vehicle). We will look at message format and requirements for different systems at the different ERSCs. Next we will look at two methods for sending data between vehicles and between vehicles and the roadway: broadcast media such as radio or directional line-of-sight links such as infrared or ultrasonics. Finally we discuss why communications systems can improve detection performance for collision avoidance.

#### **Specification of Communication Requirements**

Communication needs are specified with message content and length, the allowable delay, the frequency of transmission, the error rate and the type of communications needed.

Message Structure: Figure 96 shows the general structure of a message between cars. Most of the bits in a message are overhead. Synchronization bits allow the receiver to synchronize to an incoming message (16-32 bits). <sup>(100)</sup> The destination ID tells who should receive the message. The source ID tells who sent the message. The size of the ID fields may be quite large if each car in the United States is uniquely identified. There were 193 million cars in the U.S. in 1990 (2<sup>28</sup> = 268

# million).<sup>(77)</sup> This number can be reduced if the roadway gives each vehicle a local ID number over a section of roadway, possibly at each check-in and check-out point. Data is the variable-length contents of the message. The message ends with an error detection code. These codes check the parity of the message to look for bit errors. If the system detects an error the message is retransmitted or ignored. Error coding can help the receiver detect transmission errors. Larger numbers of bits increase the chance of detecting an error. <sup>(100)</sup>

$$P(\text{Undetected Error}) = (1 - P(s))2^{-M} = 2^{-M}$$

(A-1)

P(s) is the probability of successful reception of a message. M is the number of error detection bits.

Synchronization	Destination ID	Source ID	Message Type	Data	Error Detection
					Codes

# Figure 96. Structure of message for communications between vehicles and between vehicles and the roadway

Acceptable Delay: The delay is the time after which the received information is so late that it is no longer useful to the system. Communications between vehicle controllers may have short delays on the order of 0.05 seconds. Communications between vehicles and the roadway may have delays on the order of seconds for the roadway traffic controllers.

Transmission Speed: The transmission speed is a function of the message length, the number of messages sent by each vehicle, the number of vehicles that transmit per second and the delay. This decides the number of bits/second that the system needs to transmit. The number of vehicles on the highway and the range of the communication system give a bound on the amount of message traffic. The number of vehicles within the range of the communications system  $n_v$  is:

$$n_{v} = n_{l} \left[ R_{t} / \left( L + S_{d} \right) \right] \tag{A-2}$$

n<sub>l</sub> is the number of automated lanes.  $R_t$  is the range of the transmitter with an omnidirectional antenna. L is the average vehicle length.  $S_d$  is the average headway between vehicles. Each vehicle spends  $T_c = R_t / V$  seconds in the zone of the communications system. The communications link must be able to handle all the required traffic while the vehicle is in this communications zone.

Error Rates: The bit error rate is the probability that one bit is received incorrectly. The probability of message error  $P_m$  for a single bit modulation scheme with independent bits is <sup>(97)</sup>:

$$P_{m} = \sum_{j=1}^{M} {\binom{M}{j}} P_{be}{}^{j} (1 - P_{be})^{M-j}$$
(A-3)

M is the number of bits in the message.  $P_{be}$  is the probability of bit error. j is the number of errors in the message. This assumption is valid for fast-fading radio channels. The number of ways in which j bits out of M may be in error is:

$$\binom{M}{j} = \frac{M!}{j! (M-j)!}$$

For a message length of 50 bits and  $P_m = 10^{-4}$ ,  $P_{be} = 10^{-5}$ . More complex modulation schemes give different results for the bit error probability. The bit error rate depends on the average received signal to noise ratio per bit  $(E_b/N_o)$ .<sup>(97)</sup> The dimensionless ratio  $(E_b/N_o)$  is a standard measure for digital communications system performance. The equation for the received power can also be modeled as (85; 97).

$$P_r = P_t L_t G_t L_p G_r L_r \tag{A-4}$$

 $L_t$  and  $L_r$  are the losses at the transmitter and receiver.  $G_t$  and  $G_r$  are the gains at the transmit and receive antennas.  $P_t$  is the transmitted power.  $L_p$  is the path loss. The path loss models the propagation environment, range, and conditions.

Types of communication: The communication links can be acknowledged or unacknowledged. Important links for maneuver coordination require acknowledgments that the message has been received correctly. The security of the data link must also be specified. Certain types of vehicle to roadway communications such as vehicle identification and location must be secure to prevent unauthorized use of the data. The type of addressing tells if the communication link is one-to-one between vehicles or one-to-many such as between the roadway and other vehicles.

#### **Directional Line-of-Sight**

Vehicle-vehicle communications assist in real-time longitudinal and lateral control and maneuver coordination. There are two approaches to vehicle-to-vehicle communications. The first is the directional line-of-sight system shown in figure 97 a. These systems have a directional beam that only works when one vehicle sees another. Examples of this technology are ultrasound, microwave, and infrared. Both ultrasound and infrared have short ranges and tight power budgets. They have very high path losses in rainy or snowy conditions as well. For lower speed data networks the infrared can operate over longer ranges. This type of infrared link has been tested in the California PATH project and in the PROMETHEUS project. <sup>(31; 33)</sup>

The directivity of the link is an advantage when it reduces interference between vehicles. The directivity also requires precise alignment between the receiver and transmitter. This may require more accurate lateral control to work on the highway. This technology may be best suited for longitudinal control problems where one vehicle follows the other.



Figure 97. Possible architectures for vehicle to vehicle communications systems. (a) shows a direct point-to-point link between vehicles. (b) shows a system where the roadway acts as a relay between the two systems.

**Broadcast Systems** 

Broadcast technology such as radio can perform communications between vehicles and vehicles and the roadway. In this system a vehicle or the roadway sends a signal out in all directions at a given bandwidth. This signal tends to have a long range. This can cause interference if more than one car talks at once in the same frequency band. Broadcast systems use coding in time, frequency, or both to avoid interference and allow multiple users to share the same channel as shown in figure 98. This shared usage can increase channel capacity. <sup>(97)</sup>

The roadway may communicate with the automated vehicles with a cellular system along the

roadway or with subcarriers on existing FM or TV signals. <sup>(16)</sup> Cellular systems break the transmission area down into smaller broadcast "cells." Each base station is in the center of a cell. Figure 99 shows the architecture of this cellular system. Adjacent cells operate at different frequency bands. Physically separated cells reuse frequencies. Cell size ranges from tens of meters to 1-2 kilometers. Efficient use of spectrum is needed since there are many competing usages for wireless communications. <sup>(104)</sup> The vehicles and roadway communicate with each other through a base station. For example vehicles in an adjacent lane may send messages to one another through the base station. The destination address may designate vehicles on certain sections of roadway by lane number.

The technology for simple vehicle to roadside communications exists in some of today's automatic toll collection systems. <sup>(38; 53)</sup> An interrogator sends out a pulse. The vehicle transponder responds with the vehicle identification number and some data from a variable register. Future systems must allow for longer and more frequent messages.

TDMA and CDMA systems conserve frequency bandwidth. A typical TDMA protocol has a fixed frame length ( $T_f$ ) during which the interrogators and transponders communicate. <sup>(97)</sup> The frame has three active segments. In the first segment the reader sends a control message to activate the transponders within transmission range and send them instructions. The second segment is for data exchange. This segment has a variable number of slots for messages ( $n_s$ ). The third segment the transponders respond to the interrogation pulse by sending their identification randomly placed in one of  $n_{ID}$  slots. The reader then selects which transponders can send messages. The number of frames offered to a vehicle in a cell is:

$$n_f = \frac{T_c}{T_f}$$

The number of vehicles in the interrogator range determine how often each vehicle can successfully request a message slot  $P_c$ :

$$P_{c} = \left(\frac{n_{ID} - 1}{n_{ID}}\right)^{n_{v} - 1}$$

The average number of frames that it takes to successfully send a message request is  $N_{ave}=1/P_{c.}^{(85)}$ . This gives an upper bound on system capacity since each transponder times out for a time interval after transmitting its data.



Figure 98. Different types of signal coding to avoid interference. (a) Frequency division multiplexing (FDMA'b) Time division multiplexing (TDMA'c) Code division multiplexing (CDMA).



Figure 99. Cellular communication system for VRC. Each cell has a base station (black dot) that communicates with the vehicles within its range.

CDMA has parallel bands of TDMA time slots. Each user follows a coded sequence to choose which time and frequency slot to use. This system provides security and interference resistance since the signal moves in time and frequency. It also requires more complex hardware and switching.

#### **Communications and Detection Theory**

Collision warning systems and emergency maneuvers require that fast and accurate decisions be made about what to do and what to tell the driver. Sensors alone can detect the direction and speed of other vehicles but the vehicle can only guess at the intent of the other vehicles. With communications between vehicles this need of guessing may disappear.

The rear-end collision warning and avoidance systems demonstrate this problem. In a system with no vehicle to vehicle communications the following vehicle must detect if the vehicle ahead is braking and how much it is braking. The rear-end collision warning system in the vehicle warns the driver that a rear-end collision may occur. The driver then brakes. The collision warning system must warn the driver when a collision may occur without false alarms. High rates of false alarms can cause the driver to ignore the warnings or turn off the system. If vehicle to vehicle communications is present then the vehicle ahead sends its braking data to the vehicle behind. This data serves as an electronic brake light for the vehicle behind.

For example, if there is no communication between vehicles then the follower vehicle must wait until the lead car begins to brake and slow down before it can detect an emergency stop Æ t<sub>d</sub>:

 $\Delta t_d = b + t_{det}$ 

 $\tau_b$  is the braking actuation delay for the preceding vehicle.  $t_{det}$  is how long the vehicle behind needs to measure the closing rate until it can decide that the vehicle ahead is slowing. The figure below illustrates this problem. The dashed line shows the actual deceleration curve of the lead vehicle relative to the following vehicle. The solid line shows the actual sensor measurements taken by the following vehicle. The measurement noise is due to sensor and controller errors. Due to this noise the rear-end collision warning or avoidance system cannot decide that a collision will occur until it is sure that it is measuring an actual deceleration, not noise. If communication between vehicles is present, the detector only needs to confirm the information sent by the vehicle ahead. This takes less time than a detector working alone.



Figure 100. The dashed line shows the actual deceleration curve of the lead vehicle relative to the following vehicle. The solid line shows the actual sensor measurements taken by the following vehicle. The measurement noise is due to sensor and controller errors.  $t_{fa}$  shows a false alarm situation.  $t_{miss}$  shows a case where the detector misses the emergency braking condition.  $t_{det}$  shows when the detector can safely decide that the vehicle is braking with a low probability of false alarm.

This is equivalent to a binary detection problem between two hypotheses:  $S_0$  and  $S_1$ .  $S_0$  is the case where the car ahead is not decelerating.  $S_1$  is the case where the vehicle ahead brakes at an acceleration  $a_{bmax}$ . This gives the detection problem: <sup>(110)</sup>

$$\begin{array}{l} S_0: \ \mathcal{E}v + n \\ S_1: \ \mathcal{E}v + a_{bmax} \, t + n \end{array}$$

Æv is the closing rate between the vehicles. t (t is the time since the beginning of the lead vehicle stopping maneuver. n is the noise of the sensor measurements and the vehicle speed controller. As with any binary decision problem there are four possible outcomes that are summarized in the table below. The goal is to minimize the number of false alarms and maximize the number of correct detections. These parameters determine the performance of the detector.

Table 41.	Possible	decisions	of a	binary	detector.
-----------	----------	-----------	------	--------	-----------

True Condition	Decision by Detector	Cost of	Description
		Decision	
S <sub>0</sub> : Vehicle ahead not braking	S <sub>0</sub> : Vehicle ahead not braking	C <sub>00</sub>	Correct Decision, null condition
S <sub>0</sub> : Vehicle ahead not braking	S <sub>1</sub> : Vehicle ahead braking	<i>C</i> <sub>10</sub>	Incorrect decision; false alarm
S <sub>1</sub> : Vehicle ahead braking	S <sub>0</sub> : Vehicle ahead not braking	<i>C</i> <sub>01</sub>	Incorrect decision; missed detection
S <sub>1</sub> : Vehicle ahead braking	S <sub>1</sub> : Vehicle ahead braking	<i>C</i> <sub>11</sub>	Correct Decision, braking detected when the vehicle ahead brakes, Detection Rate

For this analysis, we assume that the noise is Gaussian with zero mean and a variance n. The probability of correctly deciding to brake  $P_D$  is <sup>(110)</sup>

$$P_D = \operatorname{erfc}\left(\frac{\ln}{d} - \frac{d}{2}\right)$$

The probability of deciding to brake when the car ahead is not decelerating  $P_{FA}$  is <sup>(110)</sup>

$$P_{FA} = \operatorname{erfc}\left(\frac{\ln}{d} + \frac{d}{2}\right)$$

d is the normalized distance between hypothesis S<sub>0</sub> and S<sub>1</sub>: d = at/n. Figure 101 shows the geometric interpretation of d. erfc is the complementary error function:

$$erfc(x) = \frac{1}{\sqrt{2}} \int_{x}^{\infty} \exp\left(-\frac{x^2}{2}\right) dx$$

is the Bayes cost function that gives the cost for false alarms, correct and missed detections (110):

$$\underline{\Delta} \; \frac{P_0(C_{10} - C_{00})}{P_1(C_{01} - C_{11})}$$

 $P_0$  and  $P_1$  are the a priori probabilities that the vehicle ahead is braking or not braking. In most cases these probabilities are unknown and set to 1/2. The terms  $C_{00}$ ,  $C_{10}$ ,  $C_{01}$ ,  $C_{11}$  are the costs of each possible decision listed in table 41. For example in a collision warning system the cost of a false alarm  $C_{10}$  is high since humans will quickly learn to ignore repeated false alarms. This gives a higher value for  $\therefore$  In a collision avoidance system where the vehicle is responsible for emergency stops the cost of a missed detection  $C_{10}$  is high  $\therefore$  This gives a lower value for  $\therefore$ 



Figure 101. The curve centered at 0 is the null hypothesis that assumes that the vehicle ahead is not executing an emergency stop. (a) In this case, the vehicle has just begun to brake, the hypotheses are not far enough away (i.e. *d* is small) to make a good decision. (b) The vehicle has been braking for some time(i.e. *d* is large) and the detector can make a correct decision with a low probability of false alarm.

Page 304

Figure 102 shows the relationship for P<sub>D</sub> and P<sub>FA</sub> with respect to d. The curves change with  $= \infty$  corresponds to the case where the receiver always decides no acceleration. = 0 corresponds to the case where the receiver always decides that acceleration is present. This shows that as t increases for constant noise and accelerations the performance of the detector improves. A good rear-end collision detection system needs a low false alarm rate and a high probability of detection.  $^{(64)}$  This may require a d on the order of 4 or higher. For example if the maximum deceleration of the lead vehicle is 8 m/s<sup>2</sup> and the noise deviation is 0.5 m/s then t = 0.25 s gives d=4. d=5 gives t=0.31 s.



Figure 102. The probability of correctly deciding that the vehicle ahead is decelerating depends on the space between the two signals *d*. As *d* increases the detection-false alarm trade-off improves.

#### **Appendix B: Roadway Traffic Controller**

For manual driving, the traffic flow is often irregular and unstable. This leads to congestion and to low traffic flow rate during minor incidents. These irregularities are due to variation of vehicle speeds and headways that are selected by the individual drivers. Congestion could be reduced and traffic flow rate could be increased, if the speed and density of the vehicles along a lane can be properly controlled by a roadway controller. In this paper we propose a roadway controller and investigate the benefits of controlling the density and speed of the vehicles along a lane by using the roadway controller. The roadway controller sends the appropriate speed commands to vehicles in each section of the lane. The vehicles are equipped to follow each other automatically and respond to the roadway commands without any driver intervention. A macroscopic traffic flow model is used for analysis and simulations. Our results demonstrate significant benefits in terms of smooth and stable traffic flow that reduce congestion considerably.

#### Introduction

Congestion is nowadays one of the main problems in urban transportation all over the world. Reduction of congestion is an important objective not only for ensuring shorter and more reliable travel times, but also in reducing its indirect consequences on pollution and fuel consumption, thus improving the quality of life.

Traditionally, on-ramp metering control strategies are employed to improve overall freeway operation by limiting, regulating and timing the entrance of vehicles from one or more ramps onto the main line Variable message signs are also an alternative way adopted at road side to set speed limits and to advise drivers of conditions on the road ahead or to advise them on the best route to follow and reach their destination. The recently developed concepts of automated highway system provide a direct control of overall freeway operation through microscopic vehicle following or platoon control laws and macroscopic traffic control laws. Research results on microscopic vehicle following or platoon control laws can be found in many papers <sup>(18,42,50,95,130)</sup>. In this paper, we concentrate on the design of roadway controllers for traffic control on the macroscopic level.

An important function of macroscopic roadway controllers is to generate the appropriate speed commands to be received by the vehicles at the various sections of a highway system. It is well known that the inhomogeneity of the traffic density distribution is the main cause of the formation of congestion and congestion amplification. Therefore, in normal traffic condition, the command should be generated so that the density distribution along the highway is uniform leading to higher and smoother traffic flow. In case of incidents the roadway controller should be able to control the traffic density distribution in an effort to minimize the effect of the incident on traffic flow.

Current research conducted in macroscopic traffic control emphasize the design of macroscopic control strategies to operate overall systems with homogeneous traffic density <sup>(89)</sup>. (The purpose of these macroscopic control laws is to avoid the formation of congestion or to avoid congestion amplification which are mainly caused by the traffic inhomogeneity.) strongly coupling nature of traffic models, the macroscopic traffic controllers proposed in the literature are still remained in the category of {\bf strategy planning} approach. No theoretical support is offered by these controllers. The efficiency of these controllers are either demonstrated by simulation results or by experimental results conducted in some particular highways. In other words, the validity of these controllers is still

based on case-by-case experience. The design of a theoretic based macroscopic controller is still an open problem in this field which needs further research effort.

Seeing these problems and in order to provide more reliable solution, we propose a roadway traffic density controller. The proposed roadway traffic density controller is capable of operating the overall highway system not only with a uniform traffic density distribution (in normal traffic condition), but also with a proper (which may not be uniform) density distribution (in case of incidents) determined by a higher level network controller based on overall network considerations.

This appendix is organized as follows. Firstly, a discrete traffic flow model is reviewed and the problem statement is given. Secondly, a roadway traffic controller is proposed. The implementation issue is then discussed and simulation results are presented.

#### **Traffic Flow Model**

Traffic models describe traffic behavior in terms of appropriate aggregated traffic variables. Due to the analogy between the mathematical description of traffic flow and fluid dynamics, the first traffic flow model was proposed by Lighthill and Withal <sup>(122)</sup> based on kinematic wave theory. In this model traffic density is the only state variable which results in poor transient behavior. Payne <sup>(126,127)</sup>, Cremer and May <sup>(119)</sup> proposed several modifications to overcome this problem. A much more sophisticated model was proposed by Papageorgiou <sup>(123,124)</sup> which has been tested, validated and reported with excellent results in the Boulevard Peripherique in Paris <sup>(123,124)</sup>. However, Papageorgiou's model exhibits several unrealistic phenomena. Due to these concerns, Karaaslan, Varaiya and Walrand <sup>(56)</sup> proposed a modified model to eliminate these unrealistic phenomena. In the following, a more detail description about this modified model is given.

Consider a freeway system which is subdivided into N sections with lengths  $L_i$ , (i = 1, ..., N). For a discrete time KT, where T is the sampling time interval, we define the following space-time discretized traffic variables:  $k_i(n)$  is the number of vehicles in the freeway section i at time nT divided by the length  $L_i$  of the section;  $v_i(n)$  is the space mean speed of vehicle in the freeway section i at time nT;  $q_i(n)$  is the number of vehicles leaving sections i during the time period [nT,(n+1)T], dividing by T; and  $r_i(n)$  and  $s_i(n)$  (No. vehicles/hour) are on-ramp and off-ramp volumes of section i respectively.

The modified freeway traffic flow model was in the following form:

$$q_i(n) = k_i(n)v_i(n) + (1 - )k_{i+1}(n)v_{i+1}(n)$$
(B.1)

$$k_{i}(n+1) = k_{i}(n) + \frac{T}{L_{i}}[q_{i-1}(n) + r_{i}(n) - s_{i}(n)]$$
(B.2)

$$v_{i}(n+1) = v_{i}(n) + \frac{T}{L} \{ V_{e}[k_{i}(n)] - v_{i}(n) \}$$
  
+ 
$$\frac{T}{L_{i}} \frac{k_{i-1}(n)}{k_{i}(n) + v_{i}(n)} [\sqrt{v_{i-1}(n)v_{i}(n)} - v_{i}(n)] - \frac{(n)T}{L_{i}} w_{i}(n)$$
(B.3)

where

$$(n) = \begin{cases} \frac{1}{k_{jam} - k_{i+1}(n) + 1} & \text{if } k_{i+1}(n) > k_i(n) \\ \frac{1}{2} & \text{otherwise} \end{cases}$$

and  $\alpha$ ,  $\rho$ ,  $\sigma$ ,  $\kappa'$ ,  $\tau$ ,  $\mu_1$ ,  $\mu_2$  are positive constants;  $V_e[k_i(\cdot)]$  represents the steady-state speed-density characteristics;  $w_i(\cdot)$  represents the average influence of vehicles' response on the mean speed evolution at sampling time nT in section i.

Typical parameter values are:

$$v_{f} = 93.1 km / h; k_{j} = 110 veh / km / lane; l = 1.86; m = 4.05;$$
  
= 0.95; = 40 veh / km / lane; ' = 4 veh / km / lane;  
\_{1} = 12 km^{2} / h; \_{2} = 6 km^{2} / h; = 120 veh / km / lane;  
= 35 veh / km / lane; and = 20.4 sec. (B.4)

The physical meaning of each term of equation (B.3) which influences the mean speed of a section  $\frac{T}{V_e[k_i(n)] - v_i(n)}$ is the relaxation term which includes

can be interpreted as follows The term is the relaxation term which includes the speed-density characteristics as a desired value according to the current density  $k_i(n)$ .  $V_e(k_i)$  denotes the steady-state speed-density characteristics. For homogeneous traffic condition on todays freeway, a fairly general formula for the steady-state speed-density relationship is given by

$$V_{e}(k_{i}) = v_{f} (1 - (\frac{k_{i}}{k_{jam}})^{l})^{m}$$
(B.5)

where l>0 and m>1 are real-valued parameters;  $v_f$  is the free speed; and  $k_{jam}$  denotes the traffic density at traffic jam. The free speed  $v_f$  represents the human driving characteristic of the particular road under consideration and its value can be estimated by calibrating with real traffic data. For a fully automated highway system under homogeneous heavy traffic conditions, the steady-state speed-density characteristic depends on the specified safety policy between two consecutive vehicles. For example, if the specified safety policy is a function of speed, i.e.

$$S_d = S(v)$$

(where  $S_d$  is the designated safety distance to be kept for two consecutive vehicles), then the steadystate speed-density relationship corresponding to this safety policy is

$$V_e(k_i) = S^{-1}(\frac{1}{k_i})$$
(B.6)

where  $S^{-1}(\cdot)$  is the inverse function of  $S(\cdot)$ , i.e.,  $x=S^{-1}(y)$  satisfies the equation S(x)=y. The term  $\frac{T}{L_i} {}_{i}(n)[v_{i-1}(n) - v_i(n)]$ is the convection term. It represents the influence of the incoming traffic on T

the mean speed evolution in segment i. The term  $-\frac{1}{L_i}w_i(n)$  is the anticipation term. It describes the driver response to the down stream density. For example, if the density downstream is lower, drivers tend to speed up and vice versa. For today's freeway system,  $w_i(n)$  can be approximately represented by

$$w_i(n) = \frac{k_{i+1}(n) - k_i(n)}{k_i(n) +};$$

It represents the influence of the traffic density downstream on the mean speed evolution. For an automated highway system, where the human driver has been replaced by an automatic control system, the anticipation term will be greatly affected by the adopted automatic control strategy and automated highway architecture. A suitable control strategy to determine the anticipation term is important for achieving high capacity and smooth traffic flow.

We use equations (B.1), (B.2), (B.3), (B.5) with

$$w_i(n) = \frac{k_{i+1}(n) - k_i(n)}{k_i(n) +};$$

to represent the traffic dynamics of current highway system. The traffic dynamics of automated highway system was represented by (B.1), (B.2), (B.3), (B.6) with a properly designed  $w_i(n)$ .

#### **Problem Statement**

To reduce congestion, the traffic management center of the automated highway system should be capable of providing the strategy to properly guide and organize vehicles. One of the possible strategies is to directly control the density and speed of the vehicles along a lane by sending the appropriate speed commands to each vehicle of the lane.

Assume that the roadway has the capability of measuring mean speeds and traffic densities at each section of a lane. The tasks of the traffic management center should be able to assess the status of the traffic and provide the appropriate speed commands to the vehicles at the various sections of the lane. For this purpose, a roadway controller should be designed to perform these tasks. The roadway controller should generate the appropriate speed commands to be received by the vehicles at the various sections of the lane. The command should be generated so that the density distribution along the lane is uniform leading to higher and smoother traffic flow. In case of incidents the roadway should be able to control the traffic density distribution in an effort to minimize the effect of the incident on traffic flow. Therefore, to be able to address the situations described above, a roadway

traffic density controller is proposed in this section. The proposed roadway traffic density controller is designed to provide proper speed command to each vehicle of the lane such that the density distribution of the lane is able to track a desired distribution determined based on overall traffic considerations.

Consider a lane which is subdivided into N sections with lengths  $L_i$ , (i = 1, ..., N). The traffic flow volume entering section 1 at sampling time nT is  $q_0(n)$  No. vehicles/hr. Based on overall traffic considerations, the desired traffic density distribution of the freeway is assumed to be determined by a higher level controller. (For example, it is known that inhomogeneous traffic density distribution is the main cause of the traffic congestion. Therefore, the higher level controller may assign a homogeneous density distribution for the dedicated lane to avoid the formation of congestion.) Let us denote the desired or optimal traffic density of section i at sampling time n as  $k_{di}$  (n).

We assume that the following condition is satisfied for each section at any sampling time.

#### Assumption

The traffic flow controllability is satisfied, i.e.,
 There exists a positive constant δ\* such that the following holds:

$$k_{i}(n+1) = k_{i}(n) + \frac{T}{L_{i}} [k_{i}(n)v_{i-1}(n) + (1-2)k_{i}(n)v_{i}(n) - (1-)k_{i+1}(n)v_{i+1}(n)]$$
  

$$\geq * > 0 \qquad \forall i, n,$$

The above condition is referred to as the condition of traffic flow controllability. If this condition is violated at sampling time n in section i, the traffic density at sampling time n+1 in section i will be less than  $\delta$ . This situation can be interpreted as not enough vehicle/traffic will be in this section at sampling time n+1; and hence, there is not enough vehicles/traffic to be controlled in this section. In this situation, we say that the traffic is loss of control and "traffic control" is meaningless in this section; and hence, the control law at this section can be switched off. Therefore, the assumption of traffic flow controllability is physically reasonable and acceptable.

Our objective is to choose a proper value of  $w_i(n)$  for section i at sampling time n such that the traffic density at section i converges to the desired traffic density  $k_{di}$  exponentially, i.e.,

$$k_i \rightarrow k_{d_i}$$
 as  $n \rightarrow \infty$ 

exponentially fast.

## **Roadway Traffic Density Controller**

In this section, we propose a macroscopic roadway traffic density controller for a single lane without on-ramp and off-ramp traffic.

Let the on-ramp and off-ramp traffic flow rates are zero, i.e.,  $r_i(n)=s_i(n)=0$ , and define

$$k_{i}(n) = k_{i}(n) - k_{d_{i}}(n)$$
  
$$k_{i}(n) = \frac{T}{L_{i}} [q_{i-1}(n) - q_{i}(n)] - k_{d_{i}}(n+1) + k_{i}(n) - c \quad (n)$$

From (B.1)-(B.3), we have

$$_{i}(n+1) = c_{i}(n) + _{i}(n)$$
  
 $_{i}(n+1) = c_{i}(n) + _{i}(n)$ 

where

$$\begin{split} {}_{i}(n) &= e_{i}(n) + \frac{(n)T}{L_{i}} [\overline{a}_{i}(n)w_{i-1}(n) + \overline{b}_{i}(n)w_{i}(n) + \overline{c}_{i}(n)w_{i+1}(n)] \\ a_{i}(n) &= \frac{T}{L_{i}} k_{i}(n) \\ b_{i}(n) &= \frac{T}{L_{i}} (1-2)k_{i}(n) \\ c_{i}(n) &= \frac{-T}{L_{i}} (1-)k_{i+1}(n) \\ d_{i}(n) &= k_{i}(n) - c_{i}(n) - k_{d_{i}}(n+1) \\ \overline{a}_{i}(n) &= a_{i}(n+1), \overline{b}_{i}(n) = b_{i}(n+1) \\ \overline{c}_{i}(n) &= c_{i}(n+1), \overline{d}_{i}(n) = d_{i}(n+1) \\ f_{i}(n) &= v_{i}(n) + \frac{T}{4} \{V_{e}[k_{i}(n)] - v_{i}(n)\} \\ &+ \frac{T}{L_{i}} \frac{k_{i-1}(n)}{k_{i}(n) + v_{i}(n)} [\sqrt{v_{j-1}(n)v_{i}(n)} - v_{i}(n)] \\ e_{i}(n) &= -c_{i}(n) - \overline{a}_{i}(n)f_{i-1}(n) + \overline{b}_{i}(n)f_{i} \\ &+ \overline{c}_{i}(n)f_{i+1} + \overline{d}_{i}(n) \quad \forall i = 2, 3, \cdots, N-1 \\ e_{N}(n) &= -c_{N}(n) - \overline{a}_{N}(n)f_{N-1}(n) + \overline{b}_{N}(n)f_{N} + \overline{d}_{N}(n) \\ c_{\eta} &> 0, \ c_{\xi} &> 0, \ |c_{\eta}| < 1, \ |c_{\xi}| < 1 \end{split}$$

Lemma 1

Consider the following discrete time system

$$z(n+1) = cz(n) + u(n), \quad z(0) = z_0$$

where c is a constant and |c| < 1, then,

$$u(n) \rightarrow 0$$
 exponentially implies  $z(n) \rightarrow 0$  exponentially.

Proof: The proof is trivial and is omitted.

For section i, at sampling time nT, if  $w_i(n)$  is designed such that

$$\kappa_i(n) = 0 \quad \forall i, n,$$

then, from Lemma 1, since  $\kappa_i(n) = 0$  and  $|c_{\eta}| < 1$ , we have  $\eta_i(n) \to 0$  as  $n \to \infty$ . Moreover, since  $|c_{\xi}| < 1$  and  $\eta_i(n) \to 0$  as  $n \to \infty$ , we have  $\xi_i(n) = k_i(n) - k_{di}(n) \to 0$  as  $n \to \infty$ . In other words, the control objective is achieved.

Based on this, we have the following theorem.

Theorem 2 Assume that the traffic flow controllability is satisfied for each section at any sampling time. There exists a  $w_i(n)$  at sampling time n, section i

$$w_i(n) = -\frac{L_i}{(n)T}u_i(n),$$

where  $u_i(n)$  satisfies

$$P(n) U(n) = E(n)$$

and

$$P(n) = \begin{bmatrix} \overline{b}_1 & 1\overline{b}_2 & 0 & \cdot & \cdot & \cdot & 0\\ 2\overline{b}_1 & \overline{b}_2 & 2\overline{b}_3 & 0 & \cdot & \cdot & 0\\ & 3\overline{b}_2 & \overline{b}_3 & 3\overline{b}_3 & 0 & \cdot & 0\\ 0 & 0 & \cdot & \cdot & \cdot & 0 & 0\\ 0 & \cdot & 0 & \cdot & \cdot & \cdot & 0\\ 0 & \cdot & 0 & \cdot & \cdot & \cdot & 0\\ 0 & \cdot & \cdot & 0 & \cdot & \cdot & N_{-1}\overline{b}_N\\ 0 & 0 & \cdot & \cdot & 0 & N\overline{b}_{N-1} & \overline{b}_N \end{bmatrix}$$

$$U(n) = \begin{bmatrix} u_{1}(n) \\ u_{2}(n) \\ u_{3}(n) \\ \vdots \\ \vdots \\ u_{N}(n) \end{bmatrix}, \quad E(n) = \begin{bmatrix} e_{1}(n) \\ e_{2}(n) \\ e_{3}(n) \\ \vdots \\ \vdots \\ e_{N}(n) \end{bmatrix}$$

that drives the traffic density at section i converges to the desired traffic density k  $_{di}$  exponentially, i.e.,  $_{k_i} \rightarrow _{k_{di}}$ 

exponentially.

Proof: The control  $w_i$  can be obtained by solving i(n) = 0. For a detail derivation, please see Chien (17).

If the velocity command, which is to be send to vehicles in each section of the lane to achieve at sampling time n+1, is chosen as

$$v_{command} = v_i(n) + \frac{T}{V_e} \{ V_e[k_i(n)] - v_i(n) \}$$
  
+  $\frac{T}{L_i} \frac{k_{i-1}(n)}{k_i(n)}, v_i(n) [\sqrt{v_{i-1}(n)v_i(n)} - v_i(n)]$   
-  $\frac{(n)T}{L_i} w_i(n)$ 

then the traffic density at section i converges to the desired traffic density k<sub>di</sub> exponentially.

#### **Implementation algorithm**

In this section, a computation algorithm for the purpose of simplifying implementation is proposed. To obtain the control law for each section, the computation involves solving a tridiagonal matrix equality.

$$P(n) U(n) = E(n)$$

Since traffic flow controllability is assumed, the non-singularity of the matrix P(n) is guaranteed. Therefore, it is trivial that the solution of the control law for each section can be computed through the obtaining of the matrix inversion.

$$U(n) = P^{-1}(n) E(n)$$

However, high computational effort is required to obtain the inversion of the N  $\times$  N matrix P(n) if the number of sections N is large. Hence, more efficient method should be employed. Several iterative

methods can be adopted to obtain the solution. However, these iterative methods have several drawbacks:

- Iterative methods require recursive computations. The computation rate may turn out to be slow when the numbers of sections N is large. Moreover, during the recursive computation stage, numerical error might accumulate and the solution may be highly distorted.
- Most recursive iterative methods require all information about the element of the matrix to be available during the recursive computing stage. Hence, to apply these recursive iterative methods to our problem, information from all sections should be combined and be passed to a central computer center. And then, based on all information from each section, the central computer decides the control action to be taken for each section and then passes the decision back to each section.

To avoid slow computation rate due to a large number of sections, large accumulated numerical error due to recursive computation, and extremely complex and high cost communication network between sections and central computer, we propose an approximation algorithm with the following features:

- Non-recursive algorithm. (This avoids the slow computation rate and large accumulated numerical error due to recursive computation.)
- No central computer center is needed. The control action is determined by a local computer center based only on the information passed from its adjacent sections in a series way. (This eliminates the need of extremely complex and high cost communication network.)

We described the approximation algorithm below: For section  $\,$  i, at sampling time n, the control law  $w_i(n)$  is replaced with

$$\hat{w}_{li}(n) = -\frac{L_i}{(n)T} \hat{u}_{li}(n)$$

where  $\hat{u}_{l_i}(n)$  is a component of  $\hat{U}_{l_i}$ , and

$$\begin{split} \hat{U}_{l}(n) &= \sum_{m=0}^{l} (-1)^{m} Z_{m}(n) \\ Z_{m}(n) &= [T^{-1}(n) \Delta(n)] Z_{m-1}(n) \ m = 1, 2, 3, \dots \\ Z_{0}(n) &= T^{-1}(n) E(n) \\ T(n) &= \\ \begin{bmatrix} \overline{b}_{1}(n) & 0 & 0 & \cdot & \cdot & 0 \\ 2\overline{b}_{1}(n) & \overline{b}_{2}(n) & 0 & 0 & \cdot & \cdot & 0 \\ 2\overline{b}_{1}(n) & \overline{b}_{2}(n) & 0 & 0 & \cdot & \cdot & 0 \\ 0 & 3\overline{b}_{2}(n) & \overline{b}_{3}(n) & 0 & 0 & \cdot & 0 \\ 0 & 3\overline{b}_{2}(n) & \overline{b}_{3}(n) & 0 & 0 & \cdot & 0 \\ 0 & 0 & \cdot & \cdot & 0 & 0 \\ 0 & 0 & \cdot & \cdot & 0 & 0 \\ 0 & 0 & \cdot & \cdot & 0 & 0 \\ 0 & 0 & \cdot & \cdot & 0 & N\overline{b}_{N-1}(n) & \overline{b}_{N}(n) \end{bmatrix} \\ \Delta(n) = \end{split}$$

0	$_1\overline{b}_2(n)$	0	•	•	•	0
0	0	$_{2}\overline{b}_{3}(n)$	0	•	•	0
0	0	0	$_{3}\overline{b}_{3}(n)$	0		0
0	0				0	0
0		0			•	0
0			0	•	0	$_{N-1}\overline{b}_{N}(n)$
0	0	•	•	0	0	0

The on-line implementation of this approximation is summarized in the following:

Step 1: Determination of  $Z_0$ Since  $Z_0 = T^{-1} E$ , it implies  $TZ_0 = E$ . Let  $z_0^1, z_0^2, z_0^3, ..., z_0^N$  are components of  $Z_0$ . Therefore, they should satisfy the following equations:

$$b_{1}z_{0}^{1} = e_{1}$$

$${}_{2}\overline{b}_{1}z_{0}^{1} + \overline{b}_{2}z_{0}^{2} = e_{2}$$

$${}_{3}\overline{b}_{2}z_{0}^{2} + \overline{b}_{3}z_{0}^{3} = e_{3}$$

$$\cdots$$

$${}_{N}\overline{b}_{N-1}z_{0}^{N-1} + \overline{b}_{N}z_{0}^{N} = e$$

In other words, all components of  $Z_0$  can be determined by a series way.

Step 2: Determination of  $Z_i$  (i = 2, 3, 4, ... l) Since  $Z_i = T^{-1} \Delta Z_{i-1}$ , it implies  $TZ_i = \Delta Z_{i-1}$ . Let  $z_i^{1}, z_i^{2}, z_i^{3}, ..., z_i^{N}$  are components of  $Z_i$ . Therefore, they should satisfy the following equations:

$$\overline{b}_{1}z_{i}^{1} = \overline{b}_{2}z_{i-1}^{2}$$

$$\overline{b}_{1}z_{i}^{1} + \overline{b}_{2}z_{i}^{2} = \overline{b}_{3}z_{i-1}^{3}$$

$$\overline{b}_{2}z_{i}^{2} + \overline{b}_{3}z_{i}^{3} = \overline{b}_{4}z_{i-1}^{4}$$

$$\cdots$$

$$\overline{b}_{N-1}z_{i}^{N-1} + \overline{b}_{N}z_{i}^{N} = 0$$

In other words, all components of  $Z_i$  can be determined by a series way. Remark It can be shown that, with the parameter set (B.4),

$$\lim_{l\to\infty} \|\frac{U(n)-\hat{U}_l}{U(n)}\| = 0$$

for some natural norm  $\|\cdot\|$ . It implies that  $U_i(n)$  can be served as a good approximation of U(n); and hence,  $\hat{W}_i(n)$  can be served as a good approximation of  $w_i(n)$ . Furthermore, it can also be shown that, even the zero-order approximation, i.e.,  $w_{0i}(n)$ , provides a good approximation of  $w_i(n)$ . In other words, the computation effort can be greatly reduced by the proposed algorithm when zero-order approximation is adopted.

#### References

- 1. American Association of State Highway and Transportation Officials (AASHTO), *A Policy on Geometric Design of Highways and Streets*, Washington, D.C.: AASHTO, 1990.
- 2. Ackermann, J. and Sienel, W., "Robust Control for Automatic Steering," in the *Proc. of the American Control Conference*, San Diego, 795-800, 1990.
- 3. Allen, R.W., "Performance and Safety Considerations in Automatic and Four Wheel Steering Control Laws," in the *Proc. of the IVHS America 1993 Annual Meeting*, 269-275, 1993.
- 4. Altan, O.D., Craig, D.B., Litkouhi, B.B., and Oberdier, L.M., "LANETRAK: A Vision-based Automatic Vehicle Steering System," General Motors Research, *Research Publication*, GMR-7835, 1992.
- 5. Bakker, E., Nyborg, L. and Pacejka, H.B., "Tyre Modelling for Use in Vehicle Dynamics Studies," Society of Automotive Engineers, *Paper*, 879421, 1987.
- 6. Baldwin, S.M. and Walrand, J., "Prototyping an Infrared Communication Link for the PATH Project," California PATH Program, Institute of Transportation Studies, University of California, Berkeley, *PATH Study*, 1990.
- 7. Baños, J.C.M., Papageorgiou, M. and Schäffner, C., "Integrated Optimal Flow Control in Traffic Networks," *European Journal of Operational Research*, Volume 71, 317-323, 1993.
- 8. Baron, S., "Pilot Control," In *Human Factors in Aviation*, edited by Earl L. Wiener and David C. Nagel, 347-385, San Diego: Academic Press, Inc., 1988.
- 9. Begault, D.R., "Head-up Auditory Displays for Traffic Collision Avoidance System Advisories: A Preliminary Investigation," *Human Factors*, Volume 35, Issue 4, 707-717, 1993.
- 10. Bhandal, A.S. and Palmquist, U.B., "Driver Information and Traffic Control Using Short Range Vehicle Roadside Communications," in the *Proc. of the Autotech '93*, 1993.
- 11. Billington, R. and Allan, R.N., *Reliability Evaluation of Engineering Systems: Concepts and Techniques*, New York: Plenum Press, 1983.
- 12. Bulkeley, D., "The Quest for Collision-free Travel." *Design News*, October 4, 1993, 108 (4), 1993.
- 13. Calson, W.L., "Crash Injury Loss: The Effect of Speed, Weight and Crash Configuration.," *Accident Analysis and Prevention*, Volume 9, 1977.
- 14. Carson, J.M. and Wierwille, W.W., "Development of a Strategy Model of the Driver in Lane Keeping," *Vehicle System Dynamics*, Volume 7, 233-253, 1978.
- 15. Caywood, W.C. and Rubinstein, N., "Ride Quality and Guideway Roughness Measurements of the Transpo '72 PRT Systems," *High Speed Ground Transportation Journal*, Volume 8, Issue 3, 213-226, 1974.

- 16. Chadwick, D.J. and Patel, V.M., "A Communication Architecture for Early Implementation of Intelligent Vehicle Highway Systems," in the *Proc. of the Transportation Research Board 72nd Annual Meeting*, January, 1993.
- 17. Chien, C.C., "Intelligent Vehicle Highway System (IVHS): Advanced Vehicle Control System.," University of Southern California, 1994.
- 18. Chien, C.C. and Ioannou, P., "Automatic Vehicle Following," in the *Proc. of the American Control Conference*, Chicago, IL, 1992.
- 19. Chien, C.C., Zhang, Y., Stotsky, A., and Ioannou, P., "Roadway Traffic Controller Design for Automated Highway Systems," To appear in the *Proc. of the 33rd IEEE Conference on Decision and Control (CDC)*, Lake Buena Vista, FL, December 14-16, 1994.
- 20. Chiu, H.Y., Stupp, G.B. and Brown, S.J., "Vehicle-Follower Control with Variable-Gains for Short-Headway Automated Guideway Transit Systems," *Journal of Dynamic Systems*, Volume 99, Issue 3, 183-189, 1977.
- Choven, J.D., Tijerina, L., Alexander, G., and Hendrick, D.L., "Examination of Lane Change Crashes and Potential IVHS Countermeasurements," U.S. Department of Transportation, DOT HS 808 071, 1994.
- 22. Church, V.M., "Technology Takes the Toll." *Popular Science*, March 1992, 78-81, 1992.
- Cugiani, C., Giubbolini, L. and Tascone, R., "Millimetre-wave Guard Rail Tracking System for Vehicle Lateral Control," in the *Proc. of the IEEE-IEE Vehicle Navigation & Information Systems* (VNIS'93) Conference, Ottawa, IEEE, 521-524, 1993.
- 24. Dettki, F., "Analyse des Fahrverhaltens hinsichtlich Zeitbedarf, Art und Akzeptanz der Uebergabeprozedur zwischen automatisch quergefuhrtem und manuell betriebenenm Fahren," Daimler-Benz, 1993.
- 25. Dickerson, J.A., Kim, H.M. and Kosko, B., "Fuzzy Control of Platoons of Smart Cars," in the *Proc. of the IEEE International Conference on Fuzzy Systems (FUZZ-IEEE)*, Orlando, Florida, 1994.
- Fancher, P.S., Ervin, R.D., Bareket, Z., Johnson, G.E., Trefalt, M., Tiedecke, J., and Hagleitner, W., "Intelligent Cruise Control: Performance Studies Based Upon an Operating Prototype," in the *Proc. of the IVHS America*, Atlanta, Georgia, 1994.
- 27. Farber, E., "Intelligent Vehicle Highway Systems Safety Benefits and Public Policy," in the *Proc.* of the 13th International Technical Conference on Experimental Safety Vehicles, 1991.
- 28. Farber, E., "Using the REAMACS Model to Compare the Effectiveness of Alternative Rear End Collision-Warning Algorithms," in the *Proc. of the Fourth Annual Meeting of IVHS America*, Atlanta, 1994.
- 29. Fenton, R.E., "A Headway Safety Policy for Automated Highway Operations," *IEEE Transactions* on Vehicular Technology, Volume VT-28, Issue 1, 22-28, 1979.

- 30. Fenton, R.E., Melocik, G.C. and Olson, K.W., "On the Steering of Automated Vehicles: Theory and Experiment," *IEEE Transaction on Automated Control*, Volume AC-21, Issue 3, 306-315, 1976.
- 31. Foreman, B., "Infrared Communication in PATH," in the *Proc. of the PATH Short Course Notes*, Berkeley, CA, 1994.
- 32. Franklin, G.F., Powell, J.D. and Emami-Naeini, A., *Feedback Control of Dynamic Systems*, Second Edition, Vol. Reading, Mass.: Addison-Wesley, 1991.
- 33. French, R.L., "Road Transport Informatics: The Next 20 Years," in the *Proc. of the Future Transportation Technology*, San Francisco, Society of Automotive Engineers, SAE-881175, 1988.
- 34. Fukuhara, H. and Kurami, K., "Essential Issues Involved in Radar-based Collision Warning/Avoidance Systems," in the *Proc. of the IVHS America*, Atlanta, 1994.
- 35. Glimm, J. and Fenton, R.E., "An Accident-Severity Analysis for a Uniform-Spacing Headway Policy," *IEEE Transactions on Vehicular Technology*, Volume VT-29, Issue 1, 96-103, 1980.
- 36. Godthelp, H., "Vehicle Control During Curve Driving," *Human Factors*, Volume 28, Issue 2, 211-221, 1986.
- 37. Graham, R. and Hirst, S., "The Effect of a Collision Avoidance System on Driver's Braking Response," in the *Proc. of the IVHS America*, Atlanta, 1994.
- 38. Greenberg, A., "AT&T Team, Amtech in Automatic Highway Toll Battle." *Electronic News*, July 20, 1992, 23, 1992.
- 39. Gustafsson, F., "Slip-Based Estimation of Tire-Road Friction," Linkoping University, Sweden, *Report*, LiTH-ISY-R-1509, 1993.
- 40. Gustafsson, F. and Ekstrom, H., "Tire-Road Friction Estimation Results from Field Trials in Arvidsjaur," Linkoping University, Sweden, LiTH-ISY-R-1476, 1993.
- 41. Gustafsson, F., Zhang, Q. and Ljung, L., "On the Estimation of Tire-Road Friction," Linkoping University, Sweden, LiTH-ISY-I-1405, 1992.
- 42. Hedrick, J.K., McMahon, D., Narendran, V., and Swaroop, D., "Longitudinal Vehicle Controller Design for IVHS Systems," in the *Proc. of the American Control Conference*, 3107-3112, 1991.
- 43. Heller, M. and Huie, M., "Vehicle Lateral Guidance using Vision, Passive Wire and Radar Sensors," in the *Proc. of the IEEE-IEE Vehicle Navigation & Information Systems (VNIS'93) Conference*, Ottawa, 505-508, 1993.
- 44. Hirata, T., Hara, T. and Kishore, A., "The Development of a New Multi-AV System Incorporating an On-Board Navigation Function," in the *Proc. of the Automotive Display Systems and IVHS*, SAE International, 1-10, 1993.
- 45. Hitchcock, T., "A Specification of an Automated Freeway with Vehicle-Borne Intelligence," California PATH Program, *PATH Report*, UCB-ITS-PRR-92-18, 1992.

- 46. Hitchcock, T., "Casualties in Accidents Occurring During Split and Merge Maneuvers," California PATH, *PATH Technical Memorandum*, 93-9, 1994.
- 47. Hsu, A., Eskafi, F., Sachs, S., and Varaiya, P., "The Design of Platoon Maneuver Protocols for IVHS," California PATH Program, Institute of Transportation Studies, University of California, Berkeley, *PATH Research Report*, UCB-ITS-PRR-91-6, 1991.
- 48. Hussain, T.M., Saadawi, T.N. and Ahmed, S.A., "Overhead infrared-sensor for monitoring vehicular traffic," *IEEE Transactions on Vehicular Technology*, Volume 42, Issue 4, 477-483, 1993.
- 49. Inoue, T., Aoki, K., Suzuki, T., and Yokoyama, T., "A Preview Steering Control Strategy Based on Computer Vision for Automated Vehicle Highway Systems," in the *Proc. of the IFAC*, Sydney, Autralia, 489-492, 1993.
- 50. Ioannou, P. and Chien, C.C., "Autonomous Intelligent Cruise Control," *IEEE Transactions on Vehicular Technology*, Volume 42, Issue November, 657-672, 1993.
- 51. Ioannou, P. and Xu, T., "Throttle and Brake Control for Automatic Vehicle Following," *IVHS Journal*, Volume 1, Issue 4, 345-377, 1994.
- 52. Ito, T., Azuma, S. and Sumiya, K., "Development of the New Navigation System-Voice Route Guidance," in the *Proc. of the Automotive Display Systems and IVHS*, SAE International, 111-119, 1993.
- 53. Kady, M. and Shloss, P., "Electronic Messaging Using VRC (Vehicle to Roadside Communications)," in the *Proc. of the IVHS America*, Atlanta, 1994.
- 54. Kantowitz, B., Becker, C.A. and Barlow, S.T., "Assessing Driver Acceptance of IVHS Components," in the *Proc. of the the 37th Annual Meeting of the Human Factors and Ergonomics Society*, 1993.
- 55. Kao, W.-W., "Integration of GPS and Dead-Reckoning Navigation Systems," in the *Proc. of the Vehicle Navigation and Information Systems (VNIS'91)*, Dearborn, MI, SAE International, 635-643, 1991.
- 56. Karaaslan, U., Varaiya, P. and Walrand, J., "Two Proposals to Improve Freeway Traffic Flow," California PATH Program, Institute of Transportation Studies, University of California, Berkeley, *PATH Research Report*, UCB-ITS-PRR-90-6, 1990.
- 57. Kelley, J.R., "Road Weather Information Systems (RWIS) for IVHS Applications and Improved Maintenance Procedures," in the *Proc. of the IVHS America*, Atlanta, Paper #94041, 1994.
- 58. Klein, L.A., "Traffic Parameter Measurement Technology Evaluation," in the *Proc. of the the IEEE-IEE Vehicle Navigation & Information Systems Conference (VNIS'93),* Ottawa, IEEE, 529-533, 1993.
- 59. Knipling, R.R., "IVHS Technologies Applied to Collision Avoidance: Perspectives on Six Target Crash Types and Countermeasures," in the *Proc. of the 1993 IVHS America Annual Meeting*, 1993.

- 60. Knipling, R.R., Wang, J.S. and Yin, H.S., "Rear-end Crashes: Problem Size Assessment and Statistical Description," *U.S. Department of Transportation*, *Technical Report*, DOT HS 807 994, 1993.
- 61. Lee, A.Y., "Vehicle Stability Augmentation Systems Design using Parameter Optimization," General Motors Research Laboratories, *Research Publication*, GMR-6166, 1988.
- 62. Lee, A.Y., "A Preview Steering Autopilot Control Algorithm for Four-Wheel-Steering Passenger Vehicles," *Journal of Dynamic Systems, Measurement, and Control*, Volume 114, 401-408, 1992.
- 63. Lenard, M., *Safety Consideration for a High Density Automated Vehicle System.*, Erie, Pennsylvania: General Electric Company, 1969.
- 64. Lerner, N.D., Kotwal, B.M., Lyons, R.D., and Gardner-Bonneau, D.J., "Preliminary Human Factors Guidelines for Crash Avoidance Warning Devices (Draft)," Office of Crash Avoidance Research, National Highway Traffic Safety Administration, *Interim Report*, DTNH22-91-C-07004, 1993.
- 65. Lewis, E.E., Introduction to Reliability Engineering, New York: John Wiley & Sons, 1987.
- 66. Limpert, R., *Motor Vehicle Accident Reconstruction and Cause Analysis*, Third Edition, Vol. Charlottesville, Va.: Michie Company, 1989.
- 67. Marans, R., "User Acceptance of Intelligent Vehicle-Highway Systems (IVHS): Directions for Future Research," In *SAE Technical Paper Series*, edited by 97-101, SAE No. 901508. Warrendale, PA: Society of Automotive Engineers, 1990.
- 68. Masaki, I., "Vision-Based Collision Avoidance Systems," in the *Proc. of the IVHS America*, Atlanta, Paper # 94113, 1994.
- 69. McLean, J.R. and Hoffmann, E.R., "The Effects of Restricted Preview on Driver Steering Control and Performance," *Human Factors*, Volume 15, Issue 4, 421-430, 1973.
- Meisner, J., Frazzini, R., Funk, H., and Zhang, W.-B., "Function Criticality, Allocation and Mechanization for AHS Health Management Precursor System Analysis," Office of Safety and Traffic Operations, R&D, Federal Highway Administration, *Working Paper, Contract #DTFH61-93-C-00197*, 1994.
- 71. Messmer, A. and Papageorgiou, M., "Automatic Control Methods Applied to Freeway Network Traffic," *Automatica*, Volume 30, Issue 4, 691-702, 1994.
- 72. Mironer, M. and Hendricks, D., "Examination of Single Vehicle Roadway Departure Crashes and Potential IVHS Countermeasures," U.S. Department of Transportation, Volpe National Transportation Center, *Report*, DOT-VNTSC-NHTSA-94-3, 1994.
- 73. Mitchell, V.A., Graham, R. and Saunderson, C.A., "The Design of the Man-Machine Interface to a Medium Range Pre-information (MRP) System," in the *Proc. of the Autotech '93*, 1993.
- 74. Müller, R. and Nöcker, G., "Intelligent Cruise Control with Fuzzy Logic," in the *Proc. of the the Intelligent Vehicles '92 Symposium*, Detroit, 173-178, 1992.

- 75. Murphy, D.O. and Woll, J.D., "A Review of the VORAD Vehicle Detection and Driver Alert System," Society of Automotive Engineers, *Report*, SAE-922495, 1992.
- 76. Najm, W.G., "A Review of IVHS Crash Avoidance Technologies Workshop on Collision Avoidance System," in the *Proc. of the IVHS AMERICA Safety and Human Factors Committee*, Reston, VA, 1994.
- 77. National Highway Transportation Safety Administration, "General Estimates System 1989-A Review of Information on Police-Reported Traffic Crashes in the United States," National Highway Transportation Safety Administration, *NHTSA Technical Report*, DOT HS 807 665, 1989.
- 78. Nguyen, H.G. and Laisne, J.Y., "Obstacle Detection using Bi-Spectrum CCD Camera and Image Processing," in the *Proc. of the Intelligent Vehicle Symposium*, 42-50, 1992.
- 79. Niehaus, A. and Stengel, R.F., "An Expert System for Automated Highway Driving." *IEEE Control Systems Magazine*, 53-61, 1991.
- 80. Olson, R.A., Gustavson, R.L., Wangler, R.J., and McConnell, R.E., "Active-infrared Overhead Vehicle Sensor," *IEEE Transactions on Vehicular Technology*, Volume 43, Issue 1, 79-85, 1994.
- 81. Pacini, P. and Kosko, B., "Lane Changing," University of Southern California, *SIPI Research Report*, 1993.
- 82. Papacostas, C.S., *Fundamentals of Transportation Engineering*, Englewood Cliffs, New Jersey: Prentice Hall, 1987.
- 83. Peacock, B. and Karwowski, W., Automotive Ergonomics, London: Taylor and Francis, 1993.
- Peng, H., Zhang, W.-B., Shladover, S., Tomizuka, M., and Arai, A., "Magnetic Marker Based Lane Keeping: A Robustness Experimental Study," *Society of Automotive Engineers*, Issue Paper Number 930556, 127-132, 1993.
- 85. Polydoros, A., Dessouky, K., Pereira, J., Sun, C., Papavassiliou, T., and Li, V., "Vehicle to Roadside Communications Study," California PATH Program, *PATH Technical Report*, 1993.
- 86. Pomerleau, D., Thorpe, C., Langer, D., Rosenblatt, J., and Sukthankar, R., "AVCS Research at Carnegie Mellon University," in the *Proc. of the IVHS America 1994 Annual Meeting*, Atlanta, Paper#94100, 1994.
- 87. Pue, A.J., "A State Constrained Approach to Vehicle Follower Control for Short Headway Automated Guideway Transit Systems," in the *Proc. of the 1977 Joint Automatic Control Conference*, San Francisco, 401-407, 1977.
- Rahimi, A., "Safety Analysis for Automated Transportation Systems," Operations Research Society, 1971.
- 89. Rao, B.S.Y., Varaiya, P. and Eskafi, F., "Investigations into Achievable Capacities and Stream Stability with Coordinated Intelligent Vehicles," California PATH Program, Institute of Transportation Studies, University of California, Berkeley, *California PATH Report*, 1992.

- Riley, V., "Human Factors Design of Automated Highway Systems—Function Allocation," Federal Highway Research Center, *Revised Working Paper*, Contract Number DTFH61-92-C-00100, 1994.
- 91. Schwarzinger, M., "Vision-Based Car-Following: Detection, Tracking, and Identification," in the *Proc. of the Intelligent Vehicles Symposium*, 24-29, 1992.
- 92. Sekine, M., Senoo, T., Morita, I., and Endo, H., "Design Method for an Automotive Laser Radar System and Future Prospects for Laser Radar," in the *Proc. of the Intelligent Vehicle Symposium*, 120-125, 1992.
- 93. Sheikholesam, S. and Desoer, C.A., "Longitudinal Control of a Platoon of Vehicles with No Communication of Lead Vehicle Information: A System Level Study," California PATH Program, *PATH Technical Memorandum*, 91-2, 1991.
- 94. Shimizu, Y. and Kawai, T., "Development of Electric Power Steering," 1991.
- 95. Shladover, S.E., Desoer, C.A., Hedrick, J.K., and al., e., "Automated Vehicle Control Developments in the PATH Program," *IEEE Transactions on Vehicular Technology*, Volume 40, Issue 1, February, 114-130, 1991.
- 96. Sims, D., "False Alarms Trigger Changes in Collision-Avoidance Code." *IEEE Software*, November, 107-108, 1992.
- 97. Sklar, B., *Digital Communications: Fundamentals and Applications*, Englewood Cliffs: New Jersey: Prentice Hall, 1988.
- 98. Sorkin, R.D., Kantowitz, B.H. and Kantowitz, S.C., "Likelihood Alarm Displays," *Human Factors*, Volume 30, Issue 4, 445-459, 1988.
- 99. Stowers, J.R., "HOV Lessons from the Dulles Toll Road," *TR News*, Volume 170, Issue Jan-Feb, 5-9, 1994.
- 100. Streisand, S.L. and Walrand, J., "A Communications Architecture for IVHS," Institute of Transportation Studies, University of California, Berkeley, *Technical Report*, 1992.
- Sun, Y. and Ioannou, P., "A Handbook on Headway Calculation for Collision Free Vehicle Following," Southern California Center for Advanced Transportation, University of Southern California, *Technical Report*, 1994.
- 102. Syverud, K., "Liability and Insurance Implications of IVHS Technology," SAE, *Technical Paper Series*, 1990.
- 103. Teramoto, T., Fujimura, K. and Fujita, Y., "Study of Laser Radar," in the *Proc. of the 12th International Technical Conference on Experimental Safety Vehicles*, 888-895, 1989.
- Transportation Research Board (TRB), "TRB Spectrum Needs for IVHS Applications Workshop," Transportation Research Board, 1993.
- 105. Tribe, R., Conlong, R. and Prynne, K., "Collision Warning," in the *Proc. of the Autotech '93*, Prometheus-Future Systems, 1993.

- 106. Triggs, T. and Harris, W.G., "Reaction time of drivers to road stimuli," Monash University, Melbourne, Australia, *Human Factors Report*, HFR-12, 1982.
- 107. Tsuji, H., Maeda, H., Shibata, A., and Morisue, F., "Evaluation of Location System Combining a GPS Receiver with Inertial Sensor," in the *Proc. of the Vehicle Navigation & Information Systems* (*VNIS '91*), Dearborn, MI, SAE International, 645-649, 1991.
- 108. Turan, L., Ioannou, P., Safonov, M.G., Smith, D., and Damos, D., "Automated Check-In/Check-Out," University of Southern California, *Report*, 94-06-02, 1994.
- 109. Ulmer, B., "VITA An Autonomous Road Vehicle (ARV) for Collision Avoidance in Traffic," in the *Proc. of the Intelligent Vehicles Symposium*, 36-41, 1992.
- 110. Van Trees, H.L., *Detection, Estimation, and Modulation Theory, Part 1*, New York: John Wiley & Sons, 1968.
- 111. Varaiya, P. and Shladover, S., "Sketch of an IVHS System Architecture," California PATH Program, Institute of Transportation Studies, University of California, Berkeley, PATH Research Report, UCB-ITS-PRR-91-3, 1991.
- 112. Walters, J., "Los Angeles: Technology and Traffic." Governing, October 1992, 43, 1992.
- 113. Wang, J.S. and Knipling, R.R., "Lane Change/Merge: Problem Size Assessment and Statistical Description," U.S. Department of Transportation, *Technical Report*, DOT Draft, 1993.
- 114. Wang, J.S. and Knipling, R.R., "Single Vehicle Roadway Departure Crashes: Problem Size Assessment and Statistical Description," U.S. Department of Transportation, Technical Report, DOT Draft, 1993.
- 115. Wierwille, W.W., Casali, J.G. and Repa, B.S., "Driver Steering Reaction Time to Abrupt-Onset Crosswinds, as Measured in a Moving-Base Driving Simulator," *Human Factors*, Volume 25, Issue 1, 103-116, 1983.
- 116. Woodson, W.E., Tillman, B. and Tillman, P., *Human Factors Design Handbook*, New York: McGraw-Hill, 1991.
- 117. Xu, T. and Ioannou, P., "Modeling of the Brake Line Pressure to Time Brake Force Subsystem," *Southern California Center for Advanced Transportation Technologies*, University of Southern California, *Technical Report*, 1993.
- 118. Yanagisawa, T., Yamamoto, K. and Kubota, Y., "Development of a Laser Radar System for Automobiles," *Electronic Display Technology and Information Systems*, Volume SAE SP-904, Issue 920745, 73-85, 1992.
- 119. Cremer, M. and May, A.D., "An extended traffic model for freeway control," University of California at Berkeley, *Research Report*, UCB-ITS-RR-85-7, 1985.
- 120. Greenberg, H., "An Analysis of Traffic Flow," Operations Research, Volume 6, 79-85, 1958.

- 121. Greenlee, T.L. and Payne, H.J., "Freeway Ramp Metering Strategies for Responding to Incidents," in the *Proc. of the IEEE Conference on Decision and Control*, 982-987, 1977.
- 122. Lighthill, J.J. and Witham, G.B., "On kinematic waves ii: A theory of traffic flow on long crowded roads," *Proceedings of the Royal Society of London, Series A*, Volume 229, 317-345, 1955.
- 123. Papageorgiou, M., Blosseville, J.M. and Hadj-Salem, H., "Macroscopic modelling of traffic flow on the boulevard peripherique in Paris," *Transportation Research*, Volume B, Issue 23, 29-47, 1989.
- 124. Papageorgiou, M., Blosseville, J.M. and Hadj-Salem, H., "Macroscopic modelling of traffic flow on the boulevard peripherique in Paris Part II Coordinated on-ramp metering," *Transportation Research*, Volume B, Issue 23, 1989.
- 125. Papageorgiou, M., *Applications of automatic control concepts to traffic flow modeling and control*, Springer-Verlag, 1983.
- 126. Payne, H.J., "Models of freeway traffic and control," in the *Proc. of the Mathematical Models of Public Systems Simulation Council*, 51-61, 1971.
- 127. Payne, H.J., "A critical review of a macroscopic freeway model," in the *Proc. of the the Conference on Research Directions in Computer Control of Urban Traffic Systems*, 251-265, 1979.
- 128. Payne, H.J. and Teener, M.D., "A Critical Review of a Macroscopic Freeway Model," *International Journal of Control*, Volume 469, 52-64, 1973.
- 129. Pipes, L.A., "Vehicle Accelerations in the Hydrodynamic Theory of Traffic Flow," *Transportation Research*, Volume 3, 229-234, 1969.
- 130. Sheikholesam, S., "Control of a Class of Interconnected Nonlinear Dynamical Systems: the Platoon Problem," Ph.D. Dissertation, University of California, Berkeley, 1991.